Classical-quantum dynamics with applications to gravity

By

Zachary Weller-Davies

A thesis submitted to University College London for the degree of Doctor of Philosophy

Department of Physics and Astronomy University College London I, Zachary Weller-Davies, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Signed

Date

Classical-quantum dynamics with applications to gravity

Zachary Weller-Davies

Doctor of Philosophy of Physics University College London Prof. Jonathan Oppenheim, Supervisor

Abstract

We develop autonomous classical-quantum dynamics by providing formalism and tools to study effective theories of interacting classical and quantum systems. We find the most general form of continuous classical-quantum master equation, which generalizes the Fokker-Planck equation of classical mechanics and the Lindblad equation of open quantum systems, allowing for coupling between the classical and quantum systems. We further show that this master equation can be unraveled by stochastic classical-quantum trajectories. The resulting dynamics is a natural generalization of the standard semi-classical equations of motion. However, because the dynamics are linear in the combined classical-quantum state, they are completely positive on all initial quantum states, providing a method to study semi-classical physics even in the presence of large quantum fluctuations.

We find a general path integral representation for classical-quantum dynamics, generalizing the Feynman-Vernon and classical stochastic path integrals, allowing for interaction between classical and quantum systems. Via path integral methods, we give the first examples of Lorentz invariant classical-quantum dynamics. We further find diffeomorphism invariant classical-quantum theories of gravity. We provide a methodology to derive the generalizations of the gravitational Hamiltonian and momentum constraints in such theories.

We prove that the consistency of classical-quantum coupling implies a general trade-off between the quantum decoherence rate and the degree of diffusion induced in the classical system, given by the back-reaction strength. Applying the trade-off relation to Newtonian gravity, we find an experimental signature of theories in which gravity is fundamentally classical. Bounds on decoherence rates arising from current interferometry experiments, combined with precision measurements of mass, place substantial restrictions on theories where Einstein's classical theory of gravity interacts with quantum matter, with part of the parameter space of such theories already squeezed out. We provide figures of merit that can be used in future experiments.

Impact Statement

Semi-classical descriptions of systems are ubiquitous in all areas of physics. In gravity, this is the regime we are interested in when studying black-hole evaporation or inflationary cosmology, where quantum fluctuations in an expanding universe give rise to the primordial seeds for all of the large-scale structures we see today. In measurement theory, a quantum system interacts with a macroscopic device, considered classical. In atomic physics and quantum chemistry, small molecules often interact with a classical environment. Furthermore, most quantum technologies utilize a mixture of classical and quantum methods. In quantum communication, one often considers the problem of speeding up classical communication using quantum mechanics – a protocol that can be viewed as a classical-quantum map. Similarly, future quantum computers will combine quantum and classical operations, using quantum devices as subroutines calculating classically intractable tasks.

Despite this, it is only recently that tools, largely borrowed from quantum information, have been used to understand classical-quantum dynamics in generality. This work uses these tools to develop the classical-quantum formalism. We derive the most general form of classical-quantum master equation continuous in the classical degrees of freedom. We find unraveling and path integral descriptions of classical-quantum dynamics, whose descriptions are often better suited to numerical simulation. The dynamics we find generalize previous methods of incorporating quantum back-reaction on classical degrees of freedom. However, unlike standard semi-classical approaches, they allow for correlations to be built up between classical and quantum degrees of freedom, being applicable in a wider regime and forming a more complete semi-classical description.

Recent experimental proposals [1, 2] have also re-ignited discussion on whether one can

fundamentally treat space-time classically. Such experiments aim to rule out semi-classical theories of gravity by measuring gravitationally induced entanglement, which is not reproducible by a classical description. Though promising, these experiments are potentially decades away from being realized. In this work, we approach the problem from a different direction by studying classical-quantum theories of gravity and the generic predictions and experimental bounds that follow from a consistent treatment. This work finds generic predictions for the Newtonian limit of any theory which treats gravity classically. All classical-quantum dynamics must obey a trade-off between decoherence and diffusion, quantified in terms of the strength of the classical-quantum coupling. We use this to show that theories with a fundamentally classical gravitational field can be tested in the near term, while current experiments already substantially restrict the parameter space of such theories.

Beyond the trade-off, this work explores the conceptual and technical consequences of having a classical gravitational field, which, since we do not have a full theory of quantum gravity, is an important question in foundational physics. In particular, this work develops theories of classical-quantum gravity, finding the first examples of Lorentz invariant and diffeomorphism invariant classical-quantum dynamics via a path integral approach, which have been long sought after and could be used to propose a fundamental theory of classical gravity interacting with quantum matter.

List of Publications and Preprints

The work presented in this thesis contains material from the following publications and preprints:

- Jonathan Oppenheim and Zachary Weller-Davies. The constraints of post-quantum classical gravity. arXiv:2011.15112. JHEP 02 (2022) 080 [3].
- Jonathan Oppenheim, Carlo Sparaciari, Barbara Soda and Zachary Weller-Davies. Gravitationally induced decoherence vs space-time diffusion: testing the quantum nature of gravity. arXiv:2203.01982. Nature Commun. 14 (2023) 1, 7910. [4]
- 3. Jonathan Oppenheim, Carlo Sparaciari, Barbara Šoda and Zachary Weller-Davies. The two classes of hybrid classical-quantum dynamics. arXiv:2203.01332. [5]
- Isaac Layton, Jonathan Oppenheim and Zachary Weller-Davies. A healthier semi-classical dynamics. arXiv:2208.11722. [6]
- 5. Jonathan Oppenheim and Zachary Weller-Davies. Path integrals from completely positive classical-quantum dynamics. arXiv:2301.04677. [7]
- Jonathan Oppenheim and Zachary Weller-Davies. Covariant path integrals for quantum fields back-reacting on classical space-time. arXiv:2302.07283. [8]
- Isaac Layton, Jonathan Oppenheim, Andrea Russo and Zachary Weller-Davies. The weak field limit of quantum matter back-reacting on classical spacetime. arXiv:2307.02557. JHEP 08 (2023) 163. [9]

Other publications and preprints by the author are:

- Jonathan Oppenheim, Carlo Sparaciari, Barbara Šoda and Zachary Weller-Davies. Objective trajectories in hybrid classical-quantum dynamics. arXiv:2011.06009. Quantum 7 (2023) 891. [10]
- Juan F. Pedraza, Andrea Russo, Andrew Svesko and Zachary Weller-Davies. Lorentzian threads as 'gatelines' and holographic complexity. arXiv:2105.12735. Phys.Rev.Lett. 127 (2021) 27, 271602. [11]
- Juan F. Pedraza, Andrea Russo, Andrew Svesko and Zachary Weller-Davies. Sewing spacetime with Lorentzian threads: complexity and the emergence of time in quantum gravity. arXiv:2106.12585. JHEP 02 (2022) 093. [12]
- Juan F. Pedraza, Andrea Russo, Andrew Svesko and Zachary Weller-Davies. Computing spacetime. arXiv:2205.05705. Honorable mention for the 2022 Essay Competition of the Gravity Research Foundation. Int.J.Mod.Phys.D 31 (2022) 14, 2242010. [13]
- Rafael Carrasco, Juan F. Pedraza, Andrew Svesko and Zachary Weller-Davies. Gravitation from optimized computation: Einstein and beyond. arXiv:2306.08503. JHEP 09 (2023) 167. [14]

Acknowledgments

It would have been impossible to complete this thesis without the support I have received from many people during the completion of the Ph.D.

First and foremost, I would like to thank Jonathan, my supervisor. The thesis is an accumulation of many discussions we, and collaborators, had during my Ph.D. years. I am very grateful for how generous he was with his time and his willingness to chat and discuss (at length!) all things physics, but also for providing a relaxed research environment where everyone can pursue their own research interests at their own tempo. From him, I learned not only about physics but also how to research physics and that one must try to understand things in three different ways.

I am grateful to everyone in the group for creating a great environment to research and learn new things: Andrea, Andy, Barbara, Carlo, Emanuele, Isaac, Joan, Jonathan, Juan, Maite, and Philipp. I would also like to thank Jonathan for our many group lunches (Chutneys being my favorite). They were always fun, with a good mix of physics and general discussion. I also want to thank the CS group for all the cross-over lunches.

In particular, I want to thank Andrea for being the best Ph.D. companion (and collaborator) one could ask for during the bulk of the Ph.D. Also, thanks for the many incredible lunches he would make while we discussed Ph.D. life and the endless discussions and complaints about constraints. I want to thank Barbara, Carlo, and Joan for making the first-year transition into Ph.D. easy. I am grateful for the time Philipp and Henrique spent with Jonathan and I discussing the aspects of gravitational constraints, geometrodynamics, and shape dynamics. Thanks to Andy and Juan for being amazing post-docs to have in the group and even better collaborators. I want to thank Isaac for the infinite discussions on all things CQ and for his

encouragement when climbing. Thanks to Alessio Serafini and Dan Carney for taking time to examine the thesis and for their helpful comments and feedback.

I want to thank all my friends for sharing the joys and pains of the Ph.D., making it more joyful. Especially my housemates Alex and Petru, and both Martin and Petru for always being willing to make a trip to the pub.

Finally, I would like to thank my family for their constant love and support throughout my life and Shilu for her love and support during the process of my undergraduate, master's, and Ph.D. I am grateful to them for always encouraging me to pursue my goals, knowing they are there to support me.

Contents

Impact statement				
Li	st of	public	cations and preprints	6
A	ckno	wledgr	nents	8
Co	onte	nts		15
Li	st of	tables	5	16
1	Int	roduct	ion	17
	The	sis cont	ributions	. 21
	Stru	icture o	f the thesis	. 24
Ι	Ba	ckgrou	ınd	27
2	Cla	ssical,	quantum, and classical-quantum formalism	28
	2.1	Stocha	astic classical dynamics	. 29
		2.1.1	The probability density and probability transition amplitude \ldots .	. 29
		2.1.2	Markovian processes	. 30
		2.1.3	Short time moment expansion and the Fokker-Plank equation	. 30
		2.1.4	Pauli rate equation	. 32
		2.1.5	Pawula theorem	. 33
		2.1.6	Stochastic differential equations	. 34

		2.1.7	Path integrals for stochastic classical dynamics	35
	2.2	Quant	um theory and open quantum systems	36
		2.2.1	Quantum theory for closed systems	37
		2.2.2	Open quantum systems	38
		2.2.3	Density matrices and ensembles	39
		2.2.4	Quantum channels	42
		2.2.5	The Lindblad (GKSL) equation	43
		2.2.6	Time local dynamics	44
		2.2.7	Decoherence	46
		2.2.8	Unraveling of Lindblad equations	47
		2.2.9	Path integrals for open quantum systems	48
	2.3	Classi	cal-quantum dynamics	50
		2.3.1	Moment expansion and the CQ master equation $\ldots \ldots \ldots \ldots \ldots$	51
		2.3.2	Physical interpretation of the moments	59
		2.3.3	Master equation examples	61
3	Har	niltoni	an formulation of GR	66
3	Har 3.1	niltoni The E	an formulation of GR	66 66
3	Har 3.1 3.2	niltoni The E The 3	an formulation of GRinstein equations+1 split	66 66 68
3	Har 3.1 3.2	niltoni The E The 3 3.2.1	an formulation of GR instein equations +1 split Curvatures associated to the foliation	66 66 68 69
3	Har 3.1 3.2 3.3	niltoni The E The 3 3.2.1 Hamil	an formulation of GR instein equations +1 split Curvatures associated to the foliation tonian formulation of GR	 66 68 69 70
3	Har 3.1 3.2 3.3	niltoni The E The 3 3.2.1 Hamil 3.3.1	an formulation of GR instein equations	 66 68 69 70 74
3	Har 3.1 3.2 3.3 3.4	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp	an formulation of GR instein equations +1 split Curvatures associated to the foliation tonian formulation of GR Symmetries in the ADM formalism orating back-reaction: the semi-classical equations	 66 68 69 70 74 76
3	Har 3.1 3.2 3.3 3.4	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp	an formulation of GR instein equations	 66 68 69 70 74 76
3 II	Har 3.1 3.2 3.3 3.4	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp	an formulation of GR instein equations	 66 68 69 70 74 76 80
3 II 4	Har 3.1 3.2 3.3 3.4 De A c	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp evelop lassica	an formulation of GR instein equations	 66 68 69 70 74 76 80 81
3 II 4	Har 3.1 3.2 3.3 3.4 De A c 4.1	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp evelop lassica Positiv	an formulation of GR instein equations	 66 68 69 70 74 76 80 81 82
3 II 4	Har 3.1 3.2 3.3 3.4 De A c 4.1 4.2	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp evelop lassica Positiv Inequa	an formulation of GR instein equations	 66 68 69 70 74 76 80 81 82 83
3 II 4	Har 3.1 3.2 3.3 3.4 De A c 4.1 4.2 4.3	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp evelop lassica Positiv Inequa A clas	an formulation of GR instein equations	66 68 69 70 74 76 80 81 82 83 85
3 II 4	Har 3.1 3.2 3.3 3.4 De A c 4.1 4.2 4.3 4.4	niltoni The E The 3 3.2.1 Hamil 3.3.1 Incorp evelop lassica Positiv Inequa A clas Discus	an formulation of GR instein equations	666 68 69 70 74 76 80 81 82 83 85 90

5	Unr	Unraveling classical-quantum dynamics 92		
	5.1	1 The standard semi-classical equations		
	5.2	2 Classical-quantum trajectories		
	5.3	Unrav	eling continuous classical-quantum dynamics	. 96
		5.3.1	Deriving the unraveling	. 99
	5.4	6.4 Hamiltonian unravelings		
	5.5	Comp	arison to measurement and feedback	. 103
		5.5.1	Continuous classical-quantum dynamics as continuous measurement $\ . \ .$. 104
		5.5.2	CQ dynamics as an effective theory of quantum measurement	. 106
	5.6	Potent	tial applications to gravity	. 109
	5.7	Discus	sion \ldots	. 111
6	6 Path integrals for classical-quantum dynamics			113
	6.1	Mome	nt expansion of the dynamics	. 117
	6.2	Deriva	ation of the path integral formalism	. 118
		6.2.1	Derivation of phase space path integral for any CQ dynamics	. 119
	6.3	.3 Path integral formulation for continuous CQ master Equations without response		
		variables		
		6.3.1	Hermitian Lindblad operators	. 128
		6.3.2	Hamiltonian drift	. 129
		6.3.3	When the trade-off is saturated	. 130
		6.3.4	A path integral for continuous measurement and Markovian feedback .	. 130
	6.4	Config	guration space path integrals	. 133
		6.4.1	An explicit derivation of a configuration space path integral	. 133
6.5 Path integrals for classical fields interacting with quantum fields		ntegrals for classical fields interacting with quantum fields \ldots \ldots \ldots	. 137	
	6.6	Discus	ssion	. 139
	. .			
11	IA	pplica	ations to gravity	142
7	Cor	nstrain	ts in classical-quantum gravity	143

	7.2	CQ theory which reproduces Einstein gravity
		7.2.1 Liouville formulation of classical gravity
		7.2.2 CQ theories of gravity $\ldots \ldots \ldots$
	7.3	Deriving constraints from gauge conditions
		7.3.1 Dirac argument for Hamiltonian systems
		7.3.2 Deriving the constraint surface of GR from the Dirac argument 157
		7.3.3 Dirac argument for CQ master equations
	7.4	Deriving constraints in post-quantum theories of Gravity
		7.4.1 A general method of arriving at constraints
	7.5	A CQ theory of gravity coupled to a scalar field
	7.6	Discussion
8	Cov	ariant path integrals 178
	8.1	Completely positive path integrals
	8.2	Comparison to classical path integrals
	8.3	A natural class of path integrals
	8.4	Lorentz invariant CQ dynamics
	8.5	Diffeomorphism invariant CQ gravity
	8.6	Discussion
Q	The	Newtonian limit of classical-quantum dynamics
J	9 1	Newtonian limit of classical GR 198
	0.1	9.1.1 Newtonian limit via a reduced action 199
		9.1.2 A stochastic classical analog of the CO theory 202
		9.1.3 A simplified starting point 204
	0.2	Hamiltonian CO dynamics reproducing the Newtonian limit
	9.2 0.3	Continuous gravitational back reaction
	9.0	0.3.1 Imposing the Newtonian constraint 2008
		9.3.2 Linearity of the dynamics and white poise
	0.4	Deriving the Newtonian limit as a gauge fiving of a complete theory 212
	9.4	Comparison to provide classical eventure theories
	9.0	Comparison to previous classical-quantum theories

	9.6	Decoherence rates	. 218
	9.7	Newtonian limit for general master equations	. 219
		9.7.1 Jumping master equation	. 221
	9.8	Discussion	. 222
10) The	e trade-off between decoherence and diffusion	224
	10.1	A general Trade off between decoherence and diffusion $\ldots \ldots \ldots \ldots \ldots$. 226
	10.2	Trade off in the presence of fields	. 230
	10.3	Physical constraints on the classicality of gravity	. 232
	10.4	Discussion	. 241
A	CQ	states with continuous classical degrees of freedom	245
	A.1	Proof of Kraus theorem for CQ dynamics	. 247
В	Con	ntinuous CP evolution with arbitrary Lindblad operators	249
С	Unr	aveling of classical-quantum field theory	251
	C.1	A gravitational CQ theory example	. 252
D	Pert	turbative methods for CQ path integrals	255
\mathbf{E}	Con	nplete postivity of classical-quantum path integrals	258
	E.1	Proof of complete positivty	. 258
	E.2	Showing the natural class of CQ dynamics is CP	. 260
	E.3	Ensuring the CQ path integral is normalized	. 261
		E.3.1 Normalization of higher derivative classical path integrals \ldots .	. 262
		E.3.2 Normalization of higher derivative Feynman-Vernon path integrals \ldots	. 263
		E.3.3 Normalization of CQ path integrals	. 264
\mathbf{F}	The	e trade-off between decoherence and diffusion coupling constants	266
G	Clas	ssical-quantum dynamics with fields	268
	G.1	CQ Kramers-Moyal expansion for fields	. 269
	G.2	Trade-off between diffusion and decoherence couplings for fields	. 270

	G.3	Obser	vational trade-off for fields	271
н	\mathbf{Rel}	ating o	decoherence rates to the observational trade-off	273
Ι	Detecting gravitational diffusion 27			277
	I.1	Table-	top experiments	278
		I.1.1	Ultra-local continuous models	279
		I.1.2	Ultra-local jumping models	280
		I.1.3	Continous Diosi-Penrose model	281
J	Pos	itivity	constraints in open quantum field theory	283
Κ	K Symmetry generators with information loss 28			
	K.1 Transformations with a classical noise process $\ldots \ldots $			286
		K.1.1	Best guess for transformed states: prediction vs. retrodiction $\ . \ . \ .$	288
	K.2	Loren	tz invariance for Lindblad equations	290
С	omm	only u	sed notation	300
\mathbf{R}	efere	ences		328

List of Tables

2.1	This table compares the general form of maps that govern the dynamics of classi-
	cal, quantum, and classical-quantum systems. It shows that CQ maps generalize
	probability transition equations and quantum operations to allow for classical-
	quantum coupling. $\ldots \ldots 52$
2.2	This table compares autonomous master equations for classical, quantum, and
	classical-quantum dynamics. It shows that classical-quantum master equations
	naturally generalize the Pauli-rate equation for open classical systems and the
	GKSL equation for open quantum systems, allowing for classical-quantum coupling. $~58$
6.1	This table compares classical, quantum, and classical-quantum path integrals. It shows that the classical-quantum path integrals we find generalize stochastic path
	integrals for open quantum systems and the Feynman-Vernon path integral of
	open quantum systems, allowing for coupling between the classical and quantum
	systems
9.1	This table gives examples of decoherence and diffusion kernels that saturate the
	decoherence diffusion trade-off
10.1	This table summarizes the current experimental bounds on non-relativistic the-
	ories of classical-quantum gravity

Chapter 1

Introduction

Effective theories are ubiquitous in physics: from particle physics to classical statistical mechanics, we often make approximations to an underlying physical theory to simplify the dynamical description. Broadly, two approaches exist to constructing effective theories [15]. In a Wilsonian approach [16, 15], one starts with a high energy theory and asks how the effective low energy description changes as high momentum modes are integrated out. Alternatively, one modifies the theory by hand to isolate the desired degrees of freedom while maintaining phenomenological accuracy. This approach usually involves changing the high-energy description to make the effective description simpler and easier to use.

Often, we are interested in the effective description of a system where one part behaves classically and the other quantum mechanically; the system is described by an effective theory of combined classical-quantum (CQ) dynamics. We do this when we model a quantum measurement via the Born rule, since we treat the measurement device as classical. Similarly, in atomic physics and quantum chemistry, small molecules often interact with an environment, or thermal reservoir, which can be treated classically. In gravity, we would like to study the back-reaction of thermal radiation emitted from black holes. While the matter fields can be described by quantum field theory, practically, we only know how to treat space-time classically. Likewise, in cosmology, vacuum fluctuations are a quantum effect that gives rise to the primordial seeds sourcing galaxy formation. However, the expanding space-time they live in can only be treated classically.

The history of defining a consistent coupling between classical and quantum systems has

been controversial [17, 18]. When the quantum system does not back-react on the classical system, the situation is simple: the dynamics are described by unitary quantum mechanics, and the quantum state evolves according to a Hamiltonian H(z) that depends on classical degrees of freedom z. However, defining consistent dynamics where the quantum system back-reacts on the classical system has been more problematic.

The most familiar example of CQ back-reaction is semi-classical gravity. The standard approach to define back-reaction is via the semi-classical Einstein equations, which source the Einstein tensor $G_{\mu\nu}$ of the gravitational metric $g_{\mu\nu}$ by the expectation value of the stress-energy tensor $T_{\mu\nu}$ [19, 20]

$$G_{\mu\nu} = \frac{8\pi G}{c^4} \langle T_{\mu\nu} \rangle. \tag{1.1}$$

Though the scope and limitations of the semi-classical Einstein equations are not precisely understood [21, 22, 23], they are commonly understood to fail when fluctuations of the stressenergy tensor are large in comparison to its mean value [24, 25, 26, 21, 27]. The problem is that the standard semi-classical equations fail to properly account for correlations between the classical and quantum degrees of freedom. One often inputs this correlation by hand, considering situations when the quantum state is fully decohered and then evolving the classical system conditioned on the quantum state being in a particular eigenvalue. However, this does not give a dynamical description of the system before the quantum state has decohered, nor does it describe how the correlations between the classical and quantum systems are built up [28, 6]. The scenarios where quantum fluctuations are significant are often the regimes we hope to understand, such as in considering the gravitational field associated with Schrodinger cat states of massive bodies [29, 30], or vacuum fluctuations during inflation [31, 32, 33, 34]. For these regimes, background field methods are not appropriate, and an alternate effective theory of the back-reaction of quantum matter on classical gravity is required.

Moving away from effective theories, the lack of success in constructing a complete theory of quantum gravity valid beyond the Planck scale, combined with the lack of low energy signatures of quantum gravity, means the question of whether or not the gravitational field is quantum is still open for debate. The widespread belief is that gravity must be quantized, partly due to various no-go theorems surrounding consistent classical-quantum coupling. Feynman famously argued that a CQ coupling would prevent superpositions [17, 35], Epply and Hannah argued that it would lead to superluminal signalling [18], and various other arguments have been offered over the years [36, 37, 38, 39, 40, 41, 42, 43, 44, 45, 46, 47]. Meanwhile, it is well known that taking the semi-classical dynamics as fundamental is inconsistent and leads to violations of the standard principles of quantum theory, inducing a breakdown of either operational no-signaling, the Born rule, or composition of quantum systems under the tensor product [48, 18, 49, 50, 51].

Despite this, various attempts have been made to study a classical theory of gravity interacting with quantum systems, for example, by using a channel or measurement-based approach [52, 49, 53, 54, 55], which lead to linear dynamics on the quantum state. More generally, it is now known that these no-go theorems are circumvented by allowing for stochastic coupling between the classical and quantum degrees of freedom. By representing the combined CQstate as a distribution over the classical degrees of freedom, and density matrix at each point in phase space, [56, 57] introduced dynamics which are linear, completely positive, trace preserving, and preserve the split between classical and quantum degrees of freedom. These are necessary and sufficient conditions for the CQ-state to give positive probability outcomes for measurements. The evolution laws have been studied in various contexts [58, 59], including gravity [60, 52, 49, 61]. Recently, it was shown that the dynamics are special cases of the master equation derived in [28], which is the most general map governing consistent classicalquantum dynamics. If the dynamics are autonomous (time-local and completely positive at all times), one can write down the most general form of master equation. The dynamics are related to the GKSL or Lindblad equation [62, 63], which for bounded dynamics, is the most general autonomous dynamics for an open quantum system.

If gravity were fundamentally classical, then the assumptions that go into autonomous CQ dynamics are reasonable: the assumptions of complete-positivity and linearity are necessary for sensible predictions for all initial classical and quantum states; the assumption of autonomy is reasonable for any theory viewed as fundamental. Viewed this way, one expects that CQ master equations provide a template to construct consistent CQ theories of gravity. Such dynamics can probe the quantum nature of gravity in an alternate direction to current experimental proposals that have been made to measure low-energy gravitational phenomena that cannot be reproduced classically. Currently, the most promising experiments include those

which aim to detect gravitationally induced entanglement in table-top experiments via spin entanglement witnesses [64, 53, 1, 2, 65, 66, 67, 68]. There have also been proposals to measure intrinsically quantum features of gravity without studying entanglement directly [69, 70]. Though undoubtedly exciting, current estimates suggest that the technology required to perform the experiments is decades away. We use the classical-quantum formalism to consider the question from the opposite direction: can we construct a consistent fundamental classicalquantum theory of gravity? Can we find experimental signatures of such theories which can be used as indirect tests for the quantum nature of gravity? These are precisely the theories that experiments measuring gravitationally induced entanglement would rule out.

One might also expect classical-quantum dynamics to be useful as an effective theory. While the assumptions that go into autonomous CQ dynamics are reasonable for a fundamental theory, none of the assumptions need to hold, at least exactly. Nonetheless, exploring the autonomous CQ dynamics in the gravitational context is worthwhile as a starting point. It may be useful in certain regimes, but more importantly, it can be used to gain insight into the challenges that may arise when attempting to construct a more complete semi-classical description.

There are two main aims of this thesis. The first is to develop classical-quantum dynamics by providing formalism and tools to study effective theories of interacting classical and quantum systems. This thesis is primarily concerned with autonomous classical-quantum dynamics, and we do not attempt to understand when and how such a limit arises from quantum theory. Indeed in the general case, the effective classical-quantum description can be non-Markovian. Nonetheless, we hope that general lessons of classical-quantum dynamics can be used to gain insight into the effective regime. We leave a fuller understanding of taking the classical-quantum limit of quantum-quantum systems as an open question for future work.

Secondly, we develop classical-quantum theories of gravity. Though we expect such a theory could be applicable as an effective theory, we take a particular interest in developing theories of classical gravity that could be considered fundamental and studying the experimental signatures of such theories, which potentially lead to near-term indirect tests of quantum gravity. Moreover, classical-quantum dynamics provides a tractable arena to understand some of the problems which arise in theories of quantum gravity, such as diffeomorphism invariance and preservation of constraints [71, 72], which may not be solely due to treating the gravitational

field as quantum; we find similar issues in classical-quantum gravity and even purely classical probabilistic theories of gravity.

We now outline some of the main contributions of the thesis:

A classical-quantum Pawula theorem

The most general form of autonomous quantum dynamics is the Lindblad, or GKLS equation [63, 62]. Meanwhile, the most general form of classical dynamics continuous in the classical degrees of freedom is the Fokker-Planck equation [73]. Respectively, they are often used to describe the physics of quantum or classical systems interacting with an environment.

A natural question is: what are the most general dynamics that allow for the interaction of classical and quantum systems which are continuous in the classical degrees of freedom? This question is of interest not only from a foundational point of view but of practical interest since we often consider interacting systems where one can be or needs to be treated classically.

In Chapter 4, we find the general form of consistent autonomous continuous dynamics between classical and quantum systems. Our work enables one to study arbitrary continuous back-reactions and reduces to previously known master equations [57, 74] for a specific choice of Lindblad couplings. The master equation we derive can be viewed as a generalization of [75] applicable to *any* continuous classical-quantum dynamics. It extends a famous theorem in stochastic dynamics – the Pawula theorem [76] – to include quantum degrees of freedom.

This work was done in collaboration with Jonathan Oppenheim, Carlo Sparaciari, and Barbara Šoda [5].

Continuous unraveling of classical-quantum dynamics

In Chapter 5, we explicitly show that the continuous classical-quantum master equation derived in Chapter 4 can be unraveled by coupled stochastic differential equations with continuous trajectories. The resulting equations of motion are natural generalizations of the standard semi-classical equations of motion. However, because the resulting dynamics are linear in the combined classical-quantum state, it does not lead to the same pathologies - it accounts for correlations between the classical and quantum systems. In addition, despite a breakdown of predictability in the classical degrees of freedom, the quantum state evolves deterministically conditioned on the classical trajectory, provided a trade-off between decoherence and diffusion is saturated. As a result, the quantum state remains pure when conditioned on the classical trajectory.

This work was done in collaboration with Isaac Layton and Jonathan Oppenheim [6].

A path integral approach to classical-quantum dynamics

In Chapter 6, we derive a general path integral representation for classical-quantum dynamics. The path integral we derive is a generalization of the Feynman path integral for quantum systems and the stochastic path integral used to study classical stochastic processes, allowing for interaction between the classical and quantum systems. When the classical-quantum Hamiltonian is at most quadratic in the momenta, we derive a configuration space path integral. In Chapter 8, we study configuration space path integrals in more detail without resorting to master equation methods. These path integrals allow one to readily impose space-time symmetries, including Lorentz invariance or diffeomorphism invariance.

This work was done in collaboration with Jonathan Oppenheim [7].

Towards a diffeomorphism invariant theory of classical gravity

We develop classical-quantum theories of gravity. In the gravitational setting, by taking the classical degrees of freedom to live in the phase space of general relativity, the CQ dynamics describe a probability distribution over 4-geometries, each associated with a quantum state $(g_{\mu\nu}, \rho(g_{\mu\nu}))$. In Chapter 7, we provide a methodology for arriving at the analogs of the Hamiltonian and momentum constraints¹ in classical-quantum theories of gravity.

In Chapter 8 we use the path integral formulation of CQ dynamics to find the first examples of classical-quantum dynamics which are Lorentz and diffeomoprhism invariant, providing proof of principle that a complete theory could exist. We introduce a diffeomorphism invariant

¹Recall, in the ADM formulation of classical gravity [77, 78], the dynamics are generated by a Hamiltonian $H_{ADM} = \int d^3x N(x) \mathcal{H}(x) + N^a(x) \mathcal{H}_a(x)$ containing freely chosen "lapse" N(x), and "shift" $N^a(x)$ functions. In order to ensure that the dynamics do not depend on this choice, one must impose $\mathcal{H} \approx 0$, $\mathcal{H}_a \approx 0$, called the Hamiltonian and momentum constraint, respectively. The notation \approx indicates that the constraints are *weakly zero*, meaning they only vanish on a subset of the phase space - called the constraint surface.

theory based on the trace of Einstein's equations. We also introduce a path integral formulation of general relativity where the space-time metric is treated classically and we posit the generalizations of the Hamiltonian and momentum constraints which must be satisfied for the theory to be gauge invariant.

This work was done in collaboration with Jonathan Oppenheim [8, 3].

Generic predictions for Newtonian limit of classical-quantum gravity

In Chapter 9 we study the Newtonian limit of classical-quantum gravity and find a generic prediction of CQ theories: the Newtonian potential diffuses away from its classical solution by an amount that depends on the decoherence rate into mass eigenstates. In order for the dynamics to be completely positive, the amount of diffusion is necessarily lower bounded by the decoherence rate into mass eigenstates. This provides a way of testing CQ theories: one lower bounds the amount of diffusion the theory must have from coherence experiments, which can then be tested by measuring the noise in precision mass experiments. We explore the experimental consequences in detail in Chapter 10. We also show that the Newtonian limit we derived agrees with the Newtonian limit of the diffeomorphism invariant theory in Chapter 8, showing that it has a Newtonian limit which describes completely positive evolution on the subset of states satisfying the Newtonian gauge approximation, giving rise to the hope that the complete theory has constraints which can be preserved in time.

This work was done in collaboration with Jonathan Oppenheim and Andrea Russo [9].

A trade-off between decoherence and diffusion: indirect tests for the quantum nature of gravity

In Chapters 10, we prove that classical-quantum necessarily results in decoherence of the quantum system and a breakdown in predictability in the classical phase space. We further prove that a trade-off between the rate of this decoherence and the degree of diffusion induced in the classical system is a general feature of all classical quantum dynamics; long coherence times require strong diffusion in phase space relative to the strength of the coupling. Applying the trade-off relation to Newtonian gravity, we find a relationship between the strength of gravitationally-induced decoherence versus diffusion of the Newtonian potential and its conjugate momenta. The trade-off provides an experimental signature of theories in which gravity is fundamentally classical. Bounds on decoherence rates arising from current interferometry experiments, combined with precision measurements of mass, place significant restrictions on theories where Einstein's classical theory of gravity interacts with quantum matter. We find that part of the parameter space of such theories is already squeezed out. We provide figures of merit that can be used in future mass measurements and interference experiments.

This work was done in collaboration with Jonathan Oppenheim, Carlo Sparaciari, and Barbara Šoda [4].

Structure of the thesis

The thesis is split into a main body plus appendices. In the first part of the thesis (Chapters 2 and 3), we introduce the necessary background material on classical-quantum dynamics and gravity. The results presented in this section are known in the literature. The rest of the thesis is split into two parts and contains research work performed during my Ph.D. The first part – consisting of Chapters 4, 5, 6 – focuses on developing the CQ formalism, while the second part – consisting of the Chapters 7, 8, 9, and 10 – applies the CQ formalism to the gravitational setting. This work was done in collaboration with both internal and external researchers, all of whom contributed significantly to the ideas developed in this thesis. Other published work, completed during my Ph.D. with collaborators but not included in the thesis, includes [11, 12, 10].

In Chapter 2, we provide an introduction to stochastic classical dynamics, open quantum dynamics, and classical-quantum dynamics. This chapter aims to provide a reference point for concepts that are unfamiliar to the reader. We introduce the general form of autonomous CQ dynamics and its Kramers-Moyal expansion, which are central objects of study in the rest of the thesis. We also study Hamiltonian classical-quantum dynamics [28] and look at simple examples of CQ dynamics.

In Chapter 3, we provide the relevant gravity background required for the thesis. We review the initial value, or ADM formulation of General relativity, which is used in Chapter 7 to study the constraints in classical-quantum theories of gravity. We also discuss standard semi-classical approaches for incorporating quantum back-reaction on a classical gravitational field. In Chapter 4, we study the consequences of the dynamics being completely positive. We use this to determine the general form of continuous CQ dynamics via a CQ Pawula theorem. This chapter is based on the paper [5], a collaboration with Jonathan Oppenheim, Carlo Sparaciari, and Barbara Šoda.

In Chapter 5, we show how one can unravel the continuous master equation in terms of coupled stochastic differential equations, which provides a concrete algorithm for simulating continuous classical quantum dynamics which go beyond the standard semi-classical equations. We also show mathematical equivalence between the general form of the continuous master equation and continuous measurement with feedback. This chapter is based on the paper [6], a collaboration with Isaac Layton and Jonathan Oppenheim.

In Chapter 6, we derive a path integral representation for classical-quantum dynamics, which includes conditions on the couplings necessary for complete positivity and trace preservation. The path integral generalizes the Feynman path integral for quantum systems and the stochastic path integral used in classical stochastic processes. When the classical-quantum action is at most quadratic in the momenta, we show we can arrive at a configuration space path integral, which we study in more detail in Chapter 8. This chapter is based on the paper [7], a collaboration with Jonathan Oppenheim.

In Chapter 7, we take steps towards constructing a complete theory of CQ gravity. We provide a methodology to derive the constraint equations of a classical-quantum theory of gravity by imposing invariance of the dynamics under time-reparametrization invariance in a geometrodynamic picture. We find generalizations of the Hamiltonian and momentum constraints for classical-quantum dynamics, and we compute their algebra for the case of a quantum scalar field interacting with gravity. This chapter is based on the paper [3], a collaboration with Jonathan Oppenheim.

In Chapter 8, we study configuration space classical-quantum path integrals in more detail. We show they can be used to construct Lorentz and diffeomorphism invariant theories, proving that such dynamics exist. We introduce a path integral formulation of general relativity where the space-time metric is treated classically. We introduce diffeomorphism invariant theory based on the trace of Einstein's equations, and another more complete theory which gives rise to all of the components of Einstein's equations. We posit a general form of constraints for this theory which look promising, leaving it as a question for future work whether they give rise to consistent dynamics. This chapter is based on the paper [8], a collaboration with Jonathan Oppenheim.

In Chapter 9, we investigate the non-relativistic limit of classical-quantum gravity, where a classical Newtonian potential interacts with quantum matter. The theory can be viewed as a gauge fixed version of the theory introduced in Chapter 8, where we only consider the scalar degrees of freedom to linear order. The theory generalizes previous discussions on Newtonian classical-quantum gravity [60, 49, 52]. We arrive at a general form dynamics, which gives rise to Poisson's equation on average but also acts to diffuse around the classical solution with associated decoherence on the quantum system. In Chapter 10, we use the Newtonian limit introduced in this chapter to study experimental tests of classical-quantum gravity in the non-relativistic, weak field regime. This chapter is based on upcoming work [9], a collaboration with Jonathan Oppenheim and Andrea Russo.

In Chapter 10, we further explore the conditions of complete positivity, and we use this to arrive at a trade-off between decoherence and diffusion. Applying the trade-off to a model of a classical, non-relativistic Newtonian potential interacting with quantum matter, we find experimental restrictions on theories where classical gravity interacts with quantum matter. We find that part of the parameter space of such theories is already squeezed out. We provide figures of merit that can be used in future mass measurements and interference experiments. This chapter is based on the paper [4], a collaboration with Jonathan Oppenheim, Carlo Sparaciari, and Barbara Šoda.

The last part of the thesis contains the appendices. We provide minor results used in the main chapters and longer proofs and derivations, which, if added to the main body, would complicate the exposition and the presentation of more important results. Part I

Background

Chapter 2

Classical, quantum, and classical-quantum formalism

In this chapter, we introduce the background material on classical-quantum coupling required for the main body of the thesis. We first briefly overview both classical stochastic and quantum mechanics before discussing the classical-quantum formalism used in the rest of the thesis. When introducing the classical and quantum background material, we include snippet proofs of many concepts. We choose to do this because many of the classical-quantum formalism's introduced in the first half of the thesis will be extensions of their classical and quantum counterparts, often combining the classical and quantum concepts naturally; we hope this chapter can be a useful reference point. For example, in Chapter 4, we extend the classical Pawula theorem [76] to the CQ case, finding the most general form of continuous CQ dynamics, extending the Fokker-Plank and Lindblad equations. In Chapter 5, we find an unraveling for the general form of CQ dynamics; this extends and combines quantum unravelings of the Lindblad equations and unravelings of the Fokker-Plank equation by classical stochastic differential equations. While in Chapter 6, we find a path integral approach to CQ dynamics, which generalizes and combines the Feynman-Vernon path integral of open quantum systems [79] and the stochastic path integral of classical stochastic dynamics [80].

2.1 Stochastic classical dynamics

This section presents the basics of stochastic classical dynamics needed to understand the rest of the thesis. We do so in a simplified form by introducing the basic concepts required without going into the mathematical details of probability theory. We refer the reader to [73, 81, 82, 80] for a more detailed analysis.

2.1.1 The probability density and probability transition amplitude

We denote a generic classical degree of freedom by $z \in \mathcal{M}$, which could be a classical degree of freedom such as a position z = q or a point in phase space z = (q, p). When z is higher dimensional, we use d to refer to the dimension of the system and we denote the components as z_i . Because we are interested in stochastic classical dynamics, we let the classical degree of freedom be described by a random variable, denoted by a capital letter Z(t) (or sometimes Z_t). At each time, the stochastic variable $Z(t) \in \mathcal{M}$ takes on the possible values of the classical variable.

In stochastic classical dynamics, the basic object of interest is the *probability density* p(z,t), which is positive $p(z,t) \ge 0$ and normalized $\int dz \ p(z,t) = 1$. It is defined by

$$p(z,t) = \mathbb{E}[\delta(z - Z(t))], \qquad (2.1)$$

where the expectation value \mathbb{E} is an ensemble average over the stochastic variable. The probability of finding the stochastic variable Z(t) in the interval $z \leq Z(t) \leq z + dz$ at time t is then given by p(z,t)dz.

The probability density in Equation (2.1) can be extended to joint distributions. The probability of finding $z_1 \leq Z(t_1) \leq z_1 + dz_1$ and $z_2 \leq Z(t_2) \leq z_2 + dz_2$... and $z_n \leq Z(t_n) \leq z_n + dz_n$ is given by

$$P_n(z_n, t_n; z_{n-1}, t_{n-1} \dots; z_1, t_1) dz_1 \dots dz_n,$$
(2.2)

where

$$P_n(z_n, t_n; z_{n-1}, t_{n-1} \dots; z_1, t_1) = \mathbb{E}[\delta(z_n - Z(t_n)) \dots \delta(z_1 - Z(t_1))].$$
(2.3)

Knowing the hierarchy of joint probability distributions is equivalent to completely knowing the stochastic process Z(t).

Given the joint probability distribution, we can define the stochastic process's conditional probability, or *probability transition amplitude*. The transition amplitude describes the probability of finding the random variable Z(t) in a given interval $z_n \leq Z(t_n) \leq z_n + dz_n$ at time t_n , given it is in the interval $z_{n-1} \leq Z(t_{n-1}) \leq z_{n-1} + dz_{n-1}$ at t_{n-1} , in the interval $z_{n-2} \leq Z(t_{n-2}) \leq z_{n-2} + dz_{n-2}$ at t_{n-2} ..., and in the interval $z_1 \leq Z(t_1) \leq z_1 + dz_1$ at t_1 . Mathematically it is defined by

$$P(z_n, t_n | z_{n-1}, t_{n-1}; \dots; z_1, t_1) = \frac{P_n(z_n; t_n; \dots; z_1, t_1)}{P_{n-1}(z_{n-1}, t_{n-1}; \dots; z_1, t_1)} = \frac{P_n(z_n; t_n; \dots; z_1, t_1)}{\int dz_n P_n(z_n, t_n; \dots; z_1, t_1)}.$$
(2.4)

2.1.2 Markovian processes

We shall be interested in Markovian processes (i.e., stochastic processes that are time-local).

For a Markovian process

$$P(z_n, t_n | z_{n-1}, t_{n-1}; \dots; z_1, t_1) = P(z_n, t_n | z_{n-1}, t_{n-1}),$$
(2.5)

so that the conditional probability for finding Z(t) in the interval $z_n \leq Z(t_n) \leq z_n + dz$ depends only on the value of Z(t) at the previous time, as opposed to its entire history. In particular, this allows us to write the *Chapman-Kolmogorov equation* for the conditional probability

$$P(z_3, t_3|z_1, t_1) = \int dz_2 P(z_3, t_3|z_2, t_2) P(z_2, t_2|z_1, t_1), \qquad (2.6)$$

and knowledge of the conditional probability $P(z, t+\delta t|z', t)$ is enough to completely understand the process. Note, by definition, the conditional probability is positive and

$$\int dz P(z,t+\delta t|z',t) = 1$$
(2.7)

ensures probabilities are normalized.

2.1.3 Short time moment expansion and the Fokker-Plank equation

For Markovian dynamics, one can derive a master equation from the transition probability amplitude. In particular, we can introduce a *Kramers-Moyal* expansion [83, 84, 73] of the master equation, which is obtained via a short time moment expansion of the probability transition amplitude $P(z, t + \delta t | z', t)$

$$P(z, t + \delta t | z', t) = \delta(z, z') + \delta t W(z | z', t).$$
(2.8)

The moments of the transition amplitude are defined via

$$M_{n,i_1\dots i_n}(z',t,\delta t) = \int dz P(z,t+\delta t|z',t)(z-z')_{i_1}\dots(z-z')_{i_n},$$
(2.9)

and their short time expansion can be calculated as

$$M_{n,i_1...i_n}(z',t,\delta t) = \delta_n^0 + \delta t \int dz W(z|z',t)(z-z')_{i_1}...(z-z')_{i_n}$$

:= $\delta_n^0 + \delta t n! D(z')_{n,i_1...i_n},$ (2.10)

where we have defined the short-time expansion of the moments

$$D(z',t)_{n,i_1\dots i_n} = \frac{1}{n!} \int dz W(z|z',t)(z-z')_{i_1}\dots(z-z')_{i_n}, \qquad (2.11)$$

which define the moments of the Kramers-Moyal expansion. The subscripts $i_j \in \{1, \ldots, d\}$ label the different components of the vectors (z - z'). For example, in the case where d = 2 and the classical degrees of freedom are the position and momenta of a particle $z = (z_1, z_2) = (q, p)$, we have $(z - z') = (z_1 - z'_1, z_2 - z'_2) = (q - q', p - p')$. The components are then given by $(z - z')_1 = (q - q')$ and $(z - z')_2 = (p - p')$.

Sometimes in the literature, a distinction is made between processes that satisfy Equation (2.5) with a time-dependent transition amplitude and those which are time-independent. The former is often called time-dependent Markovianity [85], and we will also refer to it as an *autonomous process* [28]. For notational simplicity, we shall often suppress the potential explicit time dependence of the moments D(z',t) and the short-time transition amplitude W(z|z',t), which can always be added later. We also find it helpful to refer to the moments as $D_n(z')$, by which we mean the object with components $D(z')_{n,i_1...i_n}$.

To derive the master equation for autonomous processes, we first define the characteristic function, which is the Fourier transform of the transition amplitude

$$C(u, z', \delta t) = \int dz e^{iu \cdot (z - z')} P(z, t + \delta t | z', t) = \sum_{n=0}^{\infty} \frac{(i^n) u_{i_1} \dots u_{i_n}}{n!} M_{n, i_1 \dots i_n}(z', \delta t), \qquad (2.12)$$

In Equation (2.12), and throughout the thesis, we use the summation convention, so that contracted indices are assumed to be over. Taking the inverse Fourier transform, we can relate the transition amplitude to its moments

$$P(z,t+\delta t|z',t) = \int du \ e^{-iu(z-z')}C(u,z',\delta t)$$

= $\sum_{n=0}^{\infty} \frac{M_{n,i_1...i_n}(z',\delta t)}{n!} \frac{1}{(2\pi)^d} \int du \ e^{-iu(z-z')}(i^n)u_{i_1}...u_{i_n},$ (2.13)

which, using the definition of the delta function

$$\delta(z, z') = \frac{1}{(2\pi)^d} \int du e^{-iu(z-z')},$$
(2.14)

we can write as

$$P(z,t+\delta t|z',t) = \sum_{n=0}^{\infty} \frac{1}{n!} M_{n,i_1\dots i_n}(z',\delta t) \left(\frac{\partial^n}{\partial z'_{i_1}\dots \partial z'_{i_n}}\right) \delta(z,z').$$
(2.15)

Substituting the short time moment coefficients of Equation (2.11) back into (2.15), taking the limit $\delta t \rightarrow 0$, and using the probability preserving condition in (2.7), we can write the master equation in the form

$$\frac{\partial p(z,t)}{\partial t} = \sum_{n=1}^{\infty} (-1)^n \frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \left(D_{n,i_1\dots i_n}(z) p(z,t) \right).$$
(2.16)

Equation (2.16) is known as the Kolmogorov forward equation. The moments of the master equation can be related to useful physical quantities; for example, $D_{1,i}$ governs the evolution of $\frac{d\mathbb{E}[z_i]}{dt}$ and is associated with the *drift* in the system, while $D_{2,ij}$ characterizes the amount of *diffusion* in the system and computes the second moments $\frac{d\mathbb{E}[z_iz_j]}{dt}$.

2.1.4 Pauli rate equation

In the literature, it is common to find the master equation written in a slightly different form. Instead of defining the moments of the short-time probability amplitude, one considers

$$P(z,t+\delta t|z',t) = \delta(z,z')(1-\delta t\tilde{W}(z,t))\delta t\tilde{W}(z|z',t), \qquad (2.17)$$

where the norm condition in Equation (2.7) defines $\tilde{W}(z) = \int dz \tilde{W}(z|z')$. Equation (2.17) redefines the W(z|z',t) appearing in Equation (2.8) via $W(z|z') = \tilde{W}(z|z') - \delta(z,z')\tilde{W}(z)$.

When the expansion in Equation (2.17) is used, then by direct substitution of the short time moment expansion into (2.8), we arrive at the Pauli rate equation

$$\frac{\partial p}{\partial t} = \int dz' \tilde{W}(z|z') p(z') - \tilde{W}(z) p(z).$$
(2.18)

Note, the moments $D_n, n \ge 1$ for W are the same as for \tilde{W} since they only differ by their zeroth moment, which does not contribute to the master equation. Hence, the positivity condition $\delta(z, z') + \tilde{W}(z|z', t) \ge 0$ is still the necessary and sufficient condition to preserve the positivity of probabilities. The form of the master equation in Equation (2.18) is useful since it is positive and normalized for any positive $\tilde{W}(z|z')$.

2.1.5 Pawula theorem

An important theorem of Pawula [76] says that in order for the dynamics to preserve the positivity of the probability distribution, the Kramers-Moyal expansion must terminate at second order or contain infinitely many terms; specifically, none of the even moments D_{2n} can vanish. If it terminates at second order, it gives the well-known *Fokker-Plank equation*

$$\frac{\partial p(z,t)}{\partial t} = -\frac{\partial}{\partial z_i} \left[D_{1,i}(z,t)p(z,t) \right] + \frac{\partial^2}{\partial z_i \partial z_j} \left[D_{2,ij}(z,t)p(z,t) \right],$$
(2.19)

otherwise, the system undergoes finite size jumps with non-zero probability [73]. More precisely,

Pawula Theorem. The series of moments D_n^{00} , $n \ge 1$, appearing in the Kramers-Moyal expansion of (2.16) either contains infinitely many terms, or it truncates after second order, in which case we have a Fokker-Plank equation.

We generalize the Pawula theorem to include classical-quantum coupling in Chapter 4, finding the general form of continuous classical quantum dynamics. Therefore we provide proof of the classical Pawula theorem, which can be used as a reference.

Proof. We start with the generalized Cauchy-Schwartz inequality

$$\left[\int f(\Delta)g(\Delta)P(\Delta)d\Delta\right]^2 \le \int f^2(\Delta)P(\Delta)d\Delta \int g^2(\Delta)P(\Delta)d\Delta, \qquad (2.20)$$

which holds for any non-negative distribution $P(\Delta)$ and arbitrary real valued functions $f(\Delta), g(\Delta)$. Using Equation (2.20) with

$$P(\Delta) = P(z + \Delta, t + \delta t | z), \quad f(\Delta) = \Delta_{i_1} \dots \Delta_{i_n}, \quad g(\Delta) = \Delta_{i_{n+m}} \dots \Delta_{i_{2n+2m}}$$
(2.21)

gives the inequalities

$$(M_{2n+m,i_1\dots i_{2n+m}})^2 \le M_{2n,i_1i_1\dots i_ni_n} M_{2n+2m,i_{n+m}i_{n+m}\dots i_{2n+2m}i_{2n+2m}i_{2n+2m}},$$
(2.22)

where $M_{n,i_1...i_n}(z, \delta t)$ is defined in Equation (2.10). To prove the Pawula theorem, we first relate the coefficients $M_{n,i_1...i_n}(z, \delta t)$ to the short-time expansion coefficients which appear in the master equation. Recall, we have $M_{n,i_1...i_n}(z, \delta t) = \delta_n^0 + n! D_{n_{i_1,...i_n}}(z) \delta t + O(\delta t^2)$. We have to be a little careful since $M_n(z, \delta t) = O(\delta t)$ for $n \ge 1$ but O(1) for n = 0.¹ For n = m = 0 the inequality in (2.22) is trivially satisfied, while for $n = 0, m \ge 1$, we have no constraints on the short time expansion coefficients since the right-hand side of equation (2.22) is $O(\delta t)$ while the left-hand side is $O(\delta t^2)$. For $n \ge 1, m \ge 0$ we find

$$\left[(2n+m)!D_{2n+m,i_1\dots i_{2n+m}}\right]^2 \le (2n)!(2n+2m)!D_{2n,i_1i_1\dots i_ni_n}D_{2n+2m,i_{n+m}i_{n+m}\dots i_{2n+2m}i_{2n+2m}}.$$
(2.23)

Equation (2.23) is enough to derive the Pawula theorem, which tells us that if any even moment vanishes, then all moments with $n \ge 3$ must also vanish. To see this explicitly, observe if any even moment vanishes so that $D_{2n} = 0$, then $D_{2n+m} = 0$ for all m. Hence, if any even moment is zero, all higher-order moments must also vanish. Furthermore, if $D_{2n+2m} = 0$ then it can be seen from (2.23) that $D_{2n+m} = 0$. Denoting r = n + m, then this says $D_{2r} = 0$ implies $D_{r+n} = 0$ for $n = 1 \dots r - 1$. Hence if any even moment vanishes, $D_{2r} = 0$, we deduce all higher order moments D_{2r+n} must vanish, as well as the moments D_{r+n} for $n = 1 \dots r - 1$. Except for the case r = 1, r + n will always contain an even number, and so from repeated application of this property, we deduce D_n must vanish for $n \ge 3$.

2.1.6 Stochastic differential equations

A very useful fact is that Fokker-Plank equations can be identified with stochastic differential equations [73]

$$dZ_i(t) = \mu_i(Z(t), t)dt + \sigma_{ij}(Z(t), t)dW_j(t),$$
(2.24)

¹We use the simplifying notation $M_n(z, \delta t)$, which means the matrix with components $M_{n,i_1...i_n}(z, \delta t)$

where $W_i(t)$ is a standard d dimensional Wiener process satisfying the Ito rules

$$dW_i dW_j = \delta_{ij} dt, \ dW_i dt = 0. \tag{2.25}$$

Equation (2.24) is to be interpreted, formally, as an integral equation, and to get from Equation (2.24) to (2.19) we use the Ito definition of the stochastic integral [86]. Specifically, if a stochastic process Z(t) satisfies the stochastic differential equation in (2.24) then the probability density for $p(z,t) = \mathbb{E}[\delta(z-Z(t))]$ satisfies the Fokker-Plank equation (2.19), where $D_2 = \frac{1}{2}\sigma\sigma^T$ and $D_{1,i} = \mu_i$. If we instead use the Strananovich definition of a stochastic integral, then the Fokker-Plank equation can be identified with Equation (2.24) but with a redefinition of the drift vector μ_i [73].

To make the identification between (2.24) and (2.19), we start by noting that the dynamics of Z_t induces the following evolution on the probability density $p(z,t) = \mathbb{E}[\delta(z-Z(t))],$

$$dp(z,t) = \frac{\partial p(z,t)}{\partial t} dt = \mathbb{E}[d\delta(z - Z(t))].$$
(2.26)

Because of Ito's lemma, we must go to second order in dW to find the master equation. With this in mind we find

$$\mathbb{E}[d\delta(z-Z(t))] = \mathbb{E}[\frac{\partial}{\partial Z_i}[\delta(z-Z(t))]\mu_i(Z(t),t)]dt + \mathbb{E}[\frac{1}{2}\frac{\partial^2}{\partial Z_i\partial Z_j}[\delta(z-Z(t))]\sigma_{ik}(Z_t,t)\sigma_{kj}^T(Z_t,t)]dt.$$
(2.27)

We can use some well-known facts about the delta functional to simplify Equation (2.27). Using the two identities $\partial_{Z_i}\delta(z-Z) = -\partial_{z_i}\delta(z-Z)$ and $f(Z)\delta(z-Z) = f(z)\delta(z-Z)$ for any function f, the right hand side of Equation (2.27) becomes

$$-\frac{\partial}{\partial z_i} \mathbb{E}[\delta(z-Z(t))\mu_i(z,t)]dt + \frac{\partial^2}{\partial z_i\partial z_j} \mathbb{E}[\delta(z-Z(t))D_{2,ij}(z)]dt.$$
(2.28)

Using the definition of the probability density $p(z,t) = \mathbb{E}[\delta(z - Z(t))]$ and dividing by dt, we arrive at the Fokker-Plank equation of Equation (2.19).

2.1.7 Path integrals for stochastic classical dynamics

Another representation of classical dynamics is through classical stochastic path integrals. We derive the path integral for classical-quantum dynamics in Chapter 6. As a warm-up, in this
section, we sketch a derivation of the classical path integral associated with a 1 Dimensional Fokker-Plank equation

$$\frac{\partial p(z,t)}{\partial t} = -\frac{\partial}{\partial z} \left[D_1(z,t)p(z,t) \right] + \frac{\partial^2}{\partial z^2} \left[D_2(z,t)p(z,t) \right].$$
(2.29)

The more general case can be found in [80, 87] and in Chapter 6.

To arrive at the classical path integral, one starts from the moment expansion of the transition amplitude in Equation (2.13). In particular, this tells us that the transition amplitude at time $t + \delta t$ relates to that at t via

$$\varrho(z',t+\delta t) = \frac{1}{(2\pi)} \int dudz \ e^{-iu \cdot (z'-z)} \ (1+\delta t [iuD_1p(z,t)-u^2D_2]) \varrho(z,t)$$

$$= \frac{1}{(2\pi)} \int dudz \exp\left(-iu\frac{(z'-z)}{\delta t} \delta t + \delta t iuD_1 - \delta t u^2D_2\right) \varrho(z,t).$$
(2.30)

Equation (2.30) is enough to derive the path integral. In particular, we note that by considering many time steps and taking the limit $\delta t \to 0$ we arrive at the Fokker-Plank path integral with action

$$S_C[u,z] = \int dt [-iu\frac{dz}{dt} + iuD_1 - u^2D_2].$$
 (2.31)

Because the action is quadratic in u, it is also possible to integrate out the response variables u by performing standard Gaussian integrals to arrive at a path integral for the z variables alone. In this case, one finds the action (see Chapter 6)

$$S_C[z] = \int dt \left[-\frac{1}{4} D_2^{-1} (\frac{dz}{dt} - D_1)^2 \right].$$
 (2.32)

2.2 Quantum theory and open quantum systems

This section discusses the formalism of open quantum systems, which will be used throughout the thesis. We first review the standard rules of quantum theory for closed systems before discussing open quantum systems and their corresponding dynamics, including unraveling and path integral approaches to study open systems. This section takes much inspiration from [88], and we refer the reader to [88, 82] for a detailed analysis of open quantum systems.

2.2.1 Quantum theory for closed systems

In standard quantum theory for an isolated system, a quantum state is defined as a ray in a Hilbert space; a Hilbert space is a vector space over the complex numbers \mathbb{C} which has an inner product $\langle \psi | \phi \rangle$ and is complete in the norm $||\psi|| = \langle \psi \rangle^{1/2}$. A ray is an equivalence class of vectors that differ by multiplication by a non-zero complex scalar $|\psi\rangle \sim e^{i\alpha} |\psi\rangle$. For any nonzero ray, we choose a representative of the class $|\psi\rangle$ with unit norm. States evolve dynamically under unitary evolution $\psi(t)\rangle = U|\psi(0)\rangle$, with $UU^{\dagger} = U^{\dagger}U = \mathbb{I}$.

Observables are the properties of physical systems which can be measured. In classical mechanics, an observable is any functional over the classical phase space. In quantum mechanics, observables are self-adjoint operators $O = O^{\dagger}$ on the Hilbert space. Because observables are self-adjoint, they admit a spectral representation in terms of their eigenvectors $\{|i\rangle\}$, which form a complete orthonormal basis for the Hilbert space

$$O = \sum_{i} o_i |i\rangle \langle i| = \sum_{i} o_i E_i, \qquad (2.33)$$

where $E_i = |i\rangle\langle i|$ denotes the projection operator onto the eigenvector $|i\rangle$.

When an observable O is measured in the quantum state $|\psi\rangle$, the outcome prepares an eigenstate of O. Specifically, the outcome o_i occurs with probability $p_i = \langle \psi | E_i | \psi \rangle$ and after the measurement the quantum state is given by $\frac{E_i |\psi\rangle}{\langle \psi | E_i |\psi\rangle}$. If many identical systems $|\psi\rangle$ are prepared, we can calculate the expectation value of a quantum observable $\langle O \rangle = \sum_i o_i p_i = \langle \psi | O | \psi \rangle$.

In the standard axioms of quantum theory, the puzzling distinction between deterministic unitary dynamics determining the dynamics of the quantum state and non-unitary probabilistic evolution of quantum systems when they are being measured is called the measurement problem in quantum theory. There are many equivalent descriptions and refinements of the measurement problem. In Chapter 5, we find another. We show that we can give an alternative, effective dynamical description of measurement using the classical-quantum formalism. Instead of using the Born rule directly, one can posit that in an effective theory of quantum measurement, we should specify a classical degree of freedom that we monitor (i.e., the measurement device), which couples to a quantum system to be measured. The combined CQ evolution gives rise to dynamics equivalent to the Born rule. The quantum state collapses dynamically when inferences can be made about it by measuring and conditioning the classical system. The measurement problem is then mapped to the problem of how and when some degrees of freedom classicalize.²

Quantum systems compose under the tensor product. If the Hilbert space of system A is \mathcal{H}_A and the Hilbert space of system B is \mathcal{H}_B , then the Hilbert space of the composite systems AB is the tensor product $\mathcal{H}_A \otimes \mathcal{H}_B$. If system A is prepared in the state $|\psi\rangle_A$ and system B is prepared in the state $|\psi\rangle_A \otimes |\varphi\rangle_B$, then the composite system's state is the product $|\psi\rangle_A \otimes |\varphi\rangle_B$. When writing the tensor products of states, we will suppress the tensor product notation and write $|\psi\rangle_A |\varphi\rangle_B$.

Because composite systems are defined by the tensor product, there are some states in \mathcal{H}_{AB} which do not decompose as a product of states defined on \mathcal{H}_A and \mathcal{H}_B respectively. For example, for the composition of two, two-level Hilbert spaces spanned by $\{|0\rangle, |1\rangle\}_A$ and $\{|0\rangle, |1\rangle\}_B$ the state

$$|\psi\rangle = \frac{1}{\sqrt{2}} (|0\rangle_A |0\rangle_B + |1\rangle_A |1\rangle_B)$$
(2.34)

does not decompose into a product state on A, B. Such states are called *entangled states*.

2.2.2 Open quantum systems

Because of entanglement, in open quantum systems - when we limit our attention to a part of a larger system - a different formalism is required to express the outcome of any measurement performed on a subsystem.

For example, consider a Hilbert space of a bipartite system $\mathcal{H}_A \otimes \mathcal{H}_B$, and let $\{|i\rangle_A\}, \{|\mu\rangle_B\}$ denote an orthonormal basis for \mathcal{H}_A and \mathcal{H}_B respectively, so that any quantum state can be expanded as

$$|\psi\rangle_{AB} = \sum_{i,\mu} a_{i\mu} |i\rangle_A |\mu\rangle_B.$$
(2.35)

The expectation value of any observable on the system A alone $O_A \otimes \mathbb{I}_B$ is found via

<

$$O_A \rangle = \langle \psi | O_A \otimes \mathbb{I}_B | \psi \rangle$$

= $\sum_{i,j,\mu} \langle j | O_A | i \rangle = \operatorname{Tr}_A [O_A \rho_A],$ (2.36)

 $^{^{2}}$ If one was to make the controversial statement that some degrees of freedom are fundamentally classical, for example, the gravitational field, then, in a similar way to collapse models, this provides a potential route to solve the measurement problem, since there would be no need to include the Born rule, just the dynamics of interacting classical and quantum systems.

where we have defined

$$\rho_A = \operatorname{Tr}_B\left[\rho_{AB}\right],\tag{2.37}$$

with $\operatorname{Tr}_B[]$ the partial trace operation over B and $\rho_{AB} = |\psi\rangle\langle\psi|$. The partial trace operation is a linear map that takes an operator O_{AB} on $\mathcal{H}_A \otimes \mathcal{H}_B$ to an operator on \mathcal{H}_A alone, defined by

$$\operatorname{Tr}_{B}\left[O_{AB}\right] = \sum_{\mu} \langle \mu |_{B} O_{AB} | \mu \rangle_{B}.$$
(2.38)

Both ρ_{AB} and ρ_A define density matricies on \mathcal{H}_{AB} and \mathcal{H}_A respectively. These are positive $\langle v | \sigma | v \rangle \geq 0$ $v \in \mathcal{H}$, Hermitian $\sigma = \sigma^{\dagger}$ and normalized $\operatorname{Tr}_{\mathcal{H}}[\sigma] = 1$ operators on the Hilbert space. The density matrix for a *pure state* $|\psi\rangle$ is given by $\rho = |\psi\rangle\langle\psi|$, while density matrices on subsystems are obtained via the partial trace operation. We sometimes call density matrices for subsystems *reduced density matrices*.

Density matrices are used in open quantum systems to give an operational description for all outcomes on a subsystem A, even when considered part of a larger system. Because they are positive Hermitian operators, density matrices can always be diagonalized in an orthonormal basis, in which case they can always be written the form

$$\rho = \sum_{i} p_i |i\rangle \langle i|. \tag{2.39}$$

with $p_i \ge 0$.

2.2.3 Density matrices and ensembles

Not only do density matrices describe the outcomes of measurements on subsystems, but they also combine naturally when probabilistic mixtures of quantum states are involved.

First note that given any two density matrices ρ_1, ρ_2 , any convex combination $\rho(\lambda) = \lambda \rho_1 + (1 - \lambda)\rho_2$ is also a density matrix. In particular,

$$\langle O \rangle := \lambda \langle O \rangle_1 + (1 - \lambda) \langle O \rangle_2$$

= $\lambda \operatorname{Tr} [O\rho_1] + (1 - \lambda) \operatorname{Tr} [O\rho_2]$ (2.40)
= $\operatorname{Tr} [O\rho(\lambda)].$

Consequently, density matrices are consistent with an ensemble interpretation when probabilistic mixtures of quantum states are considered. Suppose the quantum states $\rho_i = |\psi_i\rangle_{AB} \langle \psi_i|_{AB}$ are each prepared with probability p_i , then the expectation value of any quantum observable is given by

$$\langle O \rangle = \sum_{i} p_{i} \operatorname{Tr} \left[\rho_{i} \right] = \operatorname{Tr} \left[\rho O \right],$$
(2.41)

where $\rho = \sum_{i} p_i \rho_i$ defines the density matrix for the entire system, and we can interpret $\rho = \sum_{i} p_i \rho_i$ as describing the quantum state where each ρ_i is prepared with probability p_i . The probabilistic interpretation of density matrices follows because the trace operation, which computes the expectation values of quantum observables, is linear.

Similarly, the expectation value for an observable on a subsystem A alone is given by

$$\langle O_A \rangle = \sum_i p_i \operatorname{Tr}_A \left[O \rho_{i,A} \right] = \operatorname{Tr}_A \left[O_A \rho_A \right]$$
(2.42)

where $\rho_{i,A} = \text{Tr}_B[\rho_i]$ is the reduced density matrix for each of the possible prepared states and $\rho_A = \sum_i p_i \rho_{i,A}$. In other words, the density matrix ρ_A gives an operational description where each of the quantum states ρ_i is prepared with the probability p_i . The probabilistic interpretation of reduced density matrices follows because the partial trace operation, used to compute the expectation values of quantum observables on subsystems, is also linear.

Because the density matrix provides a consistent formalism even when ensembles of quantum states are considered, we can write the density matrix for a quantum system after the measurement of an observable $O = \sum_i o_i E_i$ is performed. After the measurement of O, the quantum state will be given by $|\psi'\rangle_i = \frac{E_i|\psi\rangle}{\langle\psi|E_i|\psi\rangle}$ and occurs with probability $p_i = \langle\psi|E_i|\psi\rangle$. Expressing ρ as an ensemble of pure states, we see that the measurement modifies the state according to

$$\rho' = \sum_{i} p_i |\psi_i'\rangle \psi_i' = \sum_{i} E_i \rho_i E_i.$$
(2.43)

Generically density matrices can be decomposed in terms of other density matrices in many different ways, and the ensemble they describe is ambiguous without information about the entire system. The exceptions are the pure states $\rho = |\psi\rangle\langle\psi|$, which admit a unique decomposition and cannot be written as a convex combination of other density matrices. When the state is not pure, we call it *mixed*. It can be shown that a reduced density matrix on a subsystem S is pure if and only if $\text{Tr}_S \left[\rho_S^2\right] = 1$ and mixed if $\text{Tr}_S \left[\rho_S^2\right] < 1$. This can be used to characterize entangled states since the reduced density matrix for one of its subsystems must necessarily be mixed. Though the density matrix provides a complete operational description of the statistics of any measurement on the subsystem A, the ambiguity of its decomposition means mixed states do not convey the same physical state as when they are accompanied by information about operations on the B system.

A simple example is the maximally mixed state $\rho_A = \frac{1}{2}\mathbb{I}$, which can be decomposed as $\frac{1}{2}|0\rangle\langle 0| + \frac{1}{2}|1\rangle\langle 1|$ but also as $\frac{1}{2}|+\rangle\langle +| + \frac{1}{2}|-\rangle\langle -|$, where $|\pm\rangle = \frac{1}{\sqrt{2}}(|0\rangle \pm |1\rangle$: the density matrix gives the same operational description of the system whether the states $|0\rangle, |1\rangle$ are prepared with probability 1/2, whether $|+\rangle, |-\rangle$ are prepared with probability 1/2, or whether the system $|\psi\rangle_{AB} = \frac{1}{\sqrt{2}}(|00\rangle + |11\rangle)$ is prepared, and the experimenter only has access to part of the system.

When a state is a mixed state decomposed in its diagonal form as $\rho_A = \sum_i p_i |i\rangle \langle i|$ we say that it is in an *incoherent mixture* of the states $\{|i\rangle\}$ because the relative phases of the $|i\rangle$ states are experimentally inaccessible, which appear as off-diagonal terms in the density matrix. We see that when systems become entangled, the reduced state for the subsystem is mixed, and the entanglement destroys the coherence of the subsystem; some of the phases in the superposition become inaccessible when looking at the subsystem alone. When modeling an open quantum system dynamically, as we do in Section 2.2.5, this is represented by the *decoherence* of the off-diagonal terms in the density matrix. Intuitively, as a system undergoes evolution, it becomes entangled with the environment, which destroys the available coherence to the subsystem, represented by an exponential decay in its off-diagonal elements.

One can see more explicitly that density matrices can be realized in many different ways by considering the purification of density matrices. One can always *purify* a density matrix $\rho_A = \sum_i p_i |\varphi_i\rangle \langle \varphi_i |$, realizing it as a partial trace over a larger quantum system

$$|\phi\rangle_{AB} = \sum_{i} \sqrt{p_i} |\varphi_i\rangle_A |\alpha_i\rangle_B, \qquad (2.44)$$

$$\rho_A = \operatorname{Tr}_B\left[|\phi\rangle_{AB}\langle\phi|_{AB}\right]. \tag{2.45}$$

Given that a density matrix can always be purified, it is natural to ask if they can always be realized as a proper ensemble of quantum states by measurements on a purified system.

The fact that this can always be done is immediate from Equation (2.44). By measuring an observable which projects into the $|\alpha_i\rangle_B$ basis on system *B*, the post measurement state for $|\phi\rangle_{AB}$ will be given by $|\phi'\rangle_{AB} = |\varphi_i\rangle_A |\alpha_i\rangle_B$ with probability p_i , which prepares $\rho_A =$ $\sum_i p_i |\varphi_i\rangle \langle \varphi_i|$ as a proper probabilistic ensemble of these states.

Similarly, given another decomposition of the same density matrix into pure states $\rho_A = \sum_i p'_i |q_i\rangle\langle q_i|$, we can always purify it in the form

$$|\phi_2\rangle_{AB} = \sum_i \sqrt{p_i'} |q_i\rangle_A |\beta_i\rangle_B, \qquad (2.46)$$

and realize ρ_A as a proper ensemble of $|q_i\rangle$ by orthonormal measurements in the basis $|\beta_i\rangle$ on the second system. Using the Schmidt decomposition, one can show that any two purifications of the density matrix are related by a unitary transformation on the enlarged system [89, 48]. That is

$$|\phi\rangle_{AB} = (\mathbb{I} \otimes U_B) |\phi_2\rangle_{AB}.$$
(2.47)

We see a complete operational description of the statistics of any measurement on the subsystem A, which is not the same physical state as ρ_A , accompanied with information about operations on the B system. In particular, though measurements on the B system do not appear in any change in the operational description of ρ_A , and hence cannot be used to superluminally signal, it is clear that measurements on the B system do alter the physical state of the system in a non-local way. The fact that the correlations implied by the quantum theory are not reproducible by a local physical model is the content of Bell's theorem [90].

2.2.4 Quantum channels

In open quantum systems, dynamics are no longer unitary but described by a map from density operators to density operators. These are called *quantum channels*, which are completely positive, trace preserving (CPTP), and linear map on the space on density matrices [82, 88].

Mathematically, a quantum channel is

- Linear: $\Phi(\lambda \rho_1 + (1 \lambda)\rho_2 = \lambda \Phi(\rho_1) + (1 \lambda)\Phi(\rho_2)).$
- Hermiticity preserving: $\rho = \rho^{\dagger}$ implies $\Phi(\rho) = \Phi(\rho)^{\dagger}$.
- Trace preserving: $\operatorname{Tr} [\rho] = \operatorname{Tr} [\Phi(\rho)].$
- Completely positive: $\rho \ge 0$ implies $\Phi \otimes \mathbb{I}_n(\rho) \ge 0$ for all n.

Complete positivity (CP) $\rho \ge 0$ implies $\Phi(\rho) \ge 0$ and is required if we want the channel to be positive, even if we consider it to act on just part of a larger system. This is required if we want the channel to give positive probabilities even when acting on part of an entangled state.

Linearity is required to maintain the ensemble interpretation of density matrices. In particular, suppose we consider preparing an ensemble of states with the density operator $\rho = \sum_{i} p_{i}\rho_{i}$. Then the final state after applying the quantum channel is

$$\rho' = \Phi(\sum_{i} p_i \rho_i). \tag{2.48}$$

Conversely, the final state should be found by applying the quantum channel to each state ρ_i and then averaging $\sum_i p_i \Phi(\rho_i)$. Equating these two is equivalent to asking the quantum channel to be linear. The trace and Hermitian preserving properties are required for the system to describe density matrices with normalized probability outcomes.

Quantum channels are fully characterized by the Kraus theorem [91], which states that any quantum channel can always be written in the form

$$\rho' = \Phi(\rho) = \sum_{\mu\nu} \Lambda^{\mu\nu} K_{\mu} \rho K_{\nu}^{\dagger}, \qquad (2.49)$$

where $\Lambda^{\mu\nu}$ is Hermitian, positive matrix $a^*_{\mu}\Lambda^{\mu\nu}a_{\nu} \ge 0$ for any vector a_{μ} . The K_{μ} are known as *Kraus operators* and can always be taken to describe an orthogonal set of operators on the Hilbert space. We call the matrix $\Lambda^{\mu\nu}$ the *Kraus matrix*. The preservation of trace enforces the normalization condition

$$\sum_{\mu\nu} \Lambda^{\mu\nu} K^{\dagger}_{\nu} K_{\mu} = \mathbb{I}.$$
(2.50)

2.2.5 The Lindblad (GKSL) equation

We will be interested in discussing the dynamics of time-local classical-quantum systems, so we review time-local quantum channels in this section. When the dynamics are time-local, the state at time $t + \delta t$ can be written in terms of the state at time t alone, and a master equation can be derived when the Kraus operators are taken to be bounded and trace-class. When considering the dynamics of quantum systems, we consider quantum channels $\rho(t) = \Phi_t(\rho(0))$ and allow t to vary $0 \le t \le T$ which defines a one-parameter family of dynamical CPTP maps Φ_t .

2.2.6 Time local dynamics

A family of dynamical maps Φ_t is said to be *time-local* if $\dot{\rho}_t = \mathcal{L}_t(\rho_t)$ for a linear map \mathcal{L}_t . An important feature of time-local dynamics is that they can always be written in Lindblad form [92]

$$\frac{\partial \rho_t}{\partial t} = -i[H, \rho_t] + \lambda^{\alpha\beta}(t) \left(L_\alpha \rho_t L_\beta^\dagger - \frac{1}{2} \{ L_\beta^\dagger L_\alpha, \rho_t \} \right), \qquad (2.51)$$

where the matrix $\lambda^{\alpha\beta}(t)$ is Hermitian, but in general, the conditions for complete positivity are not known [92, 93]. In Equation (2.51), we again use the summation convention, so that contracted $\alpha\beta$ are assumed to be over. The operators L_{α} appearing in Equation (2.51) are called *Lindblad operators* and the matrix $\lambda^{\alpha\beta}$ we call the *Lindblad coupling*. Though Φ_t is a completely positive map on initial states for all times t, the generator \mathcal{L}_t need not generate CP dynamics for $0 < t \leq T$: one can consider scenarios where the initial quantum state decoheres at early times, but recoeheres at late times keeping the total dynamics Φ_t completely positive [93]. In other words, \mathcal{L}_t needs only generate CP dynamics on the subset of states that are reachable at time t, { $\sigma_t : \exists \rho_0 \text{ s.t } \sigma_t = \Phi_t(\rho_0)$ }. For example, suppose that for $0 \leq t \leq T/2$ the map Φ_t acts as the perfectly depolarizing channel, sending any density matrix to the identity. We can then construct a completely positive dynamics Φ_t for $0 \leq t \leq T$ by concatenating the depolarizing channel with any map (not necessarily CP) that preserves the identity (i.e., a unital map). The resulting dynamics will be CP on all initial states, but for t > T/2 the generator of the dynamics \mathcal{L}_t doesn't need to be completely positive on all states.

Markovian and autonomous dynamics

When the coefficients $\lambda^{\alpha\beta}$ are time-independent positive matrices $a^*_{\alpha}\lambda^{\alpha\beta}a_{\beta} \geq 0$ for any a_{α} , then Equation (2.51) is the well-known Lindblad (or GKSL) equation familiar in open quantum systems [63, 62]. We call such dynamics *time-independent Markovian*. The Lindblad equation represents the most general form of allowed time-local dynamics when one also demands that the generator $\mathcal{L}_t = \mathcal{L}$ be time independent [63, 62].

When the generator is time-independent, $\mathcal{L}_t = \mathcal{L}$, it generates completely positive dynamics on *all* states. This leads to an alternative but equivalent definition of quantum Markovianity, which extends naturally to when $\lambda^{\alpha\beta}$ are time-dependent. In particular, we say dynamics are autonomous if \mathcal{L}_t generates CPTP dynamics on all states. In this case, the general form of master Equation is given by Equation (2.51) where the coefficients $\lambda^{\alpha\beta}(t)$ are positive but can now be time-dependent. When the generator is time-independent, autonomy reduces to time-independent Markovianity, while when the generator is time-dependent but CPTP on all states, the dynamics have also been coined *time-dependent Markovianity* [85, 94, 92].

Deriving the master equation for autonomous dynamics

Because we are interested in autonomous CQ dynamics and the associated CQ master equation, it is helpful to include a derivation of the associated master equation, which is given by Equation (2.51) with time-dependent coefficients $\lambda^{\alpha\beta}(t)$ which are positive Hermitian matrices.

We start by performing a short-time expansion of the Kraus form in Equation (2.49). For autonomous dynamics, this always takes the form

$$\rho'(t+\delta t) = \sum_{\mu\nu} \Lambda^{\mu\nu}(t,\delta t) K_{\mu}\rho(t) K_{\nu}^{\dagger}$$
(2.52)

where $\Lambda^{\mu\nu}(t, \delta t)$ is a positive matrix for all times.

Since we can choose the basis of Kraus operators to be an arbitrary basis on the Hilbert space, for the derivation, we take them to be $K_{\mu} = (\mathbb{I}, L_{\alpha})$ and assume they are bounded and trace-class. Since for $\delta t = 0$, we know the dynamical map reduces to the identity,

$$\Lambda^{\mu\nu}(t,\delta t) = \delta^{\mu}_{0}\delta^{\nu}_{0} + \delta t\lambda^{\mu\nu}(t), \qquad (2.53)$$

we find

$$\rho(t+\delta t) = \sum_{\mu\nu} \Lambda^{\mu\nu}(t) L_{\mu}\rho(t) L_{\nu}^{\dagger} = \rho + dt \sum_{\mu\nu} \lambda^{\mu\nu}(t) L_{\mu}\rho(t) L_{\nu}^{\dagger}$$

$$= \rho(t) + dt [\lambda^{00}\rho + \lambda^{0\alpha}\rho(t) L_{\alpha}^{\dagger} + \lambda^{\alpha0} L_{\alpha}\rho + \lambda^{\alpha\beta} L_{\alpha}\rho(t) L_{\beta}^{\dagger}], \qquad (2.54)$$

where positivity demands that the matrix

$$\Lambda^{\mu\nu}(t) = \begin{bmatrix} 1 + \delta t \lambda^{00} & \delta t \lambda^{0\beta} \\ \delta t \lambda^{\alpha 0} & \delta t \lambda^{\alpha\beta} \end{bmatrix} + O(\delta t^2), \qquad (2.55)$$

is positive. Explicitly, for any vector a_{μ} we have

$$a^*_{\mu}\Lambda^{\mu\nu}a_{\mu} = |a_0|^2 + \delta t [a^*_0\lambda^{0\alpha}a_{\alpha} + a^*_{\alpha}\lambda^{\alpha 0}a_0 + a^{\dagger}_{\beta}\lambda^{\alpha\beta}a_{\alpha}] \ge 0.$$
(2.56)

By choosing $a_0 = 0$, Equation (2.56) imposes that $\lambda^{\alpha\beta}(t)$ is a positive matrix in α, β . However, for any choice of a_{μ} , there are no additional restrictions on the off-diagonals $\lambda^{0\beta}$ since they are $O(\delta t)$ and in the limit $\delta t \to 0$ the dominant contribution arises from the Λ^{00} component which is of order O(1), enforcing positivity.

To arrive at the Lindblad form of Equation (2.51), note that the norm condition imposes that

$$\sum_{\mu\nu} \Lambda^{\mu\nu} L^{\dagger}_{\nu} L_{\mu} = \mathbb{I},$$

$$\lambda^{00} \mathbb{I} + \lambda^{0\alpha} L_{\alpha} + \lambda^{\alpha 0} L^{\dagger}_{\alpha} + \lambda^{\alpha \beta} L_{\beta} L_{\alpha} = 0,$$
(2.57)

so substituting for λ^{00} and taking the $dt \to 0$ limit gives us the GKSL or Lindblad equation [63, 62]

$$\frac{\partial \rho}{\partial t} = -i[H,\rho] + \lambda^{\alpha\beta} L_{\alpha} \rho L_{\beta}^{\dagger} - \frac{1}{2} \lambda^{\alpha\beta} \{ L_{\beta}^{\dagger} L_{\alpha}, \rho \}, \qquad (2.58)$$

where we have defined the Hamiltonian $H = \frac{i}{2}(\lambda^{\alpha 0}L_{\alpha} - \lambda^{0\alpha}L_{\alpha}^{\dagger})$. In Equation (2.58), the coefficients $\lambda^{\alpha\beta}(t)$ can be time-dependent but must be positive matrices in $\alpha\beta$. This concludes the proof that any autonomous dynamics can be brought to the form of Equation (2.58) for a specific choice of Lindblad operators (\mathbb{I}, L_{α}) and it is a well-known result [82, 95] that Equation (2.58) for arbitrary Lindblad operators L_{α} also defines autonomous dynamics. We will often write the anti-commutator $\{,\}$ as $\{,\}_+$, to distinguish it from the Poisson bracket appearing in Hamiltonian classical mechanics.

2.2.7 Decoherence

When considering an open quantum system according to the dynamics in Equation (2.58), the system generically undergoes decoherence due to its interaction with the environment. Intuitively, the system becomes entangled with its environment, destroying the subsystem's available coherence. This behavior is most easily seen when the Lindbladian operators are Hermitian. For example, consider the master equation

$$\frac{\partial \rho}{\partial t} = -i[H,\rho] + \lambda L \rho L^{\dagger} - \frac{1}{2} \lambda \{L^{\dagger}L,\rho\}, \qquad (2.59)$$

for a Hermitian operator $L = L^{\dagger}$. Because L is Hermitian, it can be diagonalized, and we can choose to decompose ρ in terms of its eigenvectors $\rho = \sum_{ij} \rho_{ij} |i\rangle \langle j|$. The action of the

Lindbladian term is to suppress the off-diagonal elements of the density matrix

$$\frac{\partial \rho_{ij}}{dt} \approx -\frac{\lambda}{2} (L_i - L_j)^2 \rho_{ij}.$$
(2.60)

2.2.8 Unraveling of Lindblad equations

In analogy with the unraveling of the Fokker-Plank equation by stochastic differential equations, the Lindblad equation can be unraveled by a set of stochastic differential equations. However, unlike the classical case, where the trajectories are objective, because of the ambiguities in the ensemble interpretation of the density matrix, the unravelings of the master equation are not unique, unless extra information about the state of the environment is provided. Nonetheless, they can be valuable tools in simulating open quantum systems [82].

One commonly found unraveling [96] is given by the stochastic differential equation for the pure quantum state

$$d|\psi\rangle_{t} = -iH|\psi\rangle_{t}dt$$

$$-\frac{1}{2}\lambda^{\alpha\beta}(L_{\beta}^{\dagger} - \langle L_{\beta}^{\dagger}\rangle)(L_{\alpha} - \langle L_{\alpha}\rangle)|\psi\rangle_{t}dt + \frac{1}{2}\lambda^{\alpha\beta}(\langle L_{\beta}^{\dagger}\rangle L_{\alpha} - \langle L_{\alpha}\rangle L_{\beta}^{\dagger})|\psi\rangle_{t}dt \qquad (2.61)$$

$$+ (\lambda^{1/2})^{\alpha\beta}(L_{\alpha} - \langle L_{\alpha}\rangle)|\psi\rangle d\xi_{\beta},$$

where $\lambda^{1/2}\lambda^{1/2\dagger} = \lambda$ is the square route of the positive Hermitian matrix λ . In Equation (5.9), the noise process is a complex noise

$$d\xi_{\alpha}d\xi_{\beta} = 0, \ d\xi_{\alpha}d\xi_{\beta}^* = \delta_{\alpha\beta}, \tag{2.62}$$

and the density matrix is found by averaging over all possible realizations of the quantum state

$$\rho(t) = \mathbb{E}[|\psi(t)\rangle\langle\psi(t)|]. \tag{2.63}$$

To see the unraveling in Equation (2.61) gives the Lindblad equation when averaged over trajectories, first note that the evolution of the density matrix takes the form

$$d\rho = \mathbb{E}[|d\psi(t)\rangle\langle\psi| + |\psi(t)\rangle\langle d\psi(t)| + |d\psi(t)\rangle\langle d\psi(t)|]$$
(2.64)

where we have to go to second order because $d\xi d\xi^* = O(dt)$.

We can compute each term individually,

$$\mathbb{E}[|d\psi(t)\rangle\langle\psi|] = [-iH\rho - \frac{1}{2}\lambda^{\alpha\beta}L^{\dagger}_{\beta}L_{\alpha}\rho + \lambda^{\alpha\beta}\langle L^{\dagger}_{\beta}\rangle L_{\alpha}\rho - \frac{1}{2}\lambda^{\alpha\beta}\langle L^{\dagger}_{\beta}\rangle\langle L_{\alpha}\rangle\rho]dt$$
(2.65)

$$\mathbb{E}[|\psi(t)\rangle\langle d\psi|] = [i\rho H - \frac{1}{2}\lambda^{\alpha\beta}\rho L^{\dagger}_{\beta}L_{\alpha} + \lambda^{\alpha\beta}\rho L^{\dagger}_{\beta}\langle L_{\alpha}\rangle + \frac{1}{2}\lambda^{\alpha\beta}\rho\langle L^{\dagger}_{\beta}\rangle\langle L_{\alpha}\rangle]dt$$
(2.66)

$$\mathbb{E}[d|\psi(t)\rangle\langle d\psi|] = [\lambda^{\alpha\beta}L_{\alpha}\rho L_{\beta}^{\dagger} - \lambda^{\alpha\beta}\rho L_{\beta}^{\dagger}\langle L_{\alpha}\rangle - \lambda^{\alpha\beta}\langle L_{\beta}^{\dagger}\rangle L_{\alpha}\rho + \lambda^{\alpha\beta}\langle L_{\beta}^{\dagger}\rangle\langle L_{\alpha}\rangle\rho]dt, \qquad (2.67)$$

and summing the contributions we find the Lindblad equation equation.

If one also includes information about the environment, one can arrive at unravelings that lead to pure state quantum evolution, which is objective, conditioned on the environment's history. Objective unravelings can be seen in [96], who interpret the Lindblad equation as a continuous measurement of an environment. We also see this more generally in Chapter 5, where we find the general form of unraveling for continuous classical-quantum dynamics; we find that, though the evolution of the quantum and classical states are stochastic, if one conditions on the entire classical trajectory the evolution of the pure quantum state is uniquely determined.

2.2.9 Path integrals for open quantum systems

In this section, we introduce the path integral formalism for open quantum systems (see [97] for an excellent review). We combine this with the path integral for stochastic classical mechanics in Chapter 6 to arrive at a combined classical-quantum path integral [8, 7].

It is well known in quantum mechanics that for a Hamiltonian system $H = \frac{p^2}{2m} + V(x)$ one can write a path integral governing the amplitude of the quantum state

$$\psi(x_f, t_f) = \int^{x(t_f) = x_f} \mathcal{D}x e^{iS[x]} \psi(x, t_i), \qquad (2.68)$$

where $\psi(t,x) = \langle x|\psi(t)\rangle$ and $S[x] = \int dt [m\frac{\dot{x}^2}{2} - V(x)] = \int dt L[x]$, with L[x] the Lagrangian of the system. Equation (2.68) can also be used to write a density matrix path integral

$$\psi(x_f^+, t_f)\psi^*(x_f^-, t_f) = \int^{x^+(t_f)=x_f^+, x^-(t_f)=x_f^-} \mathcal{D}x^- \mathcal{D}x^+ e^{iS[x^+]-iS[x^-]}\psi(x^+, t_i)\psi^*(x^-, t_i), \quad (2.69)$$

where x^+ are associated with the ket variables and x^- to the bra variables.

Similarly, we can derive a path integral representation for the Lindblad equation in Equation (2.51). To do so, we first expand the density matrix

$$\rho(t) = \int dx^+ dx^- \rho(x^+, x^-, t) |x^+\rangle \langle x^-|, \qquad (2.70)$$

and to compute the path integral, we need to evaluate

$$\rho(t+\delta t) = \rho(t) + \delta t \mathcal{L}(\rho(t)). \tag{2.71}$$

The fundamental object of interest is given by relating the components $\rho(y^+, y^-, t + \delta t)$ to those at time t, $\rho(x^+, x^-, t)$

$$\rho(y^+, y^-, t + \delta t) = \rho(x^+, x^-, t) + \delta t \langle y^+ | \mathcal{L}(\rho) | y^- \rangle.$$
(2.72)

Because the Hamiltonian can depend on momentum, we will compute $\langle y^+ | \mathcal{L}(\mathbb{I}_+ \rho \mathbb{I}_+) | y^- \rangle$ and insert resolutions of the identity in the momentum basis $\mathbb{I}_{\pm} = \int dp^{\pm} | p^{\pm} \rangle \langle p^{\pm} |$. We find

$$\rho(y^{+}, y^{-}, t + \delta t) = \int dx^{+} dx^{-} dp^{+} dp^{-} \langle y^{+} | p^{+} \rangle \langle p^{+} | x^{+} \rangle \langle x^{-} | p^{-} \rangle \langle p^{-} | y^{-} \rangle \rho(x^{+}, x^{-}, t) \times$$

$$[1 + \delta t(-iH(p^{+}, x^{+}) + iH(p^{-}, x^{-})$$

$$+ \lambda^{\alpha\beta} L_{\alpha}(x^{+}) L_{\beta}(x^{-}) - \frac{1}{2} \lambda^{\alpha\beta} L_{\beta}^{*}(x^{+}) L_{\alpha}(x^{+}) - \frac{1}{2} \lambda^{\alpha\beta} L_{\beta}^{*}(x^{-}) L_{\alpha}(x^{-}))],$$

$$(2.73)$$

which gives the path integral

$$\rho(y^{+}, y^{-}, t + \delta t) = \int dx^{+} dx^{-} dp^{+} dp^{-} \exp\left(i\frac{y^{+} - x^{+}}{\delta t}\delta t - i\frac{y^{-} - x^{-}}{\delta t}\delta t + -iH(p^{+}, x^{+}) + iH(p^{-}, x^{-})\right) + \lambda^{\alpha\beta}L_{\alpha}(x^{+})L_{\beta}(x^{-}) - \frac{1}{2}\lambda^{\alpha\beta}L_{\beta}^{*}(x^{+})L_{\alpha}(x^{+}) - \frac{1}{2}\lambda^{\alpha\beta}L_{\beta}^{*}(x^{-})L_{\alpha}(x^{-}))\right)\rho(x^{+}, x^{-}, t).$$

$$(2.74)$$

Taking the $\delta t \to 0$ limit, and considering many time intervals, we find (after skipping a few steps [97]) a path integral with action

$$iS[x^{+}, x^{-}, p^{+}, p^{-}] = \int dt \left[i(\dot{x^{+}}p^{+} - H[p^{+}, x^{+}]) - i(\dot{x^{-}}p^{-} - H[p^{-}, x^{-}]) \right]$$

$$\lambda^{\alpha\beta}L_{\alpha}(x^{+})L_{\beta}(x^{-}) - \frac{1}{2}\lambda^{\alpha\beta}L_{\beta}^{*}(x^{+})L_{\alpha}(x^{+}) - \frac{1}{2}\lambda^{\alpha\beta}L_{\beta}^{*}(x^{-})L_{\alpha}(x^{-}) \right].$$
(2.75)

Note, we can also integrate out the momentum variables, the result of which is to perform the same Legendre transformation as in standard quantum mechanics, in which case we arrive at a path integral with action

$$iS[x^{+}, x^{-}] = \int dt \left[iL[x^{+}] - iL[x^{-}] + \lambda^{\alpha\beta}L_{\alpha}(x^{+})L_{\beta}(x^{-}) - \frac{1}{2}\lambda^{\alpha\beta}L_{\beta}^{*}(x^{+})L_{\alpha}(x^{+}) - \frac{1}{2}\lambda^{\alpha\beta}L_{\beta}^{*}(x^{-})L_{\alpha}(x^{-}) \right].$$
(2.76)

The term

$$iS_{FV} = \int dt \lambda^{\alpha\beta} L_{\alpha}(x^{+}) L_{\beta}(x^{-}) - \frac{1}{2} \lambda^{\alpha\beta} L_{\beta}^{*}(x^{+}) L_{\alpha}(x^{+}) - \frac{1}{2} \lambda^{\alpha\beta} L_{\beta}^{*}(x^{-}) L_{\alpha}(x^{-}) \bigg]$$
(2.77)

is known as the *Feynman-Vernon action* [79] and incorporates Lindbladian evolution into the path integral.

As a simple example, take L(x) = x, then the action of the Feynman-Vernon term is

$$iS_{FV} = -\int dt \; \frac{1}{2}\lambda (x^+ - x^-)^2, \qquad (2.78)$$

and we see the path integral suppresses paths away from values where $x^+ \neq x^-$. i.,e, it causes the off-diagonal elements of the density matrix to be exponentially suppressed, causing decoherence of the quantum system.

2.3 Classical-quantum dynamics

We now introduce the general formalism used to describe a classical degree of freedom coupled to a quantum one, and we denote a generic classical degree of freedom by z. For example, it could be a classically treated position variable z = q or a point in phase space z = (q, p).

When considering a hybrid system, the natural set of states to consider are hybrid classicalquantum (CQ) states. Formally, a classical-quantum state associates to each classical variable an un-normalized density matrix $\varrho(z,t) = p(z,t)\sigma(z,t)$ such that $\operatorname{Tr}_{\mathcal{H}}[\varrho(z)] = p(z,t) \geq 0$ is a normalized probability distribution over the classical degrees of freedom and $\int dz \varrho(z,t)$ is a normalized density operator on a Hilbert space \mathcal{H} . Intuitively, p(z,t) can be understood as the probability density of being in the phase space point z and $\sigma(z,t)$ as the *normalized* quantum state one would have given the classical state z occurs. This is consistent with the ensemble interpretation of density matrices given in Section 2.2.3.

An example of such a *CQ-state* is the CQ qubit, where we take a 2 dimensional Hilbert space and couple to classical position and momenta. The state then takes the form of a 2×2

matrix over phase space

$$\varrho(q, p, t) = \begin{pmatrix} u_0(q, p, t) & c(q, p, t) \\ c^*(q, p, t) & u_1(q, p, t) \end{pmatrix}.$$
(2.79)

We study the CQ qubit example in Section 2.3.3. Moving away from states, we can define any CQ operator f(z) which associates a quantum operator to each point in phase space.

In analogy with the Kraus theorem for quantum mechanics and the probability transition equation for classical mechanics, it has been shown [28, 98] that any dynamics mapping CQ states onto themselves, if taken to be linear, will be completely positive if and only if it can be written in the form

$$\varrho(z,t_f) = \int dz' \Lambda(z,t_f|z',t_i)(\varrho(z',t)) = \int dz' \sum_{\mu\nu} \Lambda^{\mu\nu}(z,t_f|z',t_i) K_{\mu}\varrho(z',t_i) K_{\nu}^{\dagger}, \qquad (2.80)$$

where $\Lambda^{\mu\nu}(z, t_f | z', t_i)$ is a completely positive Hermitian matrix kernel in $\mu\nu$ for each z, z', and the K_{μ} are an arbitrary orthogonal set of Kraus operators on the Hilbert space. Specifically, we ask that for any vector $a_{\mu}(z)$

$$\int dz dz' a_{\mu}^{*}(z) \Lambda^{\mu\nu}(z, t_{f} | z', t_{i}) a_{\mu}(z') \ge 0, \qquad (2.81)$$

which also allows us to deal with the case where $\Lambda^{\mu\nu}(z, t_f | z', t_i)$ is only defined in a distributional sense, i.e, $\Lambda^{\mu\nu}(z, t_f | z', t_i) \sim \delta(z - z')$. The normalization of probabilities requires

$$\int dz \sum_{\mu\nu} \Lambda^{\mu\nu}(z, t_f | z', t_i) K_{\nu}^{\dagger} K_{\mu} = \mathbb{I}.$$
(2.82)

We compare the general dynamics for classical, quantum, and classical-quantum dynamics and their associated positivity and norm conditions in Table 2.1. Because we will everywhere deal with $\sum_{\mu\nu}$, we will hereon drop the summation and use the Einstein summation convention, implicitly assuming that contracted indices are to be summed over.

2.3.1 Moment expansion and the CQ master equation

We will be primarily interested in studying CQ master equations and will focus on time-local and autonomous dynamics. In the CQ case, we take autonomous to mean that the map in Equation (2.80) defines completely positive dynamics when applied to all states; in other words, CQ dynamics is autonomous if $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$ is a positive matrix-kernel for all times.

	Classical
Dynamics	$p(z,t_f) = \int dz' P(z,t_f z',t_i) p(z',t_i)$
Positivity condition	$P(z, t_f z', t_i) \ge 0$ for all z, z'
Norm condition	$\int dz(z,t_f z',t_i)=1$
	Quantum
Dynamics	$\rho(t_f) = \sum_{\mu\nu} \Lambda^{\mu\nu}(t_f, t_i) K_{\mu} \rho(t_i) K_{\nu}^{\dagger}$
Positivity condition	$\Lambda^{\mu u}(t_f,t_i)$ a positive matrix in μu
Norm condition	$\sum_{\mu u} \Lambda^{\mu u}(t_f, t_i) K^{\dagger}_{\nu} K_{\mu} = \mathbb{I}$
	Classical-quantum
Dynamics	$\varrho(z,t+\delta t) = \int dz' \sum_{\mu\nu} \Lambda^{\mu\nu}(z,t_f z',t_i) K_{\mu} \varrho(z',t_i) K_{\nu}^{\dagger}$
Positivity condition	$\Lambda^{\mu\nu}(z,t_f z',t_i)$ a positive matrix in $\mu\nu$ for all z,z'
Norm condition	$\int dz \sum_{\mu\nu} \Lambda^{\mu\nu}(z, t_f z', t_i) K_{\nu}^{\dagger} K_{\mu} = \mathbb{I}$

Table 2.1: A table illustrating the general dynamics governing classical, quantum, and classicalquantum dynamics. We also show the positivity conditions required for dynamics to maintain positive probabilities, as well as the norm condition, which ensures probabilities sum to one. In this sense, Equation (2.80) is a natural classical-quantum generalization of the classical transition probability equation and the quantum Kraus decomposition theorem to the hybrid case.

Assuming autonomy, one can derive the CQ master equation in a similar fashion to the classical and quantum master equations by performing a short time expansion of Equation (2.80) in the case when the L_{μ} are trace-class [28]. When the dynamics are autonomous, we can write the transition equation of Equation (2.80) in the form

$$\varrho(z,t+\delta t) = \int dz' \Lambda(z,t+\delta t|z',t)(\varrho(z',t)) = \int dz' \Lambda^{\mu\nu}(z,t+\delta t|z',t) K_{\mu}\varrho(z',t) K_{\nu}^{\dagger}.$$
 (2.83)

Moreover, we can perform a short-time expansion of the transition amplitude

$$\Lambda^{\mu\nu}(z, t + \delta t | z', t) = \delta^{\mu}_{0} \delta^{\nu}_{0} + \delta t W^{\mu\nu}(z | z', t).$$
(2.84)

We saw that in classical autonomous dynamics, the Kramers-Moyal expansion relates the master equation to the moments of the probability transition amplitude. We now perform the analogous calculation for the combined classical-quantum case. For the moment expansion, we work with the form of the dynamics in Equation (2.83), using a basis of Kraus operators on the Hilbert space which includes the identity, $K_{\mu} = \{\mathbb{I}, L_{\alpha}\} \equiv L_{\mu}$. We take the classical degrees of freedom to be d dimensional, $z = (z_1, \ldots z_d)$, and we label the components as $z_i, i \in \{1, \ldots d\}$.

We begin by introducing the moments of the CQ transition amplitude

$$M_{n,i_1\dots i_n}^{\mu\nu}(z',t,\delta t) = \int dz \ \Lambda^{\mu\nu}(z,t+\delta t|z',t)(z-z')_{i_1}\dots(z-z')_{i_n},$$
(2.85)

where $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$ are the components of the dynamics of the CP map in the basis $L_{\mu} = \{\mathbb{I}, L_{\alpha}\}$, as defined in Equation (2.99).

We define the characteristic function, which is the Fourier transform of the transition amplitude

$$C^{\mu\nu}(u,z',t,\delta t) = \int dz e^{iu \cdot (z-z')} \Lambda^{\mu\nu}(z,t+\delta t|z',t) = \sum_{n=0}^{\infty} \frac{(i^n)u_{i_1}\dots u_{i_n}}{n!} M^{\mu\nu}_{n,i_1\dots i_n}(z',t,\delta t).$$
(2.86)

Taking the inverse Fourier transform, we can relate the transition amplitude to its moments

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \int du \ e^{-iu(z-z')}C^{\mu\nu}(u,z',t,\delta t)$$

= $\sum_{n=0}^{\infty} \frac{M_{n,i_1...i_n}^{\mu\nu}(z',t,\delta t)}{n!} \frac{1}{(2\pi)^d} \int du \ e^{-iu(z-z')}(i^n), u_{i_1}...u_{i_n},$ (2.87)

which, using the definition of the delta distribution, we can write as

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \sum_{n=0}^{\infty} \frac{1}{n!} M^{\mu\nu}_{n,i_1\dots i_n}(z',t,\delta t) \left(\frac{\partial^n}{\partial z'_{i_1}\dots \partial z'_{i_n}}\right) \delta(z,z').$$
(2.88)

Looking at the short time expansion coefficients of $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$, as defined in (2.84), we have

$$M^{\mu\nu}(z',t,\delta t)_{n,i_1...i_n} = \delta^{\mu}_0 \delta^{\nu}_0 + \delta t \int dz W^{\mu\nu}(z|z',t)(z-z')_{i_1}...(z-z')_{i_n}$$

$$\equiv \delta^{\mu}_0 \delta^{\nu}_0 + \delta t n! D^{\mu\nu}_{n,i_1...i_n}(z',t) + O(\delta t^2), \qquad (2.89)$$

where we have implicitly defined the quantity $D^{\mu\nu}(z',t)_{n,i_1...i_n}$ via

$$D_{n,i_1\dots i_n}^{\mu\nu}(z',t) := \frac{1}{n!} \int dz W^{\mu\nu}(z|z',t), t(z-z')_{i_1}\dots(z-z')_{i_n}.$$
 (2.90)

Just as for the classical case, we shall occasionally find it useful to refer to the moments as $D_n(z')$, by which we mean the object with components $D^{\mu\nu}(z')_{n,i_1...i_n}$. We shall also often suppress the time dependence of $D_n(z')$, which can be added later.

Substituting the short time moment coefficients back into Equation (2.88), taking the limit $\delta t \rightarrow 0$, and using the probability preserving condition in (2.82), we can write a classicalquantum master equation in the form

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{00}_{n,i_1\dots i_n}(z,\delta t) \varrho(z,t) \right)
- i[H(z), \varrho(z)] + D^{\alpha\beta}_0(z) L_\alpha \varrho(z) L^{\dagger}_{\beta} - \frac{1}{2} D^{\alpha\beta}_0 \{ L^{\dagger}_{\beta} L_\alpha, \varrho(z) \}_+
+ \sum_{\mu\nu \neq 00} \sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{\mu\nu}_{n,i_1\dots i_n}(z) L_\mu \varrho(z,t) L^{\dagger}_{\nu} \right),$$
(2.91)

where we define the Hermitian operator $H(z) = \frac{i}{2} (D_0^{\mu 0} L_{\mu} - D_0^{0 \mu} L_{\mu}^{\dagger})$ (which is Hermitian since $D_0^{\mu 0} = D_0^{0 \mu *}$). H(z) defines a Hamiltonian which is dependent on the classical degrees of freedom.

We see the first line of Equation (2.91) describes purely classical dynamics and is fully described by the moments of the identity component of the dynamics $\Lambda^{00}(z, t + \delta t | z', t)$. The second line describes pure quantum Lindbladian evolution described by the zeroth moments of the components $\Lambda^{\alpha 0}(z, t + \delta t | z', t)$, $\Lambda^{\alpha \beta}(z, t + \delta t | z', t)$; specifically the (block) off diagonals, $D_0^{\alpha 0}(z)$, describe the pure Hamiltonian evolution, while the components $D_0^{\alpha \beta}(z)$ describe the dissipative part of the pure quantum evolution. Note that the Hamiltonian and Lindblad couplings can depend on the classical degrees of freedom, so the second line describes the action of the classical system on the quantum one. The third line contains the non-trivial classicalquantum back-reaction, where changes in the distribution over phase space are induced and accompanied by changes in the quantum state.

Equation (2.91) is a natural generalization of the Lindblad equation and classical master equation in the case of classical-quantum coupling. The positivity conditions from (2.80) transfer to positivity conditions on the master equation via (2.84). We can write the positivity conditions in an illuminating form by writing the short time expansion of the transition amplitude $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$, as defined by Equation (2.84), in block form

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \begin{bmatrix} \delta(z,z') + \delta t W^{00}(z|z',t) & \delta t W^{0\beta}(z|z',t) \\ \delta t W^{\alpha 0}(z|z',t) & \delta t W^{\alpha\beta}(z|z',t) \end{bmatrix} + O(\delta t^2),$$
(2.92)

and the dynamics will be completely positive if and only if $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ is a positive matrix. It is useful to note that from (2.92), we can immediately deduce that $\delta(z,z') + \delta t W^{00}(z|z',t)$ must be positive, which is the same positivity condition as for classical dynamics, as well as the matrix $W^{\alpha\beta}(z|z',t)$. Furthermore, if either of $W^{\alpha\beta}(z|z',t)$ or $W^{00}(z|z',t)$ vanish, then so must $W^{0\alpha}(z|z',t)$, except for its $\delta(z,z')$ component which generates pure Hamiltonian evolution. This tells us that to have non-trivial CQ coupling, we must have a non-zero $W^{\alpha\beta}(z|z',t)$. When the classical degrees of freedom are discrete, the Schur complement – assuming $W^{00}(z|z',t)$ is non-vanishing – informs us the matrix $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ will be positive if and only if $W^{00}(z|z',t)W^{\alpha\beta}(z|z',t) - W^{0\beta}(z|z',t)W^{\alpha0}(z|z',t) \succeq 0$ is a positive matrix in $\alpha\beta$ for all $z \neq z'$. We must be more careful in the continuous case since the components $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ may only be defined in a distributional sense. We explore the positivity conditions in detail in Chapters 4 and 10, and we shall see that they have important consequences for CQ dynamics, such as a general trade-off between decoherence and diffusion, which can be used to constrain classical-quantum theories of gravity.

We have derived the master equation of Equation (2.91) for a specific set of Lindblad operators $L_{\mu} = (\mathbb{I}, L_{\alpha})$. However, it can be shown that Equation (2.91), combined with the positivity condition (2.92), defines completely positive CQ dynamics for an arbitrary set of Lindblad operators $\{L_{\mu}\}$ [28, 95].

One of the contributions (Chapter 4) of this thesis is to prove that the most general form of continuous (in the classical degrees of freedom), autonomous classical-quantum dynamics takes the form

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{n=2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{00}_{n,i_1\dots i_n} \varrho(z,t) \right)
- \frac{\partial}{\partial z_i} \left(D^{0\alpha}_{1,i} \varrho(z,t) L^{\dagger}_{\alpha} \right) - \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} L_{\alpha} \varrho(z,t) \right)
- i[H(z), \varrho(z,t)] + D^{\alpha \beta}_0(z) L_{\alpha} \varrho(z) L^{\dagger}_{\beta} - \frac{1}{2} D^{\alpha \beta}_0 \{ L^{\dagger}_{\beta} L_{\alpha}, \varrho(z) \}_+,$$
(2.93)

where $2D_2^{00} \succeq D_1 D_0^{-1} D_1^{\dagger}$ and $(\mathbb{I} - D_0 D_0^{-1}) D_1 = 0$. Here, and throughout, D_0^{-1} is the generalized inverse of the positive semi-definite Lindbladian coupling $D_0^{\alpha\beta}$, D_1 is a matrix in both α, i indices with entries $D_{1,i}^{0\alpha}$, which encodes the strength of the CQ back-reaction, and D_2^{00} is a matrix in i, j with entries $D_{2,ij}^{00}$, which represents the necessity of diffusion in the classical phase space. The symbol \succeq refers to matrix positivity, and $a \succeq b$ is equivalent to a - b being a positive matrix. Equation (2.93) naturally generalizes the Fokker-Plank and Lindblad equations to the case of continuous classical-quantum coupling.

We refer to CQ dynamics undergoing dynamics according to Equation (2.93) as continuous CQ dynamics and thus undergoing the more general dynamics of Equation (2.91) as jumping CQ dynamics since it is accompanied by finite-sized jumps in the classical degrees of freedom with a finite probability.

Comments on non-Markovian CQ dynamics

When a CQ dynamics is autonomous, the generator can always be written in the form of Equation (2.91) where the moments D_n must satisfy the positivity conditions implied by the positivity conditions of $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$.

If we instead asked only for the CQ dynamics to be time-local, but relaxed the assumption that the generator $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$ be completely positive for all times, then - in analogy with the purely quantum case [92] - we still expect the dynamics to take the form of Equation (2.91), but where the positivity conditions are relaxed; indeed, the derivation of the master equation relied only on the time-local property of the dynamics. In particular, for general timelocal dynamics, we expect that for intermediate times one can have CPTP classical-quantum dynamics where the quantum degrees of freedom recohere while simultaneously the classical degrees of freedom become less diffusive. It would be interesting to explore non-Markovian CQ dynamics further since it is currently not well understood but is crucial in understanding the role of hybrid dynamics in effective theories where dynamics is generally non-autonomous.

Rate equation form

Just as in the classical case, where the master equation can be written in either a rate-equation or expanded form, the same is true for the CQ case. To write it in a rate equation form, we define $W^{\mu\nu}(z|z',t) = \tilde{W}^{\mu\nu}(z|z',t) - \delta^{\mu}_{0}\delta^{\nu}_{0}\mathcal{N}(z,t)\delta(z,z')$, where the norm condition fixes $\mathcal{N}(z',t) = \int dz \sum_{\mu\nu} \tilde{W}^{\mu\nu}(z|z',t) L^{\dagger}_{\nu}L_{\mu}.$

By substituting the short-time expansion coefficients into (2.83) and taking the limit $\delta t \to 0$, we can write the master equation in the form

$$\frac{\partial \varrho(z,t)}{\partial t} = \int dz' \; \tilde{W}^{\mu\nu}(z|z',t) L_{\mu} \varrho(z',t) L_{\nu}^{\dagger} - \frac{1}{2} \tilde{W}^{\mu\nu}(z,t) \{L_{\nu}^{\dagger}L_{\mu},\varrho\}_{+}, \tag{2.94}$$

where $\{,\}_+$ is the anti-commutator, and preservation of normalization under the trace and $\int dz$ defines

$$\tilde{W}^{\mu\nu}(z,t) = \int dz' \tilde{W}^{\mu\nu}(z'|z,t).$$
(2.95)

Following standard convention, we refer to $\tilde{W}^{\mu\nu}(z|z')L_{\mu}\varrho(z')L_{\nu}^{\dagger}$ as the *jump term* and we call $\frac{1}{2}\tilde{W}^{\mu\nu}(z)\{L_{\nu}^{\dagger}L_{\mu},\varrho\}_{+}$ the *no-event term*.³.

Note, the redefinition of $W(z|z,t) \to \tilde{W}(z|z,t)$ does not change the moments appearing in the master equation in 2.91. In particular, except for D_0^{00} , which does not appear in the master equation, the moments D_n are found to be

$$D_{n,i_1\dots i_n}^{\mu\nu}(z',t) := \frac{1}{n!} \int dz W^{\mu\nu}(z|z',t)(z-z')_{i_1}\dots(z-z')_{i_n}$$

= $\frac{1}{n!} \int dz \tilde{W}^{\mu\nu}(z|z',t)(z-z')_{i_1}\dots(z-z')_{i_n},$ (2.96)

and to ensure complete positivity of the dynamics, it is sufficient to check the positivity of the matrix

$$\begin{bmatrix} \delta(z,z') + \delta t \tilde{W}^{00}(z|z') & \delta t \tilde{W}^{0\beta}(z|z') \\ \delta t \tilde{W}^{\alpha 0}(z|z') & \delta t \tilde{W}^{\alpha \beta}(z|z') \end{bmatrix}.$$
(2.97)

³These conventions come from studying the unravellings of Lindblad equations via stochastic pure state quantum trajectories. At each time step, the quantum state either undergoes continuous evolution via an effective Hamiltonian or with some probability jumps to a new state that depends on the Lindblad operators L_{α} appearing in the master equation [99, 100, 101]

	Classical
Master equation	$\frac{\partial p}{\partial t} = \int dz' W(z z') p(z') - \int dz' W(z' z) p(z)$
Positivity condition	$P(z, t + \delta t z', t) = \delta(z, z') + \delta t W(z z') + \mathcal{O}(\delta t^2) \ge 0 \ \forall z, z'$
	Quantum
Master equation	$\frac{\partial \sigma(t)}{\partial t} = -i[H,\sigma] + h^{\alpha\beta}L_{\alpha}\sigma L_{\beta}^{\dagger} - \frac{1}{2}\left\{h^{\alpha\beta}L_{\beta}^{\dagger}L_{\alpha},\sigma\right\}_{+}$
Positivity condition	$h^{lphaeta}$ a positive matrix, $h \succeq 0$.
	Classical-quantum
Master equation	$\frac{\partial \varrho}{\partial t} = \int dz' W^{\mu\nu} \left(z z' \right) L_{\mu} \varrho \left(z' \right) L_{\nu}^{\dagger} - \frac{1}{2} \int dz' W^{\mu\nu} \left(z' z \right) \{ L_{\nu}^{\dagger} L_{\mu}, \varrho(z) \}_{+}$
Positivity condition	$ \Lambda^{\mu\nu}(z,t+\delta t z',t) = \begin{bmatrix} \delta(z,z') + \delta t W^{00}(z z') & \delta t W^{0\beta}(z z') \\ \delta t W^{\alpha 0}(z z') & \delta t W^{\alpha \beta}(z z') \end{bmatrix} + \mathcal{O}(\delta t^2) \succeq 0 \ \forall z,z' $

Table 2.2: A table illustrating the Markovian master equations governing classical, quantum, and classical-quantum dynamics. We see that the CQ master equation is a natural generalization of the classical rate equation and the Lindblad equation. We have suppressed the explicit t dependence on the transition amplitudes, but this can be back added in.

Written in the form of Equation (2.94) is useful since it is automatically normalized, whereas Equation (2.83) is not. We compare the rate equation forms for the classical, quantum, and classical-quantum master equations in Table 2.2.

Vectorization of CQ dynamics

In Chapter 4, we shall find it useful to deal with superoperators and to double the quantum degrees of freedom using the vectorization map [102]. We do so by representing the CQ density operators $\varrho(z)$ as vectors by stacking the columns, i.e., sending $|i\rangle\langle j| \rightarrow |j\rangle \otimes |i\rangle$. We denote the vectorized form as $\vec{\varrho}(z)$. Then, superoperators are matrices acting on the stacked vector $\vec{\varrho}(z)$, for example

$$\vec{\varrho}(z,t+\delta t) = \int dz' \Lambda^{\mu\nu}(z,t+\delta t|z',t) (\bar{L}_{\nu} \otimes L_{\mu}) \vec{\varrho}(z,t) = \int dz' \Lambda^{vec}(z,t+\delta t|z',t) (\vec{\varrho}(z')), \quad (2.98)$$

where we write *vec* to remind us that we should view the superoperator as a matrix on the doubled Hilbert space. This is particularly useful since it allows us to identify the components of the superoperator in any orthogonal basis of operators $(\bar{L}_{\nu} \otimes L_{\mu})$ via

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \operatorname{Tr}\left[(\bar{L}_{\nu}\otimes L_{\mu})^{\dagger}\Lambda^{vec}(z,t+\delta t|z',t)\right].$$
(2.99)

2.3.2 Physical interpretation of the moments

In classical Markovian dynamics, the moments of the short time expansion of the probability transition amplitude $P(z, t + \delta t | z', t)$ are useful since they are usually related to observable quantities. For example, the first moment characterizes the system's drift, while the second moment typically characterizes diffusion. In the CQ case, we have similar interpretations. We see from Equation (2.91) that the zeroth moments characterize the pure quantum evolution, with $D_0^{\alpha\beta}$ determining the rate of decoherence on the quantum system (and Lindbladian coupling more generally). As we shall see in Chapter 4, in order to have a non-trivial classical-quantum dynamics, positivity demands $D_0^{\alpha\beta}(z) \neq 0$ and so the classical system forces decoherence upon the quantum system. To give interpretation to the higher order moments, consider starting in a state of certainty in phase space $\varrho(z,t) = \delta(z,\bar{z})\sigma(z)$ and after some short time δt measuring the classical observable $(z - \bar{z})^n$, $n \geq 1$. In this case, using Equation (2.91), we find

$$\int dz (z-\bar{z})^n \operatorname{Tr}\left[\varrho(z,t+\delta t)\right] = \delta t n! D_n^{\mu\nu}(\bar{z}) \operatorname{Tr}\left[L_{\nu}^{\dagger} L_{\mu} \sigma\right], \qquad (2.100)$$

hence we see the coefficients $D_n^{\mu\nu}(z)$ (for $\mu\nu \neq 00$) characterize the back-reaction of the quantum system on the classical system in the presence of non-trivial CQ coupling. In particular, the first moment $D_{1,i}^{\mu\nu}$, with $\mu, \nu \neq 0$, gives the force exerted by the quantum system on the classical system, and the second moment determines the diffusion induces on the classical degrees of freedom.

More generally, we can consider the expectation value of any CQ operator O(z), $\langle O(z) \rangle := \int dz \operatorname{Tr} [O(z)\varrho]$ which, for simplicity, we assume does not have an explicit time dependence. Its

evolution law can be determined via Equation (2.91)

$$\frac{d\langle O\rangle}{dt} = \int dz \operatorname{Tr} \left[O(z) \frac{\partial \varrho}{\partial t} \right]
= \int dz \operatorname{Tr} \varrho \left[-i[O(z), H(z)] + D_0^{\alpha\beta}(z) L_{\beta}^{\dagger} O(z) L_{\alpha} - \frac{1}{2} D_0^{\alpha\beta} \{ L_{\alpha} L_{\beta}^{\dagger}, O(z) \}_+ \right.
\left. + \sum_{n=1}^{\infty} \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D_{n, i_1 \dots i_n}^{\alpha\beta}(z) L_{\beta}^{\dagger} L_{\alpha} O(z, t) \right) \right],$$
(2.101)

where we have used cyclicity of trace and integration by parts to bring the equation of motion into a form that would enable us to write a CQ version of the *Heisenberg representation* [28] for a CQ operator. If we are interested in the expectation value of phase space variables, then $O(z) = z\mathbb{I}$ and Equation (2.101) gives

$$\frac{d\langle z\rangle}{dt} = \int dz D_{1,i_1}^{\mu\nu} \operatorname{Tr}\left[L_{\nu}^{\dagger} L_{\mu} \varrho(z,t)\right], \qquad (2.102)$$

with all higher order terms vanishing, and we see that $\sum_{\mu\nu\neq00} D_{1,i_1}^{\mu\nu} \langle L_{\nu}^{\dagger}L_{\mu} \rangle$ governs the rate at which the quantum system moves the classical system through phase space. The force of this back-reaction is especially apparent if the equations of motion are Hamiltonian in the classical limit as in [28].

Specifically, defining $H_I(z) := h^{\alpha\beta} L^{\dagger}_{\beta} L_{\alpha}$ and take $D^{\alpha\beta}_{1,i} = \omega^j_i d_j h^{\alpha\beta}$ with ω the symplectic form and d_j the exterior derivative. Then Equation (2.102) is analogous to Hamilton's equations, and the CQ evolution equation, after tracing out the quantum system, has the form of a Liouville's equation to first order

$$\frac{\partial \rho(z,t)}{\partial t} = \{H_c, \rho(z,t)\} + \operatorname{tr}\left(\{H_I(z), \varrho(z)\}\right) + \dots, \qquad (2.103)$$

where $\rho(z) := \text{Tr} [\varrho(z)]$ and the the ... represent the higher order terms in the moment expansion. We call any CQ master equation with Hamiltonian drift Hamiltonian CQ-dynamics.

The significance of the second moment is also seen via Equation (2.101) to be related to the variance of phase space variables $\sigma_{z_{i_1}z_{i_2}} := \langle z_{i_1}z_{i_2} \rangle - \langle z_{i_1} \rangle \langle z_{i_2} \rangle$

$$\frac{d\sigma_{z_{i_1}, z_{i_2}}^2}{dt} = 2\langle D_{2, i_1, i_2}^{\alpha\beta} L_{\beta}^{\dagger} L_{\alpha} \rangle + \langle z_2 D_{1, z_{i_1}}^{\alpha\beta} L_{\beta}^{\dagger} L_{\alpha} \rangle - \langle z_{i_2} \rangle \langle D_{1, z_{i_1}}^{\alpha\beta} L_{\beta}^{\dagger} L_{\alpha} \rangle
+ \langle z_{i_1} D_{1, z_{i_2}}^{\alpha\beta} L_{\beta}^{\dagger} L_{\alpha} \rangle - \langle z_{i_1} \rangle \langle D_{1, z_{i_2}}^{\alpha\beta} L_{\beta}^{\dagger} L_{\alpha} \rangle.$$
(2.104)

In the case when $D_{1,z_{i_1}}$ is uncorrelated with z_{i_2} and $D_{1,z_{i_2}}$ uncorrelated with z_{i_1} , then the growth of the variance only depends on the diffusion coefficient.

The zeroth moment $D_0^{\alpha\beta}$ is just the pure Lindbladian couplings. The simplest example is the case of a pure decoherence process with a single Hermitian Lindblad operator L and decoherence coupling D_0 . Then we can define a basis $|a\rangle$ via the eigenvectors of L and

$$\langle a | \frac{\partial \varrho}{\partial t} | b \rangle = -i \langle a | [H(z), \varrho] | b \rangle - \frac{1}{2} D_0 (L(a) - L(b))^2 \langle a | \varrho | b \rangle, \qquad (2.105)$$

and we see that the matrix elements of ρ which quantify coherence between the states $|a\rangle, |b\rangle$ decay exponentially fast with a decay rate of $D_0(L(a) - L(b))^2$. For a damping/pumping process of a quantum harmonic oscillator with Hamiltonian $H = \omega a^{\dagger} a$, $L_{\downarrow} = a$, $L_{\uparrow} = a^{\dagger}$, a the creation operator, and $D_0^{\uparrow\uparrow}$, $D_0^{\downarrow\downarrow}$ the non-zero couplings, then standard calculations [82] show that an initial superposition $\frac{1}{\sqrt{2}}(|n\rangle + |m\rangle)$ with n,m large and $n \gg m$ will initially decohere at a rate of approximately $(D_0^{\uparrow\uparrow} + D_0^{\downarrow\downarrow})(m + n)/2$. The state will eventually thermalize to a temperature of $\omega/\log (D_0^{\downarrow\downarrow}/D_0^{\uparrow\uparrow})$. In this case, the Lindblad couplings determine not only the rate of decoherence but also the rate at which energy is pumped into the harmonic oscillator. In Chapter 10, we will derive a trade-off between Lindblad couplings and the diffusion, this terminology is only strictly appropriate for pure decoherence processes. More generally, it is a trade-off between Lindblad couplings and diffusion coefficients.

2.3.3 Master equation examples

In this section, we go through examples of CQ dynamics, illustrating the general properties of CQ theories discussed in this chapter.

A model with finite-sized jumps in the classical phase space

We shall now look at a model of a spin half particle interacting with a linear potential. This model was studied in detail in [10] and helps with intuition in understanding the consequences of the CQ coupling. We take the classical degrees of freedom to be position and momentum z = (q, p) and a two-dimensional Hilbert space \mathcal{H} to represent the quantum spin degrees of freedom. We take the pure classical evolution to be generated by a Hamiltonian $H_c = p^2/2m$ and an interaction Hamiltonian $H_I(q, p) = h^{\alpha} L^{\dagger}_{\alpha} L_{\alpha} = Bq\omega |0\rangle \langle 0| - Bq\omega |1\rangle \langle 1|$. We take the Lindblad operators to be $L_{\alpha=0} = |0\rangle \langle 0|, L_{\alpha=1} = |1\rangle \langle 1|$, which then defines $h^1(q, p) = \omega q B$, $h^2(q, p) =$ $-\omega qB$. We take the following dynamics for the system

$$\frac{\partial \varrho(z,t)}{\partial t} = \{H_c(z), \varrho(z,t)\} - i [H_I(z), \varrho(z,t)] \\
+ \frac{1}{\tau} \sum_{\alpha} \left(e^{\tau \{h^{\alpha}(z), \cdot\}} L_{\alpha} \varrho(z,t) L_{\alpha}^{\dagger} - \frac{1}{2} \left\{ L_{\alpha}^{\dagger} L_{\alpha}, \varrho(z,t) \right\}_{+} \right),$$
(2.106)

where the Poisson bracket in the exponential acts on the CQ state as a linear superoperator

$$\sum_{\alpha} \frac{1}{\tau} e^{\tau \{h^{\alpha}(z),\cdot\}} L_{\alpha} \varrho(z,t) L_{\alpha}^{\dagger} = \sum_{\alpha} \frac{1}{\tau} \left(1 + \tau \left\{ h^{\alpha}(z), L_{\alpha} \varrho(z,t) L_{\alpha}^{\dagger} \right\} + \ldots \right).$$
(2.107)

This is seen to be completely positive due to Equation (2.94).

It is helpful to discuss the interpretation of the dynamics briefly. Equation (2.106) is modeled on a Poisson-like process with jump rate $\frac{1}{\tau}$ jump size $\delta = Bq\tau$.⁴ In particular, a jump in phase space is accompanied by a Lindblad operator acting on the quantum state, which causes it to jump to being in the $|0\rangle$ or $|1\rangle$ state. The Lindblad operators, L_1, L_2 , cause the off-diagonals of the quantum density matrix to decay exponentially with time. Specifically, a quantum state in superposition "collapses" to the $|0\rangle$ state with a rate given by $1/\tau$ – and when it does so, there is a jump in the classical momentum of the particle by an amount $B\omega\tau$ – or the state "collapses" to $|1\rangle\langle 1|$, and there is a jump of momentum $-B\omega\tau$. It was shown [10] that this leads to objective quantum state trajectories conditioned on the classical phase space since measuring the classical degree of freedom makes it possible to determine the quantum state exactly. The model, in some sense, mimics the features of the Stern-Gerlach experiment; the magnetic fields measure the quantum state in the $|0\rangle$, $|1\rangle$ basis and kicks the particle depending on the state the spin has collapsed to. As we evolve in time, classical degrees of freedom in the state naturally diffuse according to the distribution of momentum jumps the system undergoes.

Now, we can repeat the steps in Section 2.3.1 and perform a Kramers-Moyal expansion of the master equation to obtain

$$\frac{\partial \varrho}{\partial t} = \{H_c, \varrho(z)\} - i[H_I(z), \varrho(z)] + \sum_{n=0}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots z_{i_n}}\right) \left[D^{\alpha}_{n, i_1 \dots i_n}(z) L_{\alpha} \varrho(z) L^{\dagger}_{\alpha}\right] - \frac{1}{2} D^{\alpha}_0(z) \{L^{\dagger}_{\alpha} L_{\alpha}, \varrho(z)\}_+,$$
(2.108)

⁴Indeed, it is easy to verify that the probability rate of a jump larger than δ , namely $R = \lim_{t \downarrow s} \frac{1}{t-s} \operatorname{Prob}[|z(t) - z'(s)| > \delta]$, is given by $R = \frac{1}{\tau} \operatorname{Tr}_{\mathcal{H}} \left[L_{\alpha}^{\dagger} L_{\alpha} \varrho \right]$ if $0 < \delta < \delta t \frac{\partial h^{\alpha}}{\partial q}$ and zero otherwise.

with the moments given by

$$D_{n,p\dots p}^{1} = \frac{1}{\tau} \frac{1}{n!} (\omega B\tau)^{n}, \quad D_{n,p\dots p}^{2} = \frac{1}{\tau} \frac{1}{n!} (-\omega B\tau)^{n}.$$
(2.109)

We can directly see that the zeroth moment, $D_0^{\alpha} = \frac{1}{\tau}$, is characteristic of the decoherence rate of the quantum system. We write the CQ state in component form

$$\varrho(q, p, t) = \begin{pmatrix} u_0(q, p, t) & c(q, p, t) \\ c^*(q, p, t) & u_1(q, p, t) \end{pmatrix}.$$
(2.110)

Using this, we can write an equation of motion for the components

$$\frac{\partial u_i(q,p,t)}{\partial t} = -\frac{p}{m} \frac{\partial u_i(q,p,t)}{\partial q} + \frac{1}{\tau} \left(u_i \left(q, p + (-1)^i \omega B\tau, t \right) - u_i(q,p,t) \right), \quad i \in \{0,1\}$$
(2.111)

$$\frac{\partial c(q, p, t)}{\partial t} = -2\omega i q B \omega c(q, p, t) - \frac{p}{m} \frac{\partial c(q, p, t)}{\partial q} - \frac{1}{\tau} c(q, p, t)$$
(2.112)

and using the method of characteristics, one can analytically solve for the off-diagonals of the density operator to obtain

$$c(q, p, t) = \tilde{c}(q - \frac{p}{m}t) e^{-iB(\omega_0 - \omega_1)\left(q - \frac{p}{2m}t\right)t - \frac{t}{\tau}},$$
(2.113)

which illustrates that the off-diagonal terms of the quantum state vanish after a characteristic time $t_{dec} = \tau$.

Taking $\tau \to 0$ gives us a natural classical limit of CQ dynamics: the quantum state rapidly decoheres into its pointer basis. At the same time, higher order terms D_n , $n \ge 2$ arising from the exponentiated Poisson bracket are suppressed by powers of τ . The dominating contribution to the back-reaction is given by the term $\left\{h^{\alpha}(z), L_{\alpha} \varrho(z,t) L_{\alpha}^{\dagger}\right\}$ which, under trace, describes Hamiltonian evolution on the classical degrees of freedom as in Equation (2.103). Occasionally in the thesis, we will refer to the classical limit of CQ dynamics, and this is what we have in mind.

We can find the short-time variances in position and momentum by calculating

$$\int dz \operatorname{Tr} \left[(z_i - \langle z_i \rangle) (z_j - \langle z_j \rangle) \varrho(z, \delta t) \right].$$
(2.114)

If we take an initial state to be of the form $\varrho(z,0) = \delta(q-\bar{q},p-\bar{q})\varrho$, then using integration by parts and taking the trace of the quantum system, the short time variance in momenta is easily seen to be

$$\sigma_{pp}^{2}(\delta t) = \delta t 2 \frac{\tau(\omega_{0}B)^{2}}{2} \operatorname{Tr}\left[(|0\rangle\langle 0| + |1\rangle\langle 1|)\varrho\right] = \delta t \tau(\omega B)^{2} = \delta t (D_{2,pp}^{1} + D_{2,pp}^{2}), \qquad (2.115)$$

and so the rate of diffusion in the classical phase space is characterized by the second moment in the Kramers- Moyal expansion. From equation (2.115), we see that changing the decoherence rate τ directly alters the amount of diffusion in the classical phase space. We can similarly calculate the drift

$$\langle p \rangle(\delta t) = \operatorname{Tr}\left[(D_p^1 | 0 \rangle \langle 0 | + D_p^2 | 1 \rangle \langle 1 |) \varrho \right].$$
(2.116)

Finally, we can analyze the trade-off between diffusion and decoherence, seen in simulations of the model in [10] and discussed in full generality in Chapter 10. There we show that for CQ dynamics to be completely positive, one necessarily lower bounds the diffusion in the system $D_2^{\mu\nu}$ in terms of the Lindbladian coupling $D_0^{\alpha\beta}$ and the force exerted on the classical system, the latter quantified by the rectangular matrix $(D_1^{br})^{\alpha\mu} = D_1^{\alpha\mu}$. The bound on diffusion takes the form of a matrix inequality $2D_2D_0 - D_1^{br}D_1^{br\dagger} \succeq 0$, which in the present context tells us that $2D_2^{\alpha}D_0^{\alpha} \geq (D_{i,p}^{\alpha})^2$. Reading off the moments from (2.85), we see the inequality relating the decoherence and diffusion to the Hamiltonian is satisfied and in-fact saturated

$$\frac{1}{2D_0^{\alpha}} (D_{i,p}^{\alpha})^2 = \frac{\tau}{2} (\omega B)^2 = D_{2,pp}^{\alpha}.$$
(2.117)

An example of continuous dynamics

A simple example of a continuous master equation of Equation (2.93) is given by a classical oscillator coupled to a quantum one. The classical oscillator we describe by the classical Hamiltonian

$$H_c = \frac{1}{2}p^2 + \frac{1}{2}\omega_c^2 q^2, \qquad (2.118)$$

and the quantum oscillator we describe by the quantum Hamiltonian

$$H_q = \frac{1}{2}P^2 + \frac{1}{2}\omega_q^2 Q^2.$$
 (2.119)

We consider the coupling via the interaction Hamiltonian $H_{cq} = D_1 q Q$. From Equation (2.93), the deterministic part of the dynamics is given by

$$\{H_c, \varrho\} - i[H_q, \varrho] - iD_1q[Q, \varrho] + \frac{1}{2}D_1\{qQ, \varrho\} - \frac{1}{2}D_1\{\varrho, qQ\}$$
$$= \{H_c, \varrho\} - i[H_q, \varrho] - iD_1q[Q, \varrho] + \frac{1}{2}D_1\left(Q\frac{\partial\varrho}{\partial p} + \frac{\partial\varrho}{\partial p}Q\right).$$
(2.120)

The back-reaction governed by the D_1 term and takes the form

$$\frac{1}{2}(\{H_{cq},\varrho\} - \{\varrho, H_{cq}\}),$$
(2.121)

which is known as the *Alexandrov-Gerasimenko bracket* [103, 104]. However, Equation (2.120) is not completely positive, without adding decoherence and diffusion due to Equation (2.93). The full master equaiton reads

$$\frac{\partial \rho}{\partial t} = \{H_c, \varrho\} - i[H_q, \varrho] - iD_1q[Q, \varrho] + \frac{1}{2}D_1\left(Q\frac{\partial \varrho}{\partial p} + \frac{\partial \varrho}{\partial p}Q\right) \\
+ \lambda \frac{1}{2}[Q, [\varrho, Q]] + D_2\frac{\partial^2 \rho}{\partial p^2},$$
(2.122)

where complete positivity requires the decoherence-diffusion trade-off

$$D_2 \ge \frac{D_1^2}{\lambda}.\tag{2.123}$$

Roughly speaking, when the system is coherent, the diffusion has to mask the force that the quantum system exerts on the classical one. Equation (2.122) is of the form originally studied by Diosi [57], where one can also add friction term $\gamma \frac{\partial(p\rho)}{\partial p}$ to dampen the effect of the diffusion.

Chapter 3

Hamiltonian formulation of GR

In this chapter, we introduce background material for general relativity (GR) necessary for the second part of the thesis, where we discuss applications of the classical-quantum formalism to gravity. The language of CQ dynamics presented so far is given by a master equation formalism, where one specifies an initial classical-quantum state and evolves according to the dynamics of a CQ master equation. We, therefore, review the initial value, or ADM formulation [77], of GR (see [105] for a detailed overview), which is used in Chapter 7 to study the constraints in classical-quantum theories of gravity.¹ We also discuss standard semi-classical approaches for incorporating quantum back-reaction on a classical gravitational field, which gives rise to non-linear evolution on the quantum state but can be useful in certain regimes.

3.1 The Einstein equations

The setting of Einstein gravity is a Lorentzian manifold $(M, g_{\mu\nu})$, where $g_{\mu\nu}$ is a Lorentzian metric with signature (-, +, +, +). The metric solves Einstein's equations

$$G_{\mu\nu} = \frac{8\pi G}{c^4} T_{\mu\nu},\tag{3.1}$$

which relate the curvature of space-time $G_{\mu\nu} = R_{\mu\nu} - \frac{1}{2}Rg_{\mu\nu}$ to the stress energy tensor $T^{\mu\nu}$ of the matter degrees of freedom. The Einstein equations arise from variations of the gravitational

¹We also consider a covariant approach via path integral methods in Chapter 8.

action $S = S_{EH} + S_m$, where

$$S_{EH} = \frac{c^4}{16\pi G} \int R\sqrt{-g} \, \mathrm{d}^4 x,$$
 (3.2)

is the *Einstein-Hilbert action* and S_m is a matter action; for example, for a minimally coupled scalar field Φ the matter action S_m is given by

$$S_m = \int d^4x \sqrt{g} \left[-\frac{1}{2} g^{ab} \nabla_a \Phi \nabla_b \Phi - V(\Phi) \right].$$
(3.3)

Einstein's equations are diffeomorphism invariant; a diffeomorphism $\theta: M \to M$ is a smooth map between manifolds which is one-to-one, onto, and has a smooth inverse. In particular, denoting the metric and matter degrees of freedom as (g, Φ_m) , then if (g, Φ_m) solve the Einstein's equations, then the push-forward $((g)_*, (\Phi_m)_*)$, obtained by applying the diffeomorphism to the dynamical degrees of freedom, also solve the Einsteins equations. Because diffeomorphism invariance is a local symmetry, it is treated as a gauge symmetry of the theory. Because of this gauge invariance, there are 2 degrees of freedom per space-time point²; $G_{\mu\nu}$ is a symmetric tensor containing 10 equations, the Bianchi identities $\nabla_{\mu}G^{\mu\nu} = 0$ removes 4 degrees of freedom, and gauge symmetry removes another 4.

Written in the form of Equation (3.1), the symmetries of Einstein's equations are manifest. However, they are presented in a different format than is usual when considering the dynamics of physical systems; usually, one considers a physical problem by specifying initial data, with dynamics that evolve the initial data in time to find a solution. In closed systems, the dynamics are usually considered to be Hamiltonian. In open systems, one can consider more general master equations, such as the Fokker-Plank equation for open classical dynamics or the Lindblad equation for open quantum systems. In its covariant form (3.1), it is not apparent which initial conditions are sufficient to govern the evolution of Einstein's equations, nor how to ensure the initial value problem is well posed with a unique solution because the equations have a diffeomorphism gauge symmetry.

Because much of the thesis is devoted to studying classical-quantum master equations and their applications to gravity, we now introduce the Hamiltonian (initial value) formulation of GR.

²We define the number of physical degrees of freedom we define to be the number of generalized positions (here $g_{\mu\nu}$), whose evolution is given by a second order in time differential equation.

3.2 The 3+1 split

To obtain an initial value formulation of GR, one considers a foliation of the space-time M, which we assume is globally hyperbolic $M \cong \mathbb{R} \times \Sigma$. Practically, what this means is that there exists a scalar function t(x) and a family of hypersurfaces Σ_t , such that $M = \bigcup_t \Sigma_t$ and each Σ_t is a level surface of the scalar field

$$\forall t \in \mathbb{R}, \ \Sigma_t := \{ p \in M, t(p) = t \}.$$
(3.4)

Given a foliation of space-time, one specifies initial data on a slice Σ_t which evolve according to the gravitational Hamiltonian equations of motion.

To arrive at the Hamiltonian formulation, we first need to decompose the metric into quantities that are intrinsic to the leaves of the foliation Σ_t . This is achieved by performing a 3 + 1split of the metric and writing it in a form adapted to the foliation.

To derive the 3+1 split, on each hypersurface Σ_t we define spatial co-co-ordinates x^i , and let $x^{\mu} = (t, x^i)$ define co-ordinates on M [105]. We call these coordinates the *coordinates adapted* to the foliation. We denote $n_{\mu} = -N\nabla_{\mu}t$, the unit normal to the surfaces Σ_t . In the adapted co-ordinates, $n_{\mu} = -N\delta^0_{\mu}$, and its normalization determines N via $g^{00} = -\frac{1}{N}$. The function N is called the *lapse function*.

We define the normal evolution vector $m^{\mu} = Nn^{\mu}$. This vector has a special role since the vector $\delta t m^{\mu}$ transports the hypersurface Σ_t to its neighboring one $\Sigma_{t+\delta t}$. To see this, note that the normal evolution vector satisfies $m^{\mu}\nabla_{\mu}t = 1$, so that along $\delta t m^{\mu}$ the scalar time function t changes according to $t' = t + \delta t m^{\mu} \nabla_{\mu} t = t + \delta t$.

This property is also satisfied by the vector ∂_t , which has components δ^0_{μ} in the adapted coordinates and corresponds to a simple translation of the scalar time variable. The difference between the vectors ∂_t and m is tangent to Σ_t and is called the *shift vector*, which we denote β^{μ} . In components, we have

$$\delta_0^{\mu} = m^{\mu} + \beta^{\mu} = -N^2 g^{0\mu} + \beta^{\mu}. \tag{3.5}$$

By contracting Equation (3.5) with $n_{\mu} = -N\delta_{\mu}^{0}$, we see that $\beta^{0} = 0$, which verifies that the shift vector lies tangent to Σ_{t} . We shall write $\beta^{i} = N^{i}$ and combine the lapse and shift vectors in the four vector $N^{\mu} = (N, N^{i})$.

We can use (3.5) to find the components of the metric in the coordinates adapted to the foliation. We find

$$g_{00} = g_{\mu\nu}\delta_0^{\mu}\delta_0^{\nu} = g_{\mu\nu}(-N^2g^{0\mu} + N^{\mu})(-N^2g^{0\nu} + N^{\nu}) = -N^2 + g_{ij}N^iN^j, \qquad (3.6)$$

where we have substituted in for (3.5) and used the fact that $g^{00} = -\frac{1}{N}$. Furthermore,

$$g_{0i} = g_{\mu\nu}\delta_0^{\mu}\delta_i^{\nu} = g_{\mu\nu}(-N^2g^{0\mu} + N^{\mu})\delta_i^{\nu} = N_i.$$
(3.7)

Writing $\gamma_{ij} = g_{ij}$, we arrive at the familiar 3+1 decomposition of the metric

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -N^2 dt^2 + \gamma_{ij}(dx^i + N^i dt)(dx^j + N^j dt), \qquad (3.8)$$

which is also known as the ADM decomposition [105, 77].

3.2.1 Curvatures associated to the foliation

The 3-metric γ_{ij} lies tangent to the surface Σ_t , which defines a metric on Σ_t . From now onwards, we shall use abstract index notation with Roman letters a, b to represent tensors that are intrinsic to Σ_t ; for example, we denote the 3 metric γ_{ab} to emphasize that it defines a tensor on Σ_t . This notation is consistent with Chapter 7, where we discuss constraints arising in Hamiltonian theories of CQ gravity.

Because γ_{ab} represents a metric on Σ_t , we can define a covariant derivative and associated curvature tensors. We denote D for the covariant derivative associated γ_{ab} , so that for any vector X^a tangent to Σ_t

$$D_b X^a = \partial_b X^a + \Gamma^a_{bc} X^c, \tag{3.9}$$

where Γ^a_{bc} denotes the Christoffel symbol of the covariant derivative D

$$\Gamma_{bc}^{a} := \frac{1}{2} \gamma^{ad} \left(\partial_{b} \gamma_{dc} + \partial_{c} \gamma_{db} - \partial_{d} \gamma_{bc} \right).$$
(3.10)

From D, we can also define an intrinsic Riemann tensor

$$^{(3)}R^a_{bcd} = \partial_c \Gamma^a_{bd} - \partial_d \Gamma^a_{cb} + \Gamma^a_{ce} \Gamma^e_{bd} - \Gamma^a_{de} \Gamma^e_{cb}, \qquad (3.11)$$

as well as an *extrinsic curvature* tensor

$$K_{ab} = \frac{1}{2N} \left(\dot{\gamma}_{ab} - D_a N_b - D_b N_a \right), \qquad (3.12)$$

which describes how the embedded surfaces Σ_t curve in M.

The intrinsic and extrinsic curvatures are not independent but are related by the Riemann tensor of the four geometry by the Gauss-Codazzi equations. In particular, one can use the Gauss-Codazzi equations [105] to find the G_{0a} and G_{00} projections of the Einstein tensor $G_{\mu\nu}$ in terms of K_{ab} and curvatures of γ_{ab}

$$G_{00} = \frac{1}{2} ({}^{(3)}R - K_{ab}K^{ab} + K^2) = \frac{8\pi G}{c^4} T_{00},$$

$$G_{0a} = D_b K_a^b - D_a K = \frac{8\pi G}{c^4} T_{0a}.$$
(3.13)

The Equations in (3.13) are first order in the time derivatives of γ_{ab} and the matter degrees of freedom, at least for minimally coupled matter. Consequently, (3.13) are not dynamical equations but constraints on the initial data one can specify. To elaborate on the role of constraints in gravity, we now introduce the Hamiltonian formulation of GR, where the constraint structure is manifest.

3.3 Hamiltonian formulation of GR

In this section, we will arrive at the Hamiltonian formulation of GR, also known as the ADM formalism [77, 106]. We do so explicitly for vacuum GR, without matter, and quote the end result when matter is included. We assume the space-time is asymptotically flat so that boundary terms can be ignored.

To derive the Hamiltonian formulation of vacuum GR, we start from the Einstein-Hilbert action

$$S_{EH} = \frac{c^4}{16\pi G} \int R\sqrt{-g} \, \mathrm{d}^4 x.$$
 (3.14)

We first substitute for the ADM split in Equation (3.8). After integrating by parts and assuming an asymptotically flat space-time, we find the action

$$S_{EH} = \int dt d^3 x \mathcal{L}[\gamma, N, N^a] = \frac{c^4}{16\pi G} \int dt d^3 x \sqrt{\gamma} N\left({}^{(3)}R + K_{ab}K^{ab} - K^2\right), \qquad (3.15)$$

where ${}^{(3)}R$ is the Ricci scalar of γ_{ab} , K_{ab} is the extrinsic curvature of a surface of constant t, as defined in Equation (3.12), and $K = K_{ab}\gamma^{ab}$.

The action S_{EH} is a functional of N, N^a and γ_{ab} . To introduce the Hamiltonian, we first need to determine the momenta conjugate to N, N^a and γ_{ab} . Since the action does not depend on time derivatives of N and N^a , their conjugate momenta are identically zero $P_N, P_{N^a} = 0$. The momentum conjugate to γ_{ab} is

$$\pi^{ab} \equiv \frac{\delta S}{\delta \dot{\gamma}_{ab}} = \frac{c^4 \sqrt{\gamma}}{16\pi G} \left(K^{ab} - K \gamma^{ab} \right). \tag{3.16}$$

Note that the factor of $\sqrt{\gamma}$ means that π^{ab} is not a tensor but a tensor density of weight $\frac{1}{2}$.³ The phase space is then generated by elements, (γ_{ab}, π^{ab}) which satisfy the canonical Poisson bracket relations⁴

$$\{\gamma_{ab}(x), \pi^{cd}(y)\} = \frac{1}{2} (\delta^c_a \delta^d_b + \delta^d_a \delta^c_b) \delta(x, y).$$
(3.17)

We define the Hamiltonian as the Legendre transform of the Lagrangian:

$$H = \int d^3x \left(\pi^{ab} \dot{\gamma}_{ab} - \mathcal{L} \right), \qquad (3.18)$$

which, after integrating by parts and neglecting surface terms, takes the form of the ADMHamiltonian [77, 78]

$$H_{ADM}[N,\vec{N}] = \int d^3x N^{\mu} \mathcal{H}_{\mu} = \int d^3x (N\mathcal{H} + N^a \mathcal{H}_a) =:= H[N] + H[\vec{N}], \qquad (3.19)$$

where $\mathcal{H}_{\mu} = (\mathcal{H}, \mathcal{H}_a)$. In Equation (3.19)

$$\mathcal{H} = \frac{(16\pi G)}{c^4} \pi^{ab} G_{abcd} \pi^{cd} - \frac{c^4}{16\pi G} \gamma^{1/2} R, \quad \mathcal{H}_a = -2\gamma_{ac} D_b \pi^{cb}, \tag{3.20}$$

and G_{abcd} is the *deWitt metric* defined as $G_{abcd} = \frac{1}{2\sqrt{\gamma}}(\gamma_{ac}\gamma_{bd} + \gamma_{ad}\gamma_{bc} - \gamma_{ab}\gamma_{cd})$.⁵ We can also define an extended phase space $(\gamma_{ab}, \pi^{ab}, N, P_N, N^a, P_{N^a})$, by including explicitly the constraints $P_{\mu} = (P_N, P_{N^a}) \approx 0$ in the *extended Hamiltonian*

$$H^e_{ADM} = \int d^3x (N^\mu \mathcal{H}_\mu + \lambda^\mu P_\mu). \tag{3.21}$$

³A tensor density of weight p transforms under a coordinate transformation in the same way as γ^p , the determinant of γ , times a tensor.

⁴The convention here is that $\delta(x, y)$ is a scalar in x and a scalar density in y. It is defined by its action on scalar functions $f: \Sigma \to \mathbb{R}$ as $f(x) = \int_{\Sigma} dy \delta(x, y) f(y)$. It is useful to note that as a consequence, $\int_{\Sigma} dy \nabla_a^x \delta(x, y) f(y) = \nabla_a f(x)$ and $\int_{\Sigma} dy \nabla_a^y \delta(x, y) f(y) = -\int_{\Sigma} dy \delta(x, y) \nabla_a^x f(y) = -\nabla_a f(x)$ [107].

⁵We can extend the covariant derivative to act on tensor densities of weight W, we do this by subtracting $W\Gamma_c^{ca}$ to the usual covariant derivative D_a For example $D_a\pi^{bc} = \partial_a\pi^{bc} + \Gamma_{ad}^b\pi^{dc} + \Gamma_{ad}^c\pi^{bc} - \Gamma_{da}^d\pi^{bc}$ since π^{ab} is a tensor density of weight 1.
In the Hamiltonian formalism, γ_{ab} and π^{ab} are the dynamical variables. The lapse function N and shift vector N^a appearing in Equation (3.19) are arbitrary functions and play the role of Lagrange multipliers enforcing constraints. They are non-dynamical since $P_N, P_{N^a} \approx 0$, and as a result, we see that GR is a constrained theory. In terms of the extended Hamiltonian in Equation (3.21), one sees that the time derivatives of the lapse and shift vectors are functions of the arbitrary Lagrange multipliers λ^{μ} , $\dot{N}^{\mu} = \lambda^{\mu}$, The lapse and shift vectors arose when we performed the 3+1 split of the metric in Section 3.2 and represent the gauge degrees of freedom associated to picking a foliation of space-time; in particular, a different choice of time function t and coordinates x^i will yield different lapse and shift vectors.

Asking that the constraints $P_N, P_{N^a} \approx 0$ are preserved in time leads to the Hamiltonian and Momentum constraints, $\mathcal{H} = \mathcal{H}_a \approx 0$. Conservation of these constraints is ensured via the hypersurface deformation algebra [108]

$$\{H[N], H[M])\} = H[\vec{R}]$$

$$\{H[\vec{M}], H[N]\} = H [L_{\vec{M}}N]$$

$$\{H[\vec{N}], H[\vec{M}]\} = H[L_{\vec{N}}\vec{M}],$$

$$(3.22)$$

where $R^a := \gamma^{ab} \left(N D_b M - M D_b N \right)$ and L is the Lie derivative on Σ_t .

The dynamical equations of motion are found from Hamilton's equations

$$\dot{\gamma}_{ab} = \frac{\delta H}{\delta \pi^{ab}} \quad \dot{\pi}^{ab} = -\frac{\delta H}{\delta \gamma_{ab}}.$$
(3.23)

The first equation reproduces the definition of π^{ab} , while the second equation encodes the dynamics of Einstein's equations not associated with constraints, i.e., the spatial projections of Einstein's equations.

The Hamiltonian formulation is used to solve the system of variables (γ_{ab}, π^{ab}) once a lapse and shift vector (N, N^a) is specified. Since the N^{μ} are arbitrary functions of space and time, the interpretation is that $\{\mathcal{H}_{\mu},\}$ generate gauge transformations. One way of understanding their role is that they generate gauge transformations on an initial value surface: they take initial data to other initial data, which, once the equations of motion have been solved, give rise to physically equivalent solutions [109]. To see this more explicitly, we can consider the evolution of any phase space functional $f(\gamma, \pi)$. Given it takes the value f(0) at t = 0, after a small time ϵ , it can take the values

$$f(\epsilon) = f(0) + \epsilon N^{\mu} \{ f(0), \mathcal{H}_{\mu} \}$$

$$f'(\epsilon) = f(0) + \epsilon N^{\mu'} \{ f(0), \mathcal{H}_{\mu} \},$$
(3.24)

depending on the choice of lapse and shift. Consequently, given the same initial data set, we get two equally valid descriptions of the phase space functional. If we wish to retain any notion of predictability, we must identify that action of the constraints is to give new phase space variables which should be considered physically equivalent in the sense that they will yield equivalent solutions to the dynamics [106]. The difference between the function due to different choices of lapse and shift is given by

$$\delta f(\epsilon) = \epsilon \left(N^{\mu\prime} - N^{\mu} \right) \left\{ f(0), \mathcal{H}_{\mu} \right\}, \qquad (3.25)$$

and hence one can deduce that $\{\mathcal{H}_{\mu},\}$ generates equal time gauge transformations on the phase space, associated with the different possible choices of the lapse and shift-vector. It should be emphasized that the constraints \mathcal{H}_{μ} do not *directly* map solutions of initial data to other gauge equivalent solutions, nor do they generate diffeomorphisms on solutions; this is easily seen since they do not change the values of the lapse and shift vectors, which change under diffeomorphisms [109]. Note that the extended Hamiltonian in Equation (3.21) does act as a time diffeomorphism on the space of solutions in the extended phase pace. In the next section, we discuss the relationship between the constraint generators and diffeomorphism invariance in more detail.

Understanding that the constraints generate gauge transformations, we can check that the Hamiltonian system counts the same number of degrees of freedom as Einstein's equations. Because the lapse and shift are arbitrary functions corresponding to a choice of gauge, the physical degrees of freedom are (γ_{ab}, π^{ab}) , which constitute 6 degrees of freedom, but satisfy 4 constraints, leading to two degrees of freedom per space-time point.⁶

Since the language of CQ dynamics is that of master equations, it is worth mentioning that – although it is not usually considered – we can write the dynamics of pure GR in a Liouville

⁶Note that (γ_{ab}, π^{ab}) consist of 12 phase space degrees of freedom, but since π^{ab} is defined in terms of $\dot{\gamma}^{ab}$ they lead to only 6 configuration space degrees of freedom undergoing second order time dynamics.

formulation. In particular, a phase space distribution $\rho(\gamma, \pi)$ will evolve under the dynamics as

$$\frac{\partial \rho}{\partial t} = \{H_{ADM}, \rho\},\tag{3.26}$$

subject to the constraints $\mathcal{H}\rho = \mathcal{H}_a\rho \approx 0$; that is ρ must have support only on the constraint surface.

We can, of course, add matter to the discussion. For ease of calculation, we only consider coupling scalar fields to gravity in this thesis. A classical field minimally coupled to gravity will have a Hamiltonian of the form

$$H_T[N,\vec{N}] = H_{ADM}[N,\vec{N}] + H_m[N,\vec{N}] = \int d^3x N(\mathcal{H} + \mathcal{H}_m) + N^a(\mathcal{H}_a + \mathcal{H}_{m,a}), \qquad (3.27)$$

where H_m is the Hamiltonian of the matter field. In the presence of matter, the constraint surface takes the form $\mathcal{H} + \mathcal{H}_m \approx 0, \mathcal{H} + \mathcal{H}_{m,a} \approx 0$.

For example, the Hamiltonian of the free scalar field reads

$$H_m[N,\vec{N}] = \int d^3x N(\frac{1}{2}\gamma^{-1/2}\pi^2 + \frac{1}{2}\gamma^{1/2}\gamma^{ij}\partial_i\phi\partial_j\phi + \frac{1}{2}\gamma^{1/2}m^2\phi_\phi^2) + N^i\pi_\phi\partial_i\phi.$$
(3.28)

The Liouville equation for the phase space density $\rho(\gamma, \pi_{\gamma}, \phi, \pi_{\phi}, t)$ then takes the form

$$\frac{\partial \rho}{\partial t} = \{H[N, \vec{N}], \rho\} + \{H_m[N, \vec{N}], \rho\}, \qquad (3.29)$$

where ρ must only have support on the constraint surface.

3.3.1 Symmetries in the ADM formalism

In this section, we briefly comment on the relationship between the constraint generators $\{\mathcal{H}_{\mu},\}$ and that of diffeomorphism invariance in GR, which is a gauge symmetry on the space of *solutions* to the equations of motion.

Given a choice of lapse and shift vectors $(N(t), N^a(t))$ the equations of motion, generated by Equation (7.13), give us trajectories in phase-space $(\gamma_{ab}(t), \pi^{ab}(t))$, which naturally define a Lorentzian 4 metric on M via the ADM decomposition

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -N^2 dt^2 + \gamma_{ab}(N^a dt + dx^a)(N^b dt + dx^b).$$
(3.30)

Note, Equation (3.30) is an isometric embedding so that $\gamma_{ab}(t)$ is the induced metric for surfaces of constant t, Σ_t , and we can invert $\pi_{ab}(t)$ to obtain $\dot{\gamma}_{ab} = F_{ab}[\gamma, \pi]$. We denote the set of phasespace solutions to the ADM equations by S. To discuss the relationship between diffeomorphism invariance and the hypersurface deformation algebra, consider applying a diffeomorphism θ to the solution (3.30), which acts to send $g_{\mu\nu} \rightarrow g'_{\mu\nu}$. Since GR is diffeomorphism invariant, $g'_{\mu\nu}$ will also be a solution to the Einstein equations. We can further find a 3+1 split of $g'_{\mu\nu}$

$$g'_{\mu\nu}dx^{\mu}dx^{\nu} = -N'^{2}dt^{2} + \gamma'_{ab}(t,x)(N'^{a}dt + dx^{a})(N'^{b}dt + dx^{b}), \qquad (3.31)$$

which defines a new set of variables $(\gamma'_{ab}(t), \pi^{ab'}(t), N'(t), N'^a(t))$ which will also lie in \mathcal{S} .

The fact that $g'_{\mu\nu}$ is also a solution to Einstein's equations leads naturally to a notion of diffeomorphism invariance on the phase space; more appropriately, this should be called a diffeomorphism equivariance. We say that the theory is diffeomorphism equivariant since $(\gamma_{ab}(t), \pi^{ab}(t), N(t), N^a(t))$ is a solution to the ADM equations of motion if and only if transformed solution $(\gamma'_{ab}(t), \pi^{ab'}(t), N'(t), N'^a(t))$ is also a solution to the ADM equations. We denote the mapping between the two solutions by ϕ .

In the case of the ADM equations of motion, the transformation ϕ is exactly that which arises from Noether's inverse theorem applied to gauge transformations generated by $\mathcal{H}, \mathcal{H}_a$ on the extended phase space [110]. Explicitly, on the extended phase space, ϕ is generated by [110]

$$G(t;\xi) = P_{\mu}\dot{\xi}^{\mu} + \xi^{\mu} \left(\mathcal{H}_{\mu} + N^{\varrho}C^{\nu}_{\mu\varrho}P_{\nu}\right), \qquad (3.32)$$

where $G(t;\xi)$ acts on the space of solutions S, taking one solution to a physically equivalent solution. In Equation (3.32), ξ^{μ} parameterizes the diffeomorphism, and $\{\mathcal{H}_{\mu}, \mathcal{H}_{\nu}\} = C^{\sigma}_{\mu\nu}\mathcal{H}_{\sigma}$. Specifically, ξ^{0} parameterizes transformations normal to surfaces of constant time, while ξ^{a} parameterizes the spatial diffeomorphisms.

Importantly, we see that \mathcal{H}_{μ} by themselves do not generate diffeomorphisms on the space of solutions. Instead, they act as gauge transformations, taking initial data to other initial data, which, once the equations of motion have been solved, give rise to physically equivalent solutions; one can easily verify \mathcal{H}_{μ} cannot generate diffeomorphisms on the space of solutions, since they do not change the values of the lapse and shift vectors, which change under diffeomorphisms. They do, however, partially do the job, and one can check that they generate diffeomorphisms on the phase space variables (γ_{ab}, π^{ab}); the extra terms in Equation (3.32) then account for the changes in the lapse and shift vectors. One can also check that with the substitution $\xi^{\mu} = N^{\mu}$ Equation (3.32) reduces to the extended Hamiltonian of Equation (3.21), which can be understood as generating time diffeomorphisms along the vector ∂_t . Note that the dual role of the extended Hamiltonian does not imply that the dynamics are frozen or timeless since the roles of gauge transformations and dynamical evolution are quite distinct. In particular, as the generator of dynamics, the Hamiltonian takes initial data and generates a solution $p \in S$ to the phase space equations of motion; in loose terms, the Hamiltonian acts within a single solution p. On the other hand, as a gauge transformation, it is to be understood as taking one solution $p \in S$ to another, gauge equivalent, solution $p' \in S$. We refer the reader to [109] for a more detailed discussion of these points.

3.4 Incorporating back-reaction: the semi-classical equations

We now discuss the case of incorporating back-reaction when the matter degrees of freedom are considered quantum. In the case of gravity, the standard approach to define back-reaction is via the semi-classical Einstein equations, which source the Einstein tensor $G_{\mu\nu}$ by the expectation value of the stress-energy tensor $T_{\mu\nu}$ [19, 20]

$$G_{\mu\nu} = \frac{8\pi G}{c^4} \langle T_{\mu\nu} \rangle, \qquad (3.33)$$

where the quantum state is understood to undergo unitary dynamics; the quantum state $|\psi\rangle$ at time t is determined by Hamiltonian evolution that depends on classical degrees of freedom (g, π) .

$$\frac{d|\psi\rangle}{dt} = -iH(g,\pi)|\psi\rangle. \tag{3.34}$$

The semi-classical Einstein equations can be derived from effective low energy quantum gravity when there is a dominant background gravitational field, and fluctuations around it are small [111, 112, 113, 114]. Though the scope and limitations of the semi-classical Einstein equations are not precisely understood [21, 22, 23], they are commonly understood to fail when fluctuations of the stress-energy tensor are large in comparison to its mean value [24, 25, 26, 21, 27].

Though the semi-classical equations are valid when quantum fluctuations are small, and so the quantum state is essentially classical, the case where the fluctuations are significant are often precisely the regimes we most wish to understand; for example, when considering the gravitational field associated with Schrodinger cat states of massive bodies [29, 30], or vacuum fluctuations during inflation [31, 32, 33, 34]. For these regimes, background field methods are not appropriate, and an alternate effective theory of the back-reaction of quantum matter on classical gravity is required.

The problem with using the semi-classical equations in the presence of large quantum fluctuations is that they lead to non-linear evolution on the quantum state and, more importantly, the density matrix due to the expectation value sourcing the classical degree of freedom. Consequently, they also fail when probabilistic mixtures are considered since non-linear evolution on the density matrix is inconsistent with an ensemble interpretation [6, 28]. The non-linearity of the dynamics leads to violations of the standard principles of quantum theory; as an effective theory, small violations of quantum theory may be acceptable, which is the case for an essentially classical quantum state, but away from this, the theory should be deemed inapplicable. The reason for the failure of semi-classical gravity when large quantum fluctuations are considered is that the semi-classical equations fail to account for correlations between the classical and quantum degrees of freedom. In practice, the correlation is often put in by hand by considering situations when the quantum state is fully decohered and then evolving the classical system conditioned on the quantum state being in a particular eigenvalue, but this is quite distinct from a direct application of the semi-classical equations (see [6, 28] for a more detailed discussion of these points).

We can see the non-linearity of the semi-classical equations in more detail by considering the Newtonian limit of the semi-classical equations. The Newtonian limit of semi-classical gravity is described by the single Hamiltonian constraint

$$\nabla^2 \Phi = 4\pi G \langle \psi | m | \psi \rangle = 4\pi G m | \psi(x) |^2, \qquad (3.35)$$

while the quantum state undergoes Hamiltonian evolution according to $H = H_m + H_I$,

$$\frac{d|\psi\rangle}{dt} = -iH_m|\psi\rangle - i\int d^3x \Phi(x)m(x)|\psi\rangle, \qquad (3.36)$$

where H_I is the interaction Hamiltonian

$$H_I = \int d^3x \Phi(x) m(x), \qquad (3.37)$$

and H_m denotes the matter Hamiltonian not associated with gravity. Inserting the solution for the gravitational potential

$$\Phi(x) = -G \int d^3x' \frac{\langle \psi | m(x) | \psi \rangle}{|x - x'|}$$
(3.38)

into Equation (3.36), we find the Schrodinger-Newton equation [115, 116, 117]

$$\frac{d|\psi\rangle}{dt} = -i(H_m)|\psi\rangle + iG \int d^3x \int d^3x' \frac{\langle\psi|m(x)|\psi\rangle}{|x-x'|} m(x)|\psi\rangle, \qquad (3.39)$$

which describes the evolution of the quantum state in the Newtonian limit of semi-classical gravity.

It is well known that the dynamics of Equation (3.39) is not consistent when applied to quantum states with large fluctuations [29, 49, 51], and because it leads to non-linear evolution of the density matrix, it violates the standard principles of quantum theory, inducing a breakdown of either operational no-signaling, the Born rule, or composition of quantum systems under the tensor product [48, 18, 49, 50, 51]. Though one might be willing to expect some small violations of these properties as an effective theory, the same cannot be said if it is treated as a fundamental equation. Regardless, an improved semi-classical description should be sought after; in Chapter 5, we shall show that CQ dynamics can be used to give rise to a consistent semi-classical formalism, which upholds the standard principles of quantum theory, and leads to consistent dynamics when any quantum state is considered – even when quantum fluctuations are large.

We also mention another formalism for semi-classical gravity called *stochastic gravity* [21], an extension of semi-classical gravity aimed at incorporating higher-order corrections to Einstein's equations. The main object of study in stochastic gravity is the so-called *Einstein-Langevin* equation

$$G_{ab}[g+h] + \Lambda(g_{ab} + h_{ab}) = \frac{8\pi G}{c^4} (\langle T_{ab}[g+h] \rangle + \xi_{ab}[g]), \qquad (3.40)$$

which describes the fluctuations h around a background metric g in terms of a stochastic noise source $\xi_{ab}[g]$, whose statistics are defined in terms of the two-point correlation function of the stress-energy tensor.⁷ The Einstein-Langevin equation is solved to obtain a probability

⁷It is easy to check that Equation (3.40) is invariant under linear diffeomorphisms, so long as the background metric g is a solution to the semi-classical Einstein equation.

distribution over the perturbations h, from which one can attempt to learn about the backreacting regime; for example, by studying correlation functions of metric degrees of freedom averaged over the noise $\xi_{ab}[g]$. Though useful as an effective theory in the presence of small quantum fluctuations, like the semi-classical Einstein equation, the Einstein-Langevin equation gives rise to non-linear evolution on the quantum state. Part II

Developing classical-quantum dynamics

Chapter 4

A classical-quantum Pawula theorem

Having introduced the relevant background material, we now enter into the first part of the main thesis, which is concerned with developing the classical-quantum formalism.

In this chapter, we study the positivity conditions of CQ dynamics in detail. In classical dynamics, in Chapter 2, we saw that we could write the master equation in terms of the moments of the transition probability amplitude via the Kramers-Moyal expansion [83, 84, 73]. The complete positivity of the dynamics means the transition amplitude must be positive, which can be used to derive constraints on the moments appearing in the moment expansion. Of particular relevance is the Pawula theorem [76], which states that the moment expansion either stops after the first or second moments, or else it must contain an infinite number of terms; in the former case, this restricts continuous dynamics to the well-known Fokker-Planck equation [73].¹

Here, we prove a classical-quantum version of the Pawula theorem, which follows from a combined CQ Cauchy-Schwarz inequality, Equation (4.9). We find that in order for a non-trivial classical-quantum interaction to be completely positive, the classical-quantum moment expansion must either contain an infinite number of terms or it must be of the form

¹It is important to note that if one truncates the series after n terms with $n \ge 3$, the resulting equation, although not positive, can still be used as an approximation to the dynamics in an appropriate regime. Indeed, one might attain a better approximation of certain classical dynamics by using an approximation that is not positive; one just has to be careful about the validity of the approximation [73].

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{n=2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{00}_{n,i_1\dots i_n} \varrho(z,t) \right) - \frac{\partial}{\partial z_i} \left(D^{0\alpha}_{1,i} \varrho(z,t) L^{\dagger}_{\alpha} \right) - \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} L_{\alpha} \varrho(z,t) \right) - i[H(z), \varrho(z,t)] + D^{\alpha \beta}_0(z) L_{\alpha} \varrho(z) L^{\dagger}_{\beta} - \frac{1}{2} D^{\alpha \beta}_0 \{ L^{\dagger}_{\beta} L_{\alpha}, \varrho(z) \}_+,$$

$$(4.1)$$

where $2D_2^{00} \succeq D_1 D_0^{-1} D_1^{\dagger}$ and $(\mathbb{I} - D_0 D_0^{-1}) D_1 = 0$ Here, D_0^{-1} is the generalized inverse of the matrix $D_0^{\alpha\beta}$, D_1 is a matrix in both α, i indices with entries $D_{1,i}^{0\alpha}$ and D_2^{00} is a matrix in i, j with entries $D_{2,ij}^{00}$.

Previously in the literature, examples of continuous classical-quantum master equations have been given [57, 60], but the general form was unknown [74]; just like examples of the Fokker-Plank equation were known before Pawula proved that it was the most general form of continuous master equation. Infinite moments are indicative of a jump process, and so we find that Equation (4.13) is the most general form of CQ master equation that has continuous trajectories in the classical phase space.

As a further consequence of the CQ Pawula theorem, we show that for classical-quantum dynamics to be completely positive, one must have a term representing pure Lindbladian evolution on the quantum state. In other words, the nature of completely positive dynamics necessarily results in the classical degrees of freedom inducing decoherence on the quantum state. Classicality induces classicality.

This chapter is based on the paper [5], which is work done in collaboration with Carlo Sparaciari, Barbara Šoda, and Jonathan Oppenheim.

4.1 Positivity conditions for classical-quantum interactions

We start with the master equation for autonomous classical-quantum dynamics

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D_{n,i_1\dots i_n}^{00}(z)\varrho(z,t) \right)
- i[H(z),\varrho(z)] + D_0^{\alpha\beta}(z)L_{\alpha}\varrho(z)L_{\beta}^{\dagger} - \frac{1}{2}D_0^{\alpha\beta}\{L_{\beta}^{\dagger}L_{\alpha},\varrho(z)\}_+
+ \sum_{\mu\nu\neq00} \sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1}\dots\partial z_{i_n}} \right) \left(D_{n,i_1\dots i_n}^{\mu\nu}(z)L_{\mu}\varrho(z,t)L_{\nu}^{\dagger} \right),$$
(4.2)

where we have suppressed the explicit t dependence on the moments $D_n(z,t)$. We know that the dynamics in Equation (4.2) will be positive so long as the transition amplitude $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ is a positive matrix and that positivity of $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ transfers naturally to positivity conditions on the short time moment expansions defined in Equation (2.84); for example, by considering the block form of (2.92)

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \begin{bmatrix} \delta(z,z') + \delta t W^{00}(z|z',t) & \delta t W^{0\beta}(z|z',t) \\ \delta t W^{\alpha 0}(z|z',t) & \delta t W^{\alpha \beta}(z|z',t) \end{bmatrix} + O(\delta t^2), \quad (4.3)$$

where we recall

$$D_{n,i_1\dots i_n}^{\mu\nu}(z',t) := \frac{1}{n!} \int dz W^{\mu\nu}(z|z',t), t(z-z')_{i_1}\dots(z-z')_{i_n}.$$
(4.4)

In this chapter, we also suppress the explicit time dependence of the transition amplitudes W(z|z',t), which can always be added back in.

We first note that the pure classical positivity condition, given by the 00 component $\Lambda^{00}(z, t + \delta t | z', t) = \delta(z, z') + \delta t W^{00}(z | z', t)$, leads to the well known Pawula theorem of classical autonomous dynamics discussed in Chapter 2; if any even moment D_n^{00} vanishes, then all moments with $n \geq 3$ must also vanish.

For the classical-quantum interaction to be completely positive, Equation (4.3) tells us $W^{\alpha\beta}(z|z')$ must be a positive matrix in α, β . We shall now use this fact to derive a family of inequalities which their moments $D_n^{\alpha\beta}(z)$ must satisfy, enabling us to prove a strengthened version of the Pawula theorem to CQ dynamics. In particular, we use this to show that the most general form of an autonomous classical-quantum master equation continuous in the classical phase is given by (4.13); any other master equation must contain finite-sized jumps in the phase space with non-zero probability. We further show that we *must* have a non-zero pure decoherence term; the completely positivity of the CQ interaction necessarily causes the classical system to induce decoherence on the quantum system. This, in turn, requires diffusion in phase space above a certain threshold value which we quantify further in Chapter 10.

4.2 Inequalities on the moments from positivity conditions

In this section, we derive a Cauchy-Schwartz inequality, Equation (4.9), applicable to any CQ map which is completely positive. We use it to derive a set of inequalities relating the moments

 $D_n^{\alpha\beta}(z)$ appearing in Equation (4.2). We first note that since $W^{\alpha\beta}(z|z')$ is a positive matrix, $W(z|z')(\varrho(z')) = W^{\alpha\beta}(z|z')L_{\alpha}\varrho(z')L_{\beta}^{\dagger}$ defines a completely positive operator. It will prove useful to use the vectorization map (2.99) to write the expansion coefficients $D_n^{\alpha\beta}(z)$ appearing in the dynamics of (4.2) in terms of the components of the completely positive operator W(z|z'). Explicitly,

$$D_{n,i_1\dots i_n}^{\alpha\beta}(z')L_{\alpha}\varrho(z')L_{\beta}^{\dagger} = \frac{1}{n!}\int dz \ W^{\alpha\beta}(z|z')L_{\alpha}\varrho(z')L_{\beta}^{\dagger}(z-z')_{i_1}\dots(z-z')_{i_n}$$

$$= \frac{1}{n!}\int dz \operatorname{Tr}\left[(\bar{L}_{\beta}\otimes L_{\alpha})^{\dagger}W^{vec}(z|z')\right]L_{\alpha}\varrho(z')L_{\beta}^{\dagger}(z-z')_{i_1}\dots(z-z')_{i_n}.$$

$$(4.5)$$

We could equally well write the completely positive operator W(z|z') in a different basis, which will prove useful. To that end, given an arbitrary basis on the underlying Hilbert space $\{|a\rangle\}$, we define the natural basis of operators on the Hilbert space E_{ab} via $E_{ab} = |a\rangle\langle b|$. In this basis

$$D_{n,i_1\dots i_n}^{\alpha\beta}(z)L_{\alpha}\varrho L_{\beta}^{\dagger} = D_{n,i_1\dots i_n}^{abcd}(z)E_{ca}\varrho E_{bd},$$
(4.7)

where as in Equation (4.5)

$$D_{n,i_1...i_n}^{abcd}(z') := \int dz \frac{1}{n!} \text{Tr} \left[(E_{db} \otimes E_{ca})^{\dagger} W^{vec}(z|z') \right] (z-z')_{i_1} \dots (z-z')_{i_n} \quad .$$
(4.8)

Now, let us prove a generalized form of the Cauchy-Schwartz inequality that we can use for the case of hybrid classical-quantum theories. It will take the form

$$\int d\Delta \operatorname{Tr}_{\mathcal{H}} \left[f(\Delta)^{\dagger} f(\Delta) T(\Delta) \right] \int d\Delta \operatorname{Tr}_{\mathcal{H}} \left[g(\Delta)^{\dagger} g(\Delta) T(\Delta) \right]$$

$$\geq \int d\Delta \operatorname{Tr}_{\mathcal{H}} \left[f^{\dagger}(\Delta) g(\Delta) T(\Delta) \right] \int d\Delta \operatorname{Tr}_{\mathcal{H}} \left[g^{\dagger}(\Delta) f(\Delta) T(\Delta) \right],$$
(4.9)

and it holds for any completely positive operator $T(\Delta)$ and arbitrary CQ operators $g(\Delta), f(\Delta)$. The above relation is easily derived by rearranging

$$\int d\Delta d\Delta' \operatorname{Tr}_{A,B} \left[(f_A(\Delta)g_B(\Delta') - g_A(\Delta)f_B(\Delta'))^{\dagger} (f_A(\Delta)g_B(\Delta') - g_A(\Delta)f_B(\Delta'))T_A(\Delta)T_B(\Delta') \right],$$
(4.10)

which is positive because each map $T_{A/B}(\Delta)$ acting on its share of a positive operator is a completely positive map. Using (4.9) with

$$T(z + \Delta, z) = W^{vec}(z + \Delta, z)$$

$$f(\Delta) = (E_{bb} \otimes E_{aa})\Delta_{i_1} \dots \Delta_{i_n}, \quad g(\Delta) = (E_{bd} \otimes E_{ac})\Delta_{i_{n+m}} \dots \Delta_{i_{2n+2m}},$$

$$(4.11)$$

and then integrating over z, we find the inequalities on the moments arising in the CQ equation

$$(2n!)(2n+2m)!D_{2n,i_{1}i_{1}\ldots i_{n}i_{n}}^{abab}D_{2n+2m,i_{n+m}i_{n+m}\ldots i_{2n+2m}i_{2n+2m}i_{2n+2m}}^{cdcd} \ge |(2n+m)!D_{2n+m,i_{1}\ldots i_{2n+m}}^{abcd}|^{2},$$

$$(4.12)$$

where we have used $D^{abcd} = (D^{badc})^*$, which follows from the fact that $W^{\alpha\beta}(z|z')$ is Hermitian.

4.3 A classical-quantum Pawula theorem

The inequalities in Equation (4.12) possess essentially the same structure as those arising in the classical Pawula theorem [76]. However, crucially, they must hold for all $n, m \ge 0$. The difference between the CQ and classical case arises since the zeroth moment of the map $\Lambda^{\mu\nu}(z, t+$ $\delta t|z', t)$ is of order $O(\delta t)$ for the classical-quantum interaction, while it is O(1) for the classical case due to the consistency condition at $\delta t = 0$. More precisely, for $\delta t = 0$ the CQ map in Equation (2.80) takes the form $\Lambda^{\mu\nu}(z,t|z',t) = \delta_0^{\mu}\delta_0^{\nu} + O(\delta t)$. As a result, the zeroth moment of the purely classical component of the CQ map is O(1); there can be no inequalities relating the zeroth moment of the classical dynamics to any higher-order moments since the zeroth moment always dominates. However, for the classical-quantum interaction, the zeroth moment is $O(\delta t)$, so there exist inequalities relating the zeroth moment to the higher order moments, leading to a strengthened version of the Pawula theorem – which we now state and prove. We define *non-trivial CQ evolution* to be one where $W^{\alpha\beta}(z|z')$ is somewhere positive so that the quantum system back-reacts on the classical system.

CQ Pawula Theorem. For non-trivial CQ evolution, we must have infinitely many moments defined in Equation (4.8), or else the master equation takes the form

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{n=2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{00}_{n,i_1\dots i_n} \varrho(z,t) \right) - \frac{\partial}{\partial z_i} \left(D^{0\alpha}_{1,i} \varrho(z,t) L_{\alpha}^{\dagger} \right) - \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} L_{\alpha} \varrho(z,t) \right) - i[H(z), \varrho(z,t)] + D^{\alpha\beta}_0(z) L_{\alpha} \varrho(z) L_{\beta}^{\dagger} - \frac{1}{2} D^{\alpha\beta}_0 \{ L_{\beta}^{\dagger} L_{\alpha}, \varrho(z) \}_+,$$

$$(4.13)$$

where $2D_2^{00} \succeq D_1 D_0^{-1} D_1^{\dagger}$ and $(\mathbb{I} - D_0 D_0^{-1}) D_1 = 0$ Here, D_0^{-1} is the generalized inverse of the matrix $D_0^{\alpha\beta}$, D_1 is a matrix in both α , i indices with entries $D_{1,i}^{0\alpha}$ and D_2^{00} is a matrix in i, j with entries $D_{2,ij}^{00}$. Furthermore, the zeroth moment, $D_0^{\alpha\beta}(z)$ cannot vanish.

Proof. Consider the inequality in Equation (4.12) for $n, m \ge 1$. Suppose any even CQ moment vanishes, so that $D_{2n}^{abcd} = 0$ for all a, b, c, d, then so must $D_{2n+m}^{abcd} = 0$, meaning all higher order moments also vanish. Furthermore, if $D_{2n+2m}^{abcd} = 0$, for all a, b, c, d, then $D_{2n+m}^{abcd} = 0$. Denoting r = n + m we see if $D_{2r}^{abcd} = 0$ then $D_{r+n}^{abcd} = 0$ for $n = 1 \dots r - 1$. To summarise: if any even moment vanishes $D_{2r}^{abcd} = 0$ we deduce that all higher order moments D_{2r+n}^{abcd} must vanish, as well as the moments D_{r+n}^{abcd} for $n = 1 \dots r - 1$. Except for the case r = 1, a moment expansion to order r + n will always contain an even moment, and so from repeated application of these properties, if any even moment vanishes, then $D_n^{abcd} = 0$ for all $n \ge 3$. This is the usual Pawula theorem for the coefficients $D_n^{\alpha\beta}$. However, for the CQ case, we also know that the block diagonal matrix $W^{\alpha\beta}(z|z')$ in Equation (4.3) also defines a completely positive map. We can use this to strengthen the condition. In particular, we also have the inequality (4.12) for $n = 0, m \ge 1$ which tells us

$$(2n)! D_0^{abab} D_{2m, i_1 i_1 \dots i_m i_m}^{cdcd} \ge |(m)! D_{m, i_1 \dots i_m}^{abcd}|^2.$$

$$(4.14)$$

Taking any even moment to be zero, we deduce that $D_4^{abcd} = 0$. But then from (4.14) we must then have $D_2^{abcd} = 0$, which in turn implies $D_1^{abcd} = 0$. Hence we see that if any even moment vanishes, then all of the moments $D_n^{\alpha\beta}$, $n \ge 1$ vanish.

Hence, we conclude that the block $W^{\alpha\beta}(z|z')$ describes pure quantum evolution and is determined by the zeroth moment $D_0^{\alpha\beta}$. What remains is to prove that $D_n^{0\alpha} = 0$ for $n \ge 1$ and Equation (4.3) will be positive if and only if the remaining couplings satisfy the positivity conditions $2D_2 \succeq D_1 D_0^{-1} D_1^{\dagger}$ and $(\mathbb{I} - D_0 D_0^{-1}) D_1 = 0$ where D_0^{-1} is the generalized inverse of the matrix with elements $D_0^{\alpha\beta}$, D_1 is a matrix in both α, i indices with entries $D_{1,i}^{0\alpha}$, and D_2^{00} is a matrix in i, j with entries $D_{2,ij}^{00}$.

If any of the even moments of D_n^{00} and $D_n^{\alpha\beta}$ greater than two vanish, our findings show that we can write the transition amplitude in block form (4.3) as

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \begin{bmatrix} \delta(z,z') + \delta t \sum_{n=0}^{2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}}\right) \left(D^{00}_{n,i_1\dots i_n}(z)\delta(z,z')\right) & \delta t \sum_{n=0}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}}\right) \left(D^{00}_{n,i_1\dots i_n}(z)\delta(z,z')\right) \\ \delta t \sum_{n=0}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}}\right) \left(D^{00}_{n,i_1\dots i_n}(z)\delta(z,z')\right) & \delta t D^{00}_0\delta(z,z') \end{bmatrix}$$

$$(4.15)$$

where we must remember that

$$D_0^{00}(z,t)\mathbb{I} + D_0^{0\alpha}L_{\alpha} + D_0^{\alpha0}L_{\alpha}^{\dagger} + D_0^{\alpha\beta}L_{\beta}^{\dagger}L_{\alpha} = \mathbb{I},$$
(4.16)

from the normalization condition in Equation (2.82).

To finalize the proof, we consider the positivity condition of $\Lambda^{\mu\nu}(z, t+\delta t|z', t)$ directly, which states that

$$\int dz A^*_{\mu}(z, z') \Lambda^{\mu\nu}(z, t + \delta t | z', t) A_{\nu}(z, z') \ge 0.$$
(4.17)

Since the 00 component of (4.15) contains a delta function $\delta(z, z')$ which is order O(1), whilst all other components are order $O(\delta t)$, Equation (4.17) will be always be positive unless we pick $A_0(z, z') = (z-z')^n f(z, z')$ where f(z, z) is non-zero. Since we know the blocks $\Lambda^{00}(z, t+\delta t|z', t)$ and $\Lambda^{\alpha\beta}(z, t+\delta t|z', t)$ are positive, we must consider the case in which A_{α} is non-zero, or else the off-diagonal terms of the block matrix (4.15) do not contribute. The only choice of $A_{\mu}(z, z')$, which gets rid of the leading order $\delta(z, z')$, has well-defined distributional derivatives and keeps the off-diagonal terms is $A_0(z, z') \sim (z - z')f(z, z')^n$ and $A_{\alpha} = a_{\alpha}(z, z')$, where $a_{\alpha}(z, z)$ is non-zero.

In the case in which we have many classical degrees of freedom z_i , we pick $A_0(z, z') = b^{i_1...i_n}(z, z')(z-z')_{i_1}...(z-z')_{i_n}$ for some vector $b^{i_1...i_n}(z, z')$. Choosing $A_{\mu}(z, z') = (b^i(z, z')(z-z')_{i_1}...(z-z')_{i_n}, a_{\alpha}(z, z')), n \ge 2$ gives the condition

$$\sum_{n=2}^{\infty} n! [(b^{i_1 \dots i_n *}(z, z) D^{0\alpha}_{n, i_1 \dots i_n}(z) a_{\alpha}) + ((b^{i_1 \dots i_n *}(z, z) D^{0\alpha}_{n, i_1 \dots i_n}(z) a_{\alpha})^*] \ge 0,$$
(4.18)

which can always be made negative for a suitable choice of b and a_{α} , hence we deduce $D_n^{0\alpha} = 0$ for $n \ge 2.^2$

The remaining conditions arise from choosing $A_0(z, z') = b^i(z, z')(z - z)_i$. With this choice of $A_0(z, z')$ we find the condition for positivity of $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$ is

$$2b^{i*}(z,z)D_{2,ij}^{00}b^{j}(z,z) + b^{i*}(z,z)D_{1,i}^{0\alpha}a_{\alpha}(z,z) + a_{\alpha}^{*}(z,z)D_{1,i}^{\alpha0}b^{i}(z,z) + a_{\alpha}^{*}(z,z)D_{0}^{\alpha\beta}a_{\beta}(z,z) \ge 0.$$
(4.19)

Defining D_2 to be the $n \times n$ matrix with elements $D_{2,ij}^{00}$, D_1 to be the $n \times p$ matrix in i, α with elements $D_{1,i}^{0,\alpha}$ and D_0 the $p \times p$ matrix in α, β with elements $D_0^{\alpha\beta}$ Equation (4.19) can be written in the form

$$\begin{bmatrix} b^*, \alpha^* \end{bmatrix} \begin{bmatrix} 2D_2 & D_1 \\ D_1^* & D_0 \end{bmatrix} \begin{bmatrix} b \\ \alpha \end{bmatrix} \ge 0, \tag{4.20}$$

²For example, suppose for a specific choice of α, b that Equation (4.18) was positive, then $b \to -b$ will be negative.

which is equivalent to the condition that the $(n + p) \times (n + p)$ matrix be positive semi-definite

$$D = \begin{bmatrix} 2D_2 & D_1 \\ D_1^* & D_0 \end{bmatrix} \succeq 0.$$

$$(4.21)$$

Since D_0 and D_2 must be positive semi-definite and since (4.21) is a block matrix, we know that the Schur complement of D_0 must be positive semi-definite, which implies

$$2D_2 \succeq D_1 D_0^{-1} D_1^{\dagger}, \ (\mathbb{I} - D_0 D_0^{-1}) D_1 = 0.$$
(4.22)

From Equation (4.22), we see that if D_0 vanishes, so does D_1 , and so we must have decoherence for non-trivial CQ evolution. The master equation then takes the form of Equation (4.13). In the case of a single Lindblad operator, the requirements implied by Equation (4.22) reduce to that found in [57].

We further show in Appendix B that the Lindblad operators in (4.13) can be arbitrary rather than requiring them to be traceless and orthogonal. The map will still be completely positive – so long as the conditions on the moments in Equation (4.22) are satisfied.

We have therefore found a strengthened version of the Pawula theorem for the CQ couplings – we either have infinitely many terms in the moment expansion, or else the dynamics must take the form of Equation (4.13). It is useful to note that since Equation (4.21) is in block form, it follows the two following conditions are equivalent

$$D \succeq 0 \Leftrightarrow D_2 \succeq 0, 2D_0 - D_1^{\dagger} D_2^{-1} D_1 \succeq 0, (I - D_2 D_2^{-1}) D_1 = 0 \text{ and}$$

$$D \succeq 0 \Leftrightarrow D_0 \succeq 0, 2D_2 - D_1 D_0^{-1} D_1^{\dagger} \succeq 0, (I - D_0 D_0^{-1}) D_1^{\dagger} = 0.$$
(4.23)

We shall refer to both conditions as the *decoherence-diffusion trade-off*, the consequences of which we explore in detail in Chapter 10. Although we refer to this as a trade-off between decoherence and diffusion, this terminology is only strictly appropriate for pure decoherence processes. More generally, it is a trade-off between Lindblad couplings and diffusion coefficients.

The decoherence-diffusion trade-off is precisely the reason that consistent classical-quantum dynamics exist. It necessitates stochastic unpredictability in the classical degree of freedom arising from D_2 and decoherence of quantum superpositions via D_0 , both effects of which cannot be made small when the quantum system back-reacts on the classical one via D_1 due to Equation (4.23). The trade-off between maintaining superpositions and classical uncertainty is what allows one to evade the no-go theorems of Feynman and others regarding the consistency of hybrid dynamics [36, 37, 38, 39, 40, 41, 42, 43, 44, 45, 46, 47]. The essence of arguments against quantum-classical interactions is that they would prohibit superpositions of quantum systems which source a classical field. Since different classical fields are perfectly distinguishable in principle, if the classical field is in a distinct state for each quantum system's state, causing it to decohere instantly. By satisfying the trade-off, the quantum system preserves coherence because diffusion of the classical degrees of freedom means that the state of the classical field does not determine the state of the quantum system.

In classical autonomous dynamics, the only time continuous autonomous process, in a sense that

$$\lim_{t \downarrow s} \frac{1}{t-s} \int_{|z-z'| > \delta} \mathrm{d}z \ p(t, z|s, z') = 0, \quad \forall \delta > 0,$$
(4.24)

is given by a diffusion process with dynamics described by the Fokker-Plank equation [73]. We can obtain a probability distribution for the classical degrees of freedom by taking the trace over the quantum system. To be precise, consider solving the CQ evolution; we can write $\varrho(z,t) = p(z,t)\overline{\varrho}(z,t)$ where $\sigma(z,t)$ is a normalized quantum state. The equation of motion for the classical degrees of freedom, obtained by tracing out the quantum system, is

$$\frac{\partial p(z,t)}{\partial t} = \sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(\bar{D}_{n,i_1\dots i_n}(z,t) p(z,t) \right), \tag{4.25}$$

where we define $\bar{D}_{n,i_1...i_n}(z,t) = \sum_{\mu\nu} D_{n,i_1...i_n}(z)^{\mu\nu}(z) \operatorname{Tr} \left[L_{\nu}^{\dagger} L_{\mu} \sigma(z,t) \right]$. It is worth emphasizing, however, that the resulting dynamics, although arising from autonomous dynamics linear in the CQ state, are non-Markovian and so cannot be reduced to the models of [57]. Non-Markovianity is encoded by the fact that the \bar{D}_n coefficients depend on the quantum state $\sigma(z,t)$, which can act as a memory for the full CQ dynamics. If we have an infinite number of terms in Equation (4.25), then the dynamics will contain finite-sized jumps with non-zero probability. More precisely, in the case with infinite moments in the classical-quantum dynamics, there will always exist states for which there are finite-sized jumps with finite probability in the classical phase space, take $\rho(z,t) = p(z,t)\mathbb{I}$ for example. In Chapter 5, we construct an explicit algorithm to study continuous classical-quantum dynamics in terms of continuous trajectories in phase space.

4.4 Discussion

In this chapter, we have introduced the most general form of autonomous classical-quantum master equations that are continuous in the classical phase space, given by Equation (4.13). Any other master equation necessarily causes discrete, finite-sized jumps in phase space, or else it violates the conservation of probabilities or fails to be completely positive on the quantum system. We achieved this by introducing a classical-quantum Cauchy-Schwarz inequality, which enabled us to derive various inequalities which the moments of the transition amplitude must satisfy.

The master equation we find generalizes the Fokker-Plank equation of open classical systems and the Lindblad equation of open quantum systems to the case of coupled classical-quantum systems, providing a reference for the study of hybrid classical-quantum dynamics in the future. Previously only examples of continuous classical coupling were given [57, 60], but its general form was unknown [74]. The continuous master equation will play a central role in many chapters of the thesis; in Chapter 5, we show that it can be unraveled by a set of coupled stochastic differential equations, in analogy with the unraveling of the Fokker-Plank and Lindblad equations by stochastic differential equations, which we show leads to an improved semi-classical formalism which is consistent even in the presence of large quantum fluctuations; in Chapters 6 and 8 we show that one can arrive at a path integral representation for the continuous master equation, which helps understand whether CQ dynamics can retain space-time symmetries such as diffeomorphism invariance.

Indeed, in the context of classical-quantum theories of gravity [60, 28, 49, 3], if the spacetime metric undergoes continuous dynamics, then one expects a version of Equation (4.13) to generate it. To this end, the field-theoretic version of Equation (4.13) is given in [4, 3] and Appendix G. Constructing consistent theories of CQ general relativity then amounts to an appropriate choice of Lindblad operators and couplings D_0, D_1, D_2 ; realizations of the master equation for the case of gravity have been given in [60, 28, 8], which we study in the later parts of the thesis. The fact that the amount of classical diffusion D_2 is lower bounded by the decoherence coupling D_0 leads to potential experimental signatures of a classical gravitational field [4], which we discuss in detail in Chapter 10.

Chapter 5

Unraveling classical-quantum dynamics

In analogy with the unraveling of quantum Lindblad equations [99, 101, 100], and the unraveling of classical master equations [73, 82], hybrid master equations have been unraveled by stochastic differential equations. This is true for both the continuous master equations [74], and those which contain jumps [10]. In the literature, unravelings of continuous master equations have only been derived in specific examples, usually surrounding the context of continuous measurements [52, 74, 49]. In this chapter, we find the general form of continuous classical-quantum unraveling using the continuous master equation found in the previous chapter, which provides a concrete algorithm for simulating continuous classical quantum dynamics.

The resulting equations of motion are natural generalizations of the standard semi-classical equations of motion. However, since the resulting dynamics are linear in the combined classicalquantum state, it does not lead to the pathologies which usually follow evolution laws based on expectation values. In particular, the evolution laws we present account for correlations between the classical and quantum systems, which resolves issues associated with other semi-classical approaches. It is necessary to include stochasticity in the classical part of the dynamics and a Lindbladian coupling in the quantum part of the system in order to preserve the linearity of the dynamics and the positivity of the classical-quantum state. Nonetheless, despite a break-down of predictability in the classical degrees of freedom, we find the quantum state evolves deterministically conditioned on the classical trajectory, the trade-off between the Lindbladian coupling and the diffusion coefficient found in Chapter 4 is saturated.

In the context of gravity, the solution to the dynamics is described by a probability distribution over continuous 4-geometries and an associated quantum state, which could potentially be used to simulate the *dynamics* of classical gravity interacting with quantum matter, going beyond the standard semi-classical regime. However, we do not study this in detail in this chapter.

We also prove that resulting dynamics completely parameterizes continuous measurement and (Markovian) feedback procedures, showing the equivalence of the continuous measurement [52, 74, 49] and hybrid [60, 28] approaches to continuous classical-quantum coupling, which are often treated as being mathematically distinct in discussions of classical-quantum gravity [118]. Because of the equivalence between continuous CQ dynamics, in Section 5.5.2, we propose that CQ dynamics gives rise to an effective theory of quantum measurement. One first identifies a classical system that acts as a measurement device and a quantum system to be measured and studies the effective CQ dynamics of the interacting classical and quantum systems. The details of the measurement apparatus and coupling are then encoded in a handful of phenomenological parameters, i.e., those appearing in Equation (4.13), which govern the strength of the backreaction of the quantum system and the strength of the measurement. Within this prescription, there is no need to discuss a measurement outcome, nor when or how a measurement occurs; there are only effective dynamics of continuously interacting classical and quantum systems.

This chapter is based on [6], which is work done in collaboration with Isaac Layton and Jonathan Oppenheim.

5.1 The standard semi-classical equations

For the case of gravity, we reviewed the semi-classical approach to incorporate the back-reaction of a quantum system on a classical system in Chapter 3. There, the standard approach to define back-reaction is via the semi-classical Einstein equations, which source the Einstein tensor $G_{\mu\nu}$ by the expectation value of the stress-energy tensor $T_{\mu\nu}$ [19, 20]

$$G_{\mu\nu} = \frac{8\pi G}{c^4} \langle T_{\mu\nu} \rangle, \qquad (5.1)$$

with the quantum state $|\psi\rangle$ at time t determined by a Hamiltonian H that depends on classical degrees of freedom z (here taken to be the gravitational metric)

$$\frac{d|\psi\rangle}{dt} = -iH(z)|\psi\rangle.$$
(5.2)

The dynamics represented by Equations (5.2) and (5.1) may be understood as a special case of a more general approach taken to describe back-reaction, which we shall refer to as the *standard semi-classical approach*. We take the standard semi-classical equations to be the system of equations where the quantum evolution is governed by a phase-space dependent Hamiltonian, as in Equation (5.2), and the classical evolution undergoes a back-reaction force determined by an expectation value of the quantum state,

$$dz = \{H_c, z\} + \langle \{H_I(z), z\} \rangle_{|\psi\rangle}, \tag{5.3}$$

where H_c is a purely classical Hamiltonian and H_I is a quantum interaction Hamiltonian which determines the strength of the back-reaction.

The standard semi-classical equations are often studied in molecular dynamics and are called the mean-field equations [119, 120]. As with gravity, they can be useful in some regimes where the quantum fluctuations are small, and no correlation is generated between the classical and quantum systems.

As outlined in Chapter 3, the problem with the standard semi-classical equations, away from small fluctuations, is that they fail to properly account for correlations between the classical and quantum degrees of freedom. Consequently, they give rise to non-linear evolution on the density matrix, which leads to violations of the standard principles of quantum theory [48, 18, 49, 50, 51] when the quantum fluctuations are large and correlations are dynamically generated between the classical and quantum systems. In particular, when quantum fluctuations are significant, i.e., for the case of Schrodinger cat states of massive bodies [29, 30], or vacuum fluctuations during inflation [31, 32, 33, 34], the semi-classical equations do not yield consistent dynamics, and an alternate effective theory of back-reaction is required. The standard semi-classical equations share some similarities with the Born-Oppenheimer approximation in that they do not allow correlations to build up between the classical and quantum systems. In this chapter, we will make progress towards this goal using the general form of autonomous classical-quantum dynamics found in the previous chapter. In particular, we show that CQ dynamics can be used to give rise to a consistent semi-classical formalism, which upholds the standard principles of quantum theory and leads to consistent dynamics when any quantum state is considered. Note, though the assumption of autonomy is reasonable for any theory viewed as fundamental, it may not hold in an effective theory, where the dynamics can generally be non-Markovian. Moreover, when the classical system arises as a limit of a quantum one, there may not always be a description of it in terms of continuous classical trajectories. A more complete semi-classical picture should be able to account for these issues in more detail. We discuss this in more detail in Section 5.7 when we conclude the chapter.

5.2 Classical-quantum trajectories

We start by defining the primary objects of our semi-classical description. The standard semiclassical equations describe deterministic classical-quantum trajectories. However, more generally, one can consider dynamics that generate probability distributions over classical trajectories in phase space and quantum trajectories in Hilbert space. We saw in Chapter 4 that this was necessary if the dynamics are autonomous due to the requirement of a non-zero diffusion coefficient D_2 in the master equation of Equation (5.5).

We assume that at all times, the semi-classical system is fully characterized by the pair (z, ρ) . Here z denotes the classical degrees of freedom ρ denotes a quantum state. The entire evolution of the semi-classical system is thus characterized by a classical-quantum trajectory, which we denote by $\{(Z_t, \rho_t)\}_{t>0}$. Each of the trajectories occurs with some probability. Both Z_t and ρ_t can be understood as random variables, taking values in the classical phase space and the space of density operators, respectively. These are a natural combination of the trajectories considered in Chapter 2, where we saw the Fokker-Plank equation was unraveled by stochastic differential equations governing the evolution of the Z_t , and the Lindblad equation could be unraveled by stochastic differential equations governing the evolution of quantum trajectories $\rho_t = |\psi_t\rangle\langle\psi_t|$.

We can relate the classical-quantum trajectories to the CQ state by defining

$$\varrho(z,t) = \mathbb{E}[\delta(z-Z_t)\rho_t], \tag{5.4}$$

where the expectation value is over all possible values of the underlying noise process. Equation (5.4) is easily seen to be a normalized CQ state. In particular, taking the trace of the quantum system, we arrive at the probability distribution over classical variables p(z,t) is described by $p(z,t) = \mathbb{E}[\delta(z - Z_t)]$ while integrating out the classical degrees of freedom z, we arrive at $\mathbb{E}[\rho_t]$, which is a normalized quantum state. Equation (5.4) can be understood as weighting the quantum state achieved for a particular value of the noise process by the probability it accompanies a specific classical value and then averaging over all possible values of the noise process.

5.3 Unraveling continuous classical-quantum dynamics

We now find the general classical-quantum unraveling for the continuous autonomous classicalquantum master equation found in the previous chapter

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{n=2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{00}_{n,i_1\dots i_n} \varrho(z,t) \right) - \frac{\partial}{\partial z_i} \left(D^{0\alpha}_{1,i} \varrho(z,t) L_{\alpha}^{\dagger} \right) - \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} L_{\alpha} \varrho(z,t) \right) - \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} \rho(z,t) L_{\alpha}^{\dagger} \right) - \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} L_{\alpha} \varrho(z,t) \right) - \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} \rho(z,t) \right) - \frac{\partial}{\partial z_i} \left(D$$

recalling that the conditions for complete positivity are $2D_2^{00} \succeq D_1 D_0^{-1} D_1^{\dagger}$ and $(\mathbb{I} - D_0 D_0^{-1}) D_1 = 0$, where, D_0^{-1} is the generalized inverse of the matrix $D_0^{\alpha\beta}$, D_1 is a matrix in both α, i indices with entries $D_{1,i}^{0\alpha}$ and D_2^{00} is a matrix in i, j with entries $D_{2,ij}^{00}$.

Explicitly, the dynamics of Equation (5.5) is unraveled by the pair of stochastic differential equations

$$dZ_{t,i} = D_{1,i}(Z_t)dt + \langle D_{1,i}^{\alpha 0}(Z_t)L_{\alpha} + D_{1,i}^{0\alpha}(Z_t)L_{\alpha}^{\dagger} \rangle dt + \sigma_{ij}(Z_t)dW_j,$$
(5.6)

$$d\rho_t = -i[H(Z_t), \rho_t]dt + D_0^{\alpha\beta}(Z_t)L_{\alpha}\rho L_{\beta}^{\dagger}dt - \frac{1}{2}D_0^{\alpha\beta}(Z_t)\{L_{\beta}^{\dagger}L_{\alpha}, \rho_t\}_+ dt$$

$$+ D_{1,j}^{\alpha0}\sigma_{ji}^{-1}(Z_t)(L_{\alpha} - \langle L_{\alpha}\rangle)\rho_t dW_i + D_{1,j}^{0\alpha}\sigma_{ji}^{-1}(Z_t)\rho_t(L_{\alpha}^{\dagger} - \langle L_{\alpha}^{\dagger}\rangle)dW_i,$$
(5.7)

where σ_{ij} is a positive semi-definite matrix which is defined in terms of the diffusion matrix of Equation (5.5) by $D_{2,ij}^{00} = \frac{1}{2}\sigma_{ik}\sigma_{kj}^{T}$. In Equation's (5.6) and (5.7), σ^{-1} denotes the generalized inverse of σ , and W_i is the standard multivariate Wiener process satisfying the Ito rules

$$dW_i dW_j = \delta_{ij} dt, \ dW_i dt = 0. \tag{5.8}$$

Although the expectation values appear in the equations of motion for both Z_t and ρ_t when averaged over the noise process, they give rise to a linear master equation on the CQ state; the fact that the dynamics give rise to non-linear evolution on ρ_t is not inconsistent with quantum theory. Indeed, the same is true for unravelings of the GKSL equation [48] and also for quantum measurements. For example, the conditioning of the quantum state after a measurement outcome (which we label by the classical variable *i*) is obtained is a non-linear map, $\rho \rightarrow \frac{\rho_i}{\text{Tr}[\rho_i]}$, but averaging over all possible outcomes is a linear map on the space of density matrices. In this case, the non-linearity arises purely because of the normalization of the quantum state. One can show the same is true for the expectation values appearing in the quantum part of the evolution. In [6], it is shown that one can write the quantum evolution in Equation (5.7) by considering un-normalized quantum states.

Conversely, the expectation value appearing in the classical equation of motion is not due to normalization. Instead, it represents that the drift of the classical system is generically unknown, even for pure quantum states, and the noise is required precisely so that one cannot determine the drift exactly while the quantum state has coherence. In particular, the equation for dZ_t should be interpreted as an equation that governs the *statistics* of dZ_t , not that the expectation value actually sources the drift. Again, this feature is also true for measurements on subsystems in standard quantum theory. Pure quantum states only encode the statistics of measurement outcomes; one can only say that the outcome *i* will occur with certainty if the quantum state is in an eigenvalue of the operator which is being measured. We shall show that the only case where the drift can be determined exactly in Equation (5.6), i.e., the $\sigma \to 0$ limit, is when the quantum states are in eigenstates of the drift operator.

Though the dynamics of Equation (5.7) is stochastic, an observer with knowledge of the entire classical trajectory up-to time t, $\{Z_t\}_{t\leq t}$ can deduce ρ_t from changes in Z_t . This property is shown explicitly in [6], where the quantum state evolution is presented without any noise terms, but instead in terms of changes to the classical degree of freedom dZ_i . This form of equation can be found by inverting Equation (5.6) to find dW_i in terms of dZ_i , and substituting into Equation (5.6). We also give an explicit example of this in Section 5.5.2.

On first inspection, the master Equation of (5.5), and the unraveling in Equation's (5.6) and (5.7) lead to a loss of quantum information due to the presence of the decoherence terms with coefficients D_0 . However, for an initially pure quantum state, we find that when the decoherence-diffusion trade-off is saturated, which we take to mean that $D_0 = D_1(\sigma\sigma^T)^{-1}D_1^{\dagger}$, the quantum state ρ_t remains pure, i.e., Tr $[(\rho_t + d\rho_t)^2] \rightarrow 1$ and there is no loss of quantum information [6].

We can see this more explicitly via the pure state unraveling

$$d|\psi\rangle_t = -iH(Z_t)|\psi\rangle_t dt + D_{1,j}^{\alpha 0} \sigma_{ij}^{-1}(Z_t)(L_\alpha - \langle L_\alpha \rangle)|\psi\rangle_t dW_i - \frac{1}{2} D_0^{\alpha \beta}(Z_t)(L_\beta^{\dagger} - \langle L_\beta^{\dagger} \rangle)(L_\alpha - \langle L_\alpha \rangle)|\psi\rangle_t dt + \frac{1}{2} D_0^{\alpha \beta}(\langle L_\beta^{\dagger} \rangle L_\alpha - \langle L_\alpha \rangle L_\beta^{\dagger})|\psi\rangle_t dt,$$
(5.9)

which, using the standard Ito rules

$$d\rho = d|\psi\rangle\langle\psi| + |\psi\rangle d\langle\psi| + d|\psi\rangle d\langle\psi|, \qquad (5.10)$$

is equivalent to Equation (5.7) when the decoherence diffusion trade-off is saturated. Thus, despite the loss of predictability in the classical degrees of freedom, classical-quantum theories saturating the trade-off can preserve the purity of the quantum states when conditioned on the classical trajectory; this is one important difference between classical-quantum unravelings and quantum unravelings of the GKSL equation.

In hindsight, the fact that one can preserve the purity of the quantum system conditioned on the classical trajectory is perhaps an expected feature of hybrid dynamics. In Section 5.5.1, we show that the dynamics of Equations (5.6) and (5.7) have an equivalent description in terms of a generalization of the procedure given in [96], where an auxiliary classical degree of freedom is sourced by the measurement signal of a continuous measurement, and we also allow for an auxiliary variable to have its own purely classical dynamics; in this sense, Equations (5.6) and (5.7) form a complete parameterization for continuous measurement-based classical-quantum control where one also allows for continuous control on the classical system, and are similar to measurement based feedback equations familiar in quantum control [121, 122].

The fact that the purity of quantum states can be maintained when conditioned on the classical trajectory is then an expected feature. In continuous quantum measurement, weak measurements of a pure quantum state are concatenated in time, with the dynamics of both the quantum system and the measurement signal forming a pair of stochastic differential equations similar to that of Equation (5.6) and (5.9). While, in general, the quantum state evolution will be random due to the probabilistic nature of quantum measurements, conditioned on the series of measurement outcomes, the state will remain pure. Similarly, in open quantum systems, pure states become mixed as they become entangled (correlated) with their environment, which is, practically speaking, inaccessible. However, with knowledge of both the system and the environment dynamics, the quantum state of the entire system will remain pure; here, the classical degree of freedom behaves as the environment in restoring the purity of the state.

5.3.1 Deriving the unraveling

To arrive at the unraveling of Equation's (5.6) and (5.7) we start by noting that the dynamics of Z_t and ρ_t induce the following evolution on the CQ state $\rho(z,t) = \mathbb{E}[\delta(Z_t - z)\rho_t]$,

$$d\varrho(z,t) = \frac{\partial \varrho(z,t)}{\partial t} dt = \mathbb{E}[d(\delta(Z_t - z)\rho_t)].$$
(5.11)

One must therefore calculate

$$\mathbb{E}[d(\delta(Z_t - z)\rho_t)] = \mathbb{E}[d\delta(Z_t - z)\rho_t + \delta(Z_t - z)d\rho_t + d\delta(Z_t - z)d\rho_t].$$
(5.12)

For clarity, we shall go through each term individually. Combining Ito's lemma with the equation of motion for the classical variable in (5.6), the first term in Equation (5.12) reads

$$\mathbb{E}[d\delta(Z_t - z)\rho_t] = \mathbb{E}[\frac{\partial}{\partial Z_i}[\delta(Z_t - z)]\rho_t(D_{1,i}^{00}(Z_t, t) + \langle D_{1,i}^{\alpha0}(Z_t, t)L_\alpha + D_{1,i}^{0\alpha}(Z_t, t)L_\alpha^\dagger \rangle)]dt + \mathbb{E}[\frac{1}{2}\frac{\partial^2}{\partial Z_i \partial Z_j}[\delta(Z_t - z)]\rho_t\sigma_{ik}(Z_t, t)\sigma_{kj}^T(Z_t, t)]dt.$$
(5.13)

We can use some well-known facts about the delta functional to simplify Equation (5.13). Using the two identities $\partial_{Z_i}\delta(Z-z) = -\partial_{z_i}\delta(Z-z)$ and $f(Z)\delta(Z-z) = f(z)\delta(Z-z)$ for any function f, the right hand side of Equation (5.13) becomes

$$-\frac{\partial}{\partial z_i} \mathbb{E}[\delta(Z_t-z)\rho_t(D_{1,i}^{00}(z)+\langle D_{1,i}^{\alpha0}(z)L_{\alpha}+D_{1,i}^{0\alpha}(z)L_{\alpha}^{\dagger})\rangle]dt + \frac{\partial^2}{\partial z_i\partial z_j} \mathbb{E}[\delta(Z_t-z)\rho_t D_{2,ij}^{00}(z)]dt.$$
(5.14)

Using the definition of the CQ state in Equation (5.4), we arrive at

$$\mathbb{E}[d\delta(Z_t-z)\rho_t] = (-\frac{\partial}{\partial z_i}[\varrho(z)D_{1,i}^{00}(z)] - \frac{\partial}{\partial z_i}[\langle D_{1,i}^{\alpha 0}L_{\alpha}\rangle\varrho] - \frac{\partial}{\partial z_i}[\varrho(z)\langle D_{1,i}^{0\alpha}L_{\alpha}^{\dagger}\rangle] + \frac{\partial^2}{\partial z_i\partial z_j}[\varrho(z)D_{2,ij}^{00}(z)])dt$$

$$\tag{5.15}$$

The second term in Equation (5.12) is simpler to calculate and gives the pure quantum evolution terms

$$\mathbb{E}[\delta(Z_t - z)d\rho_t] = i[H(Z_t), \rho_t]dt + D_0^{\alpha\beta}(Z_t)L_\alpha\rho L_\beta^{\dagger}dt - \frac{1}{2}D_0^{\alpha\beta}(Z_t)\{L_\beta^{\dagger}L_\alpha, \rho_t\}_+dt$$
(5.16)

For the final term in Equation (5.12), only the second-order terms $dW^2 = dt$ are relevant. We find

$$\mathbb{E}[d\delta(Z_t - z)d\rho_t] = \mathbb{E}[\frac{\partial}{\partial Z_i}[\delta(Z_t - z)]\rho_t D_{1,i}^{\alpha 0}(Z_t, t)(L_\alpha - \langle L_\alpha \rangle) + D_{1,i}^{0\alpha}(Z_t, t)(L_\alpha^\dagger - \langle L_\alpha^\dagger \rangle))]dt, \quad (5.17)$$

and again using the standard properties of the delta function

$$\mathbb{E}[d\delta(Z_t - z)d\rho_t] = -\frac{\partial}{\partial z_i} [D^{\alpha 0}_{1,i}(z)(L_\alpha - \langle L_\alpha \rangle)\varrho(z) + D^{0\alpha}_{1,i}\varrho(z)(L^{\dagger}_\alpha - \langle L^{\dagger}_\alpha \rangle)]dt.$$
(5.18)

Summing the three contributions gives the equation of motion for $\partial_t \mathbb{E}[\delta(Z_t - z)\rho_t] = \partial_t \varrho(z)$ to be that of the continuous master equation in Equation (5.5).

5.4 Hamiltonian unravelings

Having presented the general dynamics for Z_t and ρ_t , we now turn to an important example of the dynamics where the back-reaction is generated by a Hamiltonian and the decoherencediffusion trade-off is saturated. Specifically, we take the pure classical part of the drift determined by D_1^{00} to be generated by a classical Hamiltonian $H_C(z)$. For the interaction term, one can use the freedom in the choice of Lindblad operators to pick $L_{\alpha} = \{Z_{\alpha}, H_I\}$, where $H_I(z)$ is an interaction Hamiltonian, and then set $D_{1,i}^{0\alpha} = \frac{1}{2}\delta_i^{\alpha}$. In this case, we arrive at a set of equations that we called "the healed semi-classical equations" in [6]. The classical dynamics are given by

$$dZ_{t,i} = \{Z_{t,i}, H_C(Z_t)\}dt + \langle \{Z_{t,i}, H_I(Z_t)\}\rangle dt + \sigma_{ij}(Z_t)dW_j,$$
(5.19)

while the quantum evolution takes the form

$$d|\psi\rangle_{t} = -i(H_{0} + H_{I}(Z_{t}))|\psi\rangle_{t}dt + \frac{1}{2}\sigma_{ij}^{-1}(\{Z_{j}, H_{I}\} - \langle\{Z_{j}, H_{I}\}\rangle)|\psi\rangle_{t}dW_{i}$$
(5.20)
$$- \frac{1}{8}\sigma_{ij}^{-1}\sigma_{ik}^{-1}(\{Z_{j}, H_{I}\} - \langle\{Z_{j}, H_{I}\}\rangle)(\{Z_{k}, H_{I}\} - \langle\{Z_{k}, H_{I}\}\rangle)|\psi\rangle_{t}dt,$$

where in the above σ is any real matrix satisfying $(\delta_j^i - (\sigma \sigma^{-1})_j^i)\delta_i^{\alpha} = 0$, which arises due to the complete positivity condition $(\mathbb{I} - D_2 D_2^{-1})D_1 = 0$. For a given initial quantum state $|\psi_i\rangle$ and classical state z_i , these coupled stochastic differential equations determine the probability of ending up in any final pair of states z_f and $|\psi_f\rangle$. An early example of this dynamics for the special case of linear coupling between two particles, one classical and one quantum, was described in [75]. Many examples and simulations can be found in our paper [6].

We now briefly discuss features of the healed semi-classical equations. The quantum state evolves under Equation (5.20), with the first term responsible for pure unitary evolution and the final two terms describing continuous stochastic evolution that tends to drive the quantum state towards a joint eigenstate of the operators $\{Z_i, H_I\}$ [123]; Equation (5.20) is mathematically equivalent to a continuous measurement of the operator $\{Z_i, H_I\}$ [121, 122], and exponentially suppress trajectories away from the minimum value of

$$\frac{1}{8}\sigma_{ij}^{-1}\sigma_{ik}^{-1}(\{Z_j, H_I\} - \langle\{Z_j, H_I\}\rangle)(\{Z_k, H_I\} - \langle\{Z_k, H_I\}\rangle)|\psi\rangle_t,$$
(5.21)

which is when the quantum state is in an eigenstate of $\{Z_i, H_I\}$ [123]. Note, this does not necessarily mean that Equation (5.20) has a fixed point since the pure quantum evolution does not always share the same eigenvectors as $\{Z_i, H_I\}$. In this case, there is a battle between the pure Hamiltonian evolution that creates coherence and the remaining evolution that destroys it.¹

On the other hand, the classical evolution in Equation (5.19) consists of a term describing a purely classical drift, another which describes the quantum back-reaction on the classical system, and a diffusion term. Despite the appearance of an expectation value in the backreaction term, the dynamics give statistics for Z_t as if the classical system were diffusing around a force given by a random eigenstate of the operators $\{Z_i, H_I\}$. In particular, since $\{Z_i, H_I\}$ is Hermitian, we can decompose the quantum state in terms of the eigenvectors of $\{Z_i, H_I\} =$ $\sum_j h_i^j(Z) |h_i^j\rangle$ and write $|\psi\rangle = \sum_j \sqrt{p_j} |h_i^j\rangle$. We find that

$$\langle \{Z_i, H_I\} \rangle = \sum_j p_j h_i^j, \tag{5.22}$$

which can be interpreted as sources of the force by an eigenvector h_i^j with probability p_j .

¹We thank Isaac Layton for pointing this out.

The only free parameters of the model are encoded in the matrix σ_{ij} , which governs both the rate at which the quantum state evolves to an eigenstate and the rate of diffusion of the classical system. The fact that the two rates are inversely related is a consequence of the decoherence-diffusion trade-off, which ensures that there can be no quantum back-reaction without associated diffusion or decoherence of the quantum system.

The differences between the healed semi-classical and standard semi-classical equations, given by Equations (5.2) and (5.3), are evident. The classical evolution of standard semiclassical Equation (5.3) takes the form of the healed semi-classical Equation (5.19) but without the noise term. Similarly, the quantum evolution of the standard semi-classical Equation (5.2) takes the form of the healed semi-classical Equation (5.20) but without both the stochastic diffusion term and the term which drives quantum states to an eigenvector of the drift operator $\{Z_i, H_I\}$. Taking the limit of deterministic classical evolution in Equation (5.19), $\sigma \to 0$ means that the quantum states are very quickly driven to eigenstates due to the appearance of σ^{-1} in Equation (5.20). Similarly, sending $\sigma^{-1} \to 0$ to make the quantum evolution unitary results in wildly stochastic classical dynamics. The decoherence diffusion trade-off prevents the recovery of both equations in any limit.

The standard semi-classical equations are thus inconsistent if applied to all states. Using Equations (5.19) and (5.20), we can find a prescription to test the validity of the standard semi-classical equations for any given quantum state $|\psi\rangle_t$. Firstly, we note that the low noise limit $\sigma \to 0$ must be taken in Equation (5.19) to recover the standard semi-classical equation for the classical degrees of freedom. For the quantum state $|\psi\rangle_t$ to then be also effectively described by unitary evolution, as in the standard semi-classical approach in Equation (5.2), the quantum state must then generally satisfy

$$+\frac{1}{2}\sigma_{ij}^{-1}(\{Z_j, H_I\} - \langle \{Z_j, H_I\} \rangle) |\psi\rangle_t dW_i -\frac{1}{8}\sigma_{ij}^{-1}\sigma_{ik}^{-1}(\{Z_j, H_I\} - \langle \{Z_j, H_I\} \rangle) (\{Z_k, H_I\} - \langle \{Z_k, H_I\} \rangle) |\psi\rangle_t dt \approx 0.$$
(5.23)

More precisely, Equation (5.23) should be negligible in comparison to the pure Hamiltonian evolution of the quantum state, so the valid quantum states will be approximate eigenstates of $\{Z_i, H_I\}$, which minimize (5.23). It is worth mentioning that in the Newtonian limit of semi-classical gravity discussed in Chapter 3, the interaction is dominated by the mass density $\frac{\partial H_I}{\partial \Phi} = m(x)$ and we see that the standard semi-classical equations are valid only when the

quantum state is in an approximate eigenstate of the mass density operator, which excludes macroscopic superpositions, as well as states which are spatially entangled: essentially the quantum state of matter must be approximately classical [24, 25, 31].

An alternative viewpoint is that when the low noise $\sigma \to 0$ limit is taken, the quantum dynamics rapidly cause the quantum state to evolve to the eigenstate $\{Z_i, H_I\}$, with the probabilities determined by the Born rule [123]. The classical evolution is then well approximated by conditioning on eigenstates of the quantum state and then evolving the classical degree of freedom using the eigenstate. The classical system is probabilistic in this limit, but only due to the probability distribution over the decohered quantum eigenstates. In practice, the semiclassical Einstein equations are often used this way; for example, when considering statistical mixtures or when describing vacuum fluctuations during inflation where quantum states decohere on super-horizon scales and give rise to a classical probability distribution over space-time perturbations. This use is justified as a limiting case of Equations (5.19) and (5.20) when the classical noise is small. The advantage of Equations (5.19) and (5.20) is that they can more generally be used to understand what happens to the dynamics of the classical system when the quantum state still has coherence, giving rise to a consistent semi-classical theory even in the presence of large quantum fluctuations.

5.5 Comparison to measurement and feedback

In this section, we compare our result to previous methods of generating consistent classicalquantum dynamics using continuous measurement and feedback approaches [75, 49, 52]. In these approaches, the classical degree of freedom is sourced by the outcomes of a continuous measurement. By construction, such approaches are completely positive and lead to consistent coupling between classical and quantum degrees of freedom. The stochasticity of the dynamics is due to the continuous measurement, and the non-linearity is due to the state update rule, meaning the dynamics of [75, 49, 52] take a similar form to Equations (5.19) and (5.20). However, it is worth noting some differences between the previous approaches based on continuous measurement and the one we have presented in this Chapter. Firstly, the dynamics we present allow for the classical degrees of freedom to be independent of the quantum degrees of freedom and have their own dynamics, described via the purely classical evolution term $D_{1,i}^{00}$. This allows us to apply the dynamics to more complex CQ scenarios, for example, when the Hamiltonian H(z) is non-linear in z; this is necessary to consider semi-classical dynamics of gravity beyond the weak field limit. As a result, the dynamics, while autonomous on both the classical and quantum systems, can be non-Markovian on the classical and quantum systems alone. It, therefore, does not always reduce to pure Lindbladian evolution on the quantum system, such as in [49, 124, 52].

Secondly, we have taken the dynamics on the phase space degrees of freedom to be continuous. In the measurement and feedback approaches of [49, 52], the classical degrees of freedom evolve discontinuously because the classical coordinate is directly sourced by the outcome J_i of a continuous measurement which is a discontinuous stochastic random variable. To obtain continuous classical degrees of freedom, one can instead source the conjugate momenta of the canonical coordinates via the measurement signal $J_i dt$. This approach is taken in [75], which leads to a special case of our dynamics.

We now show that all continuous CQ dynamics have an equivalent description in terms of a generalization of the procedure given in [75], where an auxiliary classical degree of freedom is sourced by the measurement signal of a continuous measurement, and we also allow for an auxiliary variable to have its own purely classical dynamics. In other words, Equation's (5.6) and (5.7) form a complete parameterization for continuous measurement-based classicalquantum control where one also allows for continuous control on the classical system and are similar to measurement based feedback equations familiar in quantum control [121, 122]. It would be interesting to find a complete parameterization of the dynamics in the discontinuous case, which we do not consider here.

5.5.1 Continuous classical-quantum dynamics as continuous measurement

To show the equivalence between classical-quantum unravelings and continuous measurement, we will consider a continuous measurement described by a series of generalized measurements, also known as positive operator valued measures (POVMs). These will be described by the Kraus operators $\{\Omega_J\}$ performed in the interval [t + dt). The outcome of the measurement we label by $J_{t,k}$, and we shall assume that this drives the auxiliary classical variable through a force term $dZ_{t,k} = D_{t,k}^{00} + J_{t,k}dt$, where we also allow for purely classical dynamics $D_{t,k}^{00}$. We could choose the outcome of the continuous measurement to drive the classical system differently, but this will be sufficient for our purposes.

We consider a generalized case of [96] and here let the measurement $\{\Omega_J\}$ at time t be explicitly Z_t dependent, i.e., we allow for the measurement to depend on the classical trajectory, which we write $\{\Omega_J(Z_t)\}$.

Specifically, we consider the measurement described via

$$\Omega_J(Z_t) = 1 - iH(Z_t)dt - \frac{1}{2}D_0^{\alpha\beta}(Z_t)L_\beta^{\dagger}L_\alpha dt + \frac{1}{2}L_\alpha D_{1,i}^{\alpha0}(Z_t)(D_2^{-1})^{ij}(Z_z)J_{t,j}dt.$$
 (5.24)

The normalization condition on the measurement

$$\int d\mu_0(J)\Omega_J^{\dagger}\Omega_J = 1, \qquad (5.25)$$

is satisfied so long as we pick the measure $d\mu_0(J)$ to be such that

$$\int d\mu_0(J)(J_{t,i}dt) = 0, \ \int d\mu_0(J)(J_{t,i}dt)(J_{t,j}dt) = (\sigma\sigma^T)_{ij} = 2D_{2,ij}dt,$$
(5.26)

and we take $D_0^{\alpha\beta} = \frac{1}{2} D_{1,i}^{0\alpha} D_2^{-1ij} D_{1,j}^{\beta0}$. Equation (5.26) has the same statistics as a multivariate Gaussian random variable, so we pick $\mu_0(J)$ to be a Gaussian measure with co-variance matrix $(\sigma\sigma^T)_{ij}$.

We can calculate the mean of $J_{t,i}$ in the quantum state ρ via

$$\int d\mu_0(J) \operatorname{Tr}\left[\rho \Omega_J^{\dagger} \Omega_J\right] J_{t,i} = \langle D_{1,i}^{0\alpha}(Z_t) L_{\alpha}^{\dagger} + D_{1,i}^{\alpha 0}(Z_t) L_{\alpha} \rangle + O(dt^2), \qquad (5.27)$$

whilst we can similarly calculate the seconds moments $J_{t,i}J_{t,j}$. These turn out to be independent of the system ρ and hence equivalent to the statistics of a Gaussian random variable with variance $(\sigma\sigma^T)_{ij}$ [96]. As such, the statistics of the measurement outcomes can be described by the stochastic differential equation

$$J_{t,i}dt = dZ_t = \langle D_{1,i}^{0\alpha}(Z_t)L_{\alpha}^{\dagger} + D_{1,i}^{\alpha 0}(Z_t)L_{\alpha}\rangle dt + \sigma_{ij}(Z_t)dW_j.$$
(5.28)

Given the measurement outcome $J_{t,k}$, the conditioned density matrix takes the form

$$\rho' = \frac{\Omega_J \rho \Omega_J^{\dagger}}{Tr[\Omega_J \rho \Omega_J^{\dagger}]}.$$
(5.29)

Denoting $\tilde{L}^j = D^{\alpha 0}_{1,i}(\sigma^{-1})^{ij}(Z_t)$ we find

$$\begin{split} \rho' &= \rho - i [H(Z_t), \rho] dt - \frac{1}{2} D_0^{\alpha\beta}(Z_t) \{ L_{\beta}^{\dagger} L_{\alpha}, \rho \} dt + (\tilde{L}^i - \langle \tilde{L}^i \rangle) \rho J_{t,i} dt + \rho (\tilde{L}^{\dagger i} - \langle \tilde{L}^{\dagger i} \rangle) J_{t,i} dt \\ &+ \tilde{L}^i \rho \tilde{L}^{\dagger j} J_{t,i} dt J_{t,j} dt + D_0^{\alpha\beta}(Z_t) \langle L_{\beta}^{\dagger} L_{\alpha} \rangle dt - \langle \tilde{L}^{\dagger j} \tilde{L}^i \rangle J_{t,j} dt J_{t,i} dt \\ &+ \langle \tilde{L}^i + \tilde{L}^{\dagger i} \rangle \langle \tilde{L}^j + \tilde{L}^{\dagger j} \rangle J_{t,i} dt J_{t,j} dt. \end{split}$$

(5.30)

Substituting for $J_{t,i}dt$ in Equation (5.28), one finds the continuous CQ unraveling equation which saturates the decoherence diffusion trade-off. To obtain the general unraveling form, one can simply include Lindbladian terms in the quantum state evolution, which we do not explicitly show. In this sense, any CQ master equation which does not saturate the trade-off can be interpreted as a continuous measurement process with inefficient quantum measurements [121].

Within this framework, the decoherence-diffusion trade-off is straightforwardly interpreted as a manifestation of the information-disturbance trade-off. In this case, the strength of the measurement is parameterized by D_0 ; the weaker the continuous measurement, the less decoherence on the quantum state. However, a weak measurement leads to less information being learned about the quantum system, so there is larger noise in the outcomes of the measurements. Since the measurement outcomes drive the classical system through D_1 , this leads to greater diffusion on the classical state. The stronger the coupling is, the larger the coefficient pre-multiplying the noise is, thus explaining the appearance of the coupling strength D_1 in the trade-off.

5.5.2 CQ dynamics as an effective theory of quantum measurement

The fact that the classical-quantum unravelings form a parameterization for continuous measurementbased classical-quantum control means that they can be used to describe an effective theory of measurement dynamics. Though the unraveling equations are formally equivalent to those of continuous measurement, the equations can have a very different physical interpretation. In continuous measurement, the classical degrees of freedom are taken to be measurement outcomes, or signals, which result from applying the Born rule to the measured quantum system. Conversely, in CQ dynamics, the classical degrees of freedom can be considered any classical system. They can be dynamical; for example, they can be taken as a macroscopic system's classical position and momenta.

Hybrid formalisms to model measurement were originally introduced by Sherry and Sudarshan [125, 126, 127], though the resulting dynamics are generally not completely positive on the quantum system. Blanchard and Jadczyk [56, 128] later studied a consistent framework to study a class of completely positive CQ master equations that contain jumps in the classical phase space. Here we are able to provide a complete framework to study measurement by the interaction of continuous classical and quantum systems as an effective theory.

We can construct an effective theory of continuous quantum measurement - extending the ideas presented of [127, 125, 126, 56, 128] - as follows:

- 1. One identifies a classical system that acts as a measurement device.
- 2. One identifies the quantum system to be measured.
- 3. One takes the dynamics of the combined classical-quantum system to be generated by CQ dynamics of Equations (5.6) and (5.7), with the details of the measurement apparatus and coupling encoded in the *phenomenological* parameters D_0, D_1, D_2 , which govern the strength of the back-reaction of the quantum system and the strength of the measurement respectively.

Notably, the CQ effective theory consists only of dynamics of interacting classical and quantum degrees of freedom, with all the fine-grained details of the measurement device encoded in a few parameters: D_1 , which governs the strength of the back-reaction of the quantum system on the classical, while D_0 and D_2 are related to the strength of the measurement. When the trade-off is saturated, $2D_0 = D_1 D_0^{-1} D_1^{\dagger}$, which describes a noiseless measurement of a quantum system, while when the trade-off is not saturated, the dynamics describe a noisy quantum measurement. When the measurement is noiseless, we can describe the evolution of the quantum state by the pure state unraveling in Equation (5.9).

Within this prescription, there is no need to discuss what a measurement outcome is, nor when or how a measurement occurs; there are only effective dynamics of continuously interacting classical and quantum systems. Moreover, it removes the need to introduce auxiliary quantum probe systems. Instead, all of this information is encoded in the phenomenological parameters.
One way of interpreting this is that when measurements are included, for all practical purposes, we should be viewing quantum theory as an effective theory of interacting classical and quantum systems, with dynamics evolving according to Equations (5.6) and (5.7).

As a simple example, we can consider the case where we have a classical measurement device that we model via a classical position and momenta (x, p) of a pointer. We take the effective dynamics to be generated by

$$dx_t = \frac{p_t}{m_c} dt,\tag{5.31}$$

$$dp_t = D_1 \langle L \rangle dt + \sigma d\xi, \tag{5.32}$$

$$d|\psi\rangle_t = -iH|\psi\rangle_t dt + \frac{D_1}{2\sigma}(L - \langle L \rangle)|\psi\rangle_t d\xi - \frac{D_1^2}{8\sigma^2}(L - \langle L \rangle)^2|\psi\rangle_t dt.$$
(5.33)

Given the classical trajectory (x_t, p_t) is observed, we can invert $d\xi = \frac{1}{\sigma}[dp_t - D_1 \langle L \rangle dt]$ to uniquely determine the evolution of the quantum state, conditioned on the classical measurement apparatus

$$d|\psi\rangle_t = -iH|\psi\rangle_t dt + \frac{D_1}{2\sigma^2}(L - \langle L \rangle)|\psi\rangle_t [dp_c - D_1 \langle L \rangle] - \frac{D_1^2}{8\sigma^2}(L - \langle L \rangle)^2|\psi\rangle_t dt.$$
(5.34)

The dynamics of the quantum state described by Equation (5.33) is precisely the quantum state dynamics that arise in continuous measurement theory [121, 96, 122]. Specifically, equation (5.33) represents the dynamics for a noiseless continuous measurement of the Hermitian operator L with measurement strength determined by the ratio of $\frac{D_1}{\sigma}$, recalling that $\sigma^2 = 2D_2$. The result is to continuously collapse the state vector into eigenstates of the L operator, where the final result of the measurement can be read off from the classical variables position from Equation (5.34). Taking the strong measurement limit $\frac{D_1}{\sigma} \to \infty$, one collapses the state instantaneously, and it can be shown this leads to the probabilities described by the Born rule [123].

Note that we do not expect the classical degree of freedom representing the measurement device to interact directly with the quantum system. On the contrary, we expect that the measurement device consists of many complex quantum systems which interact with the quantum system being measured. The advantage of the classical-quantum framework is that one arrives at a consistent, effective theory of the measurement procedure, which is phenomenologically accurate, by encoding this fine-grained information in the parameters D_1, σ which are to be experimentally determined: this is what we want in an effective theory of measurement. It would be interesting to explore this effective theory of measurement in more detail. For example, one could use the path integral approach to classical-quantum dynamics which we discuss in Chapter 6 to better under the role of measurements in relativistic quantum dynamics [129, 130, 131, 132]. Furthermore, given that the classical-quantum unraveling is equivalent to the dynamics of quantum measurement, it would be interesting to see whether one could use it to give an alternative derivation of the Born rule based on the requirement of consistent coupling between classical and quantum systems; this is a subtle problem since the derivation of the classical-quantum dynamics relied on complete positivity, which is almost tantamount to assuming the Born rule to begin with [133].

5.6 Potential applications to gravity

Thus far, we have only considered continuous classical degrees of freedom. In Appendix C, we discuss how one can formally arrive at dynamics for fields – the result is the same but to replace quantities with their local counterparts and derivatives with functional derivatives. Effectively, the spatial coordinate x acts like an index of the Lindblad operators and the matrices D_n .

Our goal in this Chapter is not to reproduce a fully covariant semi-classical description of quantum gravity but rather a framework describing consistent semi-classical dynamics beyond the standard approach. We nonetheless conclude with a brief discussion of the full gravitational context. Classical-quantum dynamics in the full gravitational setting has previously been studied in [28, 3]. The idea, introduced in [28], was to take the classical degrees of freedom to be given by the Riemannian 3 metrics (on some 3 surface Σ) and their conjugate momenta $z = (\gamma_{ij}, \pi^{ij})$. One then considers completely positive dynamics, depending on some lapse Nand shift N^i , which maps hybrid states $\rho(\gamma, \pi, t)$ onto themselves, describing a geometrodynamic [108] picture of classical gravity interacting with quantum matter. One can also consider the lapse and shift and their conjugate momenta part of the phase space, in which case they enter into the Poisson bracket. While this changes nothing in the purely classical case, it offers some advantages in the CQ case. Here, by (formally) studying the unraveling of the dynamics for each realization of the noise process, we now have entire trajectories for each of the variables $(\gamma_{ij}, \pi^{ij}, N, N^i)$ each associated to a quantum state, $\rho(t|\gamma_{ij}, \pi^{ij}, N, N^i)$. This allows us to define a tuple $(g_{\mu\nu}, \rho_{\Sigma_t}(t))$ via the ADM embedding

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -N^2(t,x)dt^2 + \gamma_{ij}(t,x)(N^i(t,x)dt + dx^i)(N^j(t,x)dt + dx^j).$$
(5.35)

Equation (5.35) associates a 4-metric and quantum state to each trajectory. The unraveling thus provides a method to study the dynamics of classical gravity interacting with quantum matter.

Taking the pure dynamics to be local and Hamiltonian, in the sense of Equations (5.19), (5.20), in Appendix C we find the dynamics

$$G_{ij} = 8\pi G \langle T_{ij}[g,\pi] \rangle + \sigma_{ij}^{kl}[g,\pi] d\xi_{kl}, \qquad (5.36)$$

where $d\xi_{kl}$ is a white noise process. The evolution of the quantum state is given by Equation (5.20) where H_I is the matter Hamiltonian. Constructing a classical-quantum theory with the same degrees of freedom as GR amounts to constructing the hybrid versions of the gravitational constraints, which are the G_{00} and G_{0i} components of the Einstein tensor. In Chapter 7, we will provide the first study of hybrid classical-quantum constraints, while in Chapter 8, we introduce a diffeomorphism covariant and invariant theory of classical-quantum gravity. While we do not construct a complete theory and show it has constraints which are preserved, it serves as a proof of principle that classical-quantum theories may be made diffeomorphism invariant, which may lead to insight into the constraints. We leave whether these constraints can be preserved as a question for future research. Importantly, the decoherence-diffusion trade-off can be used to experimentally test for theories with a fundamentally classical gravitational field since they necessarily lead to diffusion in the gravitational potential and decoherence of masses in spatial superposition (see Chapter 10).

On a related note, the form of the classical evolution equation (5.36) looks similar to a Markovian version of the non-Markovian Einstein-Langevin equation [134, 135, 136, 137, 21]. The Einstein-Langevin equation is the central object of study in stochastic gravity [134, 135, 136, 137, 21] aimed at incorporating higher order corrections to Einstein's equations sourced by the quantum stress-energy tensor. Such corrections were initially motivated by studying the interaction of two linear quantum systems via a path integral approach, integrating out one of the quantum systems and looking at the $\hbar \rightarrow 0$ dynamics of the remaining system. The Einstein-Langevin equation is believed to be valid whenever the dynamics can be approximated by correlation functions which are second order. The dynamics we introduce provide a semiclassical regime that goes beyond this since we have arrived at a consistent semi-classical picture that gives rise to consistent dynamics on any quantum state: this includes quantum states in macroscopic superposition, which are not approximated well by second-order correlation functions and for which the Einstein-Langevin equation fails to be a good approximation [21].

5.7 Discussion

The equations in this chapter parameterize the general form of completely-positive, linear, autonomous, and continuous classical-quantum dynamics in terms of combined classical-quantum trajectories. Given that the initial motivation was to arrive at a healthier theory of semiclassical gravity, it is worth considering when we expect these assumptions to hold. If gravity were to be fundamentally classical, then these assumptions are reasonable: the assumptions of complete-positivity and linearity are necessary for sensible predictions for all initial classical and quantum states; the assumption of autonomous dynamics is reasonable for any theory viewed as fundamental; and the assumption of continuous classical trajectories is necessary for the dynamics to describe probability distributions over space-times. Viewed this way, one expects the field-theoretic versions of Equation (5.6) and (5.9) to provide a template to construct consistent CQ theories of gravity.

The dynamics introduced here should also be useful as an effective theory. While we expect this to be true, it is crucial to note that none of the assumptions need necessarily hold, at least exactly, or for all times. For instance, if one allows for the non-Markovian evolution that generically arises in the study of open quantum systems, we necessarily violate the assumption of autonomous dynamics. In this case, a time-local non-Markovian theory takes the same form as Equation (5.5), but without the requirement for the decoherence-diffusion trade-off to hold for all times, [28]. Alternatively, one may construct dynamics that are not completely positive on all initial classical distributions but completely positive on those permissible by a quantum theory or on some more significant set subset of quantum states than those allowed in semi-classical gravity. While such theories may be useful as effective theories, they are likely incompatible with the assumption of classical trajectories. In this regard, the dynamics we present here are likely valid in a regime of the effective theory where classical trajectories are well-defined.

A more detailed treatment of classical-quantum dynamics may shed light on some of the open problems in semi-classical physics. Of potential interest would be in understanding the role of vacuum fluctuations in cosmology and structure formation. Since we wish to investigate the role that vacuum fluctuations play in density in-homogeneity, this is a regime in which the semiclassical Einstein Equation (5.1) cannot be used. In practice, researchers consider situations in which the density perturbations have decohered [34, 33, 138, 139, 140, 141, 142] so that they can condition on their value and feed this into the Friedman-Robertson-Walker equation governing the expansion of the local space-time patch [143, 144, 145]. As already discussed, this procedure can be understood as the low noise, $\sigma \rightarrow 0$, limit of the healed semi-classical equations (5.19) and (5.20). The semi-classical dynamics we have presented provide a framework where it is possible to ask what happens earlier when there are genuine quantum fluctuations. Exploring features such as these in more realistic models would be of great interest, especially since we can consistently evolve the system before the fluctuations have decohered.

Chapter 6

Path integrals for classical-quantum dynamics

In this chapter we make use of the developments in the understanding of classical-quantum master equations presented in previous chapters to write down a classical-quantum path integral, equivalent to dynamics which is CPTP. Specifically, using the most general form of autonomous classical-quantum dynamics of Equation's (2.91) and (4.13), we associate a path integral to *any* CQ master equation, from which the conditions on complete positivity can easily be read off. The general result is given by Equation (6.16). We study classical-quantum path integrals without resorting to master equation methods in Chapter 8.

In the master equation picture, the complete positivity, and general consistency, of classicalquantum dynamics is manifest. However, in a variety of contexts a path integral approach is perhaps more useful. For example, some numerical simulations are better suited to path integral methods [146, 147, 148, 149, 150, 151, 151, 152], especially when saddle point approximations are valid. For practical applications, we saw in the previous chapter that classical-quantum dynamics can be viewed as the natural framework to discuss quantum theory when measurements are involved, which is particularly relevant for quantum control procedures. Indeed, the most general operation one is allowed to perform in standard quantum theory is described by a series of CPTP maps which are performed conditioned on the outcomes of measurements. On the other hand, CQ dynamics is the framework to consider theories with a classical field, whether fundamental or effective, and a path integral approach allows one to impose space-time and gauge symmetries, as well as the possibility to enforce the modern principles used when studying effective field theories [15].

In quantum mechanics, it is well known that one can derive the path integral approach from the Schrödinger equation, and also from the more general Lindblad equation arising from open quantum systems [79, 97]. It is perhaps less well known that one can do the same for classical dynamics, arriving at an equivalence between general master equations and path integrals [153, 154, 80, 87]. For example, a Brownian particle whose conditional probability distribution P(x, p|x', p') evolves according to the Fokker-Plank equation has an equivalent description in terms of a path integral which is (up to factors of i, \hbar) the same as the standard path integral of quantum mechanics. We showed explicitly how one can derive path integrals for open classical and quantum systems in Chapter 2, and the path integral combines them to the case of classicalquantum coupling. We compare the path integral for open quantum systems, the classical path integral for stochastic systems, and the classical-quantum hybrid path integrals we construct in this chapter in Table 2.2.

Classical-quantum path integrals have appeared previously [155, 156, 157, 158, 159]. These may be valid when applied to some initial probability densities, but generally lead to negative probabilities since the dynamics is not completely positive on all initial states. Here, we consider dynamics which is CPTP on all states at all times. Of particular relevance is the class of continuous master equations [57, 60], the most general form of which was introduced in Chapter 4. We find these path integrals have a natural decomposition into a pure classical part, representing the stochastic nature of the classical degrees of freedom, a pure quantum part, which includes a Feynman-Vernon term, and a classical-quantum part – which acts to exponentially suppress the paths which deviate from the averaged equations of motion – as summarized by Table 2.2. Under certain conditions, namely when the dynamics is at most quadratic in momenta, we can integrate out the momenta to arrive at a configuration space path integral. In the case where the dynamics is approximately Hamiltonian, as in [60, 28, 10], the configuration space path integral acts to exponentially suppress the paths which deviate from paths solving the averaged Euler-Lagrange equations.

The final form we find motivates a general form of configuration space path integrals, sum-

Table 6.1: A table representing the classical, quantum, and classical-quantum path integrals.

	Classical stochastic
Path integral	$p(q, p, t_f) = \int \mathcal{D}q \mathcal{D}p \ e^{iS_C[q, p]} \delta(\dot{q} - \frac{\partial H}{\partial p}) p(q, p, t_i)$
Action	$iS_C = -\int_{t_i}^{t_f} dt \frac{1}{4} \left(\frac{\partial H}{\partial q} + \dot{p}\right) D_2^{-1} \left(\frac{\partial H}{\partial q} + \dot{p}\right)$
CP condition	D_2^{-1} a positive (semi-definite) matrix, $D_2^{-1} \succeq 0$

(a) The path integral for continuous, stochastic classical dynamics [153, 154, 80, 87] in phase space generated by a stochastic Hamiltonian. One sums over all classical configurations (q, p) with a weighting according to the difference between the classical path and its expected force $-\frac{\partial H}{\partial q}$, by an amount characterized by the diffusion matrix D_2 . In the case where the force is determined by a Lagrangian L_C , the action S_C describes a suppression of paths away from the Euler-Lagrange equations $iS_C = -\int_{t_i}^{t_f} dt \frac{1}{4} (\frac{\delta L_C}{\delta q_i}) (D_2^{-1})^{ij} (\frac{\delta L_C}{\delta q_j})$, by an amount determined by the diffusion coefficient D_2 . The most general form of classical path integral is given by Equation (6.18).

	Quantum
Path integral	$\rho(\phi^{\pm}, t_f) = \int \mathcal{D}\phi^{\pm} \ e^{iS[\phi^+] - iS[\phi^-] + iS_{FV}[\phi^+, \phi^-]} \rho(\phi^{\pm}, t_i)$
Action	$S[\phi] = \int_{t_i}^{t_f} dt \left(\frac{1}{2}\dot{\phi}^2 + V(\phi)\right),$ $iS_{FV} = \int_{t_i}^{t_f} dt \left(D_0^{\alpha\beta} L_{\alpha}^+ L_{\beta}^{*-} - \frac{1}{2}D_0^{\alpha\beta} (L_{\beta}^{*-} L_{\alpha}^- + L_{\beta}^{*+} L_{\alpha}^+)\right)$
CP condition	$D_0^{\alpha\beta}$ a positive (semi-definite) matrix, $D_0 \succeq 0$.

(b) The path integral for a general autonomous quantum system, here taken to be ϕ . The quantum path integral is doubled since it includes a path integral over both the bra and ket components of the density matrix, here represented using the \pm notation. In the absence of the Feynman Vernon term S_{FV} [79], the path integral represents a quantum system evolving unitarily with an action $S[\phi]$. When the Feynman Vernon action S_{FV} is included, the path integral describes the path integral for dynamics undergoing Lindbladian evolution [62, 63] with Lindblad operators $L_{\alpha}(\phi)$.

	Classical-quantum
Path integral	$\rho(q, p, \phi^{\pm}, t_f) =$
	$\int \mathcal{D}q \mathcal{D}p \mathcal{D}\phi^{\pm} e^{iS_C[q,p] + iS[\phi^+] - iS[\phi^-] + iS_{FV}[\phi^{\pm}] + iS_{CQ}[q,p,\phi^{\pm}]} \delta(\dot{q} - \frac{p}{m}) \rho(q,p,\phi^{\pm},t_i)$
Action	$iS_C[z] + iS_{CQ}[z,\phi^{\pm}] = -\frac{1}{4} \int_{t_i}^{t_f} dt \ D_2^{-1} \left(\frac{\partial H_C}{\partial q} + \frac{1}{2}\frac{\partial V_I[q,\phi^{+}]}{\partial q} + \frac{1}{2}\frac{\partial V_I[q,\phi^{-}]}{\partial q} + \dot{p}\right)^2.$
CP condition	$D_0 \succeq 0, D_2 \succeq 0 \text{ and } 8D_2 \succeq D_0^{-1}$

(c) The phase space path integral for continuous, autonomous classical-quantum dynamics. The path integral is a sum over all classical paths of the variables q, p, as well as a sum over the doubled quantum degrees of freedom ϕ^{\pm} . The action contains the purely quantum term from the quantum path integral in Table 6.1b, but also includes the term $iS_C + iS_{CQ}$. This suppresses paths away from the drift, which is sourced by both purely classical terms described by the Hamiltonian H_C and the back-reaction of the quantum systems on the classical ones, described by a classical-quantum interaction potential V_I . The action acts to suppress paths which deviate from the \pm averaged Hamilton's equations (see Equation (6.49)). The most general form of classical-quantum path integral is given by Equation (6.16). Under certain conditions, namely when the classical-quantum action (6.16) is quadratic in momenta, one can arrive at a configuration space path integral, where paths deviating from the \pm averaged Euler-Lagrange equations as suppressed (see Equation (6.66)). In order for the dynamics to be completely positive, the decoherence-diffusion trade-off $8D_2 \succeq D_0^{-1}$ must be satisfied [5, 4], where D_0^{-1} is the generalized inverse of D_0 , which must be positive semi-definite. As a consequence, there must be both a Feynman-Vernon term D_0 , and deviation from paths away from their expected drift due to the diffusion coefficient D_2 , and both effects cannot be made small. When the trade-off is saturated, the path integral preserves purity of the quantum state, conditioned on the classical degree of freedom [6] (see Section 6.3.3). marized by Equation (6.72). In Chapter 8 we prove such path integrals are completely positive without resorting to master equation methods, meaning the general form is valid even when higher derivative terms are included. As a result, these path integrals provide a general framework to construct classical-quantum theories which respect space-time symmetries. In Chapter 8 we use them to construct CQ path integrals for gravity including a diffeomorphism invariant theory of CQ gravity. These CQ path integrals can be thought of as an effective theory where space-time is treated as classical. On the other hand, if taken as fundamental, the parameter space of the theory can be experimentally constrained via the decoherence diffusion trade-off [4], which has already been used to constrain theories with a fundamentally classical gravitational field. We will find that the trade-off plays a special role here. When it is saturated, the path integral takes on a particularly simple form. In this chapter we will primarily interested in deriving CQ path integrals from master equations, but for completeness we also include a brief discussion of their more general form.

This chapter is based on the paper [7], which is work done in collaboration with Jonathan Oppenheim.¹

6.1 Moment expansion of the dynamics

In this chapter, we will assume classical degrees of freedom are described by a continuous measurable space, and we will generically denote elements of the space by z. We discuss the case of fields separately, and less rigorously, in Section 6.5 and also in Chapter 8.

Recall from the background section in Chapter 2 that we can write the transition amplitude $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ for an autonomous classical-quantum map as

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \int du \ e^{-iu \cdot (z-z')} C^{\mu\nu}(u,z',\delta t) = \sum_{n=0}^{\infty} \frac{M_{n,i_1\dots i_n}^{\mu\nu}(z',\delta t)}{n!} \frac{1}{(2\pi)^d} \int du \ e^{-iu \cdot (z-z')}(i^n) u_{i_1}\dots u_{i_n}.$$
(6.1)

As a consequence, using the short-time expansion of the moments

$$M^{\mu\nu}(z',\delta t)_{n,i_1\dots i_n} = \delta^{\mu}_0 \delta^{\nu}_0 + \delta t n! D^{\mu\nu}_{n,i_1\dots i_n}(z',t) + O(\delta t^2), \tag{6.2}$$

¹The factor of 2 difference between this chapter and [7] is because in this thesis we have chosen to absorb the $\frac{1}{n!}$ appearing in the master equation of [7] into the definition of D_n . The same is true for Chapter 8.

we can write the state at $t + \delta t$ in terms of the coefficients $D_{n,i_1...i_n}^{\mu\nu}(z')$ and the state at t as

$$\varrho(z,t+\delta t) = \frac{1}{(2\pi)^d} \int du dz' \, e^{-iu(z-z')} \left(\varrho(z',t) + \sum_{n=0}^{\infty} \delta t(i^n) u_{i_1} \dots u_{i_n} D_{n,i_1\dots i_n}^{\mu\nu}(z',t) L_{\mu} \varrho(z',t) L_{\nu}^{\dagger} \right)$$
(6.3)

Equation (6.3) will be a key equation in deriving the CQ path integral. Recall, the moments appearing in Equation (6.3) can be related to physical quantities. For example, the first and second moments D_1, D_2 of the probability transition amplitude characterize the amount of drift and diffusion in the system, whilst the zeroth moment D_0 can be related to the amount of decoherence of the quantum system. One needs to remember that the moments are not independent of each other and using the conservation of probability (Equation 2.82) one can eliminate D_0^{00} in favour of the other coefficients. In this section, we shall often write D_n to describe the object with entries $D_{n,i_1...i_n}^{\mu\nu}$ and we occasionally write the master equation in short-hand as

$$\frac{\partial \varrho}{\partial t} = \mathcal{L}(\rho), \tag{6.4}$$

where the superoperator \mathcal{L} is defined via the right hand side of Equation (6.3). The formal solution to the dynamics can then be written as

$$\rho(t) = U(t, t_i) = \mathcal{T}\{e^{\int_{t_i}^t dt' \mathcal{L}(t')}\}(\rho(t_i)),$$
(6.5)

where \mathcal{T} denotes the time ordering operator, which is required since in the time dependent case $\mathcal{L}(t)$ operations do not commute with each other at different times. Since the dynamics is autonomous, it forms a semi-group [63], so that $U(t, t_i) = U(t, t')U(t', t_i)$.

6.2 Derivation of the path integral formalism

In classical autonomous dynamics, the Kramers-Moyal expansion is used to obtain a path integral representation of the dynamics. For the reader unfamiliar with classical path integrals for open classical systems we recommend [80] (see also [73, 87]). The path integral for quantum systems is found after Trotterizing [160] the dynamics and inserting position and momentum resolutions of the identity – see [97] for a review of quantum path integrals for open quantum systems. In the hybrid case, we shall do both simultaneously to arrive at a CQ path integral, using the short time representation of the dynamics appearing in Equation (6.3).

We first derive a path integral for the most general CQ master equation to arrive at a phase space path integral, which includes an integral over response variables. The result is Equation (6.16). In its most general form, the path integral is a complicated object, however, in Section 6.3 we study the path integral for the class of continuous master Equations. In this case, we find one can always integrate out the response variables to arrive at a phase space path integral alone, given by Equation (6.41).² The resulting path integral has a natural interpretation in terms of suppressing paths away from there averaged equations of motion by an amount characterized by D_2^{-1} .³ Simultaneously there is decoherence the quantum system, by an amount depending on D_0 . The decoherence diffusion trade-off [4], necessary for complete positivity of the dynamics, tells us that one cannot simultaneously make the effects of decoherence and diffusion small if there is back-reaction on the classical system.

6.2.1 Derivation of phase space path integral for any CQ dynamics

Let us now derive the CQ path integral for the master equation defined by Equation (6.3). For ease of presentation, we shall take the Lindblad operators L_{μ} to be functions of two canonically conjugate operators ϕ, π , with $[\phi, \pi] = i$, $L_{\mu}(\phi, \pi)$, but the derivation also holds if they are functions of multiple operators and we can also write a coherent state path integral using similar methods to [97]. We use the convention that $\langle \phi | \phi' \rangle = \delta(\phi - \phi')$ and $\langle \pi | \pi' \rangle = 2\pi \delta(\pi - \pi')$ so that $\langle \phi | \pi \rangle = e^{i\pi\phi}$.

To derive the path integral, we first Trotterize the dynamics [160]. Defining $t_f = t_i + K\delta t$ and $t_k = k\delta t$, we use the identity

$$U(t_{f}, t_{i}) = \lim_{K \to \infty} U(t_{f}, t_{k-1}) U(t_{k-1}, t_{k-2}) \dots U(t_{2}, t_{i})$$

=
$$\lim_{K \to \infty} (1 + \delta t \mathcal{L}(t_{k-1})) (1 + \delta t \mathcal{L}(t_{k-2})) \dots (1 + \delta t \mathcal{L}(t_{i})).$$
 (6.6)

In the time independent case, Equation (6.6) reduces to the familiar statement

$$U(t_f, t_i) = e^{\mathcal{L}(t_f - t_i)} = \lim_{K \to \infty} e^{\mathcal{L}\delta t} = \lim_{K \to \infty} (1 + \mathcal{L}\delta t)^K.$$
(6.7)

 $^{^{2}}$ In Section 6.4 we discuss the sufficient conditions to derive a configuration space path integral, namely that the classical-quantum action be at most quadratic in momenta.

³Here, and throughout, recall the ⁻¹ denotes the generalized inverse of $D_2(D_0)$, since $D_2(D_0)$ are only required to be positive *semi-definite*

We can use Equation (6.6) to write the CQ state at time t_{k+1} in terms of the state at time t_k as

$$\varrho(z_{k+1}, t_{k+1}) = \int dz_k \delta(z_{k+1} - z_k) \varrho(z_k, t_k) + \delta t \mathcal{L}(z_{k+1} | z_k) (\varrho(z_k, t_k)).$$
(6.8)

Using the definition of the delta function, we can identify the right hand side of Equation (6.8) with that of Equation (6.3) to write

$$\varrho(z_{k+1}, t_{k+1}) = \frac{1}{(2\pi)^d} \int du_k dz_k \ e^{-iu_k \cdot (z_{k+1} - z_k)} \times \left(\varrho(z_k, t_k) + \sum_{n=0}^{\infty} \delta t(i^n) u_{k, i_1} \dots u_{k, i_n} D_{n, i_1 \dots i_n}^{\mu\nu}(z_k, t_k) L_{\mu} \varrho(z_k, t_k) L_{\nu}^{\dagger} \right)$$
(6.9)

The next step is to map the Lindblad operators acting on the CQ state to c-numbers which can be exponentiated. Just as with the quantum path integral, we first write the state in terms of the ϕ basis

$$\varrho(z_k, t_k) = \int d\phi_k^- d\phi_k^+ \ \varrho(z_k, \phi_k^+, \phi_k^-, t_k) \ |\phi_k^+\rangle \langle \phi_k^-|, \tag{6.10}$$

where $\rho(z, \phi_k^+, \phi_k^-, t_k) = \langle \phi^+ | \rho(z, t_k) | \phi^- \rangle$. The +, - terms arise because we expand the density matrix in terms of both bra and ket states. The convention is such that when we calculate the expectation value of operators Tr $[O^+ \rho O^-]$, then, after using cyclicity of the trace Tr $[\rho O^- O^+] =$ Tr $[O^- O^+ \rho]$, the operators O^- are always to the left of O^+ .

Inserting Equation (6.9) into Equation (6.10), along with π^+, π^- resolutions of the identity $\mathbb{I}L_{\mu\varrho}L_{\nu}^{\dagger}\mathbb{I}$ gives the following expression for the transition amplitude

$$\varrho(z_{k+1}, \phi_{k+1}^+, \phi_{k+1}^-, t_{k+1}) = \frac{1}{(2\pi)^d} \int d\phi_k^- d\phi_k^+ d\pi_k^- d\pi_k^+ du_k dz_k e^{\delta t \mathcal{I}(\phi_k^-, \phi_k^+, \pi_k^-, \pi_k^+, u_k, z_k, t_k)} \varrho(z_k, \phi_k^+, \phi_k^-, t_k),$$
(6.11)

where we have implicitly defined the (time-discrete) classical-quantum action

$$\mathcal{I}(\phi_k^-, \phi_k^+, \pi_k^-, \pi_k^+, u_k, z_k, t_k) = -iu_k \cdot \frac{(z_{k+1} - z_k)}{\delta t} + i \frac{(\phi_{k+1}^+ - \phi_k^+)}{\delta t} \pi^+ - i \frac{(\phi_{k+1}^- - \phi_k^-)}{\delta t} \pi^- + \sum_{n=0}^{\infty} (i^n) u_{k,i_1} \dots u_{k,i_n} D_{n,i_1\dots i_n}^{\mu\nu} (z_k, t_k) L_{\mu}^+ L_{\nu}^{-*},$$
(6.12)

using the shorthand $L^+_{\mu} := L_{\mu}(\phi^+, \pi^+) = \langle \pi^+ | L_{\mu}(\phi, \pi) | \phi^+ \rangle$, and similarly for L^- . Taking the $t \to 0, K \to \infty$ limit, with $\delta t K = t_f - t_i$, we arrive at the path integral representation of the

transition amplitude

$$\varrho(z,\phi^{+},\phi^{-},t_{f}) = \lim_{K \to \infty} \int \left\{ \prod_{k=1}^{K} d\phi_{k}^{\pm} \right\} \left\{ \prod_{k=1}^{K} \frac{d\pi_{k}^{\pm}}{2\pi} \right\} \left\{ \prod_{k=1}^{K} \frac{du_{k}}{(2\pi)^{d}} \right\} \left\{ \prod_{k=1}^{K} dz_{k} \right\} \times e^{\delta t \sum_{k=1}^{K} \mathcal{I}(\phi_{k}^{-},\phi_{k}^{+},\pi_{k}^{-},\pi_{k}^{+},u_{k},z_{k},t_{k})} \varrho(z_{i},\phi_{i}^{+},\phi_{i}^{-},t_{i}),$$
(6.13)

where it should always be understood that boundary conditions for the final state have been imposed. For ease of notation, we will write this formally as

$$\varrho(z,\phi^+,\phi^-,t_f) = \int \mathcal{D}\phi^{\pm} \mathcal{D}\pi^{\pm} \mathcal{D}u \mathcal{D}z \ e^{\mathcal{I}(\phi^-,\phi^+,\pi^-,\pi^+,u,z,t_i,t_f)} \varrho(z_i,\phi_i^+,\phi_i^-,t_i), \tag{6.14}$$

where

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u,z,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[-iu \cdot \frac{dz}{dt} + i\dot{\phi^{+}}\pi^{+} - i\dot{\phi^{-}}\pi^{-} + \sum_{n=0}^{\infty} (i^{n})u_{i_{1}}\dots u_{i_{n}}D_{n,i_{1}\dots i_{n}}^{\mu\nu}(z,t)L_{\mu}^{+}L_{\nu}^{-*} \right],$$
(6.15)

and the \pm denotes integration over both the +, - variables. Finally, we can use the normalization condition to substitute in for the coefficient D_0^{00} to write the path integral in a way which reflects the structure of the master equation in (2.91), in which case we find our general expression for the *CQ action*

$$\begin{aligned} \mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u,z,t_{i},t_{f}) &= \int_{t_{i}}^{t_{f}} dt \bigg[-iu \cdot \frac{dz}{dt} + \sum_{n=1}^{\infty} (i^{n})u_{i_{1}} \dots u_{i_{n}} D_{n,i_{1}\dots i_{n}}^{00}(z,t) \\ &+ i\dot{\phi^{+}}\pi^{+} - iH^{+} - i\dot{\phi^{-}}\pi^{-} + iH^{-} + D_{0}^{\alpha\beta}(z,t)L_{\alpha}^{+}(\phi^{+},\pi^{+})L_{\beta}^{*-} - \frac{1}{2}D_{0}^{\alpha\beta}(z,t)\left(L_{\beta}^{*+}L_{\alpha}^{+} + L_{\beta}^{*+}L_{\alpha}^{+}\right) \\ &+ \sum_{\mu\nu\neq00}\sum_{n=1}^{\infty} (i^{n})u_{i_{1}}\dots u_{i_{n}} D_{n,i_{1}\dots i_{n}}^{\mu\nu}(z,t)L_{\mu}^{+}L_{\nu}^{*-}\bigg]. \end{aligned}$$

$$(6.16)$$

We can break down Equation (6.16) into its familiar parts, by writing

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u,z,t_{i},t_{f}) = iS_{C}[u,z] + iS[\phi^{+},\pi^{+}] - iS[\phi^{-},\pi^{-}] + iS_{FV}[z,\phi^{\pm},\pi^{\pm}] + iS_{CQ}[u,z,\phi^{\pm},\pi^{\pm}].$$
(6.17)

In Equation (6.17) $S_C[u, z]$ is the pure classical action [80]

$$iS_C[u,z] = \int_{t_i}^{t_f} dt \left[-iu \cdot \frac{dz}{dt} + \sum_{n=1}^{\infty} (i^n) u_{i_1} \dots u_{i_n} D_{n,i_1\dots i_n}^{00}(z,t) \right],$$
(6.18)

 $S[\phi,\pi] = \dot{\phi}\pi - H$ is a pure quantum action (written in momentum variables), which appears in the combination $iS[\phi^+,\pi^+] - iS[\phi^+,\pi^+]$ due to the bra and ket components of the density matrix. $S_{FV}[z,\phi^{\pm},\pi^{\pm}]$ is the Feynman-Vernon action familiar in the study of open quantum systems [79, 97], describing the pure Lindbladian part of the dynamics

$$S_{FV}[z,\phi^{\pm},\pi^{\pm}] = -i \int_{t_i}^{t_f} dt \ D_0^{\alpha\beta}(z,t) L_{\alpha}^{+} L_{\beta}^{*-} - \frac{1}{2} D_0^{\alpha\beta}(z,t) \left(L_{\beta}^{*+} L_{\alpha}^{+} + L_{\beta}^{*+} L_{\alpha}^{+} \right), \tag{6.19}$$

and $S_{CQ}[u, z, \phi^{\pm}, \pi^{\pm}]$ is describes the novel non-trivial CQ interaction terms

$$iS_{CQ}[u, z, \phi^{\pm}, \pi^{\pm}] = \int_{t_i}^{t_f} dt \sum_{\mu\nu\neq00} \sum_{n=1}^{\infty} (i^n) u_{i_1} \dots u_{i_n} D_{n, i_1 \dots i_n}^{\mu\nu}(z, t) L_{\mu}^{+} L_{\nu}^{*-}.$$
 (6.20)

One sees that the quantum back-reaction on the classical system is encoded by the interaction of the Lindblad operators with the response variables u through the coupling $D_{n,i_1...i_n}^{\mu\nu}(z,t)$. Equation (6.13) is the most general path integral formulation for autonomous CQ dynamics. However, the fact that there are infinitely many terms appearing in the exponent make it potentially difficult to work with, at least exactly, and represents the fact that in general classical-quantum dynamics can involve finite sized jumps in the classical phase space.

Nonetheless, the path integral becomes much simpler for the class of continuous CQ dynamics, the most general form we introduced in Chapter 4. Recall, the general form of the continuous dynamics is given by

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{n=2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D_{n,i_1\dots i_n}^{00}(z,t)\varrho(z,t) \right)
- \frac{\partial}{\partial z_i} \left(D_{1,i}^{0\alpha}(z,t)\varrho(z,t)L_{\alpha}^{\dagger} \right) - \frac{\partial}{\partial z_i} \left(D_{1,i}^{\alpha0}(z,t)L_{\alpha}\varrho(z,t) \right)$$

$$(6.21)$$

$$i[H(z) - \varrho(z,t)] + + D_{\alpha\beta}^{\alpha\beta}(z,t)L_{\alpha}\varrho(z,t)L_{\alpha}^{\dagger} - \frac{1}{2} D_{\alpha\beta}^{\alpha\beta}(z,t)[L_{\alpha}^{\dagger} - \varrho(z,t)]$$

$$-i[H(z),\varrho(z,t)] + +D_0^{\alpha\beta}(z,t)L_{\alpha}\varrho(z,t)L_{\beta}^{\dagger} - \frac{1}{2}D_0^{\alpha\beta}(z,t)\{L_{\beta}^{\dagger}L_{\alpha},\varrho(z,t)\}_{+}, \quad (6.22)$$

where completely positivity of the dynamics is equivalent to the condition that the matrix

$$D = \begin{bmatrix} 2D_2 & D_1 \\ D_1^{\dagger} & D_0 \end{bmatrix} \succeq 0 \tag{6.23}$$

is positive semi-definite. Recall that since Equation (6.23) is in block form it follows from the Schur complement that positive semi-definiteness of D is equivalent to the following conditions

$$D \succeq 0 \Leftrightarrow D_2 \succeq 0, 2D_0 - D_1^{\dagger} D_2^{-1} D_1 \succeq 0, (I - D_2 D_2^{-1}) D_1 = 0 \text{ and}$$

$$D \succeq 0 \Leftrightarrow D_0 \succeq 0, 2D_2 - D_1 D_0^{-1} D_1^{\dagger} \succeq 0, (I - D_0 D_0^{-1}) D_1^{\dagger} = 0.$$
(6.24)

Here $^{-1}$ denotes the generalized inverse, since D_2, D_0 are only required to be positive *semi*definite. For convenience, in this chapter we often write the generalized inverse of an object Xas X^{-1} but we emphasise that it should always be understood as the generalized inverse if Xis not invertible.

For the class of continuous master equations, the action of Equation (6.16) reads

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u,z,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[-iu \cdot \frac{dz}{dt} + iu_{i}D_{1,i}^{00} - u_{i}D_{2,ij}^{00}u_{j} \right] \\
i\dot{\phi^{+}}\pi^{+} - iH^{+} - i\dot{\phi^{-}}\pi^{-} + iH^{-} + D_{0}^{\alpha\beta}L_{\alpha}^{+}L_{\beta}^{*-} - \frac{1}{2}D_{0}^{\alpha\beta}\left(L_{\beta}^{*-}L_{\alpha}^{-} + L_{\beta}^{*+}L_{\alpha}^{+}\right) \\
+ iu_{i}D_{1,i}^{0\alpha}L_{\alpha}^{*-} + iu_{i}D_{1,i}^{\alpha0}L_{\alpha}^{+} \right],$$
(6.25)

and we can identify the purely classical contribution with the Fokker-Plank action [80, 87]

$$iS_C = \int_{t_i}^{t_f} dt \left[-iu \cdot \frac{dz}{dt} + iu_i D_{1,i}^{00} - u_i D_{2,ij}^{00} u_j \right], \tag{6.26}$$

whilst the CQ interaction is determined via

$$S_{CQ} = \int_{t_i}^{t_f} dt \left[u_i D_{1,i}^{0\alpha} L_{\alpha}^{*-} + u_i D_{1,i}^{\alpha 0} L_{\alpha}^{+} \right].$$
(6.27)

Equation (6.25) gives the general form of path integrals for any continuous autonomous CQ master equation. In the next section we show that these path integrals can be simplified further since the action is quadratic in u. Therefore, we are able to integrate out the response variables u to obtain a path integral over ϕ^{\pm}, π^{\pm}, z alone. In this case, the coupling between quantum and classical degrees of freedom can be written down directly, as opposed to being mediated by response variables, and we show this has a natural interpretation as suppressing classical paths which deviate away from their averaged path, determined by both pure classical drift and drift sourced by quantum back-reaction on the classical degrees of freedom.

To summarize this section, we have seen that by introducing a recursive short time moment expansion of the classical-quantum state, Equation (6.9), we are able to write down the general form of the phase space classical-quantum path integral. The general expression is given by Equation (6.16), and for continuous dynamics we arrive at the considerably simpler path integral of Equation (6.25). In the general case, the action includes an integral over response variables u. However, for the continuous master equations the response variables couple quadratically in the action; in the next section we shall integrate these out to obtain a path integral formulation without response variables.

6.3 Path integral formulation for continuous CQ master Equations without response variables

In this section, we perform the integral over response variables in Equation (6.25) to obtain a path integral representation of the continuous master equation in terms of the variables ϕ^{\pm}, π^{\pm}, z alone. Since the pure quantum parts of the action are u independent, the relevant portion of the action, i.e, the u dependent part, is given by $S_C + S_{CQ}$ – as defined in Equations (6.26) and (6.27). We have to be careful since the diffusion matrix D_2 appearing in S_C is a symmetric, positive *semi-definite* matrix, as opposed to a positive definite matrix, and so the Gaussian integral over response variables is slightly less standard.

To deal with this issue, we first diagonalize the diffusion matrix by means of an orthogonal transformation $O_{ij}(z,t)$

$$D_{2,ij}^{00}(z,t) = \sum_{k} O_{ik}(z,t)^T \lambda_k(z,t) O_{kj}(z,t).$$
(6.28)

We expect the u path integral measure to be invariant under orthogonal transformations of u and so after applying the orthogonal transformation to Equation (6.25), or more properly to each u integral in Equation (6.13), we arrive at the diagonalized action for the continuous master equation

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u,z,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[-iu_{i}O_{ij}\frac{dz_{j}}{dt} + iu_{i}O_{ij}D_{1,j}^{00} - u_{i}\lambda_{i}u_{i}\right]$$

$$i\dot{\phi^{+}}\pi^{+} - iH^{+} - i\dot{\phi^{-}}\pi^{-} + iH^{-} + D_{0}^{\alpha\beta}L_{\alpha}^{+}L_{\beta}^{*-} - \frac{1}{2}D_{0}^{\alpha\beta}\left(L_{\beta}^{*-}L_{\alpha}^{-} + L_{\beta}^{*+}L_{\alpha}^{+}\right)$$

$$+ iu_{i}O_{ij}D_{1,j}^{0\alpha}L_{\alpha}^{*-} + iu_{i}O_{ij}D_{1,j}^{\alpha0}L_{\alpha}^{+} \right].$$
(6.29)

Having diagonalized the diffusion matrix $D_{2,ij}^{00}$ the $u_i u_j$ cross terms appearing in Equation (6.25) decouple and we can perform each component u_i of the $\mathcal{D}u$ path integral separately. Since we know that $D_{2,ij}^{00}$ is a positive semi-definite matrix there are two cases to consider – when its eigenvalues vanish $\lambda_i = 0$ and when they are strictly positive $\lambda_i > 0$. We split the action into terms consisting of response variables which couple to a zero eigenvalue and those which couple to a strictly positive eigenvalue

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u,z,t_{i},t_{f}) = \mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u_{\lambda=0},z,t_{i},t_{f}) + \mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u_{\lambda>0},z,t_{i},t_{f})$$
(6.30)

Given this decomposition, we are then able to perform the integration over the response variables in turn, starting with those associated to a zero eigenvalue.

Integral over response variables u_i with $\lambda_i = 0$

When $\lambda_i = 0$ the action in Equation (6.29) is linear in u_i . We know that for the dynamics to be completely positive we must have (6.24) that

$$(I - D_2 D_2^{-1})D_1 = 0, (6.31)$$

and as a consequence

$$(I - \lambda \lambda^{-1})OD_1 = 0, (6.32)$$

where λ^{-1} is the generalized inverse of λ . We can write (6.32) in components as

$$\sum_{j} (1 - \delta(\lambda_i)) O_{ij} D_{1,j} = 0,$$
(6.33)

where $\delta(\lambda_i)$ is 0 if $\lambda_i = 0$ and one otherwise. In the case when $\lambda_i > 0$ this poses no extra restrictions. However, when $\lambda_i = 0$ the i, j component can only be satisfied if $\sum_j O_{ij} D_{1,j} = 0$. Hence, whenever $\lambda_i = 0$ the terms $u_i O_{1,j} D_{1,j}^{0\alpha}$ - and its complex conjugate term - will not contribute in Equation (6.29). This is expected: if we use a basis of classical variables such that the diffusion matrix is diagonal, then if any of the classical variables are deterministic we expect to have no quantum back-reaction on these degrees of freedom, since we know this must be of stochastic nature due to Equation (6.24) and the decoherence diffusion trade-off [4].

In fact, for vanishing λ_i , the only term involving the response variable u_i is given by the purely classical S_C action, and takes the form

$$\sum_{j,i|\lambda_i=0} \left(-iu_i O_{ij} \frac{dz_j}{dt} + iu_i O_{ij} D_{1,j}^{00} \right).$$
(6.34)

Performing the u_i path integral over (6.34) gives rise to a delta functional

$$\varrho(z,\phi^{+},\phi^{-},t_{f}) = \int \mathcal{D}\phi^{-}\mathcal{D}\phi^{+}\mathcal{D}\pi^{-}\mathcal{D}\pi^{+}\mathcal{D}u_{\lambda>0}\mathcal{D}z \prod_{k|\lambda_{k}=0} \delta(O_{kj}\frac{dz_{j}}{dt} - O_{kj}D_{1,j})) \times e^{\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},u_{\lambda>0},z,t_{i},t_{f})} \varrho(z_{i},\phi_{i}^{+},\phi_{i}^{-},t_{i}),$$
(6.35)

where we denote $\mathcal{D}u_{\lambda>0}$ as the remaining path integral over the variables with eigenvalue $\lambda_i > 0$. $\mathcal{D}u_{\lambda>0}$ should be understood as the $K \to \infty$ limit of $\prod_{k=1}^{K} \prod_{i=1}^{R} \frac{d\vec{u}_{k,i}}{2\pi}$, where $i = 1, \ldots R$ label the components of $u_{k,i}$ which have positive eigenvalue $\lambda_i > 0$.

Integral over response variables with $\lambda_i > 0$

Now let us consider the path integral for the response variables u_i which couple to a positive eigenvalue $\lambda_i > 0$. The terms in the CQ action (6.25) which involve these response variables are given by S_C and S_{CQ}

$$\sum_{i|\lambda_i \ge 0,j} \left[-iu_i O_{ij} \frac{dz_j}{dt} + iu_i O_{ij} D_{1,j}^{00} - u_i \lambda_i u_i + iu_i O_{ij} D_{1,j}^{0\alpha}(z) L_{\alpha}^{*-} + iu_i O_{ij} D_{1,j}^{\alpha 0}(z) L_{\alpha}^{+} \right].$$
(6.36)

Equation (6.36) is quadratic in u_i and the quadratic terms couple to a now positive matrix $u_i \lambda_i \delta_{ij} u_i$. As such, at each timestep t_k we can perform a standard Gaussian integral

$$\int \exp d^R u \left(-\frac{1}{2} u \cdot A \cdot u + iJ \cdot u \right) = \sqrt{\frac{(2\pi)^R}{\det A}} \exp\left(-\frac{1}{2} J \cdot A^{-1} \cdot J \right), \tag{6.37}$$

and integrate out the remaining response variables. Explicitly, at each time-step we can perform the integration over the remaining u_i variables using Equation (6.37), identifying R as the rank of the positive definite block of $D_{2,ij}$, A as the $R \times R$ matrix with elements $A_{ij} = 2(\delta t)\delta_{ij}\lambda_j$ and finally $J_i = O_{ij}D_{1,diff}$ where $D_{1,j}^{diff}$ is the vector

$$D_{1,i}^{diff}(z,\phi^{\pm},\pi^{\pm},t) = D_{1,i}^{00}(z,t) + D_{1,i}^{0\alpha}(z,t)L_{\alpha}^{*}(\phi^{-},\pi^{-}) + D_{1,i}^{\alpha0}(z,t)L_{\alpha}(\phi^{+},\pi^{+}) - \frac{dz_{i}}{dt}.$$
 (6.38)

 $D_{1,j}^{diff}$ describes the difference between the classical path $\frac{dz_i}{dt}$ and its expected drift, sourced by both quantum back-reaction $D_{1,i}^{0\alpha}(z)L_{\alpha}^*(\phi^-,\pi^-) + D_{1,i}^{\alpha 0}(z)L_{\alpha}(\phi^+,\pi^+)$, and purely classical drift $D_{1,i}^{00}$. Using Equation (6.37) we see that for each u_i integration we will pick up a contribution

$$\sqrt{\frac{(2\pi)^R}{(2\delta t)^R \det \lambda^+}} \exp\left(-\frac{1}{4}D_1^{diff} \cdot D_2^{-1} \cdot D_1^{diff} \delta t\right),\tag{6.39}$$

where det λ^+ is defined as the product of all the positive eigenvalues of $D_{2,ij}$.

It is important to emphasise that the diffusion matrix $D_2(z,t)$ can be z dependent, as can its eigenvalues, and so the $\lambda^+(z,t)$ appearing in Equation (6.39) is also generically z dependent. As such, the prefactor $\frac{1}{\sqrt{\det \lambda^+}}$ is not simply a normalization constant which we would expect to cancel when computing correlation functions, instead it is important to keep the $\lambda^+(z)$ in the path integral. This term is similar in nature to the additional curvature term which arises when studying classical path integrals in curved space, as well as when studying Langevin equations with multiplicative noise [161, 162]. For notational convenience, we will absorb the prefactor into measure dz. All together we can integrate out all the response variables u, to write the classical quantum path integral in terms of the variables ϕ^{\pm}, π^{\pm}, z alone

$$\varrho(z,\phi^{+},\phi^{-},t_{f}) = \int \mathcal{D}\phi^{-}\mathcal{D}\phi^{+}\mathcal{D}\pi^{-}\mathcal{D}\pi^{+}\mathcal{D}z \prod_{k|\lambda_{k}=0} \delta(O_{kj}\frac{dz_{j}}{dt} - O_{kj}D_{1,j})) \times e^{\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},z,t_{i},t_{f})}\varrho(z_{i},\phi_{i}^{+},\phi_{i}^{-},t_{i}),$$
(6.40)

where the action takes the form

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},z,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[-\frac{1}{4} D_{1}^{diff} \cdot D_{2}^{-1} \cdot D_{1}^{diff} + i\dot{\phi}^{+}\pi^{+} - iH^{+} - i\dot{\phi}^{-}\pi^{-} + iH^{-} + D_{0}^{\alpha\beta} L_{\alpha}^{+} L_{\beta}^{*-} - \frac{1}{2} D_{0}^{\alpha\beta} \left(L_{\beta}^{*-} L_{\alpha}^{-} + L_{\beta}^{*+} L_{\alpha}^{+} \right) \right],$$
(6.41)

and we have redefined the Dz integration measure, which due to Equation (6.39), now reads

$$\prod_{k=1}^{K} \left(\frac{\pi}{\delta t}\right)^{R/2} \frac{1}{\sqrt{\det \lambda^{+}}} dz_k.$$
(6.42)

Equation (6.41) gives the most general path integral for the entire class of continuous CQ master equations. The conditions for the underlying dynamics to be completely positive can be read off directly from Equation (6.24). In particular, the underlying dynamics will be CP if and only if the Lindbladian coefficient D_0 is positive semi-definite $D_0 \succeq 0$, $(\mathbb{I} - D_0 D_0^{-1})D_1 = 0$ and $2D_2 \succeq D_1 D_0^{-1} D_1^{\dagger}$ where D_0^{-1} is the generalized inverse D_0 .

The classical-quantum interaction of the action in Equation (6.41) is contained in the term

$$-\int_{t_i}^{t_f} dt \; \frac{1}{4} D_1^{diff}(z, \phi^{\pm}, \pi^{\pm}, t) \cdot D_2^{-1}(z, t) \cdot D_1^{diff}(z, \phi^{\pm}, \pi^{\pm}, t), \tag{6.43}$$

which, using the definition of D_1^{diff} in Equation (6.38), has a very natural interpretation as suppressing contributions to the path integral where paths $\frac{dz}{dt}$ differ from the expected drift, $D_{1,i}^{00} + D_{1,i}^{0\alpha} L_{\alpha}^{*-} + D_{1,i}^{\alpha 0} L_{\alpha}^{+}$, which is sourced by both the pure classical evolution and the quantum back-reaction on the system. We also see that the amount one is penalized for moving away from the expected classical trajectory depends on the inverse of the diffusion matrix; if there is a large amount of diffusion, then D_2^{-1} is expected to be small, and so classical paths which venture away from the expected value are penalized less. However, if there is very little diffusion then D_2^{-1} will be large, and classical trajectories are forced to stick nearby the most likely path. Since both D_2^{-1} and D_0 cannot simultaneously be made small by virtue of the complete positivity condition in Equation (6.24): there is a trade-off between the amount of diffusion in the classical system and the strength of the Lindbladian evolution of the quantum system which is characterized by the second line of Equation (6.41). To further gain intuition for the path integral, let us now discuss some cases in which the path integral in Equation (6.41) simplifies.

6.3.1 Hermitian Lindblad operators

A particularly nice interpretation of the path integral arises when the Lindblad operators L_{α} in Equation (6.41) are Hermitian. In this case we can write the pure Lindbladian term of the path integral S_{FV} in Equation (6.19) in terms of a negative semi-definite quadratic form

$$iS_{FV} = -\frac{1}{2} \int_{t_i}^{t_f} dt (L_{\alpha}^- - L_{\alpha}^+) D_0^{\alpha\beta} (L_{\beta}^- - L_{\beta}^+), \qquad (6.44)$$

and so Equation (6.41) reads

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},z,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[i\dot{\phi^{+}}\pi^{+} - iH^{+} - i\dot{\phi^{-}}\pi^{-} + iH^{-} + \frac{1}{2}(L_{\alpha}^{-} - L_{\alpha}^{+})D_{0}^{\alpha\beta}(L_{\beta}^{-} - L_{\beta}^{+}) - \frac{1}{4}D_{1}^{diff} \cdot D_{2}^{-1} \cdot D_{1}^{diff} \right].$$
(6.45)

Reminding ourselves that the path integral can be used to compute the off-diagonal elements of the density operator $\varrho(z, \phi^+, \phi^-, t_f)$, we see that the Lindbladian part of the action suppresses the off-diagonal elements by an amount dependent on $L_{\alpha}^- - L_{\alpha}^+$ and the Lindbladian coupling matrix $D_0^{\alpha\beta}$. In particular when $D_0^{\alpha\beta}$ is large, we find that paths where $L_{\alpha}^- \sim L_{\alpha}^+$ are heavily preferred, causing the density matrix to decohere into the eigenbasis dictated by the Lindblad operators. On the other hand, if the magnitude of the decoherence is small, then one can maintain superpositions for a long time. However, we know from the CP conditions that long coherence times leads to necessarily large D_2 on the classical system, with the precise relationship determined by the strength of the quantum back-reaction D_1 . Large diffusion means that D_2^{-1} will be small, and so paths can deviate away from there average without much suppression; there is a lot of classical uncertainty. In other words, if the decoherence rate is small, so that superpositions can be maintained, then measuring the classical degree of freedom necessarily gives you little information about the coherence of the quantum state.

6.3.2 Hamiltonian drift

Another simplification occurs when the drift D_1^{drift} is generated by a CQ Hamiltonian $H_C[q, p] + V_I[q, \phi]$, as in [28, 10, 60]. We assume the interaction is minimally coupled, so that the interaction potential $V_I[q, \phi]$ depends on classical and quantum position but not momenta, whilst $H_C[q, p]$ we take to be a purely classical Hamiltonian. When the drift is generated by a Hamiltonian, the back-reaction, described by the $D_1^{0\alpha}$ term in the master equation (6.21), is given by the Alexandrov-Gerasimenko bracket [163, 104]

$$\frac{1}{2}(\{V_I,\varrho\} - \{\varrho, V_I\}),\tag{6.46}$$

where for simplicity we consider only one classical degree of freedom. Furthermore, due to the trade-off in Equation (6.24), we know the full master equation must contain a term

$$\frac{\partial \varrho(q,p)}{\partial t} = \{H_C, \varrho\} - i[H_Q, \varrho]
+ \frac{\partial^2}{\partial p^2} (D_2 \varrho) + \frac{1}{2} (\{V_I(q,\phi), \varrho\} - \{\varrho, V_I(q,\phi)\}) - \frac{1}{2} D_0 \{\frac{\partial V_I}{\partial q}, \{\frac{\partial V_I^{\dagger}}{\partial q}, \varrho\}\},$$
(6.47)

where H_Q is a quantum Hamiltonian that can in general be phase space dependent and complete positivity demands $8D_2 \ge D_0^{-1}$. For the master equation in Equation (6.47), the path integral action of Equation (6.41) is

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},z,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \bigg[i\dot{\phi^{+}}\pi^{+} - iH_{Q}^{+} - i\dot{\phi^{-}}\pi^{-} + iH_{Q}^{-} + \frac{D_{0}}{2} \bigg(\frac{\partial V_{I}}{\partial q}^{-} - \frac{\partial V_{I}}{\partial q}^{+} \bigg)^{2} - \frac{1}{4} D_{1}^{diff} \cdot D_{2}^{-1} \cdot D_{1}^{diff} \bigg],$$
(6.48)

where the classical-quantum interaction takes the form

$$D_1^{diff} \cdot D_2^{-1} \cdot D_1^{diff} = D_2^{-1} \left(\{H_C, p\} + \frac{1}{2} \{V_I[q, \phi^+], p\} + \frac{1}{2} \{V_I[q, \phi^-], p\} - \dot{p} \right)^2.$$
(6.49)

Equation (6.49) thus acts to suppress paths which deviate from the \pm averaged Hamiltonian equations of motion. In Section 6.4, we show that when the action is at most quadratic in the momenta of the classical and quantum systems, one can integrate out the momentum variables to arrive at a configuration space path integral, where paths deviating from their averaged Euler-Lagrange equations are suppressed.

6.3.3 When the trade-off is saturated

We now see that when one saturates the decoherence-diffusion trade-off, a remarkable set of cancellations occur, and the path integral takes on a very simple form. In particular, it factorises, so that there is no coupling between the ϕ^+ and ϕ^- fields. This reflects the fact that when the trade-off is saturated the classical-quantum dynamics keeps quantum states pure conditioned on the classical trajectories, as we saw in Chapter 5. When the trade-off is saturated, we can expand out the classical-quantum interaction term in the path integral of Equation (6.41) to find that all of the cross terms involving \pm vanish. In particular those arising from D_0 cancel with those arising from D_2^{-1} and Equation (6.41) reduces to

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},z,t_{i},t_{f}) = \mathcal{I}_{CQ}^{+}(\phi^{+},\pi^{+},z,t_{i},t_{f}) + \mathcal{I}_{CQ}^{-*}(\phi^{-},\pi^{-},z,t_{i},t_{f}) - I_{C}(z,t_{i},t_{f}), \quad (6.50)$$

where

$$\mathcal{I}_{CQ}(\phi,\pi,z,t_i,t_f) = \int_{t_i}^{t_f} \left[i\dot{\phi}\pi - iH - \frac{1}{2} (D_{1,i}^{00} - \frac{dz_i}{dt}) (D_2^{-1})^{ij} D_{1,j}^{0\alpha} L_{\alpha}^* - \frac{1}{4} (L_{\alpha} D_0^{\alpha\beta} L_{\beta}^* + D_{1,i}^{0\alpha} L_{\alpha}^* (D_2^{-1})^{ij} D_{1,j}^{0\beta} L_{\beta}^*) \right]$$
(6.51)

$$I_C(z, t_i, t_f) = \int_{t_i}^{t_f} \left[\frac{1}{4} (D_{1,i}^{00} - \frac{dz_i}{dt}) (D_2^{-1})^{ij} (D_{1,j}^{00} - \frac{dz_j}{dt}) \right].$$
(6.52)

When the back-reaction is Hermitian, meaning that $D_{1,i}^{0\alpha}L_{\alpha}(\phi,\pi) = L_i(\phi,\pi,z)$ is Hermitian, Equation (6.51) simplifies further and after eliminating for $2D_0 = D_1D_2^{-1}D_1$ its general form can be written in terms of $(D_2^{-1})^{ij}$ alone

$$\mathcal{I}_{CQ}(\phi,\pi,z,t_i,t_f) = \int_{t_i}^{t_f} \left[i\dot{\phi}\pi - iH - \frac{1}{2} (D_{1,i}^{00} - \frac{dz_i}{dt}) (D_2^{-1})^{ij} L_j - \frac{1}{2} (L_i (D_2^{-1})^{ij} L_j) \right].$$
(6.53)

An important example of dynamics where the trade-off is saturated is an ideal continuous measurement of a Hermitian operator $Z(z_t)$, where we can also allow the choice of continuous measurement, and its strength $k(z_t)$, to depend on the measurement outcome z_t at time t.

6.3.4 A path integral for continuous measurement and Markovian feedback

In this section we arrive at the general form of path integral representation for a continuous measurement procedure. To construct a continuous measurement, we divide time into a sequence of intervals of length δt , and consider a weak measurement in each interval. To obtain a

continuous measurement, we make the strength of each measurement proportional to the time interval, and then take the limit in which the time intervals become infinitesimally short. We consider a continuous measurement of a Hermitian operator $Z(z_t)$, which can depend on the measurement signal z_t at time t, and which we take to be a functional of x, p. The measurement signal z_t is related to the measurement outcome α_t by $z_t = \alpha_t dt$ [121]. The measurement signal undergoes continuous evolution, whilst the measurement outcome α_t is wildly discontinuous, especially for weak measurements where little information is gained in each timestep. It is well known [122, 121] that for a continuous measurement the dynamics of the quantum state can be described by the set of coupled differential equations

$$d|\psi(t)\rangle = \left\{-k(z_t)(Z(z_t) - \langle Z(z_t)\rangle)^2 dt + \sqrt{2k(z_t)}(Z(z_t) - \langle Z(z_t)\rangle)d\xi\right\}|\psi(t)\rangle$$
(6.54)

$$dz_t = \langle Z(z_t) \rangle dt + \frac{d\xi}{\sqrt{8k(z_t)}},\tag{6.55}$$

where z_t is the measurement signal, or record, and $k(z_t)$ parameterizes the measurement strength, which can in general also be z_t dependent, and for simplicity we are ignoring the pure quantum evolution in comparison to the measurement dynamics. The stochastic part of the evolution, described via $d\xi_i$, is the standard multivariate Wiener process satisfying the Ito rules $d\xi_i d\xi_j = \delta_{ij} dt$, $d\xi_i dt = 0$. The measurement outcome itself undergoes white noise dynamics $\alpha_t = \langle Z(z_t) \rangle + \frac{1}{8k(z_t)} \frac{d\xi}{dt}$ and is discontinuous in time which is why the signal z_t is preferred. From the measurement signal, one can obtain the measurement record by taking the time derivative.

From the set of non-linear stochastic differential equations given in (6.54), (6.55), it is possible using the methods outlined in Chapter 5 [6] to construct a *linear* classical-quantum master equation for the combined classical-quantum state given by

$$\frac{\partial \varrho(z)}{\partial t} = -k(z)[Z(z), [Z(z), \varrho(z)]] + \frac{Z(z)}{2} \frac{\partial \varrho(z)}{\partial z} + \frac{\partial \varrho(z)}{\partial z} \frac{Z(z)}{2} + \frac{1}{16} \frac{\partial^2}{\partial z^2} (k^{-1}(z)\varrho(z)), \quad (6.56)$$

from which we identify $D_1^{0,Z} = D_1^{Z,0} = \frac{1}{2}$, $D_0(z) = 2k(z)$ and $D_2(z) = \frac{1}{16k(z)}$. We can easily check that for perfect measurements, the decoherence diffusion trade-off in Equation (6.23) is satisfied and in fact saturated. Substituting into Equation (6.45), we find the corresponding action

$$\mathcal{I}(x^{\pm}, p^{\pm}, z, t_i, t_f) = \int_{t_i}^{t_f} dt \bigg[i\dot{x}^+ p^+ - i\dot{x}^- p^- - k(z)(Z^-(x, p) - Z^+(x, p))^2 - 4k(z) \bigg(\frac{1}{2}(Z^+(x, p) + Z^-(x, p)) - \frac{dz}{dt} \bigg)^2 \bigg].$$
(6.57)

To our knowledge, such a general form of Markovian continuous measurement path integral has not appeared in the literature, and is complementary to current approaches [148, 149, 150, 151, 151, 152]. One could also write down a coherent state path integral for continuous measurement using the methods introduced in [97], which could be of use in studying problems in optical quantum feedback [164]. One can also allow for noisy measurements by including an appropriate Feynman-Vernon term in Equation (6.57). We should also emphasize that we have derived this from slightly different considerations than other approaches, namely by starting from complete positivity of classical-quantum dynamics. Such path integrals have proved useful in optimizations of quantum control tasks, especially in the strong measurement regime where saddle-point approximations are valid [148, 149].

It is important to note that the path integral in Equation (6.41) is more general than the continuous measurement path integral of Equation (6.57). Firstly, it allows for the case where there are many measurement operators and outcomes. It also allows for noisy imperfect quantum measurements. More importantly, it allows for the case where both the classical and the quantum degrees of freedom have dynamics of their own, which is encoded in the fact that in Equation (6.41) D_1^{diff} can contain purely classical evolution determined by the drift D_1^{00} , and we can also include purely Hamiltonian, and more generally Lindbladian, quantum evolution.

To summarize this section, we have shown that one can integrate out the response variables for the class of continuous CQ master equations to arrive at the path integral in Equation (6.41), which is represented in terms of the quantum phase space variables ϕ , π and the classical variables z alone. In the next section, we look at examples where we can arrive at a configuration space path integral in both quantum and classical degrees of freedom.

6.4 Configuration space path integrals

In this section, we consider configuration space path integrals. A configuration space path integral is an important tool in understanding whether or not one can construct covariant theories (fundamental or effective) of interacting classical-quantum fields. The conditions for which we can arrive at a configuration space path integral are standard. In ordinary quantum mechanics we are able to integrate out the quantum momenta if the path integral is at most quadratic in the quantum momentum variables. In the CQ case, if the action is quadratic in both the classical and quantum momenta then we can arrive at a full configuration space path integral. For all such dynamics, the methodology of arriving at a configuration space path integral is the same: at each time step, one completes the square in the action and performs a Gaussian path integral. In this section we focus on a class of CQ dynamics which end up having interesting configuration space path integrals. In particular, as in [28], we take the classical degrees of freedom to live in a phase space and we take the CQ dynamics to be generated by an interaction potential V_I . We assume the dynamics are *minimally coupled*, which we take to mean that the CQ couplings depend only on position and not on momenta, and we diffuse only in momenta. This class of dynamics proves to be interesting since one finds a configuration space path integral which can be written in terms of a CQ proto-action W_{CQ} , as summarized by Equation (6.66).

6.4.1 An explicit derivation of a configuration space path integral

Consider a classical system $z = (q_i, p_j)$, coupled to a quantum system ϕ, π . In this subsection we consider continuous master equations of the form

$$\frac{\partial \varrho(q,p)}{\partial t} = \{H_C, \varrho\} - i[H_Q, \varrho] + \frac{\partial^2}{\partial p_i \partial p_j} (D_{2,ij}(q_i)\varrho) + \frac{1}{2} \left(\{V_I(q_i,\phi), \varrho\} - \{\varrho, V_I(q_i,\phi)\}\right)
+ D_0^{\alpha\beta}(q_i) \left(L_\alpha(\phi)\varrho L_\beta^{\dagger}(\phi) - \frac{1}{2}\{L_\beta^{\dagger}(\phi)L_\alpha(\phi), \varrho\}\right),$$
(6.58)

where $H_C = \sum_i \frac{p_i^2}{2m_i} + V(q_i)$ is the purely classical Hamiltonian, $H_Q = \frac{\pi^2}{2m_Q} + V(\phi)$ is the purely quantum Hamiltonian and $V_I(q_i, \phi)$ is a classical-quantum interaction potential, a hybrid object which we take to only depend on the classical and quantum positions q_i, ϕ . We also assume that V_I is Hermitian, although in general this need not be the case. We take the Lindbladian coupling

 $D_0^{\alpha\beta}(q_i,t)$ and the diffusion coefficient $D_{2,ij}(q_i,t)$ to depend on the classical positions only. We emphasize the conditions imposed on Equation (6.58) are not necessary to get a configuration space path integral; one only requires that the momentum dependence be at most quadratic. However, including q_i diffusion, or momentum dependence π in the Lindblad operators, alters the form of the momentum integral and complicates the final form of path integral we find.

Since we can make an arbitrary selection of Lindblad operators, we use this to fix $\frac{1}{2}D_{1,i}^{0\alpha} = \delta_{\alpha}^{i}$, and take a basis of Lindblad operators which includes $L_{i} = \frac{\partial V_{I}}{\partial q_{i}}$. With this choice, Equation (6.58) becomes

$$\frac{\partial \varrho(q,p)}{\partial t} = \{H_C, \varrho\} - i[H_Q + V_I, \varrho] + \frac{\partial^2}{\partial p_i \partial p_j} (D_{2,ij}(q_i)\varrho) + \frac{1}{2} (\{V_I(q_i,\phi), \varrho\} - \{\varrho, V_I(q_i,\phi)\}) \\
+ D_0^{ij}(q_i) \left(\frac{\partial V_I}{\partial q_i} \varrho \frac{\partial V_I^{\dagger}}{\partial q_j} - \frac{1}{2} \left\{\frac{\partial V_I^{\dagger}}{\partial q_j} \frac{\partial V_I}{\partial q_i}, \varrho\right\}\right) + \mathcal{L}_{Lindblad}(\varrho),$$
(6.59)

where $\mathcal{L}_{Lindblad}$ denotes the collection of pure Lindbladian terms one could include in Equation (6.59) that are not associated to $\frac{\partial V_I}{\partial q_i}$. Since these will not be accompanied by any back-reaction on the classical degrees of freedom we will ignore them, focusing on terms associated to back-reaction alone. The $\mathcal{L}_{Lindblad}$ terms could easily be added back to the final configuration space path integral by a suitable choice of Feynman-Vernon action. Since we have fixed $D_{1,i}^{0\alpha} = \frac{1}{2}\delta_{\alpha}^{i}$ the complete positivity condition of Equation (6.24) is now that $8D_2 \succeq D_0^{-1}$.

The transition amplitude for Equation (6.59) reads

$$\varrho(q, p, \phi^+, \phi^-, t_f) = (\prod_i m_i) \int \mathcal{D}\phi^{\pm} \mathcal{D}\pi^{\pm} \mathcal{D}q \mathcal{D}p\delta(m_i \dot{q}_i - p_i)) \ e^{\mathcal{I}(\phi^-, \phi^+, \pi^-, \pi^+, q, p, t_i, t_f)} \varrho(q, p, \phi_i^+, \phi_i^-, t_i),$$
(6.60)

which can be read off from Equation (6.40). The corresponding action is given by

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},q,p,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[i\dot{\phi^{+}}\pi^{+} - iH^{+} - iV_{I}^{+} - i\dot{\phi^{-}}\pi^{-} + iH^{-} + iV_{I}^{-} \right. \\
\left. - \frac{1}{2} \left(\frac{\partial V_{I}^{-}}{\partial q_{i}} - \frac{\partial V_{I}^{+}}{\partial q_{i}} \right) D_{0}^{ij}(q,t) \left(\frac{\partial V_{I}^{-}}{\partial q_{j}} - \frac{\partial V_{I}^{+}}{\partial q_{j}} \right) \\
\left. - \frac{1}{4} \left(\dot{p}_{i} + \frac{\partial H_{C}}{\partial q_{i}} + \frac{1}{2} \frac{\partial V_{I}^{-}}{\partial q_{i}} + \frac{1}{2} \frac{\partial V_{I}^{+}}{\partial q_{i}} \right) (D_{2}^{-1})^{ij}(q,t) \left(\dot{p}_{j} + \frac{\partial H_{C}}{\partial q_{j}} + \frac{1}{2} \frac{\partial V_{I}^{-}}{\partial q_{j}} + \frac{1}{2} \frac{\partial V_{I}^{+}}{\partial q_{j}} \right) \right].$$
(6.61)

Since there is no q_i diffusion, the pure Hamiltonian part of the classical evolution enforces the constraint $p_i = m_i \dot{q}_i$ via the delta functional in Equation (6.60). Combined with the form of

back-reaction in Equation (6.61), we see the action suppresses paths which deviate from the \pm averaged Hamilton's equations, just as in Section 6.3.2. We can now perform the classical momentum integration, including over the final momenta, to get a path integral over the classical configuration space. Doing this, Equation (6.61) becomes

$$\mathcal{I}(\phi^{-},\phi^{+},\pi^{-},\pi^{+},q,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[+i\dot{\phi^{+}}\pi^{+} - iH^{+} - iV_{I}^{+} - i\dot{\phi^{-}}\pi^{-} + iH^{-} + iV_{I}^{-} - \frac{1}{2}\left(\frac{\partial\Delta V_{I}}{\partial q_{i}}\right)D_{0}^{ij}\left(\frac{\partial\Delta V_{I}}{\partial q_{j}}\right) - \frac{1}{4}\left(m_{i}\ddot{q}_{i} + \frac{\partial H_{C}}{\partial q_{i}} + \frac{\partial\bar{V}_{I}}{\partial q_{i}}\right)\left(D_{2}^{-1}\right)^{ij}(q)\left(m_{j}\ddot{q}_{j} + \frac{\partial H_{C}}{\partial q_{j}} + \frac{\partial\bar{V}_{I}}{\partial q_{j}}\right)\right].$$
(6.62)

In Equation (6.62) we have introduced the notation $\bar{V}_I = \frac{1}{2}(V_I^- + V_I^+)$, the \pm average potential and $\Delta V_I = V_I^- - V_I^+$ as the difference in the potential along the \pm branches. If we further define the classical Lagrangian $L_C(q) = \sum_i p_i \dot{q}_i - H_C(q_i, p_i)$ then we recognise the CQ interaction term in Equation (6.62) as the Euler-Lagrange equations which result from varying $\int_{t_i}^{t_f} dt (L_C - \bar{V}_I)$.

We can write the action in a more compact form by defining the CQ proto-action W_{CQ} via

$$W_{CQ}[q,\phi] = \int_{t_i}^{t_f} dt L_Q(\phi) + L_C(q) - V_I(q,\phi) := \int_{t_i}^{t_f} L_{CQ}(q,\phi),$$
(6.63)

where $L_Q(\phi)$ is the quantum Lagrangian, $L_C(q)$ the classical Lagrangian, and $V_I(q, \phi)$ is the interaction potential. We can rewrite the CQ interaction in Equation (6.62) in terms of variations of the \pm averaged CQ proto-action \bar{W}_{CQ}

$$-\frac{1}{4}\left(m_i\ddot{q}_i + \frac{\partial H_C}{\partial q_i} + \frac{\partial \bar{V}_I}{\partial q_i}\right)(D_2^{-1})^{ij}(q)\left(m_j\ddot{q}_j + \frac{\partial H_C}{\partial q_j} + \frac{\partial \bar{V}_I}{\partial q_j}\right)$$

$$= -\frac{1}{4}\frac{\delta}{\delta q_i}(\bar{W}_{CQ}[q,\phi^{\pm}])(D_2^{-1})^{ij}\frac{\delta}{\delta q_j}\bar{W}_{CQ}[q,\phi^{\pm}]).$$
(6.64)

In order to get a full configuration space path integral, all that remains is to do the integrals over the momentum of the quantum system. As with the standard quantum path integral, the technical requirement to be able to do this is that the action in Equation (6.62) is quadratic in π^+ , π^- so that we can perform the π^+ , π^- integral exactly by completing standard Gaussian integrals. Since we have taken the simplest case where the only momentum dependence π^- , π^+ comes from the Hamiltonian, the result of the momentum integration is to perform a Legendre transformation. We end up with the configuration space CQ path integral representation of the transition amplitude

$$\varrho(q,\phi^+,\phi^-,t_f) = \mathcal{N} \int \mathcal{D}\phi^- \mathcal{D}\phi^+ \mathcal{D}q \ e^{\mathcal{I}(\phi^-,\phi^+,q,t_i,t_f)} \varrho(q,\phi_i^+,\phi_i^-,t_i), \tag{6.65}$$

where $\mathcal{N} = \prod_i m_i$ is a normalization constant arising from the classical momentum path integral, and we have absorbed the usual factors from the Gaussian integrals into the definition of $D\phi^{\pm}$ to obtain the standard path integral measures. The action in Equation (6.65) takes its final form

$$\mathcal{I}(\phi^{-},\phi^{+},q,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \left[iL_{CQ}[q,\phi^{+}] - iL_{CQ}[q,\phi^{-}] - \frac{1}{2} \left(\frac{\delta \Delta W_{CQ}[q,\phi^{\pm}]}{\delta q_{i}} \right) D_{0}^{ij} \left(\frac{\delta \Delta W_{CQ}[q,\phi^{\pm}]}{\delta q_{j}} \right) - \frac{1}{4} \frac{\delta}{\delta q_{i}} (\bar{W}_{CQ}[q,\phi^{\pm}]) (D_{2}^{-1})^{ij} \frac{\delta}{\delta q_{j}} (\bar{W}_{CQ}[q,\phi^{\pm}]) \right].$$
(6.66)

Equation (6.66) is remarkably simple. All of the classical-quantum interaction is encoded in variations of a single classical-quantum proto-action W_{CQ} and, due to the choice of Lindblad operators, the complete positivity condition is that $8D_2 \succeq D_0^{-1}$. The classical trajectories are suppressed away from the \pm averaged equations of motion which arise from varying the proto-action by an amount depending on D_2 , whilst there is simultaneous decoherence by an amount which depends on the difference in the equations of motion between the \pm branches. In Appendix D we discuss a simple toy example of the configuration space path integral and illustrate how one can use perturbative methods familiar in quantum theories to calculate CQ path integrals using CQ Feynman diagrams.

This direct derivation of the path integral was valid for the family of master equations given by Equation (6.58), which couple less than quadratically in the momenta. However, given the final and suggestive form of Equation (6.66) it is tempting to take it as a *definition* of classicalquantum dynamics and let W_{CQ} be an arbitrary functional of q, ϕ^{\pm} and their derivatives. Although the mapping from the path integral to the master equation may not be completed analytically, one might expect that the condition that $8D_2 \succeq D_0^{-1}$ is sufficient for the dynamics to be completely positive. One often does something similar in quantum theory by taking the path integral formulation to be the fundamental object of study, which often includes higher derivative terms in the action even though the mapping between master equation's and path integral can only be computed exactly when the master equation is at most quadratic in momenta. Inspired by the action in Equation (6.66), in Chapter 8 we prove directly from path integral methods that *any* configuration space path integral of the form

$$\mathcal{I}(\phi^{-},\phi^{+},q,t_{i},t_{f}) = \mathcal{I}_{CQ}(q,\phi^{+},t_{i},t_{f}) + \mathcal{I}_{CQ}^{*}(q,\phi^{-},t_{i},t_{f}) - \mathcal{I}_{C}(q,t_{i},t_{f}) + \int_{t_{i}}^{t_{f}} dt d\bar{x} \sum_{\gamma} c^{\gamma}(q,t,\bar{x}) (L_{\gamma}[\phi^{+}]L_{\gamma}^{*}[\phi^{-}])$$
(6.67)

defines completely positive CQ dynamics (not necessarily normalized). In Equation (6.67) $c^{\gamma} \geq 0$, $L_{\gamma}[\phi^{\pm}]$ can be any functional of the bra and ket variables, \mathcal{I}_{CQ} determines the CQ interaction on each of the ket and bra paths and $\mathcal{I}_C(q, t_i, t_f)$ is the purely classical action which takes real values. In order for the path integral to be convergent we impose that \mathcal{I}_C is positive (semi) definite, as well as asking that the real part of \mathcal{I}_{CQ} be negative (semi) definite. The term on the second line of Equation (6.67) can be thought of the path integral version of a Kraus operation, and allows one to incorporate a loss of quantum information into the path integral through Lindbladian terms when the decoherence-diffusion trade-off is not saturated. Equation (6.66) is a special case of Equation (6.67) when $8D_0 \succeq D_2^{-1}$ is satisfied, which can be seen by expanding out the CQ action and grouping terms by $8D_0 - D_2^{-1}$ [8]. This is true for an *arbitrary* CQ proto-action W_{CQ} .

This result of Equation (6.67) allows us to study consistent classical-quantum path integrals, even when we cannot perform the momentum integration exactly from the Hilbert space picture. Instead, using Equation (6.67) as a starting point, we can write down CP CQ dynamics by an appropriate choice of W_{CQ} . In doing so one must be careful to ensure that the action dynamics is also normalized. Normalization is ensured if one includes the appropriate classical and quantum kinetic terms, which we discuss in more detail in Chapter 8.

6.5 Path integrals for classical fields interacting with quantum fields

In this section, we comment on the path integral for classical fields interacting with quantum fields. This provides a natural arena to study the renormalization properties of classicalquantum dynamics, as well as covariant properties of classical-quantum field theories. We study this in more detail in Chapter 8. We treat the path integral as a formal object, making no attempt to prove anything rigorously, as is often the case with field theories.

The path integral remains largely unchanged for the case of fields. Starting with a classicalquantum master equation involving fields, one can (formally) insert various resolutions of the identity and arrive at analogous formulas for Equations (6.16), (6.41), (6.66). We do not reproduce these steps here, since they are identical to those in rest of the chapter. Instead we quote the final result for *ultra-local* classical-quantum dynamics, which is to send all the couplings appearing in the action $D_{n,i_1...i_n}^{\mu\nu}(z,t) \to D_{n,i_1...i_n}^{\mu\nu}(z,x)$, the classical variables appearing in the action $z_i \to z_i(x)$, the quantum variables to $(\phi, \pi) \to (\phi(x), \pi(x))$ and finally we one must integrate over all space in the action $\int dt \to \int dx$, where $x = (t, \vec{x})$. To be explicit in understanding the field theoretic case, in this section we consider a master Equation which has a Lorentz invariant path integral.⁴

For a quantum field $\phi(x)$ coupled to a classical field q(x), the field theoretic version of the master Equation in (6.58) is

$$\frac{\partial \varrho(q,p)}{\partial t} = \{H_C, \varrho\} - i[H_Q, \varrho] + \frac{1}{2} \left(\{V_I(q,\phi), \varrho\} - \{\varrho, V_I(q,\phi)\}\right) \\
+ \int d\vec{x} \, \frac{\delta^2}{\delta p_i(\vec{x}) \delta p_j(\vec{x})} (D_{2,ij}(q,t,\vec{x})\varrho) \\
+ \int d\vec{x} \, D_0^{ij}(q,t,\vec{x}) \left(\frac{\delta V_I}{\delta q_i(\vec{x})} \varrho \frac{\delta V_I^{\dagger}}{\delta q_j(\vec{x})} - \frac{1}{2} \{\frac{\delta V_I^{\dagger}}{\delta q_j(\vec{x})} \frac{\delta V_I}{\delta q_i(\vec{x})}, \varrho\}\right),$$
(6.69)

where $V_I[q, \phi] = \int d\vec{x} \mathcal{V}_I[q, \vec{x}]$ is an interaction potential and we take the purely classical part of the dynamics to be generated by the action $S_C(q) = \int dt \int d\vec{x} \mathcal{L}_C[q, x]$. It should be noted that Equation (6.69) needs regularizing, since there are multiple functional derivatives acting at the same point x. This corresponds to the fact that in the field theoretic case the path integral will require renormalization. For the choice of dynamics in Equation (6.69), the path integral

$$\mathcal{I}(\phi^{-},\phi^{+},t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt \bigg[i\mathcal{L}_{Q}^{+}(x) - i\mathcal{L}_{Q}^{-}(x) - \frac{D_{0}}{2} \int dx \left(\phi^{-}(x) - \phi^{+}(x)\right)^{2} \bigg].$$
(6.68)

 $^{^{4}}$ In the purely quantum case, Lorentz invariant open systems, and their renormalization properties, have recently been studied in [165]. A simple example of Lorentz invariant quantum dynamics, which was shown to be renormalizable [165], is given by the Lorentz invariant Lindblad Equation which has the action

action is again found to be of the form in Equation (6.66)

$$\mathcal{I}(\phi^{-},\phi^{+},q,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt d\vec{x} \Big[i\mathcal{W}_{CQ}^{+}(x) - i\mathcal{W}_{CQ}^{-}(x) \\
- \frac{1}{2} (\frac{\delta \Delta W_{CQ}}{\delta q_{i}}) D_{0}^{ij}(z,t,\vec{x}) (\frac{\delta \Delta W_{CQ}}{\delta q_{j}}) - \frac{1}{4} \frac{\delta}{\delta q_{i}} (\bar{W}_{CQ}[q,\phi^{\pm}]) (D_{2}^{-1})^{ij}(z,t,\vec{x}) \frac{\delta}{\delta q_{j}} (\bar{W}_{CQ}[q,\phi^{\pm}]) \Big],$$
(6.70)

where now

$$W_{CQ}[q,\phi] = \int dt d\vec{x} (\mathcal{L}_Q[\phi] + \mathcal{L}_C[q] - \mathcal{V}_I[q,\phi]) := \int dt \mathcal{W}_{CQ}[q,\phi]$$
(6.71)

is a space-time CQ proto-action.

The path integral enables us to construct CQ theories with space-time symmetries. For example, Equation (6.70) will describe Lorentz invariant CQ dynamics when W_{CQ} is chosen to be a Lorentz invariant scalar. We study field theoretic path integrals in more detail in Chapter 8 [8]. There, we start with Equation (6.67) and prove the resulting dynamics is CP. Given the form of Equation (6.70), it is natural to consider the class of dynamics

$$\mathcal{I}(\phi^{-},\phi^{+},q,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt d\vec{x} \bigg[i\mathcal{W}_{CQ}^{+}(x) - i\mathcal{W}_{CQ}^{-}(x) \\
- \frac{1}{2} \frac{\delta \Delta W_{CQ}}{\delta q_{i}} D_{0}^{ij}(q,t,\vec{x}) \frac{\delta \Delta W_{CQ}}{\delta q_{j}} - \frac{1}{4} \frac{\delta \bar{W}_{CQ}}{\delta q_{i}} (D_{2}^{-1})^{ij}(q,t,\vec{x}) \frac{\delta \bar{W}_{CQ}}{\delta q_{j}} \bigg],$$
(6.72)

where we impose the restriction $8D_0 \succeq D_2^{-1}$ to ensure the action takes the form in Equation (6.67) and is therefore completely positive. Note, the action in Equation (6.72) takes the same form as Equation (6.70) but now we let W_{CQ} be an *arbitrary* CQ proto-action. In Chapter 8 we show one can use Equation (6.72) to construct examples of Lorentz and diffeomorphism invariant dynamics.

6.6 Discussion

In this chapter we have discussed the path integral for general classical-quantum master equations, emphasising the necessary and sufficient conditions for the dynamics to be consistent and completely positive on the quantum system. In the general case we find a path integral representation of the dynamics with response variables, given by Equation (6.16), whilst for the class of continuous master equations, we were able to integrate out the response variables to arrive at the phase space path integral of Equation (6.41). Under certain conditions, namely when the action of Equation (6.41) is at most quadratic in classical (quantum) momenta, we can integrate out the classical (quantum) momenta to arrive at a configuration space path integral. For the case of minimally coupled Hamiltonian theories, we end up with a simple path integral representation, Equation (6.66), where the dynamics is completely encoded via the proto-action W_{CQ} . Given its final form, we posited that the resulting CQ action should be completely positive for an arbitrary proto-action, a result we prove via path integral methods in Chapter 8. We then studied the classical-quantum path integral for fields.

Applications of the CQ path integral. It would be interesting to explore possible applications of the path integral to standard quantum mechanical scenarios. Generally, we expect CQ dynamics to be a good effective description of a quantum system when one part behaves effectively classical [166]. A particularly relevant scenario is perhaps measurement based quantum control [96, 167], or coherent quantum control with dissipative resources [121, 166]. In Appendix 6.3.4 we introduced the path integral for the most general Markovian continuous measurement procedure one can perform. In this context, these path integral can be understood as an extension of [148, 149] which has proved useful in simulating quantum control tasks, particularly in the strong measurement limit where saddle point approximations are valid. We also expect that the path integral could be useful for certain systems in quantum chemistry where hybrid classical-quantum coupling has previously been used to study systems beyond the mean field approximation [120, 119].

Lorentz invariant collapse models via a classical field and relativistic measurement. We saw that the CQ path integral corresponded to a Lorentz invariant path integral which causes decoherence of the quantum state via interaction with a classical field. Using Equation (6.67) we are able to write down families of covariant models with a fundamentally classical field which naturally give rise to a decoherence mechanism on the quantum state. It would be interesting to explore such theories further in the context of relativistic collapse models [168, 169, 170, 171, 123, 172, 173]. The main difference between the CQ dynamics considered here and standard collapse models is that here, the quantum system becomes classicalised through it's interaction with a dynamical physical field, rather than an unobservable auxiliary field.

Renormalization of classical-quantum field theories. The renormalization of open quantum

field theories was recently studied in [165, 174, 61]. It was found that open ϕ^4 theory was perturbatively renormalizable, and the complete positivity condition for the Lindblad Equation was peturbatively preserved under renormalization citebaidya2017renormalization.⁵ On the other hand, open Yukawa theory was found to be non-renormalizable [174]. It would be interesting to explore the renormalization of classical-quantum field theories. This is even more highly constrained than the renormalization of open QFT, since not only does the Lindbladian coupling need to remain positive semi-definite, but so does the diffusion coupling, and for complete positivity one also demands that these be inversely related. Having a renormalizable theory of interacting classical and quantum fields would by itself be an interesting result. On the contrary, if it was found that CQ field theories are not renormalizable, and only valid as effective theories when a physical cutoff is imposed, this would have important consequences for theories which treat the gravitational field as being fundamentally classical.

Implications for classical-quantum gravity. Finally, it would be interesting to explore further the consequences of the path integral in understanding theories where the gravitational field is treated classically, both as a fundamental theory, but also as an effective theory where non-Markovian effects could be incorporated. In Chapter 8 we show one can construct diffeomorphism invariant theories of CQ gravity and it is would be worthwhile to explore these models further to better understand this type of dynamics and constraints it imposes.

⁵Though it should be said that the most general form of dynamics introduced in [165] is not CP because of the negative definite Lindbladian momentum coupling $\partial_{\mu}\phi^{+}\partial^{\mu}\phi^{-}$ which enters into the master equation. Specifically, when one goes to the master Equation picture the resulting Lindbladian is not CP since the Gaussian momentum integrals are altered due to additional momentum couplings in the Lindbladian. We show this explicitly in Appendix J

Part III

Applications to gravity

Chapter 7

Constraints in classical-quantum gravity

Thus far, we have studied and developed the general formalism of completely-positive, linear, autonomous, classical-quantum dynamics. In Chapter 4, we arrived at the general form of continuous classical-quantum dynamics, while in Chapter's 5 and 6, we found unraveling and path integral formalism for CQ coupling. Armed with this formalism, we now study applications of the classical-quantum dynamics to gravity. The motivations are two-fold. Firstly, the lack of success in constructing a complete theory of quantum gravity valid beyond the Planck scale, combined with the lack of low energy signatures of quantum gravity, means the question of whether or not the gravitational field is quantum is still open for debate.

Many recent proposals have been made to measure low-energy gravitational phenomena that cannot be reproduced classically. Currently, the most promising experiments include those which aim to detect gravitationally induced entanglement in table-top experiments via spin entanglement witnesses [64, 53, 1, 2, 65, 66, 67, 68]. There have also been proposals to measure intrinsically quantum features of gravity without studying entanglement directly [69, 70]. Though undoubtedly exciting, current estimates suggest that the technology required to perform the experiments is decades away. We use the classical-quantum formalism to consider the question from the opposite direction: can we construct a consistent fundamental classicalquantum theory of gravity? Can we find experimental signatures of such theories which can be
used as indirect tests for the quantum nature of gravity? These are precisely the theories that experiments measuring gravitationally induced entanglement would rule out.

One might also expect classical-quantum dynamics to be useful as an effective theory. While the assumptions that go into autonomous CQ dynamics are reasonable for a fundamental theory, none of the assumptions need to hold exactly, as an effective theory. For instance, if one allows for the non-Markovian evolution that generically arises in the study of open quantum systems, we necessarily violate the assumption of autonomous dynamics. Nonetheless, exploring the autonomous CQ dynamics in the gravitational context is worthwhile as a starting point. It may be useful in certain regimes, but more importantly, it can be used to gain insight into the challenges that may arise when attempting to construct a more complete semi-classical description.

In this chapter, we take a first step towards studying the constraints which arise in a full theory of classical space-time coupled to a quantum field. In the later parts of the thesis, the models and assumptions entering this chapter will ultimately be refined to produce more appealing models. In particular, in Chapter 8, we study a path integral approach to coupling classical and quantum fields, which looks more promising than a master equation approach, and in Chapter 9, we consider the Newtonian limit of CQ theories, which leads to a constraint which violates the assumption of purely deterministic classical dynamics which enters into this section. Part of the reason for this is that in this chapter, we study constraints for the jumping master equations with infinite moments, which take a different form to the continuous master equation considered in Chapters 8 and 9; at the time this work was completed, we had not yet discovered the general form of the continuous master equation.

In this chapter, we take a geometrodynamic approach and study gauge symmetries of classical-quantum dynamics using master equations and the ADM formalism discussed in Chapter 3. It is here that other attempts to reconcile quantum theory with gravity have failed. In loop quantum gravity, the constraint algebra does not close off-shell [71, 72]. However, it is hoped that one can find an operator ordering such that it will. String theory is background dependent, although it is hoped that string field theory will resolve this.

To recap, in the ADM formulation of classical gravity [77, 78], the dynamics is generated by a Hamiltonian $H_{ADM} = \int d^3x N(x) \mathcal{H}(x) + N^a(x) \mathcal{H}_a(x)$ containing freely chosen lapse N(x), and shift $N^a(x)$ functions. In order to ensure that the dynamics do not depend on this choice, one must impose $\mathcal{H} \approx 0$, $\mathcal{H}_a \approx 0$, called the Hamiltonian and momentum constraint, respectively. When classical matter is included, one finds the constraints $\mathcal{H} + \mathcal{H}_m \approx 0$, $\mathcal{H}_a + \mathcal{H}_{m,a} \approx 0$.

We wish to study the regime where the matter degrees are treated quantum, but back-react on a classical space-time undergoing the combined evolution according to an autonomous CQ master equation. To do so, we must understand the analogous constraints in the combined classical-quantum case, which we study in this chapter. In particular, we study a class of classical-quantum theories of gravity that retain some desirable properties, namely local time reparameterization invariance and spatial diffeomorphism invariance. In such models, by applying a classical-quantum version of the Dirac argument, we can derive a set of constraints that must be satisfied for the theory to be independent of the choice of lapse and shift. In a purely classical theory, one can restrict phase space variables to lie in the constraint surface. In a quantum theory, one can impose the constraints as projectors onto a subspace of the Hilbert space. For a theory that combines classical and quantum degrees of freedom, a different method is required, especially since it is not, as far as we know, easily derived from an action principle - though we discuss progress towards this in the Chapter 8. We find that independence of the choice of lapse and shift requires "commutation" (in a certain sense) of the spatial and temporal parts of the equations of motion. For this commutation to hold, one finds a condition on the CQ-state analogous to the momentum constraint. Conservation of this condition in time leads to the Hamiltonian constraint. This methodology is the central result of the chapter.

We study these constraints explicitly in a theory of a quantum scalar field coupled to classical gravity. Although we are unable to construct a realization of the theory which satisfies the constraint conditions, at least without a further restriction on the choice of lapse and shift, we expect that the general procedure for deriving the constraints will enable further study CQ theories of gravity in a concrete setting. This chapter studies constraints for the jumping master equations with infinite moments. However, we emphasize that the methodology here should apply to the case of continuous master equations as well.

This chapter is based on the paper [3], which is work done in collaboration with Jonathan Oppenheim.

7.1 Hybrid dynamics with a Hamiltonian limit

In this chapter, we will consider the set of autonomous jumping CQ master equations with a Hamiltonian limit. Recall, the general form of master equation takes the form of Equation (2.94)

$$\frac{\partial \varrho(z,t)}{\partial t} = \int dz' \ W^{\mu\nu}(z|z',t) L_{\mu}\varrho(z') L_{\nu}^{\dagger} - \frac{1}{2} W^{\mu\nu}(z,t) \{L_{\nu}^{\dagger}L_{\mu},\varrho\}_{+}, \tag{7.1}$$

where the bracket $\{,\}_+$ is quantum anti-commutator, which should be distinguished from the classical Poisson bracket, which we write as $\{,\},$ and W(z) is defined as

$$W^{\mu\nu}(z,t) = \int dz' W^{\mu\nu}(z'|z,t).$$
 (7.2)

In Equation (7.1), the Lindblad operators L_{μ} are arbitrary. One can easily verify that (7.1) is trace-preserving, and together with the condition that

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \begin{bmatrix} \delta(z,z') + \delta t W^{00}(z|z',t) & \delta t W^{0\beta}(z|z',t) \\ \delta t W^{\alpha 0}(z|z',t) & \delta t W^{\alpha \beta}(z|z',t) \end{bmatrix} + O(\delta t^2)$$
(7.3)

be a positive matrix in μ, ν for all z, z', guarantees that the dynamics is completely positive. To isolate the purely classical degrees of freedom, we choose to decompose Equation (7.1) in terms of the operators $L_{\mu} = (\mathbb{I}, L_{\alpha})$, where L_{α} is an arbitrary Lindblad operator. In this case, the master equation can be written in the form

$$\frac{\partial \varrho(z,t)}{\partial t} = \int dz' \ W^{00}(z|z',t)\varrho(z') - \frac{1}{2}W^{00}(z,t)\varrho(z),
+ \int dz' \ W^{\alpha\beta}(z|z',t)L_{\alpha}\varrho(z')L_{\beta}^{\dagger} - \frac{1}{2}W^{\alpha\beta}(z,t)\{L_{\beta}^{\dagger}L_{\alpha},\varrho\}_{+}.$$
(7.4)

While the master equation (7.1) is completely general, we wish to restrict to dynamics which becomes approximately Hamiltonian in the classical limit. This form of dynamics was introduced in [28]. We shall review the formalism since we use it to construct CQ theories of gravity, particularly those which reproduce the Hamiltonian formulation of Einstein's gravity once we also take the classical limit of the quantum system.

We take the classical degrees of freedom to live in a phase space $\Gamma = (\mathcal{M}, \omega)$, where ω is the symplectic form. We further assume the pure classical evolution to be generated by a classical

Hamiltonian H_c . While we believe one may need to consider dynamics where the purely classical dynamics is also stochastic (see Chapters 8 and 9), we here consider the restrictive and simpler case where it is deterministic. In this case, we can perform a moment expansion of Equation (7.4) and write it as

$$\frac{\partial \varrho(z,t)}{\partial t} = \{H_c, \varrho(z,t)\} - i[H(z), \varrho(z)] + D_0^{\alpha\beta}(z)L_\alpha\varrho(z)L_\beta^{\dagger} - \frac{1}{2}D_0^{\alpha\beta}\{L_\beta^{\dagger}L_\alpha, \varrho(z)\}_+ \\
+ \sum_{\mu\nu\neq00}\sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1}\dots\partial z_{i_n}}\right) \left(D_{n,i_1\dots i_n}^{\alpha\beta}(z)L_\alpha\varrho(z,t)L_\beta^{\dagger}\right).$$
(7.5)

Now, we can define a phase space Hamiltonian vector field via $X_h^{\alpha\beta,i} = (\omega^{-1})^{ij} d_j h^{\alpha\beta}$. Here, $h^{\alpha\beta}(z)$ is some phase space functional, and d_i is the exterior derivative on the phase space. Here, the form of $h^{\alpha\beta}(z)$ is motivated by wanting to reproduce an interacting Hamiltonian of the form

$$H_I(z) = h^{\alpha\beta} L^{\dagger}_{\beta} L_{\alpha}. \tag{7.6}$$

By picking the interaction term $W^{\alpha\beta}(z|z')$ in (7.4) to be such that the vector of first moments takes the form $D_{1i}^{\alpha\beta}(z) = X_h^{\alpha\beta,j}$, the interacting part of the dynamics in (7.5) becomes

$$D_{0}^{\alpha\beta}(z)L_{\alpha}\varrho(z,t)L_{\beta}^{\dagger} - \frac{1}{2}D_{0}^{\alpha\beta}(z)\left\{L_{\beta}^{\dagger}L_{\alpha},\varrho(z,t)\right\}_{+} + \left\{h^{\alpha\beta},L_{\alpha}\varrho(z,t)L_{\beta}^{\dagger}\right\} + \frac{\partial^{2}}{\partial z_{i_{1}}\partial z_{i_{2}}}(D_{2,i_{1}i_{2}}^{\alpha\beta}L_{\alpha}\varrho(z,t)L_{\beta}^{\dagger}) + \dots$$

$$(7.7)$$

We see that defining $H_I(z) = h^{\alpha\beta}(z)L^{\dagger}_{\beta}L_{\alpha}$ and taking the trace over the quantum system gives us an effective classical equation of motion

$$\frac{\partial p(z,t)}{\partial t} = \{H_c, \varrho(z,t)\} + \operatorname{Tr}\left[\{H_I(z), \varrho(z)\}\right] + \dots , \qquad (7.8)$$

where the ... denote the higher order terms in the moment expansion. In order to reproduce Hamiltonian dynamics, we imagine the scenario in which the higher order moments in (7.8) are suppressed by some order parameter [28, 10], such as for the example given in Section 2.3.3. Recall, this example was generated by a diagonal Hamiltonian $H_I(z) = h^{\alpha}(z)L^{\dagger}_{\alpha}L_{\alpha}$

$$\frac{\partial \varrho(z,t)}{\partial t} = \{H_c(z), \varrho(z,t)\} - i [H_I(z), \varrho(z,t)] \\
+ \frac{1}{\tau} \sum_{\alpha} \left(e^{\tau \{h^{\alpha}(z), \cdot\}} L_{\alpha} \varrho(z,t) L_{\alpha}^{\dagger} - \frac{1}{2} \left\{ L_{\alpha}^{\dagger} L_{\alpha}, \varrho(z,t) \right\}_{+} \right),$$
(7.9)

with τ an order parameter suppressing higher-order contributions to the dynamics.

For clarity, we can compare the Hamiltonian limit of CQ dynamics to the case where we have two classical systems, (z_1, z_2) , which interact via an interaction Hamiltonian H_I . If the dynamics of the total system are given by

$$\frac{\partial p(z_1, z_2, t)}{\partial t} = \{H_1, \rho\} + \{H_2, \rho\} + \{H_I, \rho\}$$
(7.10)

then integrating out the second system, and defining $\bar{\rho}(z_1) = \int dz_2 \rho(z_1, z_2)$, we get an effective equation of motion

$$\frac{\partial \bar{\rho}(z_1)}{\partial t} = \{H_1, \bar{\rho}(z_1)\} + \int \mathrm{d}z_2 \{H_I, \rho(z_1, z_2)\},\tag{7.11}$$

which justifies (7.8) as the appropriate classical limit to reproduce Hamiltonian dynamics. It also ensures the classical degrees of freedom undergo Hamiltonian evolution on average – a type of Eherenfest theorem for CQ dynamics. There are some ambiguities in the construction; for example, there is no unique decomposition of the Hamiltonian into Lindblad operators L_{α} , but this shall not be relevant to the discussion in this chapter.

A natural question, then, is to ask whether or not this formalism can be used to construct a consistent theory of CQ gravity, where the gravitational field is considered classical, while the matter fields are considered quantum. In particular, the CQ theory should reproduce general relativity in the classical limit of the quantum system. The subtleties arise since gravity is a gauge theory; not only do the degrees of freedom undergo dynamics generated by the pure gravity Hamiltonian but they must also live on the constraint surface. Hence, although we can use the formalism introduced in this section to construct CQ dynamics which reproduces that of gravity, we must find a way of studying the constraints in CQ theories.

7.2 CQ theory which reproduces Einstein gravity

This section reviews how one can use the CQ formalism introduced to construct CQ models of gravity, which reproduce Einstein's gravity [28]. We here restrict ourselves to the case where the CQ equation has the purely classical evolution generated by the ADM Hamiltonian, the pure quantum evolution generated by the Klein-Gordon (KG) Hamiltonian, and the interaction term a CQ dynamics, whose first moment is such that it approximates the Hamiltonian formulation of gravity in the classical limit.

7.2.1 Liouville formulation of classical gravity

Recall from Chapter 3 that in classical gravity, the relevant configuration space is given by the space of Riemannian metrics on a surface Σ and their conjugate momenta. Elements of the phase space are denoted (γ_{ab}, π^{ab}) and are taken to satisfy the canonical Poisson bracket relations

$$\{\gamma_{ab}(x), \pi^{cd}(y)\} = \frac{1}{2} (\delta^c_a \delta^d_b + \delta^d_a \delta^c_b) \delta(x, y)$$
(7.12)

The dynamics are generated by the ADM Hamiltonian [77, 78]

$$H_{ADM}[N,\vec{N}] = \int d^3x N^{\mu} \mathcal{H}_{\mu} = \int d^3x (N\mathcal{H} + N^a \mathcal{H}_a) = H[N] + H[\vec{N}]$$
(7.13)

where

$$\mathcal{H} = (16\pi G)\pi^{ab}G_{abcd}\pi^{cd} - \frac{1}{16\pi G}\gamma^{1/2}R, \quad \mathcal{H}_a = -2\gamma_{ac}D_b\pi^{cb}$$
(7.14)

 G_{abcd} is the deWitt metric defined as $G_{abcd} = \frac{1}{2\sqrt{g}}(\gamma_{ac}\gamma_{bd} + \gamma_{ad}\gamma_{bc} - \gamma_{ab}\gamma_{cd})$ and D_a the covariant derivative with respect to the metric γ_{ab} on Σ . In this chapter, we use units in which c = 1 to simplify our formulas.

The lapse function N and shift vector N^a appearing in Equation (7.13) are arbitrary functions of (t, x). They arise when performing the 3+1 split of space-time and represent the gauge degrees of freedom associated with picking a foliation of space-time. They are non-dynamical since $P_N, P_{N^a} = 0$, and as a result, the Hamiltonian formulation of GR is a constrained theory. Asking that the constraints $P_N, P_{N^a} = 0$ be preserved in time leads to the Hamiltonian and Momentum constraints, $\mathcal{H} = \mathcal{H}_a = 0$. Conservation of these constraints is ensured via the hypersurface deformation algebra [108]

$$\{H[N], H[M])\} = H[\vec{R}]$$

$$\{H[\vec{M}], H[N]\} = H [L_{\vec{M}}N]$$

$$\{H[\vec{N}], H[\vec{M}]\} = H[L_{\vec{N}}\vec{M}]$$

$$(7.15)$$

where $R^a := \gamma^{ab} (ND_bM - MD_bN)$ and L is the Lie derivative on Σ .

Since we study gravitational CQ master equations, writing the dynamics of pure GR in a Liouville formulation is useful. In particular, a phase space distribution $\rho(\gamma, \pi)$ will evolve under the dynamics as

$$\frac{\partial \rho}{\partial t} = \{H_{ADM}, \rho\} \tag{7.16}$$

subject to the constraints $\mathcal{H}\rho = \mathcal{H}_a\rho = 0$; that is ρ must have support only on the constraint surface. There are a few things to be wary of when using a Liouville formalism. Firstly, we must remember that ρ is a distribution, so its action is only defined once smeared over phase space test functions. Furthermore, in the Liouville picture, we solve for $\rho(\gamma, \pi, t)$ given a choice of lapse and shift vector N^{μ} . The solution can then be interpreted as describing a probability density over trajectories $(\gamma_{ab}(t), \pi^{cd}(t))$, which, using the ADM split.

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -N^2 dt^2 + \gamma_{ij}(dx^i + N^i dt)(dx^j + N^j dt), \qquad (7.17)$$

we can use to define a probability distribution over 4-geometries $g_{\mu\nu}$. Since the trajectories are deterministic, we can imagine starting in a state of certainty $\rho \sim \delta(\bar{\gamma}, \gamma)\delta(\bar{\pi}, \pi)$, we can imagine picking a different lapse and shift for each point in phase space $N(t, \gamma, \pi), N^a(t, \gamma, \pi)$. Here, we take the weakest thing we can ask for, which is that the lapse and shift only depend on x, t, and we do not consider the more general case where they can be phase-space dependent.

We now consider coupling matter to gravity; for ease of calculation, we will study scalar fields coupled to gravity. A classical field minimally coupled to gravity will have a Hamiltonian of the form

$$H_T[N,\vec{N}] = H_{ADM}[N,\vec{N}] + H_m[N,\vec{N}] = \int d^3x N(\mathcal{H} + \mathcal{H}_m) + N^a(\mathcal{H}_a + \mathcal{H}_{m,a}),$$
(7.18)

where H_m is the Hamiltonian of the matter field. In the presence of matter, the constraint surface takes the form $\mathcal{H} + \mathcal{H}_m = 0, \mathcal{H} + \mathcal{H}_{m,a} = 0.$

For example, the Hamiltonian of the free scalar field reads

$$H_m[N,\vec{N}] = \int d^3x N(\frac{1}{2}\gamma^{-1/2}\pi^2 + \frac{1}{2}\gamma^{1/2}\gamma^{ij}\partial_i\phi\partial_j\phi + \frac{1}{2}\gamma^{1/2}m^2\phi_\phi^2) + N^i\pi_\phi\partial_i\phi.$$
(7.19)

The Liouville equation for the phase space density $\rho(\gamma, \pi_{\gamma}, \phi, \pi_{\phi}, t)$ then takes the form

$$\frac{\partial \rho}{\partial t} = \{H[N, \vec{N}], \rho\} + \{H_m[N, \vec{N}], \rho\},$$
(7.20)

where ρ must only have support on the constraint surface.

In order to gain parallels to the CQ theory, in particular when looking at the gravitational degrees of freedom alone, it is insightful to integrate out the ϕ , π_{ϕ} degrees of freedom to get an effective equation for the evolution of the gravitational degrees of freedom. Defining $\bar{\rho}(\gamma, \pi_{\gamma}) = \int D\phi D_{\pi_{\phi}} \rho(\gamma, \pi_{\gamma}, \phi, \pi_{\phi})$ then integrating (7.20) over the matter degrees of freedom gives an effective equation of motion

$$\frac{\partial\bar{\rho}(\gamma,\pi_{\gamma},t)}{\partial t} = \{H,\bar{\rho}\} + \int D\phi D_{\pi_{\phi}}\{H_m,\rho\},\tag{7.21}$$

which is the version of Equation (7.8) if the matter were to be treated classically.

7.2.2 CQ theories of gravity

We now construct CQ theories of gravity whose dynamics become approximately that of Einstein's gravity in the classical limit. To be slightly more precise: following [28] and the toy models in [10], the interaction between the classical and quantum degrees of freedom causes the quantum state to change while at the same time causing back-reaction on the classical degrees of in a way which approximates Hamiltonian evolution generated by the ADM Hamiltonian.

To that end, we consider the CQ dynamics of Equation (7.5), where we take the pure classical dynamics to be generated by H_{ADM} , while we take the pure quantum evolution to be generated by the Klein-Gordon Hamiltonian H_m

$$H_m[N,\vec{N}] = \int d^3x N h^{\alpha\beta} L^{\dagger}_{\beta} L_{\alpha} + N^a p^{\alpha\beta}_a L^{\dagger}_{\beta} L_{\alpha} = \int d^3x N(x) \mathcal{H}_m(x) + N^a(x) \mathcal{H}_{m,a}(x), \quad (7.22)$$

which we now take to be a quantum object. Here we have defined

$$h^{\pi\pi} := \frac{1}{2} \gamma^{-1/2}, \quad h^{\phi\phi} = \frac{1}{2} \gamma^{1/2}, \quad h^{ab} = \frac{1}{2} \gamma^{1/2} \gamma^{ab}$$

$$p^{a\pi} = 1/2, \quad p^{\pi a} = 1/2$$

$$L_{\pi}(x) = \pi_{\phi}(x), \quad L_{\phi}(x) = \phi(x), \quad L_{a}(x) = \partial_{a} \phi(x),$$
(7.23)

and the now quantum field operators satisfy the canonical commutation relations

$$[\phi(x), \phi(y)] = 0, \quad [\pi_{\phi}(x), \pi_{\phi}(y)] = 0, \quad [\phi(x), \pi_{\phi}(y)] = i\delta(x, y).$$
(7.24)

The $\delta(x, y)$ is defined as a scalar in x and a scalar density in y. We shall not construct the precise nature of the Hilbert space \mathcal{H} ; ultimately, we only exploit the algebraic properties of

the canonical commutation relations, so the details of the Hilbert space are not of primary importance. However, let us briefly comment on the problem of defining the Hilbert space, an important open problem in the study of classical-quantum gravity.

Even for a free field in a fixed background $g_{\mu\nu}$, infinitely many unitarily in-equivalent representations of the commutation relations on a Hilbert space exist. In general, there is no notion of a preferred state for which a Hilbert space representation can be defined [175, 176]. As such, the modern view takes an algebraic approach when studying quantum fields in curved space. One instead views the algebraic (commutation) relations satisfied by the field observables as fundamental, which are taken to belong to an algebra \mathcal{A} . States ω are then defined as positive linear functionals on \mathcal{A} . In this viewpoint, the algebraic structure is unique, but there are infinitely many *representations* of the algebra on a Hilbert space. In particular, the GNS construction [177, 178] shows that every state ω on the algebra defines a Hilbert space \mathcal{H} , a representation of the algebra on the Hilbert space, and a Hilbert space vector corresponding to ω . Thus, though technically equivalent, the algebraic approach allows one to formulate Quantum field theory (QFT) in curved space in a way that is independent of the representation of the algebra and does not require one to single out a preferred state in the theory.

The classical-quantum case has a similar problem in defining the Hilbert space. It is simple enough to define some preferred Hilbert space \mathcal{H} and take the field operators to act on \mathcal{H} : as an example, we could take the standard Hilbert space of free field in Minkowski space, defined by the existence of a Poincare invariant vacuum state [176]. We could then consider classical-quantum dynamics in this Hilbert space, but this arbitrary choice of Hilbert space is unsatisfactory. As for quantum fields in curved space, we can view classical-quantum dynamics and the commutation relations (7.24) algebraically – for example, by using the Heisenberg representation of classical-quantum dynamics introduced in [28]. What needs to be added in the CQ case is a version of the GNS construction: given a classical-quantum state which enables us to calculate probabilities for observables, we need to determine how to construct a Hilbert space representation for the dynamics uniquely. Since this is potentially a challenging and technical problem, we view the operators acting on the Hilbert space as formal expressions. We leave an algebraic formulation of classical-quantum dynamics as an interesting open problem for future research.¹

Now that we have specified the purely classical and quantum dynamics, all that remains is to specify the classical-quantum interaction term, which describes the back reaction of the quantum system on the classical degrees of freedom. We call a specific choice of the CQ coupling a *realization* of a CQ theory. Firstly, we demand that the first moments of the Kramers-Moyal expansion in (7.5) contain a term

$$\int d^3x N(x) \frac{\delta h^{\alpha\beta}}{\delta\gamma_{cd}} L_{\alpha} \varrho L_{\beta}^{\dagger}$$
(7.25)

in order to yield Einstein's gravity in the classical limit. Consequently, we view Equation (7.25) as a condition on CQ theories of gravity that a sensible realization of CQ dynamics must satisfy. Here the α, β indices run over $\phi, \pi_{\phi}, \nabla_a \phi$. In order to study CQ dynamics in a concrete setting, we shall make some further assumptions about the realizations. Firstly, we assume that the realizations are local so that the CQ interaction is fully specified by the set of transition amplitudes $W^{\alpha\beta}(z|z',x), W^{\alpha\beta}_a(z|z',x)$, representing the generalizations of the quantum matter Hamiltonian density $\{\mathcal{H}_m,\}$ and momentum density $\{\mathcal{H}_{m,a},\}$. Secondly, we shall focus on a natural class of dynamics, namely those with CQ couplings $W^{\alpha\beta}(z|z')$, which are linear in N and N^a .

There is a good reason to do this. From a physical standpoint, the free functions N and N^a represent the local time reparameterization invariance and space-like diffeomorphism invariance of the underlying theory. A natural question is whether we can have a CQ theory of gravity that upholds these symmetries. Furthermore, from a technical point of view, if we were to consider non-linear couplings in the lapse and shift – say we had an N^2 coupling – then the method we use to derive the constraints would lead to a constraint which itself depends on N. Preservation of such a constraint then leads to N becoming dynamical, which generically leads to a gauged fixed theory; this happens, for example, in Horava gravity [179, 180, 181]

¹Note that the problem in defining a Hilbert space representation is compounded relative to the semi-classical case because the space-time metric is now dynamical and background independent: solutions are described by a probability distribution over 4-metrics, where each metric is associated with a quantum state. This is perhaps both a blessing and a curse; having a dynamical space-time makes it harder to find an algebraic formulation of classical-quantum dynamics, but it perhaps provides an interesting intermediate arena to study properties of QFT beyond the semi-classical regime.

(although there are exceptions, where additional secondary constraints fix N up to a global reparameterization invariance, for example in shape dynamics [182, 183, 184]). We leave gauge fixed CQ theories as a possible area for further study.

Assuming that realizations are linear in the lapse and shift, we are led to study dynamics of the form

$$\frac{\partial \varrho(z)}{\partial t} = \{H_{ADM}, \varrho(z)\} - i[H_m, \varrho(z)]
+ \int dz' \int d^3x \Big[(NW^{\alpha\beta}(z|z', x) + N^a W^{\alpha\beta}_a(z|z', x)) L_\alpha(x) \varrho(z') L^{\dagger}_{\beta}(x) \Big]
- \frac{1}{2} \int d^3x [(NW^{\alpha\beta}_0(z) + N^a W^{\alpha\beta}_{0a}(z)) \{L^{\dagger}_{\beta}(x) L_\alpha(x), \varrho(z)\},$$
(7.26)

where the first moments of the realizations satisfy Equation (7.25). Here, z labels points in the phase space of GR, $z = (\gamma_{ab}, \pi^{cd})$, and writing out equation (7.26) in full we obtain

$$\frac{\partial \varrho(\gamma,\pi)}{\partial t} = \{H_{ADM}, \varrho(\gamma,\pi)\} - i[H_m, \varrho(\gamma,\pi)]
+ \int \mathcal{D}\gamma' \mathcal{D}\pi' \int d^3x \Big[(NW^{\alpha\beta}(\gamma,\pi|\gamma',\pi',x) + N^a W^{\alpha\beta}_a(\gamma,\pi|\gamma',\pi',x)) L_\alpha(x) \varrho(\gamma',\pi') L^{\dagger}_{\beta}(x) \Big]
- \frac{1}{2} \int d^3x [(NW^{\alpha\beta}_0(\gamma,\pi) + N^a W^{\alpha\beta}_{0a}(\gamma,\pi)) \{L^{\dagger}_{\beta}(x) L_\alpha(x), \varrho(\gamma,\pi)\}.$$
(7.27)

In Equation (7.27), the integral over $\int \mathcal{D}\gamma \mathcal{D}\pi$ is to be treated as a formal integral over all configurations of Riemmanian 3 metrics γ_{ab} and their conjugate momenta π^{ab} , and we do not make any attempt to justify its existence rigorously. To simplify notation, we will often suppress the phase space integrals and write, for example, $\int dz' W^{\alpha\beta}(z|z',x)\varrho(z') = W^{\alpha\beta}(x)(\varrho)$ indicating that $W^{\alpha\beta}(x)$ is a differential (or in other cases, a CQ CP map) acting on ϱ . Furthermore, we will often write the master equation in a more compact form as

$$\frac{\partial \varrho(\gamma, \pi)}{\partial t} = \int d^3 x N \mathcal{L}(\varrho) + N^a \mathcal{L}_a(\varrho), \qquad (7.28)$$

implicitly defining

$$\mathcal{L}(x)(\varrho) = \{\mathcal{H}(x), \varrho\} - i[\mathcal{H}_m(x), \varrho] + W^{\alpha\beta}(x)L_{\alpha}(x)\varrho L^{\dagger}_{\beta}(x) - \frac{1}{2}W^{\alpha\beta}_0(z, x)\{L^{\dagger}_{\beta}(x)L_{\alpha}(x), \varrho\}$$
(7.29)

$$\mathcal{L}_{a}(x)(\varrho) = \{\mathcal{H}_{a}(x), \varrho\} - i[\mathcal{H}_{m,a}(x), \varrho] + W_{a}^{\alpha\beta}(x)L_{\alpha}(x)\varrho L_{\beta}^{\dagger}(x) - \frac{1}{2}W_{0a}^{\alpha\beta}(z,x)\{L_{\beta}^{\dagger}(x)L_{\alpha}(x), \varrho\}.$$
(7.30)

Equation (7.26) gives us a class of hybrid theories that give the dynamics of GR in their classical limit. Of course, things are more complicated than this since GR is a constrained system; not only do the degrees of freedom undergo dynamics generated by the ADM Hamiltonian, but they must also lie on the constraint surface. Hence, we expect constraints to enter any CQ theory of gravity that gives all the components of Einstein's equations.

In the usual picture, the constraints come directly from an action principle – which we do not have here, though we make progress towards this in Chapter 8. How, then, can we impose constraints on the CQ theory? It is known in classical GR that it is possible to derive the constraints by exploiting the algebroid nature of the hypersurface deformation algebra [108] (see also [106, 185, 186, 187, 188, 189, 107, 190], for discussion of the deformation algebra in other contexts). Similar consistency conditions can be found by considering the Dirac argument. If the equations of motion contain arbitrary functions of time, such as the lapse and shift, then to retain predictability, two solutions related by a different choice of the arbitrary function must be gauge equivalent [106]. The subsequent section shows how it leads to a constraint surface in GR. We then apply this to the CQ theory and use it to derive generalized Hamiltonian and momentum constraints in CQ theories of gravity.

7.3 Deriving constraints from gauge conditions

In this section, we briefly review the Dirac argument [106] for gauge theories. By applying a Dirac-like argument to GR, we then show that one can arrive at a set of consistency conditions that lead to constraints on the theory. The mathematics is similar to that used in HKT [108], which exploits the algebroid nature of GR to arrive at the constraint surface. We then extend this method to the case of CQ master equations which provides a methodology to derive a set of constraints for CQ theories that are linear in N, N^a , which is a central result of this chapter.

7.3.1 Dirac argument for Hamiltonian systems

Hamiltonian gauge theories generically have actions of the form [106, 191]

$$I[q^{i}, p_{j}, \lambda^{a}] = \int dt (p_{i} \dot{q}^{i} - H_{0}(p_{i}, q^{j}) - \lambda^{a} \phi_{a}(p_{i}, q_{i})), \qquad (7.31)$$

which define equations of motion and constraints

$$\dot{q}^{i} = \frac{\partial H_{0}}{\partial p_{i}} - \lambda^{a} \frac{\partial \phi_{a}}{\partial p_{i}}, \quad \dot{p}_{i} = -\frac{\partial H_{0}}{\partial q_{i}} + \lambda^{a} \frac{\partial \phi_{a}}{\partial q_{i}}, \quad \phi_{a} = 0.$$
(7.32)

Preservation of the constraints requires that

$$\frac{d\phi_a}{dt} = [\phi_a, H_0] - [\phi_a, \phi_b]\lambda^b \approx 0.$$
(7.33)

where \approx means the equation must be weakly zero, meaning it must vanish on the constraint surface [106]. Gauge theories are characterized by having $[\phi_a, \phi_b] \approx 0$, in which case the constraints are said to be *first class*. In this case, the $\lambda^a(t)$ remains undetermined, and the equations of motion contain arbitrary functions of time, as will their solutions. Gauge theories then generically have Hamiltonian's of the form

$$H_T = H_0 + \lambda^a(t)\phi_a,\tag{7.34}$$

where the $\lambda^{a}(t)$ are arbitrary and they multiply constraints. Usually, it is said that first-class constraints generate equal time gauge transformations. We can run the Dirac argument [106] as follows to see why this is. Suppose we have some initial data (q^{i}, p_{j}) at t = 0. Since the $\lambda(t)$ is undetermined, we can use either $\lambda(t)$ or $\lambda'(t)$ to solve the equations of motion. Let $f(q^{i}, p_{j})$ be any functional over the phase space. Then at time $t = \epsilon$

$$f(\epsilon) = f(0) + \epsilon \{ f(0), H_0 \} + \epsilon \lambda^a \{ f(0), \phi_a \}$$
(7.35)

$$f'(\epsilon) = f(0) + \epsilon \{ f(0), H_0 \} + \epsilon \lambda^{a'} \{ f(0), \phi_a \}.$$
(7.36)

Consequently, given the same initial data set, we get two equally valid descriptions. We must identify these descriptions as physically equivalent to retain predictability. The difference between the two solutions is

$$\delta f(\epsilon) = \epsilon (\lambda' - \lambda) \{ f(0), \phi_a \}, \tag{7.37}$$

and hence $\{\phi_a\}$ generates equal time gauge transformations. As discussed in Chapter 3, there is a subtle difference between first-class constraints generating gauge transformations on initial data - or data on a time slice - which, once the dynamics are solved, give classes of physically equivalent solutions, and gauge transformations which act on the space of solutions directly. The equal time gauge transformations are to be understood in the former sense – as gauge transformations between initial data configurations [109].

Given that constraints generate gauge transformations, it should then be true that the generators form an algebra (which follows from linearity and demanding that the gauge transformations be transitive). For first-class constraints, we have

$$\{\phi_a, \{\phi_b, f\}\} - \{\phi_b, \{\phi_a, f, \}\} = \{C^c_{ab}\phi_c, f\} = \{C^c_{ab}, f\}\phi_c + C^c_{ab}\{\phi_c, f\} \approx C^c_{ab}\{\phi_c, f\}, \quad (7.38)$$

so that the gauge generators close as an algebra on the constraint surface. Note that this holds even for the case of GR, where the C_{bc}^{a} are structure functions that depend on the phase space.

7.3.2 Deriving the constraint surface of GR from the Dirac argument

We now show how using the Dirac argument for classical gravity leads to constraints. Suppose we have a state $\rho(\gamma, \pi)$ which evolves according to the ADM equations of motion

$$\frac{\partial \rho}{\partial t} = \int d^3 x \{ N \mathcal{H} + N^a \mathcal{H}_a, \rho \}.$$
(7.39)

We will not take N(t, x), $N^a(t, x)$ to have a functional dependence on γ , π , the weakest condition we can ask for. We will have to be careful since the states $\rho(\gamma, \pi)$ are defined only in the distributional sense, so things like Poisson brackets are only defined by their action on test functions. We denote the smeared distributions as

$$\langle A, \rho \rangle = \int D\gamma D\pi \rho(\gamma, \pi, t) A(\gamma, \pi, t).$$
 (7.40)

Since the lapse and shift functions are arbitrary $\{\mathcal{H},\}$ and $\{\mathcal{H}_a,\}$ generate equal time gauge transformations. As a consequence

$$\langle A, \rho \rangle \sim \langle A, \rho \rangle + \epsilon \langle \{A, \mathcal{H}_a(x)\}, \rho \rangle.$$
 (7.41)

Since all observables, $A(\gamma, \pi, t)$ must be independent of the choice of gauge, they must satisfy $\langle \{A(\gamma, \pi, t), \mathcal{H}_a(x)\}, \rho \rangle = 0$. In particular, picking $\rho = \delta(\bar{\gamma}, \gamma)\delta(\bar{\pi}, \pi)$, we see observables must

satisfy

$$\{A, \mathcal{H}_a(x)\} \approx 0. \tag{7.42}$$

Note, we are not saying that time evolution is a gauge transformation and observables are frozen. The gauge generators take one initial configuration to another; both dynamically evolved in time. Furthermore, allowing for the case where the observables contain an explicit time dependence – which usually occurs once one has gauge fixed [192] – means they evolve in time even though the Poisson bracket with the Hamiltonian vanishes.

Demanding that the gauge transformations close as an algebra, which follows from asking that the equivalence relation be linear and transitive, we must have

$$\{H[N], \{H[M], \rho\}\} - \{H[M], \{H[N], \rho\}\} \approx \text{Gauge}(\rho).$$
(7.43)

When Equation (7.43) is smeared over observables, we see that the left-hand side must vanish since we know that observables Poisson commute with the constraints.

We now use the Jacobi identity and the hypersurface deformation algebra defined in Equation (7.15) to write Equation (7.43) as

$$\{\{H[N], H[M]\}, \rho\} = \int d^3x \{\gamma^{ab} \delta N_b \mathcal{H}_a(x), \rho\} \approx \text{Gauge}(\rho), \tag{7.44}$$

where we have defined $\delta N_b = N \partial_b M - M \partial_b N$. Smearing Equation (7.44) over a phase space smearing function A, one finds the left hand side of Equation (7.44) is

$$\int D\gamma D\pi \int d^3x \{\gamma^{ab} \delta N_b \mathcal{H}_a(x), \rho\} A = \int d^3x \langle \delta N_b \{A, \gamma^{ab}\}, \mathcal{H}_a \rho \rangle + \langle \{A, \mathcal{H}_a\} \delta N_b \gamma^{ab}, \rho \rangle.$$
(7.45)

When A is an observable, the final term in Equation (7.45) vanishes as it acts as a gauge transformation, and so we see that for the algebra of gauge transformations to close

$$\langle \{A, \gamma^{ab}\}, \mathcal{H}_a \rho \rangle \approx 0.$$
 (7.46)

In other words, we require $\langle \{A, \gamma^{ab}\}, \mathcal{H}_a \rho \rangle = 0$ whenever A is a phase space observable. This condition is satisfied by the constraints of GR since $\langle A, \mathcal{H}_a \rho \rangle = 0$ for all phase space functions A when on the constraint surface. Preservation of this condition is guaranteed by the Hamiltonian constraint $\langle A, \mathcal{H}\rho \rangle = 0$, as can be seen by applying the evolution equation on it and using the Jacobi identity. However, the consistency condition in Equation (7.46) is weaker than asking

the constraints hold. As a trivial example, we can consider a theory in which the only allowed observable is A = I, then Equation (7.46) holds for any ρ on the phase space.

In general, we shall ask: is there a set of sensible observables and states for which consistency conditions – such as that in equation (7.46) – hold and which are preserved in time? This is similar to the *on-shell closure* studied in loop quantum gravity [188]. For dynamics generated by the ADM Hamiltonian, it seems there is no non-trivial set of states/observables other than that of GR for which Equation (7.46) holds. The problem is that we can keep applying gauge transformations to generate more and more independent constraints. For example, suppose that Equation (7.46) is satisfied, then applying a spatial gauge transformation, we also require that

$$\langle \{\{\gamma^{ab}, A\}, \mathcal{H}_c\}, \mathcal{H}_b \rho \rangle \approx 0,$$
(7.47)

and similarly for the gauge transformation generated by the Hamiltonian. We can seemingly continue to do this indefinitely. We return to this subtlety for the CQ theory at the end of Section (7.5). For now, we will take the consistency conditions in a weak form asking whether or not there exists a sensible set of observables and states for which they hold.

7.3.3 Dirac argument for CQ master equations

The Dirac argument extends naturally to the CQ theory. In particular, suppose we are given a CQ master equation of the form

$$\frac{\partial \varrho}{\partial t} = \mathcal{L}_0(\varrho) + \lambda^a(t)\mathcal{L}_a(\varrho), \qquad (7.48)$$

where the $\lambda^a(t)$ are undetermined functions of time. We can run the Dirac argument on ρ . Without loss of generality, consider having an initial CQ state $\rho(z,0)$ at t = 0. The master equation depends on some arbitrary functions of time $\lambda^a(t)$. Consider picking two different functions $\lambda^a, \lambda^{a'}$. Then at time $t = \epsilon$ the solutions are

$$\varrho(\epsilon) = \varrho(0) + \epsilon \mathcal{L}_0(\varrho(0)) + \epsilon \lambda^a \mathcal{L}_a(\varrho)$$
(7.49)

$$\varrho'(\epsilon) = \varrho(0) + \epsilon \mathcal{L}_0(\varrho(0)) + \epsilon \lambda^{a'} \mathcal{L}_a(\varrho(0)).$$
(7.50)

The difference between the two solutions is

$$\delta\varrho(\epsilon) = \epsilon(\lambda^{a\prime} - \lambda^a)\mathcal{L}_a(\varrho(0)), \qquad (7.51)$$

from which we conclude that $\mathcal{L}_a(\varrho)$ is generating equal time gauge transformations. We should then ask that the generators close as an algebra

$$\mathcal{L}_a(\mathcal{L}_b(\varrho)) - \mathcal{L}_b(\mathcal{L}_a(\varrho)) \approx C_{ab}^c \mathcal{L}_c(\varrho).$$
(7.52)

Since the classical part of the CQ state may only be defined in a distributional sense, we can re-state equation (7.52) as asking that

$$\langle A, \mathcal{L}_a(\mathcal{L}_b(\varrho)) - \mathcal{L}_b(\mathcal{L}_a(\varrho)) \rangle \approx 0$$
 (7.53)

whenever A is an observable. Similar to the case of classical GR in the previous subsection, we will see that in CQ theories of gravity, the generators only close as an algebra if we are on a constraint surface. We will ask that the constraints are satisfied in the weakest sense. That is, we will look for a non-trivial set of states and observables for which the consistency condition is satisfied and preserved in time. Using this method, we can derive a generalized momentum constraint for the CQ theory; asking that this constraint is preserved will lead to a generalized Hamiltonian constraint.

7.4 Deriving constraints in post-quantum theories of Gravity

In this section, we will use the Dirac argument to derive consistency conditions for CQ theories of gravity. We expect the methods used here to extend to all CQ theories of the form introduced in Section 7.2, namely those linear in the lapse and shift. We briefly outline a general procedure for generating consistency conditions and constraints in such theories. We spend the remainder of the chapter performing the explicit calculations for a specific class of CQ theories of a quantum scalar field coupled to gravitational degrees of freedom.

7.4.1 A general method of arriving at constraints

We consider CQ dynamics linear in the lapse and shift, reproducing Hamiltonian evolution in the classical limit. In particular, we will consider we consider the simplified theory of Equation (7.26)

$$\frac{\partial \varrho(\gamma, \pi)}{\partial t} = \int d^3 x N \mathcal{L}(\varrho) + N^a \mathcal{L}_a(\varrho) = \mathcal{L}[N] + \mathcal{L}[\vec{N}], \qquad (7.54)$$

where

$$\mathcal{L}(x)(\varrho) = \{\mathcal{H}(x), \varrho\} - i[\mathcal{H}_m(x), \varrho] + W^{\alpha\beta}(x)L_{\alpha}(x)\varrho L^{\dagger}_{\beta}(x) - \frac{1}{2}W^{\alpha\beta}_0(x)\{L^{\dagger}_{\beta}(x)L_{\alpha}(x), \varrho\}$$
(7.55)

$$\mathcal{L}_{a}(x)(\varrho) = \{\mathcal{H}_{a}(x), \varrho\} - i[\mathcal{H}_{m,a}(x), \varrho] + W_{a}^{\alpha\beta}(x)L_{\alpha}(x)\varrho L_{\beta}^{\dagger}(x) - \frac{1}{2}W_{0a}^{\alpha\beta}(x)\{L_{\beta}^{\dagger}(x)L_{\alpha}(x), \varrho\}.$$
(7.56)

The Dirac argument tells us that data, or CQ states, related by different choices of N, N^a must lead to gauge equivalent solutions when the dynamics are solved, i.e., that $\mathcal{L}(x), \mathcal{L}_a(x)$ generate equal time gauge transformations.

As a consequence, denoting the dual operators (under $\int dz$) to \mathcal{L} and \mathcal{L}_a as $\mathcal{L}^*, \mathcal{L}_a^*$ respectively, this implies that

$$\int Dg D\pi \mathcal{L}^*(A)\varrho \approx 0, \quad \int Dg D\pi \mathcal{L}^*_a(A)\varrho \approx 0 \tag{7.57}$$

for all observables A^2 . In Section 7.5, we will only consider realizations that diffuse in the conjugate momenta π^{ab} . Hence, we expect that any constraints will be solved by functions of the form $\rho(\gamma, \pi) = \delta(\gamma, \bar{\gamma})\rho_c(\gamma, \pi)$ since the commutation relations will never lead to γ_{ab} diffusive terms.³ As a consequence, we expect that

$$\int D\pi \mathcal{L}^*(A)(\bar{\gamma},\pi)\varrho_c(\bar{\gamma},\pi) \approx 0, \quad \int D\pi \mathcal{L}^*_a(A)(\bar{\gamma},\pi)\varrho_c(\bar{\gamma},\pi) \approx 0.$$
(7.58)

Multiplying Equation (7.58) by an arbitrary function of the metric $f(\bar{\gamma})$, putting back the delta $\delta(\gamma, \bar{\gamma})$, and integrating over γ , we expect more generally that

$$\int D\gamma D\pi \mathcal{L}^*(A) f(\gamma) \varrho \approx 0, \quad \int D\gamma D\pi \mathcal{L}^*_a(A) f(\gamma) \varrho \approx 0, \tag{7.59}$$

so that the gauge transformation can have a metric dependence on the lapse and shift. The requirement that the gauge transformations are transitive leads to consideration of the algebra of generators

$$[\mathcal{L}_a(x), \mathcal{L}_b(y)], \quad [\mathcal{L}_a(x), \mathcal{L}(y)], \quad [\mathcal{L}(x), \mathcal{L}(y)], \tag{7.60}$$

²Here, the dual operators are defined via integration by parts $\int Dg D\pi A \mathcal{L}(\varrho) = \int Dg D\pi \mathcal{L}^*(A) \varrho$.

³This assumption turns out to be violated by the improved class of theories we discuss in Chapters 8 and 9.

where once again we remind the reader that $\mathcal{L}, \mathcal{L}_a$ are classical-quantum operators acting on ϱ , in the sense of Equation (7.54), where the couplings $W^{\alpha\beta}(x)$ are to be interpreted as kernels $W^{\alpha\beta}(x)(\varrho) = \int dz' W^{\alpha\beta}(z|z',x)\varrho(z')$, and similarly for $W^{\alpha\beta}_a(x)$. For example, $[\mathcal{L}_a(x), \mathcal{L}_b(y)](\varrho) = \mathcal{L}_a(x)(\mathcal{L}_b(y)(\varrho)) - \mathcal{L}_b(x)(\mathcal{L}_a(y)(\varrho))$.

We require that the algebra closes; the commutator of two gauge transformations is weakly another gauge transformation, which vanishes when smeared over an observable $A \in O_{obs}$. Similarly to GR, we will see that demanding this leads to a notion of a constraint surface. We will often deal with the spatially smeared versions of these transformations and denote $\mathcal{L}[N] = \int d^3x N(x) \mathcal{L}(x)$ and $\mathcal{L}[\vec{N}] = \int d^3x N^a(x) \mathcal{L}_a(x)$.

7.5 A CQ theory of gravity coupled to a scalar field

Now that we have a general method of deriving constraints in CQ theories, we spend the remainder of the chapter exploring the consistency conditions for a quantum scalar field interacting with a classical gravitational field. We make several simplifying assumptions so that the theory considered here is a special case of the one derived in [28]. We outline all of our assumptions as follows

Assumption 1. We take the evolution to be linear in the lapse and shift and consider the choice of N and \vec{N} to be pure gauge.

Assumption 2. We will take the $W_a^{\alpha\beta}(x) = 0$ so that $\mathcal{L}_a(\varrho) = \{\mathcal{H}_a(x), \varrho\} - i[\mathcal{H}_{m,a}(x), \varrho]$ generates spatial diffeomorphisms and the theory will be spatially diffeomorphism invariant.

Assumption 3. We take the CQ couplings $W^{\alpha\beta}(z|z')$ to have the same Lindblad structure as the Hamiltonian so that the interaction terms can be written

$$W^{\alpha\beta}(x)L_{\alpha}(x)\varrho L^{\dagger}_{\beta}(x) = W^{\phi\phi}(x)\phi(x)\varrho\phi(x) + W^{\pi\pi}(x)\pi_{\phi}(x)\varrho\pi_{\phi}(x)$$

+ $W^{ab}(x)\partial_{a}\phi(x)\varrho\partial_{b}\phi(x).$ (7.61)

In particular, $W^{\alpha\beta}(x)$ is taken to be local in x, while the more general case could be non-local and include a regulator $W^{\alpha\beta}(x-y)L_{\alpha}(x)\varrho L_{\beta}^{\dagger}(y)$. **Assumption 4.** We take the CQ coupling with the scalar field, in analogy with classical gravity, to have a functional dependence on the spatial metric only; $W^{\alpha\beta}[\gamma](x)$. We call such a coupling minimal coupling.

Assumption 5. We take the first moment of $W^{\alpha\beta}(x)$ to reduce to General Relativity in the classical limit so that $\operatorname{Tr}\left[W^{\alpha\beta}(x)L_{\alpha}(x)\varrho L_{\beta}^{\dagger}(x)\right] = \operatorname{Tr}\left[\{H_m,\varrho\}\right]$. Since one generally considers matter Hamiltonians which only depend on γ_{ab} and not π^{ab} , this motivates our previous assumption of minimal coupling.

Assumption 6. The CQ term couples states with different momenta π^{ab} only; we only jump in momentum. We call such theories π – dispersive, while in the more general case, one can have both dispersion in the π^{ab} and dispersion in γ_{ab} . If both terms are present, then the relationship between π^{ab} and $\dot{\gamma}_{ab}$ exists only on the level of expectation values.

Assumption 7. We take the pure gravity part of the master equation to be deterministic and given by general relativity $\{H_{ADM}, \varrho\}$. In the more general case, the pure gravity evolution can be stochastic.

Assumption 1 is respected in Einstein's theory of General Relativity, but Einstein also considered what is now known as the unimodular theory of gravity [193, 194, 195, 196, 197, 198, 199] where N is chosen so that the cosmological constant becomes an integration constant. Assumption 1 also doesn't hold in Horava gravity [179, 180, 181] and shape dynamics [182, 183, 184], but here we explore the consequences of taking the full gauge symmetry. Assumption 3 seems reasonable when considering the jumping master equations. When computing the gauge transformations' commutators, it must be the case that $W^{\alpha\beta}$ term must transform like the Hamiltonian, or else the pure classical part and pure quantum part of the evolution will transform differently to the CQ coupling. This assumption is, however, violated for the continuous master equations since they have a different structure, which implements back-reaction via the off-diagonal elements such as $D_1 1, i^{\alpha 0} L_{\alpha \varrho}(z) + c.c.$ One could also imagine a theory with alternative, higher order, Lindblad operators that have the correct transformation properties – such as $W^{\phi\phi\phi\phi}(x)\phi\phi\rho\phi\phi$ – this type of higher order coupling seems to be implied by the covariant path integral approach we discuss in Chapter 8. However, we do not discuss such theories in this chapter. We can summarize Assumptions 4 and 6 as considering CQ couplings which take the form $W^{\alpha\beta}(z|z',x) = W^{\alpha\beta}(\gamma,\pi|\gamma',\pi',x)$, where the moments of $W^{\alpha\beta}(\gamma,\pi|\gamma,\pi',x)$ depend only on γ_{ab} and not π^{ab} . These assumptions are motivated twofold. Firstly, by analogy to pure classical gravity. There, when one considers minimally coupled scalar fields, the interaction term is of the form

$$W_{classical}^{\alpha\beta}(x)(\rho) = \int d^3y \frac{\delta \mathcal{H}_m(x)}{\delta \gamma^{ab}(y)} \frac{\delta \rho}{\delta \pi^{ab}(y)} = \int d^3y \frac{\delta h^{\alpha\beta}(x)}{\delta \gamma^{ab}(y)} L^{\dagger}_{\beta} L_{\alpha} \frac{\delta \rho}{\delta \pi^{ab}(y)}, \tag{7.62}$$

which only couples states with different momenta, and the coupling only has a functional dependence on the spatial metric through $\frac{\delta h^{\alpha\beta}}{\delta\gamma^{ab}}$. Secondly, since we end up calculating the commutation relations in (7.60), which includes Poisson brackets with the pure classical Hamiltonian and momentum, Assumptions 3, 4, 6 seem natural, though we shall find in Chapters 8 and 9 that more promising CQ theories violate the assumption of being only π diffusive.

Having made these simplifying assumptions, we are now ready to study their constraints. We have yet to specify the CQ coupling explicitly, except for some Lindbladian structure and functional dependence we would like it to have. As we shall see in the next section, there are various transformation properties that any realization must satisfy. We then study the constraints in such realizations, which arise from studying the algebra of equal time gauge generators in Equation (7.60). In particular, we find that the $[\mathcal{L}_a(x), \mathcal{L}_b(y)]$ closes, which is an expected artifact of the fact we are assuming (Assumption 2) \mathcal{L}_a generates spatial diffeomorphisms. We find the $[\mathcal{L}_a(x), \mathcal{L}(y)]$ generator closes so long as the couplings $W^{\alpha\beta}(z|z', x)$ satisfy certain transformation rules; essentially telling us that $W^{\alpha\beta}(z|z', x)$ must transform correctly under spatial diffeomorphisms. Finally, we study the $[\mathcal{L}(x), \mathcal{L}(y)]$ commutator, which leads to a generalization of the momentum constraint to CQ dynamics. We find that the preservation of this constraint gives rise to a CQ analog of the Hamiltonian constraint.

$[\mathcal{L}_a(x), \mathcal{L}_b(y)]$ commutator for the scalar field

Due to Assumption 2, we are considering the case where $W_a^{\alpha\beta} = 0$, so that the total CQ momentum generator \mathcal{L}_a reads

$$\mathcal{L}_a(x)(\varrho) = \{\mathcal{H}_a(x), \varrho\} - i[\mathcal{H}_{m,a}(x), \varrho].$$
(7.63)

One can then easily verify that

$$[\mathcal{L}[\vec{N}], \mathcal{L}[\vec{M}]] = \mathcal{L}[\vec{N}, \vec{M}], \qquad (7.64)$$

which vanishes when smeared over an observable since \mathcal{L}_a is treated as gauge. Since Equation (7.64) represents the lie algebra of spatial diffeomorphisms, this verifies that Equation (7.63) is the generator of spatial diffeomorphisms.

$[\mathcal{L}_a(x), \mathcal{L}(y)]$ commutator for the scalar field

We now compute the $[\mathcal{L}_a(x), \mathcal{L}(y)]$ commutator. With the interpretation that \mathcal{L}_a generates spatial diffeomorphisms, we find that the algebra closes so long as we pick the realizations to have the correct transformation properties under spatial diffeomorphisms. In total, we find the (smeared) commutation relation

$$\begin{aligned} \left[\mathcal{L}[\vec{N}], \mathcal{L}[N]\right](\rho) &= \int d^{3}x \left[N^{a}D_{a}N\mathcal{L}(\rho)\right] \\ &+ \int d^{3}xNN^{a} \left[L_{\alpha}D_{a}(W^{\alpha\beta}\varrho)L_{\beta}^{\dagger} - \frac{1}{2}\{L_{\beta}^{\dagger}L_{\alpha}, D_{a}(W_{0}^{\alpha\beta}\varrho)\}\}\right] \\ &+ \int d^{3}xD_{a}N^{a} \left[W^{\phi\phi}\phi\varrho\phi - \frac{1}{2}\{W_{0}^{\phi\phi}\phi^{2}, \varrho\}\right] - D_{a}N^{a} \left[W^{\pi\pi}\pi\varrho\pi - \frac{1}{2}\{W_{0}^{\pi\pi}\pi^{2}, \varrho\}\right] \\ &+ \int d^{3}x \left[(D_{c}N^{c}W^{ab} - D_{c}N^{b}W^{ca} - D_{c}N^{a}W^{cb})(D_{a}\phi\varrho D_{b}\phi - \frac{1}{2}\{D_{b}\phi D_{a}\phi, \varrho\})\right] \\ &+ \int d^{3}xN(x) \left[\{H[\vec{N}], W^{\alpha\beta}L_{\alpha}\varrho L_{\beta}^{\dagger} - \frac{1}{2}W_{0}^{\alpha\beta}L_{\beta}^{\dagger}L_{\alpha}, \varrho\}\} \\ &- W^{\alpha\beta}L_{\alpha}\{H[\vec{N}], \varrho\}L_{\beta}^{\dagger} - \frac{1}{2}W_{0}^{\alpha\beta}L_{\beta}^{\dagger}L_{\alpha}, \{P[\vec{N}], \varrho\}\}\right]. \end{aligned}$$

$$(7.65)$$

Although somewhat daunting, we see that so long as $W^{\alpha\beta}$ satisfies certain transformation properties, specifically if

$$\begin{aligned} \{\mathcal{H}[\vec{N}], W^{\pi\pi}(\varrho)\} - W^{\pi\pi}(\{\mathcal{H}[\vec{N}], \varrho\}) &= \int d^3x D_a N^a W^{\pi\pi}(\varrho) - N^a D_a W^{\pi\pi}(\varrho), \\ \{\mathcal{H}[\vec{N}], W^{\phi\phi}(\varrho)\} - W^{\phi\phi}(\{\mathcal{H}[\vec{N}], \varrho\}) &= \int d^3x - D_a N^a W^{\phi\phi}(\varrho) - N^a D_a W^{\phi\phi}(\varrho), \\ \{\mathcal{H}[\vec{N}], W^{ab}(\varrho)\} - W^{ab}(\{\mathcal{H}[\vec{N}], \varrho\}) \\ &= \int d^3x [D_c N^b W^{ca}(\varrho) + D_c N^a W^{cb}(\varrho) - D_c N^c W^{ab}(\varrho)] - N^a D_a W^{ab}(\varrho), \end{aligned}$$
(7.66)

then the algebra will close

$$[\mathcal{L}[\vec{N}], \mathcal{L}[N]](\rho) = \int d^3x N^a D_a N \mathcal{L}(\rho) = \mathcal{L}[L_{\vec{N}}M](\rho), \qquad (7.67)$$

and the theory will be spatially diffeomorphism invariant. Here $L_{\vec{N}}N$ is the Lie derivative of N along N^a . The conditions in Equation (7.66) are demystified somewhat when one realizes they are analogous to terms arising in classical gravity. It is a general property of minimally coupled field theories that [108]

$$\{\mathcal{H}_{ma}(x), \mathcal{H}_{m}(y)\} = 2D_{b}\left(\frac{\delta\mathcal{H}_{m}(y)}{\delta\gamma_{ab}(x)}\right) + \mathcal{H}_{m}(x)\partial_{a}\delta(x, y)$$
(7.68)

$$\{\mathcal{H}_a(x), \mathcal{H}_m(y)\} = -2D_b \left(\frac{\delta \mathcal{H}_m(y)}{\delta \gamma_{ab}(x)}\right),\tag{7.69}$$

and these combine so that the matter Hamiltonian transforms as a scalar under spatial diffeomorphisms

$$\{\mathcal{H}_a + \mathcal{H}_{ma}(x), \mathcal{H}_m(y)\} = \mathcal{H}_m(x)\partial_a\delta(x, y).$$
(7.70)

Since we are assuming that the CQ Lindbladian has the same structure as the matter Hamiltonian, we often find terms that look similar to (7.68) and (7.69) – essentially the anomalous terms in (7.65). We expect these to cancel with terms arising from the Poisson bracket – the terms in the final line of Equation (7.65) – enforcing conditions on the allowed realizations. We interpret this as telling us that $W^{\alpha\beta}(z|z')L^{\dagger}_{\beta}L_{\alpha}$ must transform like the Hamiltonian under the action of the Poisson bracket. From now on, we will assume that these can be satisfied without reference to an explicit realization. Some realizations can be found in [28, 3]. We will now derive the constraints which arise from the final component of the algebra – the [$\mathcal{L}(x), \mathcal{L}(y)$] commutator.

$[\mathcal{L}(x), \mathcal{L}(y)]$ commutator for the scalar field

We now move on to study the final commutator in algebra. So far, we have found restrictions on the realizations of the CQ theory. In this section, we will find that we will need to impose constraints in order for the theory to be consistent. We find

$$[\mathcal{L}[N], \mathcal{L}[M]](\varrho) = \int d^3x (N\partial_a M - M\partial_a N) \left[\{\gamma^{ab} \mathcal{H}_b, \varrho\} - i[\gamma^{ab} \mathcal{H}_{m,b}, \varrho] + \bar{\mathcal{C}}^a(\varrho) \right], \quad (7.71)$$

where

$$\bar{\mathcal{C}}^{a}(\varrho) = C_{J}^{ab}(D_{b}\phi(x)\varrho\pi(x) + \pi(x)\varrho D_{b}\phi(x)) - \frac{1}{2}C_{N}^{ab}\{D_{b}\phi(x)\pi(x) + \pi(x)D_{b}\phi(x),\varrho\}_{+}$$
$$-iC_{H}^{ab}[\mathcal{H}_{a},\varrho] + iC_{JN}^{ab}(D_{a}\phi(x)\varrho\pi(x) - \pi(x)\varrho D_{a}\phi(x)).$$
(7.72)

In Equation (7.72), we have defined the couplings⁴

$$C_{J}^{ab} = 2(h^{ab}W^{\pi\pi} + h^{\pi\pi}W^{ab}), \quad C_{N}^{ab} = 2(W_{0}^{ab}h^{\pi\pi} + W_{0}^{\pi\pi}h^{ab})$$
$$C_{H}^{ab} = \frac{1}{4}(W^{\pi\pi}W^{ab} - W_{0}^{ab}W_{0}^{\pi\pi}), \quad C_{JN}^{ab} = (W^{ab}W_{0}^{\pi\pi} - W_{0}^{ab}W^{\pi\pi}).$$
(7.73)

The first two terms of Equation (7.71) give rise to a component that is a gauge transformation. To see this, we smear over a phase space observable A to find

$$\langle A, \{\gamma^{ab}\mathcal{H}_b, \varrho\} - i[\gamma^{ab}\mathcal{H}_{m,b}, \varrho] \rangle = \langle \{A, \gamma^{ab}\}, \mathcal{H}_b \varrho \rangle + \langle \{A, \mathcal{H}_b\}\gamma^{ab}, -i[\gamma^{ab}\mathcal{H}_{m,b}, \varrho] \rangle$$

$$= \langle \{A, \gamma^{ab}\}, \mathcal{H}_b \varrho \rangle + \langle \mathcal{L}_b^*(A)\gamma^{ab}, \varrho \rangle,$$

$$(7.74)$$

which, since the final term in (7.74) is a gauge transformation, is weakly equal to

$$\langle \{A, \gamma^{ab}\}, \mathcal{H}_b \varrho \rangle.$$
 (7.75)

As a consequence, we find

$$[\mathcal{L}[N], \mathcal{L}[M]](\varrho) \approx \int d^3x (N\partial_a M - M\partial_a N) \left[\{\gamma^{ab}, \mathcal{H}_b \varrho\} + \bar{\mathcal{C}}^a(\varrho) \right],$$
(7.76)

and so, in order for the algebra of generators to close, we need to impose the CQ momentum constraint

$$\mathcal{C}^{a} = \{\gamma^{ab}, \mathcal{H}_{b}\varrho\} + \bar{\mathcal{C}}^{a}(\varrho) \approx 0, \qquad (7.77)$$

which should hold when smeared over observables A. Using the definitions in Equation (7.73), we can write this out in full as

$$\mathcal{C}^{a}(\varrho) = \{\gamma^{ab}, \mathcal{H}_{b}\varrho\} - iC_{H}^{ab}[\mathcal{H}_{m,a}, \varrho] + C_{J}^{ab}(D_{b}\phi(x)\varrho\pi(x) + \pi(x)\varrho D_{b}\phi(x)) - \frac{1}{2}C_{N}^{ab}\{\mathcal{H}_{m,a}, \varrho\}_{+} + C_{JN}^{ab}(D_{a}\phi(x)\varrho\pi(x) - \pi(x)\varrho D_{a}\phi(x)).$$
(7.78)

It is useful to perform a quick sanity check on Equation (7.78) to see if it gives the correct momentum constraint in the classical limit. First, taking the trace over the quantum system and defining $\text{Tr} [\varrho(z)] = \rho(z)$, one sees

$$\operatorname{Tr}\left[C^{a}(\varrho)\right] = \left\{\gamma^{ab}, \mathcal{H}_{b}\rho\right\} + \operatorname{Tr}\left[\left(2C_{J}^{ab} - C_{N}^{ab}\right)\pi D_{b}\phi\varrho\right].$$
(7.79)

⁴Here the labels J, N, H, JN stand for "Jump", "no-event", "Hamiltonian", "jump-no-event". Specifically, the C_J^{ab} term takes the form of a Jump term in a Lindblad equation, while C_N^{ab} takes the form of a no-event term in the Lindblad equation. The C_H^{ab} term takes the form of a Hamiltonian term. The final term C_{JN}^{ab} has no analog with the Lindblad equation but arises due to the commutation of the jump and no-event term in \mathcal{L} .

Performing a Kramers-Moyal expansion of the CQ couplings $W^{\alpha\beta}(z|z',x)$ to first order, one finds the zeroth order term cancels in Equation (7.79) and, remembering that the first moment is such that Einstein's equations hold, we are left with

$$\operatorname{Tr}\left[C^{a}(\varrho)\right] = \left\{\gamma^{ab}, \mathcal{H}_{b}\rho\right\} + \operatorname{Tr}\left[D_{b}\phi\pi\{\gamma^{ab}, \varrho\}\right] + \dots$$
(7.80)

Smearing over an observable A, Equation (7.80) becomes

$$\int Dg D\pi A \operatorname{Tr} \left[C^{a}(\varrho) \right] = \int Dg D\pi \{A, \gamma^{ab}\} (\mathcal{H}_{b}\rho + \pi \operatorname{Tr} \left[D_{b}\phi\varrho \right] + \dots) .$$
(7.81)

Comparing this CQ constraint to the standard momentum constraint of GR, we see that it gives a sensible constraint in the classical limit, i.e., one that is satisfied by classical gravity (recall that $\mathcal{H}_{m,a} = \pi D_b \phi$). We also see that we get a sensible constraint in the limit where the matter remains quantum. It appears it can be satisfiable even though it contains both functionals of the classical degrees of freedom and quantum operators. In particular, one could have been concerned that we would find that we had to satisfy the naive CQ constraint $\mathcal{H}_a + \pi D_a \phi \rho \approx 0$ when restricted to the constraint surface; this would have required setting a c-number equal to an operator equation. Instead, we get a CQ-equation equivalent to finding mixed fixed points of some dynamics, which is at least possible to hold in principle.

Conservation of the momentum constraint C^a for the scalar field

Now that we have a momentum constraint, we must check to see if it is preserved in time. In the classical case, preservation of the momentum constraint gives rise to the Hamiltonian constraint, and we expect something analogous for the CQ theory. Indeed, we find we get a constraint which is the standard Hamiltonian constraint in the classical limit – although it appears to require additional constraints or a restriction on the lapse and shift if one is to hope that it will be preserved in time. We discuss the possible implications for this when we conclude in Section 7.6.

Conservation of the momentum constraint requires calculating the quantity $C^a(\frac{\partial \varrho}{\partial t}) = C^a(\mathcal{L}[N, \vec{N}](\varrho))$. For calculation purposes, it is slightly simpler to consider the commutator $[C^a, \mathcal{L}[N, \vec{N}]](\varrho)$, noting that the difference between the commutator and the evolution of the constraint is given by the term $\int Dg D\pi C^{a*}(\mathcal{L}^*(A))\varrho$; that is the momentum constraint but

smeared over the phase space operator $\mathcal{L}^*(A)$ instead of A. When performing the calculation, we will smear the momentum constraint C^a with a lower index spatial smearing function $M_a(x)$ and write $C^a \mathcal{C}[\underline{M}] = \int d^3x M_a \mathcal{C}^a$.

We first calculate the commutator with the spatial part of the evolution equation $[C[\underline{M}], \mathcal{L}[\vec{N}]](\varrho)$; in other words, to check whether or not the momentum constraint transforms correctly under spatial diffeomorphisms. Assuming that the realization has the transformation properties defined in Equation (7.66), one finds⁵

$$[\mathcal{C}[\underline{M}], [\mathcal{L}[\vec{N}]](\varrho) = -\int d^3x N^c D_c M_a \mathcal{C}^b(\varrho) + M_a D_c N^a \mathcal{C}^c(\varrho), \qquad (7.82)$$

which vanishes on the constraint surface. We are then left to calculate the commutation with $\mathcal{L}[N]$, which, in analogy to the classical case, we expect to give a Hamiltonian constraint.

It shall be useful to split up the generators into the purely classical part involving the Poisson bracket and everything else. To that end, we write $\mathcal{L}(\varrho) = \{\mathcal{H}, \varrho\} + \bar{\mathcal{L}}(\varrho)$ and $C^a(\varrho) = \{\gamma^{ab}, \mathcal{H}_b \varrho\} + \bar{C}^a(\varrho)$. The evolution of the smeared constraint reads

$$[C[\underline{\mathbf{M}}], \mathcal{L}[N]](\varrho) = \int d^3x d^3y M_a(y) N(x) [\{\gamma^{ab}(y), \{\mathcal{H}_b(y), \mathcal{H}(x)\}\varrho\} + \{\gamma^{ab}, \mathcal{H}_b \bar{\mathcal{L}}(\varrho)\} - \bar{\mathcal{L}}(\{\gamma^{ab}, \mathcal{H}_b \varrho\}) + [\bar{\mathcal{C}}^a, \bar{\mathcal{L}}](\varrho)) + \{\{\gamma^{ab}(y), \mathcal{H}(x)\}, \mathcal{H}_b(y)\varrho\} + \bar{\mathcal{C}}^a(\{\mathcal{H}, \varrho\} - \{\mathcal{H}, \bar{\mathcal{C}}^a(\varrho)\}].$$
(7.83)

Much is going on in Equation (7.83), so we will break it into pieces and discuss what we expect to get back from each term before presenting our findings. We first comment on the third line of Equation (7.83), which consists of the term

$$\{\{\gamma^{ab}(y), \mathcal{H}(x)\}, \mathcal{H}_b(y)\varrho\} + \bar{\mathcal{C}}^a(\{\mathcal{H}, \varrho\}) - \{\mathcal{H}, \bar{\mathcal{C}}^a(\varrho)\}.$$
(7.84)

Firstly, we note that Equation (7.84) has the Lindblad structure of the momentum constraint.⁶ To gain some intuition, it is useful to take the trace of (7.84) and look at the first order in the Kramers-Moyal expansion. Explicitly, Equation (7.84) becomes

$$\{\{\gamma^{ab}(y), \mathcal{H}(x)\}, \mathcal{H}_b(y)\rho\} + \operatorname{Tr}\left[D_b\phi\pi\{\{\gamma^{ab}(y), \mathcal{H}(x)\}, \varrho\}\right] + \dots$$
(7.85)

⁵This can be made to look similar to the $\mathcal{L}_a, \mathcal{L}_b$ commutator by using integration by parts on the second term of (7.82) where it differs slightly because $D_b C^a(\rho)$ is not vanishing

⁶By this, we mean that the expression in (7.84) contains quantum operators with the same structure as the momentum constraint in Equation (7.78), i.e., those which come in the format $\sim \partial_a \phi \rho \pi$.

and smearing this over an observable gives

$$\int Dg D\pi \{A, \{\gamma^{ab}(y), \mathcal{H}(x)\}\} (\mathcal{H}_b \rho + \operatorname{Tr} [D_b \phi \pi \varrho] + \dots) .$$
(7.86)

We can compare (7.86) to the momentum constraint in (7.81), and we note that although almost identical, they are not quite the same. To be precise, Equation (7.86) is smeared over $\{A, \{\gamma^{ab}, \mathcal{H}\}\}$ whilst Equation (7.81) is instead smeared over $\{A, \gamma^{ab}\}$. Of course, both the constraints are not independent; both vanish if we satisfy the effective classical constraint $\mathcal{H}_b\rho + \text{Tr} [D_b\phi\pi\varrho] = 0$. We, therefore, posit that in a sensible realization, Equation (7.84) should be weakly zero whenever the momentum constraint is satisfied. Otherwise, one is forced to view Equation (7.84) as a separate constraint to the momentum constraint, which must be preserved in time by itself. One then faces a similar issue looking at the time evolution of Equation (7.84) and must impose more constraints in a series that seems unlikely to terminate. We view the condition that Equation (7.84) should be weakly zero whenever the momentum constraint is satisfied as a transformation rule which any realization $W^{\alpha\beta}(z|z')$ must obey, telling us how they CQ couplings must transform under the action of the pure gravity Hamiltonian $\{\mathcal{H}, \}$ – just as the transformation rules defined in Equation (7.66) place conditions on the realizations in order for the theory to be spatially diffeomorphism invariant.

We now study the remaining terms in Equation (7.83), given by the first two lines, which we expect to give rise to a Hamiltonian constraint. Before presenting the result, getting some intuition for each of the terms appearing is helpful. Using the Dirac algebra, defined in (7.15), the purely classical term, $\{\gamma^{ab}(y), \{\mathcal{H}_b(y), \mathcal{H}(x)\}\varrho\}$, can be written as $\partial_b^y \delta(y, x)\{\gamma^{ab}(y), \mathcal{H}(y)\varrho\}$, giving rise to the classical Hamiltonian part of the constraint. It is less obvious what we expect $[\bar{\mathcal{C}}^a, \bar{\mathcal{L}}](\varrho)$ to give back. Taking a step back and looking at the analogous term in classical gravity, one has instead of $[\bar{\mathcal{C}}^a, \bar{\mathcal{L}}](\varrho)$ the term

$$\{\mathcal{H}_{ma}(x), \mathcal{H}_{m}(y)\} = 2D_{b}\left(\frac{\delta\mathcal{H}_{m}(y)}{\delta\gamma_{ab}(x)}\right) + \mathcal{H}_{m}(x)\partial_{a}\delta(x, y)$$
(7.87)

which gives rise to the Hamiltonian constraint and a term $2D_b\left(\frac{\delta\mathcal{H}_m(y)}{\delta\gamma_{ab}(x)}\right)$. In calculating the full Poisson bracket between the momentum constraint and the Hamiltonian constraint, this anomalous term cancels with a term arising from the Poisson bracket between the pure gravity momentum and the matter Hamiltonian $\{\mathcal{H}_a, \mathcal{H}_m\}$ so that in combination $\{\mathcal{H}_a + \mathcal{H}_{ma}, \mathcal{H}_m\} \sim \mathcal{H}_m$ gives back the matter part of the Hamiltonian constraint.

By virtue of the Lindblad structure of the constraints, due to Assumption 3, we expect that under commutation \bar{C}^a transforms like momentum and \mathcal{L} like a Hamiltonian. As a consequence, we expect to find (morally)

$$[\bar{C}^a, \bar{\mathcal{L}}] \sim \mathcal{R} + \mathcal{L}_{constraint}(x)\partial_a \delta(x, y), \tag{7.88}$$

where \mathcal{R} is the CQ version of $2D_b\left(\frac{\delta \mathcal{H}_m(y)}{\delta \gamma_{ab}(x)}\right)$ and $\mathcal{L}_{constraint}(x)$ is the CQ Hamiltonian constraint. Finally, in analogy with the classical case, we expect the \mathcal{R} term appearing in Equation (7.88) to cancel with the Poisson bracket term arising in the second line of (7.83), namely $\{\gamma^{ab}, \mathcal{H}_b \bar{\mathcal{L}}(\varrho)\} - \bar{\mathcal{L}}(\{\gamma^{ab}, \mathcal{H}_b \varrho\})$, so that the first two lines of Equation (7.83) gives rise to the CQ generalization of the Hamiltonian constraint.

We now present the full Hamiltonian constraint – the first two lines of Equation (7.83). In doing so, it is first useful to introduce notation for a certain combination of terms that frequently arises. We define \mathcal{R}_{AB} via

$$\mathcal{R}_{AB}^{\alpha\beta}L_{\alpha}\varrho L_{\beta} = \int d^{3}x N \left[W_{B}^{\phi\phi}\phi D_{b}(M_{a}C_{A}^{ab}\varrho)\phi - W_{B}^{\pi\pi}\pi D_{b}(M_{a}C_{A}^{ab}\varrho)\pi \right. \\ \left. + D_{e}\phi(D_{b}(M_{a}C_{A}^{af}\varrho)W_{B}^{eb} + D_{b}(M_{a}C_{A}^{ae}\varrho)W_{B}^{bf} - D_{b}(M_{a}C_{A}^{ab}\varrho)W_{B}^{ef})D_{f}\phi \right. \\ \left. + M_{a}C_{A}^{ab}L_{\alpha}D_{b}(W_{B}^{\alpha\beta}\varrho)L_{\beta} \right].$$

$$(7.89)$$

Here the A sub-index denotes terms coming from the momentum constraint and is associated with C_J, C_H, C_{JN}, C_N . In contrast, the B sub-index denotes terms coming from the Hamiltonian constraint and is associated with the couplings $W_J^{\alpha\beta} = W^{\alpha\beta}(z|z'), W_N^{\alpha\beta} = W_0^{\alpha\beta}(z), W_H^{\alpha\beta} = h^{\alpha\beta}.^7$ These terms are the CQ analogy of the $2D_b\left(\frac{\delta \mathcal{H}_m(y)}{\delta \gamma_{ab}(x)}\right)$ terms, which arise from integration by parts due the transformation properties of $\overline{C}^a, \overline{\mathcal{L}}$. A lengthy but straightforward calculation

⁷Again, the J, H, JN, N stand for jump, Hamiltonian, jump-no event and no-event and are an attempt to label sensibly the terms arising in the constraint. We remind the reader that the jump and no-event terms have the structure defined just before equation (7.4), while the jump-no event term appearing in the momentum constraint arises from a commutation of jump and no-event terms appearing in the $[\mathcal{L}(x), \mathcal{L}(y)]$ commutator.

then gives the total Hamiltonian constraint

$$\begin{split} \mathcal{L}_{constraint} &= \int d^3 x M_a D_b N \left[(-2i C_H^{ab} h^{\alpha\beta} + \frac{i}{2} C_N^{ab} W_0^{\alpha\beta} - \frac{i}{2} C_J^{ab} W^{\alpha\beta}) [L_{\beta}^{\dagger} L_{\alpha}, \varrho] \\ &+ \left\{ \gamma^{ab}(y), \mathcal{H}(y) \varrho \right\} + 2 (C_J^{ab} h^{\alpha\beta} + C_H^{ab} W^{\alpha\beta}) L_{\alpha} \varrho L_{\beta}^{\dagger} - (C_N^{ab} h^{\alpha\beta} + C_H^{ab} W_0^{\alpha\beta}) \{L_{\beta}^{\dagger} L_{\alpha}, \varrho\}_+ \right] \\ &+ \int d^3 x N M_a \mathcal{H}_b(\{\gamma^{ab}, \mathcal{H}_b \bar{\mathcal{L}}(\varrho)\} - \bar{\mathcal{L}}(\{\gamma^{ab}, \mathcal{H}_b \varrho\}) \\ &+ (\frac{i}{2} \mathcal{R}_{NN}^{\alpha\beta} - \frac{i}{2} \mathcal{R}_{JJ}^{\alpha\beta} - 2i \mathcal{R}_{HH}^{\alpha\beta}) [L_{\beta}^{\dagger} L_{\alpha}, \varrho] + (2 \mathcal{R}_{HJ}^{\alpha\beta} + 2 \mathcal{R}_{JH}^{\alpha\beta}) L_{\alpha} \varrho L_{\beta}^{\dagger} \\ &- (\mathcal{R}_{NH}^{\alpha\beta} + \mathcal{R}_{HN}^{\alpha\beta}) \{L_{\beta}^{\dagger} L_{\alpha}\}_+ \\ &+ \int d^3 x N M_a [(C_{JN}^{ab} (W_0^{\phi\phi} - 2i h^{\phi\phi}) - C_J^{ab} W_0^{\phi\phi} + C_N^{ab} W^{\phi\phi}) D_b \phi \varrho \phi + \\ &+ (C_{JN}^{ab} (W_0^{\phi\phi} + 2i h^{\phi\phi}) + C_J^{ab} W_0^{\phi\phi} - C_N^{ab} W^{\phi\phi}) \rho \varrho D_b \phi \\ &+ (C_{JN}^{ab} (-W_0^{\pi\pi} + 2i h^{\pi\pi}) - W_0^{\pi\pi} C_J^{ab} + C_N^{ab} W^{\pi\pi}) D_b \pi \varrho \pi + (C_{JN}^{ab} (-W_0^{\pi\pi} - 2i h^{\pi\pi})) \\ &+ (W_0^{\pi\pi} C_J^{ab} - C_N^{ab} W^{\pi\pi}) \pi \varrho D_b \pi \\ &+ (C_{JN}^{ab} (-W_0^{\phi\phi} + 2i h^{\phi\phi}) - C_N^{ab} W^{ef} - W_0^{ef} h^{ef} C_J^{ab}) (D_e D_f \phi) \varrho D_b \phi \\ &+ (C_{JN}^{ab} (-W_0^{\phi\phi} + 2i h^{\phi\phi}) - C_N^{ab} W^{ef} + W_0^{ef} C_J^{ab}) D_b \phi \varrho (D_e D_f \phi)] \\ &\int d^3 x [N W^{\pi\pi} \{\pi D_b (M_a \pi C_{JN}^{ab}, \varrho)\}_+ - N M_b C_{JN}^{ab} W^{\phi\phi} \{D_b \phi \phi, \varrho\}_+ \\ &+ M_a C_{JN}^{ab} \{D_b \phi D_d (N W^{cd} D_c \phi, \varrho)\}_+] \\ &- 2 \int d^3 x [N W_0^{\pi\pi} \pi D_b (C_{JN}^{ab} M_a \varrho) \pi + M_b D_d N W_0^{cd} C_{JN}^{ab} D_c \phi \varrho D_b \phi] \approx 0, \end{split}$$

which needs to be weakly zero for the theory to be gauge invariant. Again, a lot is going on in Equation (7.90), so we now summarize what each term in the constraint tells us.

• The first two lines of Equation (7.90) look like a potentially sensible Hamiltonian constraint. Indeed, if we take their quantum trace, we end up with

$$\{\gamma^{ab}(y), \mathcal{H}(y)\rho\} + 2(C_J^{ab}h^{\alpha\beta} + C_H^{ab}W^{\alpha\beta} - C_N^{ab}h^{\alpha\beta} - C_H^{ab}W_0^{\alpha\beta})\operatorname{Tr}\left[L_{\beta}^{\dagger}L_{\alpha}\rho\right].$$
(7.91)

Performing the Kramer's-Moyal expansion to first order, we find this gives

$$\{\gamma^{ab}(y), \mathcal{H}(y)\rho\} + \operatorname{Tr}\left[h^{\alpha\beta}L^{\dagger}_{\beta}L_{\alpha}\{\gamma^{ab}, \varrho\}\right] + \dots$$
(7.92)

and smeared over an observable A reads

$$\int Dg D\pi \{A, \gamma^{ab}(y)\} (\mathcal{H}(y)\rho + \operatorname{Tr}\left[h^{\alpha\beta}L^{\dagger}_{\beta}L_{\alpha}\varrho\right] + \dots$$
(7.93)

and looks exactly like what we might expect as a Hamiltonian constraint in the classical limit.

• The third line is the part coming from the Poisson bracket of the constraint with $\bar{\mathcal{L}}$, which, as mentioned, we expect to cancel with the \mathcal{R} terms (in the fourth and fifth lines) because $\bar{\mathcal{C}}^a$ transforms like momentum and $\bar{\mathcal{L}}$ like a Hamiltonian. This does not appear to happen here. Taking the trace over the quantum system, the third and fourth lines of Equation (7.90) combine to give

$$\int d^{3}x N M_{a} \bigg[\operatorname{Tr} \bigg[\bar{\mathcal{L}}(\{\gamma^{ab}, \varrho\}) - \bar{\mathcal{L}}(\mathcal{H}_{b}\{\gamma^{ab}, \varrho\}) + (2\mathcal{R}_{JH}^{\alpha\beta} - 2\mathcal{R}_{NH}^{\alpha\beta})(\varrho) L_{\beta}^{\dagger} L_{\alpha}) \bigg] + \operatorname{Tr} \bigg[(2\mathcal{R}_{HJ}^{\alpha\beta} - 2\mathcal{R}_{HN}^{\alpha\beta})(\varrho) L_{\beta}^{\dagger} L_{\alpha} \bigg] \bigg].$$
(7.94)

If we perform a Kramers Moyal expansion to first order, we see that the first term in Equation (7.94) vanishes so that we are left with the final term, $\text{Tr}\left[(2\mathcal{R}_{HJ}^{\alpha\beta}-2\mathcal{R}_{HN}^{\alpha\beta})(\varrho)L_{\beta}^{\dagger}L_{\alpha}\right]$. Recalling the definition of R_{HJ}, R_{HN} in Equation (7.89), as well as the form of C_N in Equation (7.73), we see that the R_{HN} term cancels with the zeroth moment of the R_{HJ} term and we are left with the first moment of the $2\mathcal{R}_{HJ}^{\alpha\beta}$ term alone. Explicitly, $\mathcal{R}_{HJ}^{\alpha\beta}$ is written

$$\mathcal{R}_{HJ}^{\alpha\beta} = \int d^3x N \left[W^{\phi\phi} D_b (M_a C_H^{ab} \varrho) - W^{\pi\pi} D_b (M_a C_H^{ab} \varrho) + (D_b (M_a C_H^{af} \varrho) W^{eb} + D_b (M_a C_H^{ae} \varrho) W^{bf} - D_b (M_a C_H^{ab} \varrho) W^{ef} \right] + M_a C_H^{ab} L_\alpha D_b (W^{\alpha\beta} \varrho) L_\beta \right],$$
(7.95)

and one can verify if the first moments of the Kramers-Moyal expansion are to give GR, this will not identically be zero. One might hope that one could include it in the Hamiltonian constraint. However, since it comes with different smearing functions to the (would be) Hamiltonian constraint in the first two lines of (7.90), we must either restrict the lapse and shift or Equation (7.95) must be imposed as a separate constraint, which would appear to be over-constraining the system. We also see offending terms of the form $D_b(C_H^{ab}\varrho)$; these are interesting since we do not get these in pure GR; the matter part of the momentum constraint contains no metric degrees of freedom. It is worth noting that we get these violating terms even in classical analogs of the CQ theory. To be precise, one can ask the question: can one have an autonomous master equation on the phase space of GR, which contains noise and is gauge invariant? For example, one can study a

Fokker-Plank type equation linear in the lapse and shift and apply the same arguments as outlined in this chapter to derive a momentum constraint, which must then be preserved in time. In doing so, we still find violating terms of the form $D_b(C^{ab}\varrho)$.

• The rest of the terms in Equation (7.90) are purely CQ terms with no classical analog. They come with different smearing functions to those in the (would be) Hamiltonian constraint in the first two lines of Equation (7.90). Consequently, they must be imposed as a separate constraint, itself preserved in time, or we could impose restrictions on the choice of lapse and shift; we do not check what this gives here. Given the form of the remaining terms, it seems they render the constraint non-satisfiable without restrictions on the lapse and shift.

All of these terms come from the C_J, C_{JN} and C_N parts of the momentum constraint, which in turn come from the Jump, and no-event parts of the evolution equation when calculating the $[\mathcal{L}[N], \mathcal{L}[M]]$ commutation, i.e., from the consideration of the gauge algebra under two CQ jumps. This is where one might have expected the current demands for gauge invariance to break down. One might hope that if no no-event term was associated with CQ back-reaction, then the expression for the remaining terms in (7.90) is greatly simplified. This is the case for the class of continuous classical-quantum dynamics introduced in chapter 4. More generally, it would seem that the elimination of these terms will require us to weaken or change assumptions outlined in Section 7.5. We study improved theories in Chapters 8 and 9, which violate several of the assumptions considered here.

To summarize, although we have found a sensible-looking momentum constraint, when looking at its conservation in time we find Equation (7.90), which either requires a restriction on the choice of lapse and shift or additional constraints to be satisfied, which almost surely over-constrain the system.

We have broken down Equation (7.90) into multiple terms with different interpretations. In particular, we find several terms with different smearing functions, meaning we do not find a single constraint. We must either impose multiple constraints or restrict the lapse and shift so that the extra constraints vanish. Even though we impose constraints in a weak form, which can be viewed as a constraint on the moments of any allowed distribution ρ , imposing additional constraints, themselves preserved in time, is likely to over-constrain the system.

Take, for example, the pure gravity momentum constraint as derived in Equation (7.46), $\langle \{\gamma^{ab}, A\}, \mathcal{H}_b \rho \rangle \approx 0$. As mentioned, we can take this in its weakest form; it needs to vanish when smeared over observables instead of arbitrary smearing functions. However, we can keep applying gauge transformations to the weak form of the constraints to get more and more constraints on the moments of the distribution ρ . It would appear that one is forced eventually to impose a stronger version of the constraint. Instead of viewing Equation (7.46) as a constraint on the moments of the distribution, one is likely forced to impose a constraint on the state space, which must hold for any smearing function A. In this case, imposing multiple constraints will over-constrain the theory to no longer have 2 degrees of freedom per space-time point.

7.6 Discussion

In this chapter, we have presented a methodology to study the gauge invariance of a class of autonomous CQ theories of gravity linear in the lapse and shift N, N^a , and that reproduces the dynamics of GR in the classical limit. The theory could be regarded as fundamental or an effective theory of quantum gravity in the classical limit of the gravitational degrees of freedom. We have derived the constraints on the level of the equations of motion. The theory is invariant under spatial diffeomorphisms. We then demanded that the theory be invariant under an arbitrary choice of N and N^a , from which we can derive the theory's constraints, including an analog of the momentum and Hamiltonian constraint. The momentum constraint arises as a condition required for the theory to be invariant under the choice of lapse function. At the same time, the Hamiltonian constraint arises from demanding that the momentum constraint be preserved in time. Unlike classical GR, the constraints do not correspond to a constraint surface of the phase space but rather as an operator equation acting on the CQ state. This is reminiscent of approaches in quantum gravity where the super-Hamiltonian and super-momentum operators are applied to the state, and one constrains the wave function to be annihilated by these operators. However, here, we do not assume that the classical constraints are promoted to operators but derive them from the symmetry considerations.

Here, we have asked for the full invariance of the dynamics under the lapse and shift while

still consistent with General Relativity. Full gauge invariance appears too strong a condition for the models considered in this chapter: it seems unlikely that the constraints can be solved, even in the weak sense. In Chapters 8 and 9, we study different models of CQ dynamics based on the form of the continuous master equation, which are more promising. These theories violate several assumptions presented in this chapter. The continuous master equations have a different Lindblad structure to the jumping models, sourcing back-reaction via the off-diagonal $D_{1,i}^{\alpha 0} L_{\alpha \varrho} + c.c$ components, violating Assumption 3. In the covariant path integral approach of Chapter 8, we also find that higher order Lindbladian terms are required for covariance, again violating Assumption 3. In both Chapter 8 and Chapter 9, we find that there is diffusion in the metric itself, not just the conjugate momenta, violating the Assumption 7 of deterministic classical dynamics; when studying the Newtonian limit of CQ theories in Chapter 9, we see that diffusion of the spatial metric is necessary in order to preserve the Newtonian constraint.

When studying the path integral approach in Chapter 8, we will show that it is possible to construct diffeomorphism invariant theories of CQ gravity, and we give an example where the trace of Einstein's equations are satisfied on average, showing that diffeomorphism invariance can be upheld in CQ theories. However, this theory is not constrained. We also attempt to construct a full theory that gives rise to the G_{0i}, G_{00} components of Einstein's equations in the classical limit. For this theory, we posit a set of CQ constraints, but we are unable to prove that they give rise to CPTP dynamics on the constraint surface.⁸

It could be that asking for both full gauge invariance and the classical limit of Einstein's equations is too strong a condition. Several theories, such as Horava gravity, shape dynamics, and unimodular gravity, fix the lapse and shift. The gauge group of the theory is then smaller or different than the gauge group of general relativity. One could, for example, be satisfied with foliation diffeomorphisms alone and impose that N must be spatially constant. In that case, the constraint algebroid would close since the smearing functions always appear with divergences acting on them. However, one hopes a weaker one can be found.

Another possibility is that if one thinks of this theory as the classical limit of quantum 8 In Chapter 9, we study the Newtonian limit of CQ theories, which can be understood as a gauge fixed, weak field limit of a complete theory. We show that it is possible to have constraints that are preserved in time, leading to a CQ version of Poisson's equation.

gravity, we should lift the autonomous assumption since by taking the classical limit, we are throwing away quantum information, which could act as a memory for the evolution leading to non-Markovian dynamics. In this case, one hopes the methodology of studying CQ constraints could shed light on a possible theory of quantum gravity. Indeed the algebra shares some features of quantum algebra since the matter fields are quantized. At the same time, the classical nature of the gravity part allows for a much more tractable set of calculations, especially for the continuous master equation. Along the way, one sees that a number of conceptual issues that prove difficult to resolve in quantum gravity also occur in theories where one has a probability density over gravitational degrees of freedom.

Chapter 8

Covariant path integrals

So far, we have not studied classical-quantum dynamics in a manifestly covariant framework. The problem with studying gravity, or any field theory, in a master equation picture, is twofold. Firstly, from a practical point of view, field theories are generally better suited to path integral methods. Secondly, in a master equation picture, it is difficult to impose symmetries directly on the master equation. Indeed, writing down master equations for classical-quantum fields directly, without knowing whether or not they are covariant or uphold space-time symmetries, seems to go against much of the principles of modern physics, where one starts with actions based on symmetry principles. If one took the position that there is a fundamentally classical field, such as the gravitational field, it is also not obvious how one could couple it to the standard model while simultaneously ensuring symmetry principles and renormalizability are upheld.

This chapter studies configuration space path integrals for quantum fields interacting with classical fields. We show that this can be done consistently by proving that the dynamics are completely positive directly, without resorting to master equation methods. This is especially important since, in general, we saw in Chapter 6 that it was only possible to go from a master equation picture to a path integral picture when the master equation is less than quadratic in classical or quantum momenta. The path integrals allow one to readily impose space-time symmetries, including Lorentz invariance or diffeomorphism invariance.

Our results have consequences for any theory with a degree of freedom that behaves classically, whether effective or fundamental. With this in mind, we provide a possible template for studying CQ field theories. We introduce a class of classical-quantum actions motivated by Chapter 6, which can be used to construct theories with a sensible classical limit. The corresponding path integral can be understood in terms of summing over all classical and quantum paths, where the classical paths deviating too much from their semi-classical configuration are suppressed by the coupling D_0 , which also governs the strength of the quantum decoherence. Since we do not have a full theory of quantum gravity, of particular relevance is to construct an effective theory of quantum matter back-reacting on classical space-time, and we discuss the application of our work to the gravitational setting. We introduce a path integral formulation of general relativity where the space-time metric is treated classically and a diffeomorphism invariant theory based on the trace of Einstein's equations.

This chapter is based on the paper [8], which is work done in collaboration with Jonathan Oppenheim.

8.1 Completely positive path integrals

We saw in Chapter 6 that we could derive CQ path integrals from autonomous CQ master equations, and we could arrive at a configuration space path integral analytically when the action was at most quadratic in momenta. In this section, we shall show that one can prove the complete positivity of autonomous classical-quantum dynamics directly from the path integral. As a corollary, our result can also be to prove the complete positivity of the Feynman-Vernon path integral without resorting to the Lindblad equation. This is important, since often higher derivative path integrals are considered in the literature which turn out to not be completely positive (see Appendix J).

The path integral tells us how the components of the density matrix evolve. Including a classical variable z, the path integral should tell us how to evolve the components of a classicalquantum state

$$\varrho(z,t) = \int dz d\phi^+ d\phi^- \varrho(z,t,\phi^+,\phi^-) |\phi^+\rangle \langle \phi^-|, \qquad (8.1)$$

where ϕ represents a continuous quantum degree of freedom and $\varrho(z, t, \phi^+, \phi^-) = \langle \phi^+ | \rho(z, t) | \phi^- \rangle$ are the components of the CQ state. Writing Equation (8.1) out explicitly, generically, a path integral will take the form

$$\rho(z_f, \phi_f^+, \phi_f^-, t_f) = \int \mathcal{D}z \mathcal{D}\phi^+ \mathcal{D}\phi^- e^{\mathcal{I}[\phi^+, \phi^-, z, t_i, t_f]} \rho(z_i, \phi_i^+, \phi_i^-, t_i).$$
(8.2)
In Equation (8.2), it is implicitly understood that boundary conditions are to be imposed at t_f . In the purely quantum case, one has $\mathcal{I}[\phi^+, \phi^-, t_i, t_f] = iS[\phi^+, t_i, t_f] - iS[\phi^-, t_i, t_f]$ and the path integral is doubled since we are considering density matrices so we must sum over all bra and ket paths.

When the action contains higher derivatives, we can also include additional initial conditions on the time derivatives of the fields in Equation (8.3) [200]. For example, if the action contains terms with second-time derivatives in the classical degrees of freedom, we can write down the action

$$\rho(z_f, \dot{z_f}, \phi_f^+, \phi_f^-, t_f) = \int \mathcal{D}z \mathcal{D}\phi^+ \mathcal{D}\phi^- e^{\mathcal{I}[\phi^+, \phi^-, z, t_i, t_f]} \rho(z_i, \dot{z}_i, \phi_i^+, \phi_i^-, t_i).$$
(8.3)

Having introduced the classical-quantum formalism, let us now state and prove our main result:

Any time-local classical-quantum path integral with action of the form

$$\mathcal{I}(\phi^{-}, \phi^{+}, z, t_{i}, t_{f}) = \mathcal{I}_{CQ}(\phi^{+}, z, t_{i}, t_{f}) + \mathcal{I}_{CQ}^{*}(\phi^{-}, z, t_{i}, t_{f}) - \mathcal{I}_{C}(z, t_{i}, t_{f}) + \int_{t_{i}}^{t_{f}} dt \sum_{\gamma} c^{\gamma}(z, t) (L_{\gamma}[\phi^{+}]L_{\gamma}^{*}[\phi^{-}])$$
(8.4)

defines completely positive CQ dynamics when the terms in Equation (8.4) have the following properties: $L_{\gamma}[\phi^{\pm}]$ can be any functional of the bra and ket variables, $c^{\gamma} \geq 0$, \mathcal{I}_{C} is positive (semi) definite, and the real part of \mathcal{I}_{CQ} is negative (semi) definite. We implicitly assume that c^{γ} is chosen so that the path integral converges. The dynamics described by Equation (8.4) is CP but not necessarily norm preserving. However, via Table 2.2 time-local CP dynamics can always be normalized in a linear way, and including this we find normalized CP dynamics given by Equation (8.13).

In the field-theoretic case, the final line of Equation (8.4) is replaced by

$$\int_{t_i}^{t_f} dx \sum_{\gamma} c^{\gamma}(z, x) (L_{\gamma}[\phi^+](x) L_{\gamma}^*[\phi^-](x)),$$
(8.5)

and the resulting path integral in Equation (8.4) will be completely positive so long as $c^{\gamma}(z, x)$ is positive.

In Equation (8.4) \mathcal{I}_{CQ} determines the CQ interaction on each of the ket and bra paths and $\mathcal{I}_{C}(z, t_{i}, t_{f})$ is a purely classical action which takes real values. The above requirements on positive definiteness have been imposed for the path integral to converge. This condition also arose when studying path integrals associated with CQ master equations Chapter 6; for example, one can take the classical action \mathcal{I}_C to be the action associated with the path integral of the Fokker-Planck equation (8.16) [153, 87, 80] which must be positive (semi) definite in order for the path integral to converge. The term on the final line of Equation (8.4) contains cross terms between the bra and ket branches ϕ^+ , ϕ^- , which sends pure states to mixed states and corresponds to including additional noise in the dynamics. It takes the form of a Kraus map acting on the CQ state, which ensures complete positivity, and allows one to include classical-quantum Feynman-Vernon [79, 165] terms into the action.

If all the $c^{\gamma} = 0$, the ϕ^+ and ϕ^- integrals factorize in Equation (8.4), meaning the path integral preserves the purity of the quantum state conditioned on the classical trajectory. In this case, the absence of cross terms in the action, despite the requirement of Lindblad terms in the hybrid master equation, is a consequence of saturating the decoherence-diffusion trade-off. We shall primarily focus on this case; it can be shown that any CQ dynamics which does not preserve the purity of the quantum state conditioned on the classical degree of freedom can be embedded into a larger classical space where the quantum state remains pure, in a CQ version of purification [6].

It is useful to split \mathcal{I}_{CQ} into its real and imaginary components $\mathcal{I}_{CQ} = \mathcal{R}_{CQ} + i\mathcal{S}_Q$. Then Equation (8.4) (with $c_n = 0$) reads

$$\mathcal{I}^{\pm} = \mathcal{R}^+_{CQ} + \mathcal{R}^-_{CQ} + i(\mathcal{S}^+_{CQ} - \mathcal{S}^-_{CQ}) - \mathcal{I}_C, \qquad (8.6)$$

and we can get some intuition for each term. Heuristically expanding the actions, or more properly their Lagrangian's, in terms of their field dependence $S_{CQ} \sim \sum_m a_m(z)s_m(\phi)$ and $\mathcal{R}_{CQ} \sim \sum_m b_m(z)r_m(\phi)$ we see that

$$S_{CQ}^{+} - S_{CQ}^{-} \sim \sum_{m} a_{m}(z)(s_{m}(\phi^{+}) - s_{m}(\phi^{-}))$$

$$\mathcal{R}_{CQ}^{+} + \mathcal{R}_{CQ}^{-} \sim \sum_{m} b_{m}(z)(r_{m}(\phi^{+}) + r_{m}(\phi^{-})).$$
(8.7)

Hence, the imaginary part of the integral is associated with things like coherence, which depend on the difference between the ket and bra components of the density matrix. In contrast, the real part of the action depends on the sum of the left and right components of the density matrix, which are things like its expectation value. Moreover, conditioned on a classical trajectory $\bar{z}(t)$ - which can be represented by inserting a delta function $\delta(z(t) - \bar{z}(t))$ into the classical part of the path integral - we see that the evolution of the quantum state factorizes between the ϕ^{\pm} integrals and hence keeps pure quantum states pure.

The back-reaction of the quantum system on the classical one is contained in the real components of the CQ action \mathcal{R}_{CQ}^{\pm} . Indeed, when $\mathcal{R}_{CQ}^{\pm} = 0$, the path integral in Equation (8.6) reduces to the standard quantum path integral for the density matrix but also includes a classical variable which can undergo its own autonomous dynamics due to the inclusion of the classical action \mathcal{I}_C . However, whenever there is back-reaction, Equation (8.4) necessarily describes non-unitary evolution: we saw this in Chapter 4 using master equation methods.

To prove that the dynamics described by Equation (8.4) gives rise to consistent CQ dynamics, we must show that it leads to completely positive dynamics preserving the positivity of the CQ state.

Recall that positivity of the CQ state means that for any Hilbert space vector $|v(z)\rangle$ we have $\text{Tr} [|v(z)\rangle\langle v(z)|\varrho(z)] \ge 0$. In components, complete positivity is equivalent to asking that for any vector $|v(z)\rangle$ with components $v(\phi, z) = \langle \phi | v(z) \rangle$ we have

$$\int d\phi^+ d\phi^- v(\phi^+, z)^* \varrho(\phi^+, \phi^-, z) v(\phi^-, z) \ge 0.$$
(8.8)

A CQ dynamics Λ is positive if it preserves the positivity of CQ states and completely positive if $\mathbb{I} \otimes \Lambda$ is positive when we act with the identity on any larger system.

Since we assume the dynamics are time-local, we can perform a short-time expansion of the path integral. For the action in Equation (8.4) the path integral integrand always factorizes into the form

$$[e^{\mathcal{I}^{+}[\phi^{+},z]}(e^{\mathcal{I}^{-}[\phi^{-},z]})^{*} + \delta t \sum_{\gamma} c^{\gamma} (L_{\gamma}^{+} e^{\mathcal{I}^{+}[\phi^{+},z]}) (L_{\gamma}^{-} e^{\mathcal{I}^{-}[\phi^{-},z]})^{*}] e^{-\mathcal{I}_{C}[z]} + \dots,$$
(8.9)

Because Equation (8.9) consists of a sum of terms that factorize between \pm branches, even when higher order terms in the expansion are included, it is manifestly completely positive, which can be seen from the definition of complete positivity in Equation (8.8). Equation (8.9) is almost immediate from the form of path integral in Equation (8.4), and for completeness we show this in detail in Appendix E.1. It is important to note that because of the exponentials, Equation (8.9) is always strictly positive, meaning that we do not encounter zero norm states. Instead, the problem of negative norm states and ghosts is mapped to the problem of convergence of the path integral [200].

The path integral in Equation (8.4) is CP, but it is not always norm preserving. However, in the time-local case we can always normalize a CP map in a linear manner to arrive at a CP norm preserving dynamics.

To see this, first recall that in time-local classical dynamics a positive map

$$\frac{\partial p(z)}{\partial t} = \int dz' W(z|z') p(z') \tag{8.10}$$

is normalized by subtracting W(z)p(z), where $W(z) = \int dz' W(z'|z)$. Similarly, in time-local quantum dynamics, the CP dynamics

$$\frac{\partial \sigma}{\partial t} = \lambda^{\mu\nu} L_{\mu} \sigma L_{\nu}^{\dagger} \tag{8.11}$$

is normalized by subtracting the no-event term $\frac{\lambda^{\mu\nu}}{2} \{L_{\nu}^{\dagger}L_{\mu}, \sigma\}$ appearing in the Lindblad equation. For combined CQ dynamics, we know from Table 2.2 that any positive CQ map

$$\frac{\partial \varrho(z)}{\partial t} = \int dz' W^{\mu\nu} \left(z|z' \right) L_{\mu} \varrho \left(z' \right) L_{\nu}^{\dagger}$$
(8.12)

can be normalized by subtracting $\frac{1}{2} \int dz' W^{\mu\nu} (z'|z) \{L^{\dagger}_{\nu}L_{\mu}, \varrho(z)\}_{+}$ to yield CP norm preserving dynamics.

With this in mind, for time-local dynamics, Equation (8.4) can always be normalized and taking this into account we can include a normalization factor $I_{\mathcal{N}}[\phi^+, \phi^-, z, t_i, t_f]$ in the CQ path integral

$$\rho(z_f, \phi_f^+, \phi_f^-, t_f) = \int \mathcal{D}z \mathcal{D}\phi^+ \mathcal{D}\phi^- e^{\mathcal{I}[\phi^+, \phi^-, z, t_i, t_f] - I_{\mathcal{N}}[\phi^+, \phi^-, z, t_i, t_f]} \rho(z_i, \phi_i^+, \phi_i^-, t_i).$$
(8.13)

The path integrals studied in Chapter 6 where normalized since they correspond to CPTP master equations. In Section 8.3 we introduce a 'natural class' of path integrals that are based on a CQ proto action and are motivated by Chapter 6; these are the path integrals of interest in this thesis. In Section 8.3 we show that normalization can be guaranteed by including appropriate classical and quantum kinetic terms in the action. However, more generally, starting from Equation (8.4), finding a general closed form for the normalization function seems difficult. For this reason we leave an in-depth study of the normalization general CQ path integrals for

future work. Note that we do not expect that this changes the discussion in this chapter since we know that the normalization factor can always be found. Rather, there may be mathematical tricks which can be utilized to write the normalization factor in a neat and more expressive form.

8.2 Comparison to classical path integrals

The path integral action we introduce in Equation (8.4) is general. Therefore, finding CQ actions that give rise to dynamics with a sensible physical interpretation is useful. We saw examples in Chapter 6, and we will here study a simple example of the path integral associated with the Fokker-Plank equation

$$\frac{\partial p(z)}{\partial t} = -\frac{\partial}{\partial z_i} [D_{1,i}(z)p(z)] + \frac{\partial^2}{\partial z_i \partial z_j} [D_{2,ij}(z)p(z)], \qquad (8.14)$$

where $z = (q_1, p_1, \dots, q_n, p_n)$ for an *n* dimensional system [73] and p(z) is the classical probability density p(z)

In Equation (8.14), the coefficient $D_{1,i}$ characterizes the amount of drift in the system and is equal to the evolution of the expectation value of z, $\partial_t \langle z_i \rangle$. If D_{1,q_i} also depends on p_i , it contributes a friction term. The matrix $2D_{2,ij}$ characterizes the amount of diffusion in the system and characterizes $\partial_t \langle z_i z_j \rangle$. The corresponding path integral is given by [153, 154, 80, 87]

$$p(z,t_f) = \int \mathcal{D}z \ e^{-\mathcal{I}_C(z,t_i,t_f)} p(z_i,t_i), \qquad (8.15)$$

where

$$\mathcal{I}_C(z, t_i, t_f) = \frac{1}{4} \int_{t_i}^{t_f} dt \left[\frac{dz_i}{dt} - D_{1,i}(z) \right] D_{2,ij}^{-1} \left[\frac{dz_j}{dt} - D_{1,j}(z) \right].$$
(8.16)

The path integral has a natural interpretation in suppressing classical paths that deviate from their expected drift D_1 by an amount that depends on the inverse of the diffusion coefficient D_2^{-1} . If D_2 is z dependent, Equation (8.16) can also contain an anomalous contribution, arising from the z dependence of $D_2(z)$, as in Chapter 6, but we shall not include it here since (8.16) still defines positive classical dynamics.

The simplest non-trivial case is where one diffuses only in momenta. In this case, $\dot{q}_i = \frac{p_i}{m_i}$ and the momentum integral acts to enforce a delta function over $\delta(p_i - m_i \dot{q}_i)$. Integrating out the momentum variables, the result is a path integral over only the configuration space variables q_i with action

$$\mathcal{I}_C(q, t_i, t_f) = \frac{1}{4} \int_{t_i}^{t_f} dt [m_i \frac{d^2 q_i}{dt^2} - D_{1,i}(q)] D_{2,ij}^{-1} [m_j \frac{d^2 q_j}{dt^2} - D_{1,j}(q)],$$
(8.17)

from which we see that the path integral acts to suppress paths away from their expected equations of motion with the amount depending on D_2 .

Taking the expected classical equation of motion to itself be generated by an action S_C , the action in (8.17) can be re-written as

$$\mathcal{I}(q, t_i, t_f) = \frac{1}{4} \int_{t_i}^{t_f} dt \frac{\delta S_C}{\delta q_i} D_{2,ij}^{-1} \frac{\delta S_C}{\delta q_j}.$$
(8.18)

Since S_C itself appears in the path integral action $\mathcal{I}(q, t_i, t_f)$, we shall henceforth refer to S_C as the classical proto-action, as in Chapter 6. It is important to note that, generally, one can and should include non-Lagrangian friction terms in the path integral, represented by a more general drift coefficient, as in Equation (8.17).

8.3 A natural class of path integrals

The classical action in Equation (8.18) generalizes to the combined classical-quantum case. A natural class of configuration space path integrals are those derivable from a classical-quantum proto-action that is the sum of a quantum Lagrangian, a classical Lagrangian, and a CQ interaction term

$$W_{CQ}[q,\phi] = \int d^4x \mathcal{W}_{CQ}[q,\phi] = \int d^4x \mathcal{L}_Q[\phi] + \mathcal{L}_C[q] - V_{CQ}[q,\phi], \qquad (8.19)$$

$$\mathcal{I}(q,\phi^{-},\phi^{+},t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dx \bigg[i\mathcal{W}_{CQ}^{+}(x) - i\mathcal{W}_{CQ}^{-}(x) - \frac{1}{2} \frac{\delta \Delta W_{CQ}}{\delta q_{i}(x)} D_{0,ij}(q,x) \frac{\delta \Delta W_{CQ}}{\delta q_{j}(x)} - \frac{1}{4} \frac{\delta \bar{W}_{CQ}}{\delta q_{i}(x)} D_{2,ij}^{-1}(q,x) \frac{\delta \bar{W}_{CQ}}{\delta q_{j}(x)} \bigg],$$

$$(8.20)$$

where we denote the configuration space classical variable by q, and we take $D_0(q, x)$, $D_2(q, x)$ to be symmetric, positive semi-definite real matrices. We impose the matrix restriction $8D_0 \succeq D_2^{-1}$ to ensure the action takes the form of Equation (8.4) and hence is completely positive. We show this explicitly in Appendix E.2. When $8D_0 = D_2^{-1}$, the path integral preserves purity on the quantum system, as shown in Chapter 5 using unraveling methods. In Equation (8.20) $W_{CQ}[q,\phi]$ is a real classical-quantum proto-action which generates the dynamics, and we have use the notation $\bar{W}_{CQ} = \frac{1}{2}(W_{CQ}[q,\phi^+] + W_{CQ}[q,\phi^-])$ for the \pm averaged proto-action and $\Delta W_{CQ} = W_{CQ}[q,\phi^-] - W_{CQ}[q,\phi^+]$ for the difference in the proto-action along the \pm branches. As in the classical case, one can add friction terms to Equation (8.20) though we shall not do this in the present work. For simplicity, we here deal with theories with local correlation kernels, but we also expect our results to extend to the case where D_0, D_2 are positive semi-definite matrix kernels $D_0(x, y), D_2(x, y)$ which have some range [4].

This form of action is motivated by the study of path integrals in Chapter 6 for CQ master Equations whose back-reaction is generated by a Hamiltonian [60, 28, 6], as well as the purely classical path integral in Equation (8.18).

Written in the form of Equation (8.20), we see that the action of D_2 is to suppress paths that deviate from the \pm averaged Euler-Lagrange equations, which themselves follow from varying the bra-ket averaged proto-action \bar{W}_{CQ} , while the effect of the D_0 term is to decohere the quantum system. The decoherence diffusion trade-off $8D_0 \succeq D_2^{-1}$ [5, 4], required for the dynamics to be CP, means that if coherence is maintained for a long time, then there is necessarily lots of diffusion in the classical system away from its most likely path, with the amount depending on both D_0 and the strength of the coupling which enters in W_{CQ} .

One must further ensure that Equation (8.20) is normalized. When the CQ coupling does not involve higher derivative kinetic terms, it was shown in Chapter 6 that the dynamics generated by Equation (8.20) is normalized. In Appendix E.3, we extend this result and show that any CQ path integral of the form

$$I[q,\phi^{+},\phi^{-}] = \int dt i \dot{\phi}_{+}^{2} + iV(\phi^{+}) - i \dot{\phi}_{+}^{2} - iV(\phi^{-}) - \frac{D_{0}(q,\dot{q},\phi^{+})}{2} (\ddot{q} + f(q,\dot{q},\phi^{+}))^{2} - \frac{D_{0}(q,\dot{q},\phi^{-})}{2} (\ddot{q} + f(q,\dot{q},\phi^{-}))^{2},$$
(8.21)

is normalized. In the case $D_0 > 0$ has a functional dependence on the fields, one must make sure to also include a factor of $\sqrt{\det(D_0(q, \dot{q}, \phi))}$ in the path integral measure [153, 161]. We further show that any higher derivative CQ path integral of the form

$$\begin{split} I[q,\phi^{+},\phi^{-}] &= \int dt i \ddot{\phi}_{+}^{2} + i V(\phi^{+},\dot{\phi}^{+}) - i \ddot{\phi}_{+}^{2} - i V(\phi^{-},\dot{\phi}^{-}) \\ &- \frac{D_{0}(q,\dot{q},\phi^{+},\dot{\phi}^{+})}{2} (\ddot{q} + f(q,\dot{q},\phi^{+},\dot{\phi}^{+}))^{2} - \frac{D_{0}(q,\dot{q},\phi^{-},\dot{\phi}^{-})}{2} (\ddot{q} + f(q,\dot{q},\phi^{-},\dot{\phi}^{-}))^{2}, \end{split}$$

$$(8.22)$$

is also normalized up to constant factors. In the case $D_0 > 0$ has a functional dependence on the fields, one must again include a factor of $\sqrt{\det(D_0(q, \dot{q}, \phi, \dot{\phi}))}$ in the path integral measure [153, 161]. Note, if D_0 has a functional dependence on the fields, it is possible to re-exponentiate $\sqrt{\det(D_0)}$ through a Faddeev-Popov type action [201]

$$S[a,\bar{b},b] = \int dt D_0(a^2 + \bar{b}b), \qquad (8.23)$$

where a is bosonic and b, \bar{b} are anti-commuting Fermions. The integral over b, \bar{b} yields det (D_0) , whilst the integral over a yields $(\det(D_0))^{-1/2}$ [201]. We do not consider this explicitly here.

Equation's (8.21) and (8.22) are very generic type of action one gets through Equation's (8.20) when varying a CQ proto action that has second order equations of motion in the classical degree of freedom.

8.4 Lorentz invariant CQ dynamics

As a simple example, we can consider a classical field q(x) coupled to a quantum field $\phi(x)$ with a manifestly Lorentz invariant proto-action

$$W_{CQ} = \int d^4x \Big[\mathcal{L}_Q(\phi) - \frac{1}{2} \partial_\mu q \partial^\mu q - \frac{1}{2} m_q^2 q^2 - \frac{\lambda}{2} q^2 \phi^2 \Big].$$
(8.24)

In this case, assuming $8D_0 = D_2^{-1}$, we find the expressions for the CQ coupling terms

$$\frac{\delta \Delta W_{CQ}}{\delta q} D_0 \frac{\delta \Delta W_{CQ}}{\delta q} = \lambda^2 D_0 q^2 ((\phi^+)^2 - (\phi^-)^2)^2$$
(8.25)

$$\frac{\delta W_{CQ}}{\delta q} D_2^{-1} \frac{\delta W_{CQ}}{\delta q} = 4D_0 (\partial^\mu \partial_\mu q + m_q^2 q + \lambda q ((\phi^+)^2 + (\phi^-)^2))^2.$$
(8.26)

We see that Equation (8.25) acts to decohere the quantum system into the $|\phi\rangle$ basis by suppressing configurations away from $\phi^+ = \phi^-$ by an amount proportional to $D_0\lambda^2$, where λ characterizes the back-reaction on the quantum system. On the other hand, Equation (8.26) acts to suppress configurations away from their semi-classical equations of motion - found from varying $\frac{\delta \bar{W}_{CQ}}{\delta q}$ - by an amount also proportional to D_0 . Note that this does not depend on the coupling strength so that in the regime where the back-reaction is small, one can maintain coherence without deviating too much from the expected classical equations of motion. This can be used to evaluate CQ path integrals by working perturbatively in the back-reaction coupling (see Appendix D). Equation (8.25) takes the form of Equation (8.21) if one includes a quantum Lagrangian (such as the Klein-Gordon Lagrangian) and so defines normalized dynamics. Lorentz invariant or covariant pure Linbladians have been studied in [202, 203, 165].

As another example, in Equation (8.20), we could pick a proto-action based on the stressenergy tensor of the quantum matter $T_{\mu\nu}$. For example, by choosing

$$W_{CQ} = \int dx \mathcal{L}_Q(\phi) - \frac{1}{2} \partial_\mu q \partial^\mu q - \eta^{\mu\nu} T_{\mu\nu}(x) q(x), \qquad (8.27)$$

where $T = \eta^{\mu\nu}T_{\mu\nu}$ is the trace of the stress energy tensor. In this case, we find Lorentz invariant dynamics that causes decoherence of the quantum state according to the stress energy tensor

$$\mathcal{I}(\phi^{-},\phi^{+},q,t_{i},t_{f}) = \int_{t_{i}}^{t_{f}} dt d\vec{x} \bigg[i\mathcal{L}_{Q}^{+}(x) - i\mathcal{L}_{Q}^{-}(x) - \frac{\lambda^{2}D_{0}[q]}{2} (T^{-}(x) - T^{+}(x))^{2} - \frac{1}{4D_{2}[q]} \left(-\partial_{\mu}\partial^{\mu}q(x) + \frac{\lambda}{2} (T^{+}(x) + T^{-}(x)) \right)^{2} \bigg].$$
(8.28)

Such dynamics essentially amount to a Lorentz invariant collapse model, where in Equation (8.28), the collapse occurs dynamically due to interaction with a classical field. In particular, when the proto-action is based on the stress-energy tensor, there is an amplification mechanism by which states with small energy maintain coherence while macroscopic objects decohere; this is related to the amplification mechanism used in spontaneous collapse models [168, 169, 170, 171, 48, 172, 173]. Equation (8.28) contains higher derivative Lindblad operators due to the decoherence being according to the stress energy tensor. Hence, according to Equation (8.22), the dynamics will be normalized dynamics if a higher derivative quantum Lagrangian is included, which is indicative of the theory being an effective theory [15]. We leave the study of effective CQ theories to future work. We can further arrive at diffeomorphism invariant CQ dynamics by taking the CQ interaction potential W_{CQ} to be related to a gravitational action, which we now show.

8.5 Diffeomorphism invariant CQ gravity

Let us now comment on some of the consequences of classical-quantum theories of gravity. The goal is to attempt to construct a covariant classical-quantum dynamics theory that approximates Einstein's equations. Since in Equation (8.20) the paths away from $\frac{\delta}{\delta q_i}(\bar{W}_{CQ}[q, \phi^{\pm}])$ are exponentially suppressed by an amount depending on D_2^{-1} , the most likely path will be those for which

$$\frac{\delta}{\delta q_i}(\bar{W}_{CQ}[q,\phi^{\pm}]) \approx 0.$$
(8.29)

To get a theory that agrees with Einstein's gravity on average, we could therefore try to take $W_{CQ}[g,\phi]$ to be the sum of the Einstein Hilbert action $S_{EH}[g] = \frac{1}{16\pi G} \int \sqrt{g}R$ (in units where c = 1), and a matter action $S_m[g,\phi]$ including a cosmological constant. In the case where $W_{CQ} = S_{EH} + S_m$ we have

$$\frac{\delta}{\delta g_{\mu\nu}}(W_{CQ}[g,\phi]) = -\frac{\sqrt{-g}}{16\pi G_N}(G^{\mu\nu} - 8\pi G_N T^{\mu\nu}), \qquad (8.30)$$

Thus, paths would be exponentially suppressed away from (a \pm branch average of) Einstein's equations. Explicitly, taking the classical degree of freedom to be $g_{\mu\nu}$, the decoherence part of the CQ interaction in Equation (8.20) is given by

$$\frac{\delta\Delta W_{CQ}}{\delta g_{\mu\nu}} D_{0,\mu\nu\rho\sigma} \frac{\delta\Delta W_{CQ}}{\delta g_{\rho\sigma}} = \det(-g) \frac{1}{4} (T^{\mu\nu+} - T^{\mu\nu-}) D_{0,\mu\nu\rho\sigma} (T^{\rho\sigma+} - T^{\rho\sigma-}), \tag{8.31}$$

whilst (assuming $8D_0 = D_2^{-1}$) the diffusion part takes the form

$$\frac{\delta \bar{W}_{CQ}}{\delta g_{\mu\nu}} D_{2,\mu\nu\rho\sigma}^{-1} \frac{\delta \bar{W}_{CQ}}{\delta g_{\rho\sigma}} = \frac{1}{64\pi^2 G_N^2} \det(-g) (G^{\mu\nu} - 8\pi G_N \bar{T}^{\mu\nu}) D_{0,\mu\nu\rho\sigma} (G^{\rho\sigma} - 8\pi G_N \bar{T}^{\rho\sigma}).$$
(8.32)

The dynamics take the form of Equation (8.4); thus, the dynamics are completely positive, and the quantum state of the fields remains pure conditioned on the metric. The full action, without assuming the trade-off is saturated, takes the form

$$\mathcal{I}[\phi^{-},\phi^{+},g_{\mu\nu}] = \int dx \bigg[i\mathcal{L}_{Q}^{+} - i\mathcal{L}_{Q}^{-} - \frac{\det(-g)}{8} (T^{\mu\nu+} - T^{\mu\nu-}) D_{0,\mu\nu\rho\sigma} (T^{\rho\sigma+} - T^{\rho\sigma-}) \\ - \frac{\det(-g)}{1024\pi^{2}} (G^{\mu\nu} - \frac{1}{2} (8\pi (T^{\mu\nu})^{+} + 8\pi (T^{\mu\nu})^{-}) D_{2,\mu\nu\rho\sigma}^{-1} [g] (G^{\rho\sigma} - \frac{1}{2} (8\pi (T^{\rho\sigma})^{+} + 8\pi (T^{\rho\sigma})^{-}) \bigg],$$

$$(8.33)$$

where \mathcal{L}_Q is the quantum Lagrangian density. Just as for the Lorentz invariant theory that decoheres according to the stress energy tensor, Equation (8.33) contains higher derivative Lindblad operators through $T^{\mu\nu}$. Hence, according to Equation (8.22), to normalize the dynamics means that a higher derivative quantum Lagrangian should be included – this implies that such as theory may only be valid as an effective theory to be considered up to some energy scale. Because the path integral in Equation (8.33) contains both decoherence and diffusion, it does not suffer from the same pathologies as the standard semi-classical Einstein's equation's $G_{\mu\nu} = 8\pi G \langle T_{\mu\nu} \rangle$ [28, 6]. In particular, it includes the correlation between the matter and gravitational degrees of freedom via the CQ interaction term on the second line. For example, consider starting in an initial state describing a planet in a superposition of left and right $|L\rangle$, $|R\rangle$ states. The action of the decoherence term will be to enforce (in the Newtonian limit) that the quantum state decoheres into mass eigenstates – meaning that after the decoherence time, the planet will be found on either the left or the right. Because of the CQ interaction, paths where the quantum state decoheres into being on the left are correlated with the classical paths in which the gravitational field is sourced by a planet on the left $T_L^{\mu\nu}$, and similarly for paths which decohere to the planet being found on the right.

When the trade-off is saturated $8D_2 = D_2^{-1}$, the action is fully characterized by the tensor density $D_{0,\mu\nu\rho\sigma}$. There are two possible demands one could make on this tensor. The first would be to require that it be a positive semi-definite matrix in the sense that $v^{\mu\nu}D_{0,\mu\nu\rho\sigma}v^{\rho\sigma} \ge 0$ for any matrix $v^{\rho\sigma}$. This condition would ensure that the dynamics are completely positive and normalizable on any initial state, and classical paths close to Einstein's equations are more probable. Constructing diffeomorphism invariant classical-quantum theories of gravity then amounts to trying to find a tensor $D_0^{\mu\nu\rho\sigma}$, which gives rise to a path integral which defines completely-positive dynamics.

To meet this demand, the simplest thing one can try is to take $D_{0,\mu\nu\rho\sigma} = D_0 g^{-1/2} g_{\mu\nu} g_{\rho\sigma}$, in which case one finds a diffeomorphism invariant CQ theory of gravity in which paths deviating from the *trace* of Einstein's equations are suppressed, moreover, according to Equation (8.31) the quantum state decoheres into eigenstates of the trace of the stress-energy tensor. In the Newtonian limit, where the trace of the stress-energy tensor is dominated by its mass term, it decoheres the quantum state into mass eigenstates. This decoherence is again related to the amplification mechanism used in spontaneous collapse models. However, here the decoherence mechanism arises as a consequence of treating the gravitational field classically and imposing diffeomorphism invariance on the CQ action. Furthermore, although the quantum state decoheres, it remains pure if we condition it on the classical trajectory.

The equations we find demonstrate that a diffeomorphism invariant CQ theory of gravity

is possible, though it may only be valid as an effective theory. The challenge in constructing a complete theory is obtaining the transverse parts of the Einstein equation, which are the constraints, while ensuring the path integral over classical metrics remains negative definite so that the path integral converges. In Lorentzian signature, this does not appear possible within the current framework since it amounts to constructing a positive definite metric out of the metric tensor alone. One could instead choose a $D_{0,\mu\nu\rho\sigma}$, which is non-purely geometric, in which case it either introduces a preferred background or must be made dynamical. The former suggests an effective theory in which one obtains a classical metric by adding decoherence or tracing out degrees of freedom in some reference frame. In the latter case, one should add terms proportional to $D_{0,\mu\nu\rho\sigma}g^{\mu\nu}g^{\rho\sigma}$ and $D_{0,\mu\nu\rho\sigma}g^{\mu\sigma}g^{\nu\rho}$ into the classical part of the action. One then must ensure that such terms do not conflict with experimental bounds, which we explore in Chapter 10

Alternatively, we could relax the requirement that $D_{0,\mu\nu\rho\sigma}$ be positive semi-definite but merely require that the path integral be normalizable and preserve positivity on the physical degrees of freedom of a restricted class of initial states. It need not concern us if the negative eigenvalues of $D_{0,\mu\nu\rho\sigma}$ correspond to gauge degrees of freedom or if initial states violate the general constraints relativity evolve into distributions with negative probabilities. In this case, we could impose constraints conserved in the weak sense, similar to Chapter 7 and in loop quantum gravity [190].

As an example, we can consider a classical analogy via the time-independent Fokker-Plank equation in Equation (8.14). The stationary states $p_S(z)$ for the equation are given by

$$p_S(z) = \frac{C}{D_2(z)} \exp\left(\int^z dz' \frac{D_1(z')}{D_2(z')}\right).$$
(8.34)

Now we can consider the dynamics of Equation (8.14) with $D_2 \rightarrow -D_2$. This does not define positive dynamics on all classical states, which is represented by the fact that the Fokker-Plank path integral diverges. However, it does define positive dynamics on a subset of classical states. The most natural set of states to consider are the stationary states, which now read

$$p_S(z) = \frac{C}{D_2(z)} \exp\left(-\int^z dz' \frac{D_1(z')}{D_2(z')}\right).$$
(8.35)

If one starts off in a state of the form in Equation (8.35) the dynamics does not lead to inconsistent probabilities. In the combined classical-quantum case, we could therefore hope that imposing similar constraints on the state space, will lead to completely dynamics for onshell configurations.

With this in mind, a general form of ultra-local diffusion matrix is then proportional to the generalized Wheeler-deWitt metric in 3 + 1 dimensions

$$D_{0,\mu\nu\rho\sigma} = \frac{D_0}{2} (-g)^{-1/2} \left(g_{\mu\rho} g_{\nu\sigma} + g_{\mu\sigma} g_{\nu\rho} - 2\beta g_{\mu\nu} g_{\rho\sigma} \right), \qquad (8.36)$$

with D_0 a positive constant. Equation (8.36) is not a positive matrix, so does not define sensible dynamics acting on all CQ states. However, we would like to consider the case where we consider a constrained set of possible states. The situation is tenable in part because the different components in the CQ action represent both dynamical Fokker-Planck type terms and also components that act to enforce a diffusive, CQ version of the constraints of General Relativity. We posit that these have the form

$$\rho(C|\phi^+, \phi^-) = \frac{1}{N} e^{-\int \det(-g)C(x)dx},$$
(8.37)

with

$$C(x) := \frac{1}{2} \Delta C_{\mu} \sigma^{\mu\nu} \Delta C_{\nu} + \bar{C}_{\mu} \sigma^{\mu\nu} \bar{C}_{\nu}, \qquad (8.38)$$

$$\bar{C}_{\mu}(g,\phi^+,\phi^-) := \bar{T}_{\mu0} - \frac{G_{\mu0}}{8\pi G_N},\tag{8.39}$$

$$\Delta C_{\mu}(\phi^{+},\phi^{-}) := \left(T_{0\mu}^{+} - T_{0\mu}^{-}\right), \qquad (8.40)$$

with ϕ^{\pm} to be understood as a convenient notation for all fields over some initial Cauchy slice. We would like $\sigma^{\mu\nu}$ to be positive semi-definite so that classical states have probability distributions that are peaked around the constraints of general relativity. The dynamics would then suppress large violations of the constraints. The diffusion element $D_{0,}^{0000}$ associated with the Hamiltonian constraint is non-negative whenever $\beta \leq 1$.

If one chooses $\beta < 1/3$ [204, 205] then the sub-tensor $D_{0,ijkl}$ of spatial-indices is positive semi-definite. The elements of the diffusion matrix associated with the momentum constraint, as well as terms connecting the dynamical and constraint equations, depend on the choice of g^{0i} , which is usually associated with a gauge degree of freedom and the shift vector in the Hamiltonian formalism. This gives some plausibility to the conjecture that one can choose a diffusion tensor that is positive semi-definite on the true degrees of freedom. The part of the path integral corresponding to the constraints can also be viewed as telling us that initial distributions must be of the form corresponding to Equation (8.37). The constraints then define allowed distribution in terms of the matrix elements ϕ^+, ϕ^- . The full initial state $\rho(g, \phi^+, \phi^-)$ is given by further specifying a quantum state of the field $\rho(\phi^+, \phi^-)$, and a distribution of the metric g conditional on C, ϕ^+, ϕ^- i.e.

$$\rho(g,\phi^+,\phi^-) = \rho(g,\phi^+,\phi^-|C,\phi^+,\phi^-)\rho(C|\phi^+,\phi^-)\rho(\phi^+,\phi^-).$$
(8.41)

This state can be pure conditioned on the metric. We then want to restrict ourselves to initial states of the above form, which remain positive and normalizable under the action of the path integral. We leave as an open question whether evolution preserves the constraint in general. In Chapter 9, we study the Newtonian limit of CQ theories, which can be understood as a non-relativistic gauge fixed version of the theories introduced in this chapter. We arrive at a CQ version of the Newtonian constraint, which takes a form similar to Equation (8.37), and show that it is possible to preserve this constraint in time, meaning the path integral in Equation (8.33) gives rise to sensible completely positive on the subset of CQ states which satisfy the Newtonian gauge.

Alternatively, we could impose complete positivity and normalizability of the evolution of these initial states by only summing over paths for which this holds. We regard this as unsatisfactory.

It is also possible to consider a $D_{0,\mu\nu\rho\sigma}(x,x')$, a positive-definite kernel in space-time coordinates x, x' in which case one has stochastic processes which are correlated in space-time. The CQ interaction terms then take the form

$$-\frac{1}{2}\int dxdx' \frac{\delta\Delta W_{CQ}}{\delta g_{\mu\nu}(x)} D_{0,\mu\nu\rho\sigma}(x,x') \frac{\delta\Delta W_{CQ}}{\delta g_{\mu\nu}(x')} -\frac{1}{4}\int dxdx' \frac{\delta\bar{W}_{CQ}}{\delta g_{\mu\nu}(x)} D_{2,\mu\nu\rho\sigma}^{-1}(x,x') \frac{\delta\bar{W}_{CQ}}{\delta g_{\mu\nu}(x')}.$$
(8.42)

In this case, one is breaking the autonomous property of the dynamics since it is not timelocal, which is suggestive of an effective theory rather than a fundamental one.

8.6 Discussion

In this chapter, we have introduced a general path integral for classical-quantum dynamics, given by Equation (8.4), which opens up the way to study classical degrees of freedom coupled to quantum ones via path integral methods. The path integral provides an approach to study covariant theories of classical fields coupled to quantum ones. We have given an explicit example of a Lorentz invariant CQ theory and discussed potential applications to classical-quantum theories of gravity.

In particular, we have arrived at a diffeomorphism invariant theory of CQ gravity - summarized by Equations (8.31), (8.32) - which acts to suppress paths that deviate from the trace of Einstein's equations, while simultaneously decohering the quantum system according to the trace of the stress-energy tensor. This provides a first example of diffeomorphism invariant classical-quantum dynamics and, more generally, is a first example of diffeomorphism invariant collapse dynamics [168, 169, 170, 171, 48, 172, 173], where the loss of coherence is a derived consequence of the interaction of a quantum system with a classical dynamical variable. We have also proposed a theory reproducing all of Einstein's equations as a limiting case. The theory is diffeomorphism invariant, but we have not proven that the dynamics preserve the proposed constraints: this may necessitate the further study of the theory's constraint algebroid, a project we initiated in Chapter 7.

The theories introduced in this chapter violate several assumptions which went into the scalar field model considered in Chapter 7. Firstly, the dynamics in this chapter are motivated by the continuous master equation, which has a different Lindblad structure than the jumping models; the jumping models are generally associated with path integrals containing an infinite number of couplings. Secondly, Equations (8.31), (8.32) are higher order in the stress-energy tensor, they contain products ~ T^2 , and these higher order terms were not considered in Chapter 7. Thirdly, the dynamics is diffusive in the configuration space variable $g_{\mu\nu}$ and hence in the 3-metric γ_{ij} , while the model considered in Chapter 7 was assumed to undergo deterministic classical dynamics. The danger with the model introduced in this chapter is that we have not shown that the constraints will be preserved in time. Furthermore, if the standard constraints of GR are violated too much, this will leave the vacuum state unstable [206]. The fact that large deviations from the standard constraints can be suppressed via the $D_0^{0\mu,0\nu}$ terms suggests

that these effects can be made small, but this and other experimental checks need to be further explored.

Let us conclude by discussing some further possible extensions of the present work. Firstly, exploring the renormalization properties of classical-quantum dynamics would be interesting, which we have not touched upon here. Since the resulting classical-quantum action is essentially indistinguishable from a standard quantum field theory, we expect that similar methods could be used. Though effective theories can be non-renormalizable, the renormalizability of CQ dynamics has important foundational consequences for theories with a fundamentally classical field. For the classical-quantum theories generated by a proto-action (8.20), the classical part of the dynamics is generated by a higher derivative kinetic term. For example, in the Lorentz invariant dynamics of Equation (8.32), the propagator associated with the higher derivative term is given by $(\partial_{\mu}\partial^{\mu}q)^2$, which we expect to scale like $\sim \frac{1}{(p^2)^2}$ for large momentum; this appears to significantly help with the renormalization properties of any purely classical terms, as well as those associated with back-reaction. However, the gravitational action of Equation's (8.31) and (8.32) are not power counting renormalizable due to the terms which are quadratic in the stress-energy tensor, though, as noted in [165, 174], one must be careful with power counting renormalization when considering the density matrix path integral.

Secondly, we have approached CQ dynamics starting from the description of a system in terms of classical and variables and writing down dynamics which leads to consistent evolution. As an effective theory, it would be interesting to arrive at classical-quantum theories from a top-down approach. That is, starting from a quantum-quantum system, we should be able to arrive at an effective CQ description. We expect this occurs via some decoherence mechanism on one of the systems and is closely related to the quantum to classical transition [207, 208, 209, 210].

We have here given a general construction by which one can write down CQ path integrals that uphold space-time and gauge symmetries. It would be worthwhile to explore this further with concrete examples, and to study the mapping between covariant CQ path integrals and master equations in detail. One open question is to determine what the symmetries generators are, since they should necessarily be altered in a non-unitary theory [211]. In particular, for Lorentz invariant theories, we expect the symmetry generators should form a CQ generalization of the Lorentz algebra. In Appendix K, we make some progress towards this, by studying the symmetry generators in a simple example of a Lorentz invariant open quantum system. For classical-quantum gauge theories, which could be useful in an effective theory of light-matter interactions when there is classical back-reaction, the killing form provides a natural choice for D_0 since for a compact lie group, the killing form is positive semi-definite [212]. In the context of gravity, the theory presented here could be regarded as a theory with a fundamentally classical gravitational field. However, we expect it to be useful as an effective theory of semi-classical gravitational physics when back-reaction is involved. If such a theory were renormalizable, it would form a consistent way of coupling classical and quantum gravity with a sensible UV limit. Such a theory necessarily deviates from quantum mechanics; when there is back-reaction on the gravitational field, the dynamics are no longer unitary, and the coupling necessarily induces collapse of the wavefunction due to the decoherence term parameterized by D_0 in Equation (8.20), while there must also be diffusion in the gravitational field which leads to experimentally testable signatures of a classical gravitational field which we explore in Chapter 10 [4].

Chapter 9

The Newtonian limit of classical-quantum dynamics

This chapter studies the non-relativistic of a classical Newtonian potential interacting with quantum matter. This limit can be viewed as a gauge fixed version of both [28] and the theory introduced in Chapter 8, where we only consider the scalar degrees of freedom to linear order. Our results generalize previous discussions of Newtonian classical-quantum gravity, mainly studied using continuous measurement and feedback approaches [60, 49, 124, 52, 213].

Our goal will be to study the non-relativistic limit of general classical-quantum theories of gravity to provide a template for theorists and experimentalists to develop and test CQ theories. In particular, in Chapter 10, we generalize the decoherence-diffusion trade-off discussed so far in this thesis (Chapter 4) to obtain bounds relating the decoherence rate and diffusion of any CQ theory, given in terms of the strength of the back-reaction. By studying the non-relativistic limit, we can understand the predictions of CQ theories in the Newtonian regime. Indeed, we find a generic prediction of CQ theories: the Newtonian potential diffuses away from its classical solution by an amount that depends on the decoherence rate into mass eigenstates. The decoherence-diffusion trade-off, therefore, provides a way of testing CQ theories: one lower bounds the amount of diffusion the theory must have from coherence experiments, which can then be tested by measuring the noise in precision mass experiments. We explore this in detail in Chapter 10.

In this chapter, we consider the general case of CQ Newtonian gravity by reducing the degrees of freedom to scalar perturbations. In this "bottom-up" approach, we assume that the relevant dynamical degrees of freedom are scalar perturbations of the metric, i.e., the Newtonian potential. We shall also allow for vector perturbations of the shift vector at higher order in c, which we find are necessary to construct consistent dynamics. We impose phenomenological constraints on the dynamics and ask that Newton's equation for the gravitational field is satisfied on expectation. With these assumptions, we can construct the Newtonian limit of CQ theories via a reduction - even without a complete theory. In particular, by first identifying the relevant degree of freedom as the Newtonian potential, we use the master equation and unraveling formalism to construct CQ dynamics parameterized by the moments D_0, D_2 appearing in the dynamics. We verify our findings by showing that the Newtonian limit we derived agrees with the Newtonian limit of the theory introduced in Chapter 9.

This chapter is based on upcoming work [9], which is work done in collaboration with Jonathan Oppenheim and Andrea Russo.

9.1 Newtonian limit of classical GR

In this section, we study the Newtonian limit of classical general relativity (GR), which motivates our study of the Newtonian limit of classical-quantum theories of gravity. By the Newtonian limit, we mean the linearised expansion of the metric around a flat Minkowski background, where the $c \to \infty$ limit is taken, discarding terms higher order in c.

The Newtonian limit of GR is represented by a non-dynamical scalar perturbation of flat Minkowski spacetime expressed through the metric:

$$ds^{2} = -c^{2}\left(1 + \frac{2\Phi}{c^{2}}\right)dt^{2} + \left(1 - \frac{2\Phi}{c^{2}}\right)\delta_{ij}dx^{i}dx^{j},$$
(9.1)

where Φ satisfies the gravitational Poisson equation. The usual derivation of this limit arises from a gauge fixing of the full Einstein theory. There, one starts with a generalized scalar-vector tensor perturbation of the metric in the form [214] of

$$ds^{2} = -c^{2} \left(1 + \frac{2\Phi}{c^{2}}\right) dt^{2} + \frac{w_{i}}{c} (dt dx^{i} + dx^{i} dt) + \left[\left(1 - \frac{2\psi}{c^{2}}\right)\delta_{ij} + \frac{2s_{ij}}{c^{2}}\right] dx^{i} dx^{j}, \qquad (9.2)$$

where $\partial_i w^i = \partial_i s^{ij} = 0$ and takes the infinite *c* limit of Einstein's equations. When the stressenergy tensor is chosen to represent a pressureless dust distribution, or a point particle, only one non-dynamical scalar perturbation Φ remains at the end, and it is constrained to obey the gravitational Poisson's equation.

We instead present a derivation of the Newtonian limit by directly reducing the degrees of freedom to scalar perturbations. In the reduced degrees of freedom approach, we *first* assume that the relevant physical degrees of freedom are scalar perturbations. We shall also allow for vector perturbations of the shift vector at higher order in c, which we find are necessary to construct a consistent CQ theory. In the classical case, we see we can set these to zero. This allows us to construct the Newtonian $c \to \infty$ limit of CQ theories by considering only scalar perturbations of the gravitational degrees of freedom - even without a complete theory, though with assumptions on its completion.

9.1.1 Newtonian limit via a reduced action

We now arrive at the Newtonian limit of GR via a reduced Hamiltonian. We take as a starting point the linearised Einstein Hilbert Lagrangian density, which is equivalent to the Fierz-Pauli action [215] for the metric perturbation $g_{\mu\nu} = \eta_{\mu\nu} + h_{\mu\nu}$

$$S[h_{\mu\nu}] = \frac{c^4}{16\pi G} \int d^4x \, \mathcal{L}(h_{\mu\nu}), \tag{9.3}$$

$$\mathcal{L}(h_{\mu\nu}) = -\frac{1}{2}\partial_{\mu}h^{\mu\nu}\partial_{\nu}h + \frac{1}{2}\partial_{\mu}h^{\rho\sigma}\partial_{\rho}h^{\mu}_{\sigma} - \frac{1}{4}\eta^{\mu\nu}\partial_{\mu}h^{\rho\sigma}\partial_{\nu}h_{\rho\sigma} + \frac{1}{4}\eta^{\mu\nu}\partial_{\mu}h\partial_{\nu}h.$$
(9.4)

We are interested in constructing CQ dynamics for a Newtonian theory, so we further make a Newtonian approximation of the metric. We take the ADM decomposition of the metric

$$ds^{2} = -(Nc dt)^{2} + g_{ij} (dx^{i} + N^{i}c dt) (dx^{j} + N^{j}c dt), \qquad (9.5)$$

and we assume that

$$N = \left(1 + \frac{\Phi}{c^2}\right), \ N^i = \left(0 + \frac{n^i}{c^3}\right), \ g_{ij} = \left(1 - \frac{2\psi}{c^2}\right)\delta_{ij}.$$
(9.6)

The extra factor of c in the choice of shift-vector is related to the fact that, classically, the h_{0i} component occurs at a higher order than the h_{00} , h_{ij} components [214]. Since in the ADM formalism, the physical degrees of freedom are the spatial metric g_{ij} and its conjugate momenta,

assuming that $g_{ij} = \left(1 - \frac{2\psi}{c^2}\right) \delta_{ij}$ amounts to assuming the physical degrees of freedom are scalar perturbations of the metric. We assume that all fields vanish at infinity. In the purely classical case, we find that when the stress-energy tensor $T_{0i} = 0$, then $n^i = 0$, but we will show that in the combined CQ case $n^i \neq 0$ even in the absence of the stress-energy tensor. Instead, this correction is required to preserve the theory's Hamiltonian constraint.

With the gauge fixing of Equation (9.6), the linearized action in Equation (9.3) is

$$S = \frac{1}{8\pi G} \int d^4x \left[-\frac{3\dot{\psi}^2}{c^2} + \frac{\partial_i n^i}{2c^2} (\dot{\Phi} - \dot{\psi}) - \frac{\dot{n}^i}{2c^2} (\partial_i \Phi + 3\partial_i \psi) - \frac{1}{4c^2} \partial_i n^j \partial_j n^i + \partial_i \psi \partial^i \psi - 2\partial_i \Phi \partial^i \psi + \frac{1}{4c^2} \partial_i n^j \partial_j n^i \right].$$
(9.7)

To go to the Hamiltonian picture, we first calculate the functional derivatives with respect to $\dot{\psi}$, $\dot{\phi}$ and \dot{n}^i to find the conjugate momenta

$$\pi_{\psi} = -\frac{1}{16c^2 G \pi} (12\dot{\psi} + \partial_i n^i), \ \pi_{\Phi} = \frac{\partial_i n^i}{16\pi c^2 G}, \ \pi_i = -\frac{1}{16\pi c^2 G} (\partial_i \Phi + 3\partial_i \psi).$$
(9.8)

Equation (9.8) defines two primary constraints given by

$$\Pi_{\Phi} = \pi_{\Phi} - \frac{\partial_i n^i}{16\pi c^2 G} \approx 0, \tag{9.9}$$

$$\Pi_i = \pi_i + \frac{1}{16\pi c^2 G} (\partial_i \Phi + 3\partial_i \psi) \approx 0.$$
(9.10)

In the $c \to \infty$ limit, Equations (9.9) and (9.10) become the constraints $\pi_{\Phi}, \pi_i \approx 0$, which enforce the Hamiltonian and momentum constraints.

One might worry that the constraints appearing in Equations (9.9) and (9.10) appear to be different concerning the constraints one obtains by first considering the full ADM formalism constraints and then linearising them. This difference has been studied in [216], where it was shown that the two forms are related by a canonical transformation. Alternatively, one could follow the approach of [106] and add a specific non-covariant term to the linearised action of Equation (9.3). The additional term vanishes on-shell and simplifies the primary constraints to match them with those derived from the ADM formalism. Since we are interested in the $c \to \infty$ limit, these distinctions do not matter, as we end up imposing the constraints $\pi_{\Phi}, \pi_i \approx 0$, which are equivalent to the primary constraints $\pi_N, \pi_{N_i} \approx 0$, where N, N^i are the lapse and shift vectors. Using the definitions of conjugate momenta in Equation (9.8), and working to leading order in c, we arrive at the Newtonian Hamiltonian

$$H = \int d^3x \left[-\frac{2\pi Gc^2}{3} \pi_{\psi}^2 - \frac{1}{12} \pi_{\psi} \partial_i n^i + \frac{\partial_i \psi \partial_i \Phi}{4\pi G} - \frac{\partial_i \psi \partial_i \psi}{8\pi G} + \lambda_{\Phi} \pi_{\Phi} + \lambda^i \pi_i \right].$$
(9.11)

To find the Newtonian interaction Hamiltonian, we need to couple gravity to matter. We shall consider the matter distribution to be that of a particle with mass density m(x), in which case to leading order in c only T_{00} contributes to the gravitational equations. The corresponding interaction Hamiltonian can then be written as

$$H_I = \int d^3x \Phi(x) m(x). \tag{9.12}$$

The total Hamiltonian is given by $H_{tot} = H + H_I$:

$$H_{tot} = H_m \int d^3x \left[-\frac{2\pi Gc^2}{3} \pi_{\psi}^2 - \frac{1}{12} \pi_{\psi} \partial_i n^i + \frac{\partial_i \psi \partial_i \Phi}{4\pi G} - \frac{\partial_i \psi \partial_i \psi}{8\pi G} + \lambda_{\Phi} \pi_{\Phi} + \lambda^i \pi_i + \Phi(x) m(x) \right],$$
(9.13)

where the dynamics associated to H_{tot} is given by:

$$\dot{\psi} = -\frac{4G\pi c^2 \pi_{\psi}}{3} - \frac{1}{12}\partial_i n^i, \ \dot{\pi}_{\psi} = \frac{\nabla^2 (\Phi - \psi)}{4\pi G}, \ \dot{\Phi} = \lambda_{\Phi}, \ \dot{\pi}_{\Phi} = \frac{\nabla^2 \Phi}{4\pi G} - m, \ \dot{n}^i = \lambda^i, \ \dot{\pi}_i = -\frac{1}{12}\partial_i \pi_{\psi}.$$
(9.14)

We arrive at the Newtonian limit by imposing the constraints $\pi_i, \pi_{\Phi} \approx 0$ and solving Equation (9.14). Note the constraint $\pi_{\Phi} \approx 0$ imposes

$$\frac{\nabla^2 \Phi}{4\pi G} - m \approx 0 \quad \Rightarrow \quad \Phi(t, x) = -G \int d^3 x' \, \frac{m(x)}{|x - x'|},\tag{9.15}$$

on the potential Φ i.e., Φ must solve Poisson's equation. On the other hand, the constraint $\pi_i \approx 0$ imposes $\pi_{\psi} \approx 0$, where we have used the fact that π_{ψ} vanishes at infinity. Preservation of the $\pi_{\psi} \approx 0$ constraint imposes that $\Phi = \psi$. Moreover, the time derivative of the Newtonian potential directly dictates the Lagrange multiplier via $\lambda_{\Phi} = \dot{\Phi}$ and the divergence part of the shift vector via $\dot{\Phi} = -\frac{1}{12}\partial_i n^i$.

Since we assume a stationary source, where only T_{00} contributes, $\partial_i n^i = 0$. This equation only partially fixes the shift n^i since related by different choices of the shift vector will be gauge equivalent. In the classical theory, it is common to assume the gauge $n^i = 0$, in which case we arrive at the Newtonian metric of Equation (9.1), where Φ satisfies Poisson's equation. We have arrived at the Newtonian limit of general relativity by making the Newtonian approximation on the metric in Equation (9.6) and then deriving the dynamics in the $c \rightarrow \infty$ limit. While deriving the Newtonian limit from a full GR approach requires a complete diffeomorphism invariant theory, we have seen that we can construct a consistent reduced theory by first identifying the correct degrees of freedom (in this case, scalar perturbations of the metric) and then writing down their dynamics according to a reduced Hamiltonian.

9.1.2 A stochastic classical analog of the CQ theory

In the CQ case, we will construct the Newtonian limit by assuming the relevant degrees of freedom are scalar perturbations of the metric of the form in Equation (9.6) and then considering a reduced CQ master equation governing the dynamics of the perturbations. Since we will be interested in describing the non-relativistic limit of a quantum mass interacting with classical gravity, the back-reaction on the gravitational field from the quantum matter is dominated by the T_{00} component. Any classical-quantum momentum constraint should be unchanged since it does not involve matter. In particular, the back-reaction of the quantum system on the classical system enters through π_{Φ} in Equation (9.14), through the action of the interaction Hamiltonian $H_I = \int d^3x \Phi(x)m(x)$. Because quantum back-reaction must necessarily involve diffusion, in the CQ case, the equation of motion for π_{Φ} will be modified to include a stochastic term.

To gain some intuition, we can consider the classical analog of the CQ theory by considering a Langevin equation for $\dot{\pi}_{\Phi}$

$$\dot{\pi}_{\Phi} = \frac{\nabla^2 \Phi}{4\pi G} - m - \sigma \xi, \qquad (9.16)$$

where $\sigma(x)$ is a coefficient and $\xi(t, x)$ is a white noise process

$$\mathbb{E}[d\xi(x)] = 0, \quad \mathbb{E}[\xi(t,x)\xi(t',y)] = \delta(t,t')\delta(x,y)dt.$$
(9.17)

Note, multiplying Equation (9.16) by dt, the equation takes the same form as the stochastic unravelings studied in Chapter 5, but with $\langle m \rangle \to m$ since we consider the classical analog of the back-reaction. We have also chosen to define $\sigma \to -\sigma$ so that the noise term in Equation (9.16) can be interpreted as a random contribution to the mass term.

With the modified dynamics for π_{Φ} , we find the constraint $\pi_{\Phi} \approx 0$ imposes the stochastic

Newtonian constraint

$$\nabla^2 \Phi = 4\pi G(m + \sigma\xi) \tag{9.18}$$

on the potential Φ .

The full set of dynamics then takes the form

$$\dot{\psi} = -\frac{4G\pi c^2 \pi_{\psi}}{3} - \frac{1}{12}\partial_i n^i, \ \dot{\pi}_{\psi} = \frac{\nabla^2 (\Phi - \psi)}{4\pi G}, \ \dot{\Phi} = \lambda_{\Phi}, \ \dot{\pi}_{\Phi} = \frac{\nabla^2 \Phi}{4\pi G} - m - \sigma\xi, \tag{9.19}$$

$$\dot{n}^i = \lambda^i, \ \dot{\pi}_i = -\frac{1}{12}\partial_i \pi_\psi. \tag{9.20}$$

Since the back-reaction is in T^{00} , the momentum constraint π_i remains unchanged from the deterministic case. Its preservation imposes the constraint $\pi_i \approx 0$, which further imposes the constraint $\Phi = \psi$. The constraint $\pi_{\Phi} \approx 0$ gives the Newtonian constraint. However, with the addition of the noise, the Newtonian potential is no longer stationary but instead solves the randomly sourced Poisson equation

$$\Phi = -G \int dx' \, \frac{m(x') + \sigma(x', t)\xi(x', t)}{|x - x'|}.$$
(9.21)

Equation (9.21) then determines λ_{Φ} , and using that $\Phi = \psi$, also determines $\partial_i n^i$ and λ^i through the Equation

$$\dot{\psi} = -\frac{1}{12}\partial_i n^i. \tag{9.22}$$

In particular, with the gauge choice given by Equation (9.6), we see that n^i is required for the theory to be consistent. The presence of diffusion in the equation of motion for π_{Φ} makes the Newtonian potential $\psi = \Phi$ fluctuate, and we see the shift vector $\partial_i n^i$ is required to account for this fluctuation. Had we not included it, we would have found an inconsistent set of equations since $\dot{\psi}$ would have been vanishing through Equation (9.22). However, the stochastic constraint requires it to fluctuate.

We point out once again that this does not fix the shift n^i uniquely since we are free to add a divergenceless term and get the same solution to the equation of motion. Moreover, in a complete calculation, we expect that contributions from T_{0i} will also determine the components n^i without affecting the Newtonian contribution, given by the h_{00} component. Regardless, the set of Equation's (9.19) are together consistent, and performing higher-order calculations is beyond the scope of the current work.

9.1.3 A simplified starting point

In the stochastic case, we still find that the dynamics set $\Phi = \psi$. Hence, one can instead start with the metric perturbation:

$$N = \left(1 + \frac{\Phi}{c^2}\right), \ N^i = \left(0 + \frac{n^i}{c^3}\right), \ g_{ij} = \left(1 - \frac{2\Phi}{c^2}\right)\delta_{ij}.$$
(9.23)

and consider the dynamics obtained by setting $\Phi = \psi$ in Equation (9.11). One can also remove the kinetic term $-\frac{2\pi Gc^2}{3}\pi_{\psi}^2$, which does not contribute to the equations of motion on the constraint surface, and the Lagrange multiplier involving π_{Φ} , which also vanishes on the constraint surface. In this case, the Hamiltonian reads

$$H + H_I = \int d^3x \left[\frac{(\nabla \Phi)^2}{8\pi G} + m\Phi - \frac{1}{12} \pi_{\Phi} \partial_i n^i \right], \qquad (9.24)$$

subject to the constraint $\pi_{\Phi} \approx 0$. Because Equation's (9.23) and (9.24) are considerably simpler than Equation's (9.6) and (9.11) but result in the same dynamics, we will use the Hamiltonian in Equation (9.24) to describe the Newtonian limit of CQ theories.

Since the noise process is white noise, technically, the metric perturbation will describe a probability measure, and we should use it to compute averaged quantities; this is true of both Φ and the shift vector n^i , which are now both stochastic quantities. In particular, averaging over a timescale ΔT and length scale ΔL , $\Delta L/\Delta T \ll c$, we have that $\frac{\Delta \Phi}{\Delta T} \sim \frac{n^i}{\Delta L}$, so that $\frac{\Delta L\Delta\Phi}{\Delta T} \sim n^i \ll c^3$ which verifies our initial assumption to include the perturbation h_{0i} as $\frac{n^i}{c^3}$ in Equation (9.6).

With this in mind, we now study the Newtonian limit of the full CQ theory. In the $c \to \infty$ limit, we arrive at Poisson's equation on average. However, because of the CQ interaction, the Newtonian limit also predicts diffusion around this solution according to Equation (9.21), with simultaneous decoherence on the quantum system.

Before discussing how a quantum system's back-reaction on the classical Newtonian field is implemented through diffusion processes, we would like to comment on the choice of gauge. The end goal of this chapter is to formulate the Newtonian limit of gravity for CQ-hybrid theories; we are still determining if a complete CQ theory can be made fully diffeomorphism invariant in a way that also accounts for the constraints of the theory. Regardless, our choice of gauge is motivated by the need to preserve the gravitational constraints. By choosing the gauge as in Equation (9.6), we know that we have a way of consistently selecting trajectories that stay on the constraint surface, where the conjugate momenta vanish as described in this section. We verify our findings by showing that with the gauge fixing described by Equation (9.6), the theory introduced in Chapter 8 agrees with the Newtonian limit we find using a bottom-up approach.

9.2 Hamiltonian CQ dynamics reproducing the Newtonian limit

Having discussed in detail the Newtonian limit of GR, we are now in a position to discuss how to write down classical-quantum theories that give rise to the Newtonian interaction on average. Before discussing the specifics of continuous and discrete master equations, we shall outline the general procedure and assumptions.

Assumption 1. We assume that the evolution of the combined classical-quantum system undergoes autonomous CQ dynamics.

We expect this assumption to hold if CQ is treated as a fundamental theory. However, as an effective theory, this assumption may break down since the dynamics can be non-Markovian. We comment on the differences between a fundamental and effective theory in Section 9.8 (see also [6]).

Assumption 2. We take a bottom-up approach and assume that in the weak field $c \to \infty$, the appropriate gravitational degrees of freedom are the perturbations of the metric in the form of Equation (9.23).

In particular, the leading order contribution which governs the geodesics of test particles is described by $g_{00} = (1 + \frac{\Phi}{c^2})$.

Assumption 3. We take the variables $(\Phi(t), \pi_{\Phi}(t), n^{i}(t))$ to be classical stochastic variables coupled to a stochastic quantum state $\rho(t)$. We assume that the purely classical part of the evolution is generated by the reduced Hamiltonian (9.24), that the interaction between classical and quantum degrees of freedom is Hamiltonian and that it is governed by the reduced interaction Hamiltonian in Equation (9.12), where the constraints $\pi_{\Phi}(t) \approx 0$ should also be imposed. Specifically, we require that the first moment $D_{1,\pi_{\Phi}}$ is picked to reproduce the Newtonian back-reaction on average:

$$\operatorname{Tr}\left[\{H_{I}, \varrho\}\right] = \int d^{3}x \,\operatorname{Tr}\left[m(x)\frac{\delta\rho}{\delta\pi_{\Phi}(x)}\right],\tag{9.25}$$

so that the dynamics are approximately Hamiltonian on average.

Since the back-reaction of the quantum system on the classical system is associated with T_{00} , we expect the CQ momentum constraint to be unchanged from its classical counterpart, as it is not associated with any back-reaction. One caveat, however, to keep in mind is that relativistic corrections at high energy may affect the low-energy behavior of the theory. In particular, general relativity has yet to be tested at distances shorter than the millimeter scale. Here we assume it holds to arbitrarily short distances.

As a consequence of Assumption 3, we know from the decoherence-diffusion trade-off that there must be diffusion in the classical variable and also Lindbladian evolution on the quantum state. In particular, in the next chapter we generalize the decoherence-diffusion trade-off found for the continuous master Equations in Chapter 4 to show that for any completely positive autonomous CQ master equation, the moments $D_n^{\mu\nu}$ appearing in the master equation of Equation (2.91) must satisfy $2D_{2,\pi\Phi\pi\Phi} \succeq D_{1,\pi\Phi}^{br} D_0^{-1} (D_{1,\pi\Phi}^{br})^{\dagger}$, where $D_{2,\pi\Phi\pi\Phi}$ is the full matrix of diffusion coefficients $D_{2,\phi\Phi\pi\Phi}^{\mu\nu}$, $D_{1,\pi\Phi}^{br} = D_{1,\pi\Phi}^{\mu\alpha}$ is the matrix describing the drift of the back-reaction of the quantum system and $(D_0^{-1})_{\alpha\beta}$ is the generalized inverse of the Lindbladian coefficient $D_0^{\alpha\beta}$.

Assumption 4. In this chapter, we will take the coefficients D_n entering the master equation to be minimally coupled, by which we mean they depend only on the Newtonian potential Φ , $D_n(\Phi)$ and not their conjugate momenta π_{Φ} .

This assumption is motivated by the fact that in Einstein's gravity, the mass density couples to the Newtonian potential, not its conjugate momenta, and we are imposing the constraint that $\pi_{\Phi} \approx 0$. Nonetheless, one could generalize the master equations to the non-minimally coupled case by considering couplings $D_n(\Phi) \to D_n(\Phi, \pi_{\Phi})$ in all of the equations.

Note, Assumptions 1-4 violate those given in Chapter 7 and enable us to find constraints preserved in time. In particular, Assumption 2 amounts to a choice of gauge, while the theories in Chapter 7 were assumed to be gauge independent. It may be that no consistent gauge invariant theory exists, which gives rise to the full Einstein's equations, but we can find gauge fixed dynamics that give rise to its Newtonian limit. More importantly, via Assumption 3, we let the Newtonian potential be stochastic, which amounts to letting γ_{ij} be a stochastic variable. Conversely, in Chapter 7, we assumed deterministic classical dynamics for the configuration space variable γ_{ij} . Here, we find that stochasticity in the Newtonian potential is required to find consistent constraints that close, perhaps shedding light on why the theory considered in Chapter 7 fell short.

We now consider the dynamics consistent with our assumptions. We discuss the case of continuous back-reaction in detail since, in this case, our assumptions fully determine the dynamics up to a choice of Lindbladian strength. We then discuss how one can use the decoherencediffusion trade-off to reason about the dynamics in the case of jumping master equations.

9.3 Continuous gravitational back-reaction

In Chapter 4 [5], it was shown that there are two classes of CQ master equations, and the general form of the continuous master equation was found to be Equation (4.13). We also found the unraveling representation of such master equations in Chapter 5, characterized only by the couplings D_0, σ, D_1 , with $\sigma\sigma^T = 2D_2$. In this case, our assumptions on Hamiltonian back-reaction are enough to specify the combined classical-quantum system's dynamics fully. They take the form of the Hamiltonian unraveling of Chapter 5 (see also Appendix C for a discussion of unravelings with fields)

$$d\Phi = -\frac{1}{12}\partial_{i}n^{i}dt, d\pi_{\Phi} = \frac{\nabla^{2}\Phi}{4\pi G}dt - \langle m(x)\rangle dt - \int d^{3}y \,\sigma(\Phi; x, y)dW(y), d\rho(t) = -i[H_{m} + H_{I}, \rho]dt + \frac{1}{2}\int d^{3}y D_{0}(\Phi; x, y) \left([m(x), [\rho, m(y)]]\right)dt + \frac{1}{2}\int d^{3}y \,\sigma^{-1}(\Phi; x, y) \left(m(x)\rho + \rho \,m(x) - 2\rho\langle m(x)\rangle\right)dW(y).$$
(9.26)

In Equation (9.26) H_m is a purely quantum Hamiltonian, m(x) is the quantum mass density operator, $\langle O \rangle$ is the expectation value of the normalized quantum state ρ , Tr [ρO] and $W_i(x)$ is a Wiener process in space-time satisfying

$$\mathbb{E}[dW(x)] = 0, \quad \mathbb{E}[dW(x)dW(y)] = \delta(x,y)dt. \tag{9.27}$$

For notational simplicity, we will suppress the dependence of the couplings D_0, D_2 on the Newtonian potential and write $D_0(x, y), D_2(x, y)$. The Lindbladian term, characterized by D_0 , and the diffusion term D_2 are required for the back-reaction to be completely positive, which can be seen from the decoherence diffusion trade-off [4] apparent in Equation (4.13).

In Equation (9.26), σ is related to the diffusion coefficient D_2 appearing in the master equation via

$$2D_2(\Phi; x, y) = \int dw \ \sigma(\Phi; x, w) \sigma(\Phi; w, y).$$
(9.28)

and σ^{-1} the generalized inverse of σ .

The choice of the possible master equation is therefore fully constrained up to the functional choices of the couplings $D_0(\Phi; x, y)$, $D_2(\Phi; x, y)$. Both $D_2(\Phi; x, y)$, $D_2(\Phi; x, y)$ are required to be positive kernels, where a positive kernel f(x, y) is a kernel such that $\int dx dy a^*(x) f(x, y) a(y) \ge 0$ for any function a(x). They are also constrained to satisfy the decoherence diffusion trade-off:

$$8D_2 \succeq D_0^{-1},$$
 (9.29)

where Equation (9.29) is to be understood as a matrix kernel equation:

$$\int dxdy \ a(x)^* \big[8D_2(\Phi; x, y) - D_0^{-1}(\Phi; x, y) \big] a(y) \ge 0, \tag{9.30}$$

which must hold for an arbitrary function a(x). We give example kernels that satisfy the decoherence-diffusion trade-off in Table 9.3.

9.3.1 Imposing the Newtonian constraint

To arrive at the classical-quantum version of Poisson's equation, we must impose the constraint $\pi_{\Phi} \approx 0$ according to the Hamiltonian in Equation (9.24). We impose the constraint on the level of trajectories.¹ In practice, what we are really doing is choosing n^i stochastically such

¹This is conceptually very similar to what is done in quantum theory when constraints are imposed via a path integral approach, where one associates to each path a measure given by the action, then selects only the paths which satisfy the constraint. Take for example, a Hamiltonian with $H(q,p) = H_0(q,p) + \lambda C(q,p)$. The phase space partition function for the theory is represented by $\mathcal{Z} = \int \mathcal{D}q \mathcal{D}p \mathcal{D}\lambda e^{\frac{i}{\hbar} \int dt [\dot{q}p - H(q,p) - \lambda C(q,p)]}$. Since the Hamiltonian is linear in λ , the path integral over the Lagrange multiplier in λ enforces a delta function over

Master Equation	Decoherence $D_0^{\alpha\beta}(x,y)$	Diffusion $D_{2,\alpha\beta}(x,y)$
Continuous (ultra-local)	$D_0^{\alpha\beta}(x,y) = D_0^{\alpha\beta}\delta(x,y)$	$D_{2,\alpha\beta}(x,y) = \frac{1}{8}(D_0^{-1})_{\alpha\beta}\delta(x,y)$
Continuous (Gaussian)	$D_0^{\alpha\beta}(x,y) = D_0^{\alpha\beta}g_{\mathcal{N}(x,y)}$	$D_{2,\alpha,\beta}(x,y) = \frac{(D_0^{-1})_{\alpha\beta}}{8} F(x,y) g_{\mathcal{N}(x,y)}$
Continuous (D.P)	$D_0^{\alpha\beta}(x,y) = \frac{D_0^{\alpha\beta}}{ x-y }$	$D_{2,\alpha\beta}(x,y) = \frac{1}{8} \frac{(D_0^{-1})_{\alpha\beta}}{4\pi} \nabla_y^2(\delta(x,y))$

Table 9.1: Possible choices of kernels for the continuous master equations and the resulting diffusion/decoherence coefficients, assumed to saturate the trade-off in Equation (9.29). In the table, D.P stands for the Diosi-Penrose kernel, $g_{\mathcal{N}(x,y)}$ is a normalized Gaussian distribution, and $F(x,y) = \prod_{i=1}^{d} \sum_{n=0}^{\infty} c_n(r_0) H_{2n}(\frac{x_i - y_i}{r_0})$ where $c_n(r_0) = \frac{(-1)^n (r_0)^{2n} n!}{2^n}$, d is the spatial dimension and H_{2n} denote the Hermite polynomials [217]. In particular $g_{\mathcal{N}}^{-1}(x,y) = F(x,y)g_{\mathcal{N}}(x,y)$.

that $\dot{\pi}_{\Phi} \approx 0$. Doing so turns Φ_t into a white noise variable with values given by the solution of Equation (9.31). However, naively replacing all occurrences of Φ_t with its solution in terms of an Itô white noise variable, in particular, that which appears in H^I , does not lead one to a CPTP dynamics. Before the constraints are imposed, the dynamics of Φ_t are continuous, and thus any back reaction from the quantum matter on Φ_t only returns to affect the quantum matter degrees of freedom at later times. To ensure that this time-ordering is maintained even in the limit that Φ_t no longer evolves continuously, one must be careful to ensure the action of H^I occurs after the other stochastic terms [49]. One possible way to ensure this is to first write the unravelling of the density matrix in the Stratonovich form and then impose the constraint that turns Φ_t , and hence H^I , into white noise. This allows us to correctly rewrite the unravelling such that when converting back into the Itô formalism, we pick up an extra decoherence term given by the backreaction of the stochastic gravitational field and allows us to get rid of the non-linear evolution term picked up from the solution of the noise Poisson equation.² This then gives the final form of unravelling in the Newtonian limit:

$$\frac{\nabla^2 \Phi_t}{4\pi G} = \langle m(x) \rangle + \int d^3 y \,\sigma(\Phi_t; x, y) \xi_t(y),$$
(9.31)
$$d\rho_t = -i \left[H_m + V_m, \rho_t\right] dt - i \int d^3 x \, d^3 y \, d^3 y' \left[-G \,\frac{m(x)\sigma(\Phi_t, y, y')}{|x - y|}, \rho_t\right] dW_t(y')$$

$$+ \frac{1}{2} \int d^3 x \, d^3 y \, D_0(\Phi_t; x, y) \left([m(x), [\rho_t, m(y)]]\right) dt$$

$$+ \frac{1}{2} \int d^3 x \, d^3 y \, d^3 y' \left[\sigma(\Phi_t; x, y), [\rho_t, \sigma(\Phi_t; x, y')]\right] dt$$

$$+ \frac{1}{2} \int d^3 x \, d^3 y \, \sigma^{-1}(\Phi_t; x, y) \left(m(x)\rho_t + \rho_t \, m(x) - 2\rho_t \langle m(x) \rangle\right) dW_t(y),$$
(9.31)
(9.31)

where $\xi_t(x) = \frac{dW_t(x)}{dt}$ is the formal definition of white noise, and

$$V_{m} = -\frac{G}{2} \int d^{3}x \, d^{3}y \frac{m(x)m(y)}{|x-y|},$$

$$\sigma(\Phi_{t}; x, y) = -G \int d^{3}y'' \frac{m(x)\sigma(\Phi_{t}, y, y'')}{|x-y|}.$$
(9.33)

 $\delta(C(q,p))$ so that the partition function reads $\mathcal{Z} = \int \mathcal{D}q \mathcal{D}p \delta(C(q,p)) e^{\frac{i}{\hbar} \int dt [\dot{q}p - H(q,p)]}$ which can be interpreted as summing over all paths with weight $e^{\frac{i}{\hbar} \int dt [\dot{q}p - H(q,p)]}$ and then selecting only those that satisfy the constraint C(q(t), p(t)) = 0.

²In the present context, this extra decoherence term was discovered by Isaac Layton

These equations were first written down by Tilloy and Diósi in [49] and their derivation from a fundamental theory is a central result of this chapter. The Newtonian limit of CQ theories is described by a Newtonian potential diffusing around Poisson's equation by an amount defined by D_2 , while simultaneously, the density matrix decoheres into the mass eigenbasis by an amount determined by the Lindbladian coefficient D_0 .

The details of the functional dependence of σ and D_0 on Φ_t have been left unspecified. When they are independent of Φ_t the equations coincide with those of a continuous measurement of a quantum mass, where the measurement outcome is used to source the Newtonian potential, as given by Equation (24) of [49]. In this case, the additional decoherence term appearing in Equation (9.26) proportional to $\sigma^2 \sim D_0^{-1}$ was studied in [124], where the kernels σ , D_0 leading to the minimal amount of decoherence where found. On the other hand, one can consider the couplings to be a general Markovian functional of Φ_t . This will generically lead to additional terms, as was observed with H^I above, which may not preserve the CPTP property of the dynamics. Exploring the details of these functional dependencies is an interesting question which we leave open for future work.

The spatial dependence of the couplings have also been left unspecified. The simplest example of Equation (9.31) is found when the couplings D_0, D_2 are taken to be *ultra-local*, which we take to mean that $D_0, D_2 \sim \delta(x, y)$, and also saturate Equation (9.29). For the case of ultra-local couplings, Equation (9.31) has an equivalent path integral description and we can arrive at the Newtonian limit by considering a gauge fixed Newtonian limit of the covariant theory introduced in Chapter 8. We show this in Section 9.4.

9.3.2 Linearity of the dynamics and white noise

Though the dynamics of Equation (9.31) gives rise to completely positive dynamics, this is surprising at first glance since the expectation value of the quantum state directly sources the Newtonian potential. This is in contrast to the unravellings in Chapter 5, where the change in the classical variable dz was sourced by the expectation value of a quantum operator multiplied dt.

For example, suppose that Alice entangles the state of a mass in the left and right with the

state of a massless qubit, preparing the entangled state

$$|\psi\rangle_{AB} = \frac{1}{\sqrt{2}} (|0\rangle|L\rangle + |1\rangle|R\rangle).$$
(9.34)

Then naively, it appears this could be used to violate the linearity of quantum theory.

In particular, in the state $|\psi\rangle_{AB}$, the Newtonian potential has an average around $\mathbb{E}[\nabla^2 \Phi] = \frac{1}{2}(m_L + m_R)$, with a probability distribution $u_0(\Phi)$. Now consider if Alice measures the $|0\rangle, |1\rangle$ basis. After the measurement, with probability half, the state collapses to $|0\rangle|L\rangle$, and with probability half, it collapses to $|1\rangle|L\rangle$. Thus after the measurement, the Newtonian potential will be sourced either by m_L , with a probability distribution $u_1(\Phi)$ with $\mathbb{E}[\nabla^2 \Phi] = m_L$, or by m_R , with a probability distribution $u_2(\Phi)$ with $\mathbb{E}[\nabla^2 \Phi] = m_R$. At first glance, this appears to be in contradiction with the linearity of the dynamics; the initial reduced density matrix $u_0(\Phi)\mathbb{I}$ is mapped to $u_0(\Phi)\mathbb{I} \to \frac{1}{2}u_1(\Phi)|L\rangle\langle L| + \frac{1}{2}u_2(\Phi)|R\rangle\langle R|$.

The resolution to this puzzle has to do with the properties of white noise, which is the timederivative of a Wiener process. Recall that Wiener increments δW are Gaussian distributed with variance δt

$$P(\delta W) \sim \exp\left(-\frac{(\delta W)^2}{2\delta t}\right).$$
 (9.35)

Since the Newtonian potential undergoes a white noise process, the signal $\alpha = \delta t \nabla^2 \Phi$, $\alpha = \langle m \rangle \delta t + \delta W$ is a Gaussian random variable with mean $\langle m \rangle \delta t$. Hence α is distributed according to

$$P(\alpha) \sim \exp\left(-\frac{(\alpha - \langle m \rangle \delta t)^2}{2\delta t}\right).$$
 (9.36)

Consequently, the distribution for Φ at any given time is given by

$$P(\Phi) \sim \exp\left(-\frac{\delta t (\nabla^2 \Phi - \langle m \rangle)^2}{2}\right).$$
 (9.37)

Equation (9.37) describes a wildly stochastic process with variance $\frac{1}{\delta t}$. In particular, in the $\delta t \to 0$ limit, one learns nothing about the distribution of the Newtonian potential, and the distributions $p_0(\Phi)$, $u_1(\Phi)$, $u_2(\Phi)$ all coincide. The timescale to learn about the distribution depends on the strength of the ratio between the diffusion coefficient σ and the back-reaction, which determines the strength of the collapse of the continuous measurement. In this sense, linearity is restored because by the time one could - in theory - learn about Alice's entangled measurement, the reduced density matrix on the local quantum system will have decohered due

to the appearance of the Lindbladian evolution in the mass operators, which act to continuously measure the mass density.

9.4 Deriving the Newtonian limit as a gauge fixing of a complete theory

In this section, we show that we can arrive at the Newtonian limit of (9.31) by taking the Newtonian limit of the theory introduced in Equation (8.33) of Chapter 8. This acts as a sanity check, both for the Newtonian limit of Equation (9.31) and that the theory introduced in the previous Chapter has constraints with a sensible Newtonian limit. In the Newtonian limit, keeping only the highest order terms in c, we find that the problematic off-diagonal terms appearing in Equation (8.33) disappear. In other words, we show the dynamics of Equation (8.33) defines CP dynamics on the subset of states defined by the Newtonian limit. We leave it as a question for further work whether the CQ constraints would be preserved in the more general case, and in particular, if the dynamics of Equation (8.33) lead to stable dynamics which preserves the Newtonian limit once higher order terms are c are considered.

We start by noting that for ultra-local couplings, Equation (9.31) has an equivalent description in terms of the classical-quantum path integral

$$\rho(t_f, \Phi_f, m_f^+, m_f^-) = \int \mathcal{D}\Phi \mathcal{D}m^{\pm} \exp\left[\int_{t_i}^{t_f} dt d^3x \left(i\mathcal{L}_Q[m(x)^+, \Phi] - i\mathcal{L}_Q[m(x)^-, \Phi]\right) - \frac{D_0[\Phi]}{2} \left(m(x)^+ - m(x)^-\right)^2 - \frac{1}{4D_2[\Phi]} \left(\frac{\nabla^2 \Phi}{4\pi G} - \frac{1}{2} \left(m(x)^+ + m(x)^-\right)\right)^2\right] \rho(t_i, \Phi_i, m_i^+, m_i^-),$$
(9.38)

where $\rho(t, \Phi, m^+, m^-) = \langle m^+ | \varrho(t, \Phi) | m^- \rangle$ denote the components of the CQ state in the mass eigenstate basis, and a boundary condition is imposed on $\Phi(t_f) = \Phi_f$. We show this in detail in [9] but this formula can also be derived by performing a weak continuous measurement on the mass and using the measurement outcome to source Φ [49]. Equation (9.38) describes an integral over paths of the classical Newtonian potentials and a doubled path integral over the quantum mass eigenstates m^{\pm} . The Newtonian constraint is encoded via a combination of non-dynamical terms in the path integral (i.e., they do not contain time derivatives). This path integral contains precisely the constraints we posited in Chapter 8. We now show that Equation (9.38) can be understood as a gauge fixed version of the full theory introduced in Chapter 8. For simplicity, we assume that the decoherence trade-off has been saturated. Consider the full theory of Equation (8.33), which, when coupled to a quantum mass density, has an action of the form

$$\mathcal{I}[m^{+}, m^{-}, g_{\mu\nu}] = \int dx \bigg[i\mathcal{L}_{Q}^{+} - i\mathcal{L}_{Q}^{-} - \frac{\det(-g)}{8} (T^{\mu\nu+} - T^{\mu\nu-}) D_{0,\mu\nu\rho\sigma} (T^{\rho\sigma+} - T^{\rho\sigma-}) - \frac{\det(-g)c^{8}}{128G^{2}\pi^{2}} (G^{\mu\nu} - \frac{4\pi G}{c^{4}} ((T^{\mu\nu+} + T^{\mu\nu-}) D_{0,\mu\nu\rho\sigma}[g] (G^{\rho\sigma} - \frac{4\pi G}{c^{4}} (T^{\rho\sigma+} + 8\pi T^{\rho\sigma-}) \bigg],$$

$$(9.39)$$

where \mathcal{L}_Q is the quantum Lagrangian density.

For simplicity, we take D_0, D_2 to saturate the trade-off. We take the couplings to be ultralocal

$$D_{0,\mu\nu\rho\sigma} = \frac{D_0}{2c^5} (-g)^{-1/2} \left(g_{\mu\rho} g_{\nu\sigma} + g_{\mu\sigma} g_{\nu\rho} - 2\beta g_{\mu\nu} g_{\rho\sigma} \right), \qquad (9.40)$$

where the factor of $\frac{1}{c^5}$ is to ensure that D_0 has the same units as that appearing in Equation (9.38). To obtain the Newtonian limit, we write the path integral in an ADM form, described by summing over all paths N, N^i, γ and considering the action as a functional of the variables appearing in the ADM decomposition $\mathcal{I}[m^+, m^-, N, \vec{N}, \gamma]$. The Newtonian limit can then be understood as a Gauge fixing of the complete theory, computing the transition amplitudes between the CQ states defined on surfaces Σ_t

$$\rho(t_f, m_f^+, m_f^-, \gamma_f) = \int \mathcal{D}\gamma dN d\vec{N} \mathcal{D}m^+ \mathcal{D}m^- \delta(\gamma_{ij} - (1 - \frac{2\Phi}{c^2}\delta_{ij})\delta(N - (1 + \frac{\Phi}{c^2}))\delta(N^i - \frac{n^i}{c^3})) \times e^{\mathcal{I}[m^+, m^-, N, \vec{N}, \gamma]} \rho(t_i, m_i^+, m_i^-, \gamma_i).$$
(9.41)

We now substitute for the Newtonian gauge by performing the delta functional integrals. In particular, we have $g_{00} = c^2 N = (c^2 + \Phi)$, whilst $g_{ij} \sim (1 - \frac{2\Phi}{c^2})\delta_{ij}$ and $g_{0i} = \frac{n^i}{c^3}$. The components of the Einstein tensor are calculated as

$$G^{00} = \frac{2\nabla^2 \Phi}{c^4},\tag{9.42}$$

$$G_{0i} = -\frac{2}{c_{5}^{5}}\partial_{0}\partial_{i}\Phi + \frac{1}{2c^{5}}\nabla^{2}n_{i}, \qquad (9.43)$$

$$G_{ij} = -\frac{2}{c^4} \partial_t \partial_t \Phi. \tag{9.44}$$

Similarly, noting that $det(-g) = c^2$, we see that due to the powers of c, the de-Witt metric is dominated by its 0000 component, which to leading order is given by

$$D_{0,0000} = D_0 c^3 (1 - \beta). \tag{9.45}$$

Keeping only terms leading order in c, the path integral action in Equation (9.39) is dominated by terms only involving D_{0000} and reduces to

$$\mathcal{I}[m^+, m^-, \Phi] = \int dx \left[i\mathcal{L}_Q^+ - i\mathcal{L}_Q^- - \frac{D_0(1-\beta)}{8} (m^+ - m^-)^2 - \frac{D_0(1-\beta)}{2} (\frac{\nabla^2 \Phi}{4\pi G} - \frac{1}{2} (m^+ + m^-))^2 \right]$$
(9.46)

Redefining $D'_0 = \frac{D_0(1-\beta)}{4}$ gives the same path integral as we found for the general Newtonian limit derived from Equation (9.38).

Hence, provided $\beta \leq 1$, we find the dynamics $c \to \infty$ limit of the full theory described by Equation (9.39) gives rise to complete positive dynamics, which describe a randomly sourced Poisson's equation, with associated decoherence on the quantum state. This provides a sanity check for the Newtonian limit derived in Equation (9.31) and gives rise to the hope that the theory in Chapter 8 has constraints that are preserved in time. However, we note that here we have arrived at the Newtonian limit by a gauge fixing of the full theory after neglecting all the terms higher order in c. Because we neglected terms higher order in c, we have eliminated the potential violating terms in the full path integral (9.39), which are not positive semi-definite. Specifically, the violating terms involving D_{000i} , D_{0i0i} still arise, but they are higher order in c; what we have shown the dynamics of Equation (9.39) defines CP dynamics on the subset of states defined by the Newtonian limit. We have not shown that the dynamics are consistent away from this limit, for example they could be unstable for finite c, and we leave this as a question for future research. A more general consideration is essential since we have not shown that the complete theory preserves the form of the Newtonian limit. It could be that by including higher-order terms in the calculation, one causes it to deviate from it in some way; for example, in deriving the Newtonian limit, we have assumed that Poisson's equation holds, on average, at any scale. By including higher-order terms, one might find that it is altered in the low-energy regime.
9.5 Comparison to previous classical-quantum theories

This section compares the Newtonian limit derived in the previous section to previously considered classical-quantum theories.

The Equations described via (9.26) have previously been studied in [49]: when the couplings σ and D_2 are independent of Φ they describe the same equations as a continuous measurement of a quantum mass, where the measurement outcome is used to source the Newtonian potential. Here we have shown that such dynamics arise on general grounds for all continuous classicalquantum theories of gravity, which are completely positive and agree with the Newtonian limit on expectation. As described in [49], because the Newtonian potential is sourced by a white noise process, one must be careful when implementing the quantum state evolution in Equation (9.31) since the couplings can depend on the Newtonian potential. In implementing the evolution when the couplings σ and D_2 are dependent on Φ , one can then be led to extra decoherence on the quantum state ρ due to Itô calculus involving multiplicative terms in Φ , potentially leading to faster decoherence rates than is implied by D_0 alone. However, studying this in detail is technically challenging and beyond the scope of the current work.

A variation of Equation (9.26) was also studied in [55], in a theory of strongly incoherent gravity. This model can be understood similarly to [49]. However, instead of performing a series of frequent weak measurements described by a finite coupling σ , one considers a series of strong measurements that happen infrequently and probabilistically. When the measurement dynamics are not occurring, the evolution of the quantum state is unitary, and there is no backreaction on the gravitational field: it is stationary. Conversely, when the strong measurement occurs, this can be understood as taking the parameter $\sigma \to 0$, in which case $\nabla^2 \Phi = 4\pi G \langle m \rangle$, and the quantum state instantaneously collapses into its mass eigenstates.

In Equation's (9.26) and (9.31), we have taken the drift to be local in x. At the same time, we allow for the decoherence and diffusion terms to have some range. The interaction law between the classical and quantum systems is still local in this case, but non-local correlations can be created [218]. Importantly, if the Lindbladian coupling D_0 has some range, then even though the CQ interaction is local, the master equation can, in principle, generate entanglement between two spatially separated quantum systems via the Lindbladian coefficient. However, this effect is likely to be small. If the Lindbladian coupling in Equation (9.26) D_0 is ultra-local, the dynamics do not generate entanglement between spatially separated regions, meaning that the models with local couplings parameterize the general form of the continuous master equation which would be ruled out by gravity induced entanglement (GIE) experiments. We give three examples of kernels D_0 , D_2 for continuous master equations in Table 9.3. It should be noted that with an appropriate choice of D_0 , the pure Lindbladian evolution appearing in (9.26) can be taken to resemble the Lindbladian part of spontaneous collapse models [219, 168, 169, 170, 171] except that, in our model, there is no need to think about any ad-hoc field, nor think of the collapse as a physical process. Rather, one necessarily gets decoherence of the wave function for free via gravitationally induced decoherence [220, 49, 124, 28].

Models with ultra-local couplings form perhaps the most natural class of CQ dynamics. However, in Chapter 10, we show that considerations of the decoherence diffusion trade-off have already ruled out non-relativistic versions of these models. In other words, (in line with assumptions 1-4) classical-quantum theories, which have continuous gravitational degrees of freedom with local interactions and correlations, are already ruled out by experiment. Note, this does not necessarily rule out the theory of Equation (8.33) described in Chapter 8 since we have not shown that higher order corrections can change the Newtonian limit at small distance scales. Rather, we have shown that assuming 1-4 hold on the CQ state, the dynamics of Equation (8.33) give rise to a consistent Newtonian limit with the same dynamics of Equation (9.31).

Specifically, one can constrain CQ theories via Equation (9.31) and the decoherence diffusion trade-off in Equation (9.29). The Newtonian limit of CQ gravity predicts diffusion of the Newtonian potential by an amount depending on D_2 . This can be upper bounded by precision mass experiments, which precisely measure the acceleration of particles. Conversely, coherence and heating experiments can be used to upper bound the inverse Lindbladian coefficient D_0^{-1} , which gives a lower bound on D_2 via the decoherence diffusion trade-off. Hence, when combined, it is possible to get an experimental squeeze on CQ theories [4].

9.6 Decoherence rates

For completeness, we briefly discuss the decoherence rates which follow from Equation (9.31), since they will be used in Chapter 10. We shall compute the decoherence rate for a quantum mass initially in a partially decohered superposition of state $|L\rangle$ and $|R\rangle$. We describe the quantum state using creation and annihilation operators $\psi(x), \psi^{\dagger}(x)$ on a Fock space, related to the usual momentum-based Fock operators as $\psi(x) = \int dp e^{i\vec{p}\cdot\vec{x}}a_{\vec{p}}$. The mass density operator is defined via $m(x) = m\psi^{\dagger}(x)\psi(x)$, where m is the mass of the particle. We assume that the state remains well approximated by a state with a fixed particle number. The superposition can be taken to be distributions centered around $x = x_L$ and $x = x_R$ with total mass m, i.e., for a one-particle state we could take $|L/R\rangle = \int d^3x f_{L/R}(x)\psi^{\dagger}(x)|0\rangle$. We will take them to be well separated so that $f_L(x)f_R(x) \approx 0$. With this orthogonality condition, the result of the equation's quantum part is to decohere the quantum state by an amount D_0 . We can compute the off-diagonal elements of Equation (9.31) to see this. We shall assume that the range of $D_0(x, y)$ is smaller than the length scale entering the superposition, so that $D_0(x, y)m_L(x)m_R(y) \approx 0$, and ignore the effects of the pure Hamiltonian evolution – though these can lead to additional decoherence [49]. The result is that $\rho_{LR} = \langle L|\rho|R\rangle$ undergoes dynamics according to

$$\frac{d\rho_{LR}}{dt} \approx -\frac{1}{2} \int d^3x d^3y D_0(\Phi; x, y) (m_L(x)m_L(y) + m_R(x)m_R(y))\rho_{LR}.$$
(9.47)

In particular, we see leading order off-diagonals decay exponentially at a rate determined by

$$\lambda = \frac{1}{2} \int d^3x d^3y D_0(\Phi; x, y) (m_L(x)m_L(y) + m_R(x)m_R(y)).$$
(9.48)

In practice, one does not have access to the entire history of the Newtonian potential. So one should also integrate out the Newtonian potential to find the true decoherence of the quantum state. This calculation is generally complicated since the couplings $D_0(\Phi)$, $D_2(\Phi)$ can depend on the Newtonian potential. Nonetheless, Equation (9.47) gives a lower bound for the decoherence of the quantum state if one has full knowledge of the trajectories in the unraveling. Alternatively, in the presence of a background Newtonian potential, such as that of the Earth's, and assuming polynomial dependence of the D_0 on the Newtonian potential, then we to leading order we can approximate D_0 by its background value, which gives Equation (9.47) as the decoherence rate. We use Equation (9.47) in Chapter 10 to constrain CQ theories of gravity experimentally.

As an example of a decoherence kernel, we can take $D_0(x, x')$ to be the *Diosi-Penrose kernel* (D.P kernel) defined via $D_0(x, y) = \frac{D_0}{|x-y|}$. In this case, the off-diagonals decay exponentially with a rate proportional to the Diosi-Penrose decoherence rate

$$\lambda = \frac{1}{2} \int d^3x d^3y \frac{D_0}{|x-y|} (m_L(x)m_L(y) + m_R(x)m_R(y)).$$
(9.49)

For identical spherical distributions of radius R and total mass M, and mass density ρ , Equation (9.49) can be re-written as

$$\lambda = -\frac{1}{G} \int d^3x D_0 \Phi(x) \rho(x), \qquad (9.50)$$

which is proportional to the average gravitational self-energy of the mass distribution $\lambda = \frac{3D_0M^2}{5R}$.

For a composite particle of mass M, made up of N constituents each of radius R, the mass density will be represented by a sum over all of the particles $m(x) = \sum_{i} m_i(x)$. The decoherence rate, in this case, is given by

$$\lambda = \frac{1}{2} \int d^3x d^3y \frac{D_0}{|x-y|} (\sum_{i,j} m_{L,i}(x) m_{L,j}(y) + \sum_{i,j} m_{R,i}(x) m_{R,j}(y)).$$
(9.51)

Since the cross terms involving i, j are suppressed by a factor of inter-atomic scales to leading order, the contribution to the decoherence rate is lower bounded by the i = j component of the sums in Equation (9.51), which gives an extra factor of N relative to the single-particle case $\lambda = \frac{3D_0NM^2}{5R}$.

Similarly, we can calculate the decoherence rate for ultra-local couplings $D_0(\Phi; x, y) = D_0(\Phi)(x)\delta(x, y)$. We find that the decoherence rate is calculated as

$$\lambda = \frac{D_0 M^2}{V},\tag{9.52}$$

where V is the volume of the particle. For N composite particles, the decoherence rate is additive in N and found to be $\lambda = \frac{ND_0M^2}{V}$.

9.7 Newtonian limit for general master equations

So far in this chapter, we have only considered the Newtonian limit of theories with continuous back-reaction. This section considers the more general case, which includes when the master equation and back-reaction are associated with jumps in the Newtonian potential.

As a consequence of Assumption 3, we know from the decoherence-diffusion trade-off that there must be diffusion in the classical variable and also Lindbladian evolution on the quantum state. As mentioned, in the next chapter, we generalize the decoherence-diffusion trade-off found for the continuous master Equations in Chapter 4 to show that for any completely positive autonomous CQ master equation, the moments $D_n^{\mu\nu}$ appearing in the master equation (2.91) must satisfy

$$2D_{2,\pi_{\Phi}\pi_{\Phi}} \succeq D_{1,\pi_{\Phi}}^{br} D_{0}^{-1} (D_{1,\pi_{\Phi}}^{br})^{\dagger}, \qquad (9.53)$$

where $D_{2,\pi_{\Phi}\pi_{\Phi}}$ is the full matrix of diffusion coefficients $D_{2,\phi_{\Phi}\pi_{\Phi}}^{\mu\nu}$, $D_{1,\pi_{\Phi}}^{br} = D_{1,\pi_{\Phi}}^{\mu\alpha}$ is the matrix describing the drift of the back-reaction of the quantum system and $(D_0^{-1})_{\alpha\beta}$ is the generalized inverse of the Lindbladian coefficient $D_0^{\alpha\beta}$.

We can use Equation (9.53) to understand general predictions about the second moments of the dynamics generated by general CQ master equations. In particular, from the general form of the CQ master equation in Equation (2.91) (see Appendix G for a discussion on field-theoretic master equations) that the first moment of π_{Φ} is described by

$$\partial_t \langle \pi_\Phi \rangle = \operatorname{Tr} \left[D_1^{\alpha\mu} L_\alpha \varrho(z) L_\nu^\dagger \right] + \operatorname{Tr} \left[D_1^{\mu\alpha} L_\mu \varrho(z) L_\alpha^\dagger \right] = \operatorname{Tr} \left[\{ \pi_\Phi, H_I \} \right] = \langle m \rangle, \tag{9.54}$$

while the second moment is described via

$$\partial_t \langle \pi_{\Phi}(x) \pi_{\Phi}(y) \rangle = 2 \operatorname{Tr} \left[D_2^{\mu\nu}(x,y) L_{\mu}(x) \varrho L_{\nu}^{\dagger}(y) \right].$$
(9.55)

As a consequence, the statistics of the first two moments of π_{Φ} can be described by adding a stochastic random process to the equation of motion π_{Φ}

$$\dot{\pi}_{\Phi} = \frac{\nabla^2 \Phi}{4\pi G} - \langle m(x) \rangle - u(\Phi, m)\xi(t, x), \qquad (9.56)$$

where, at each time, the noise process satisfies

$$\mathbb{E}_{m,\Phi}[u\xi(t,x)] = 0, \quad \mathbb{E}_{m,\Phi}[u\xi(t,x)u\xi(y,t')] = 2\langle D_2(x,y,\Phi)\rangle\delta(t,t'), \tag{9.57}$$

where we have defined $\langle D_2(x, y, \Phi) \rangle = \text{Tr} \left[D_2^{\mu\nu}(x, y, \Phi) L_{\mu}(x) \rho L_{\nu}^{\dagger}(y) \right]$, and ρ is the quantum state at time *t*. In Equation 9.57, the *m*, Φ subscripts of $\mathbb{E}_{m,\Phi}$ allow for the possibility that the statistics of the noise process can be dependent on the Newtonian potential and mass distribution of the particle.

In the non-relativistic limit, where $c \to \infty$, we impose the momentum constraint $\pi_{\Phi} \approx 0$, and we again recover Poisson's equation for gravity, but with a stochastic contribution to the mass

$$\Phi(t,x) \approx -G \int d^3x' \frac{[\langle m(x',t) \rangle - u(\Phi,m)J(x',t)]}{|x-x'|}.$$
(9.58)

If $u\xi(x,t)$ is Gaussian, Equation (10.28) completely determines the noise process, but in general, higher-order correlations are possible, but we can use Equation (9.57) to bound the effects due the second moments of Φ , which are fully described by $D_2^{\mu\nu}(x, y, \Phi)$.

In particular, from the trade-off, we know that there must be an accompanying decoherence on the quantum state given by the coupling $D_0^{\alpha\beta}$, which lower bounds the amount of diffusion through (9.53). In Chapter 10, we see that this can be used to constrain CQ dynamics which contain jumps experimentally.

9.7.1 Jumping master equation

An example of a jumping master equation satisfying Assumptions 1-4 is

$$\begin{split} \frac{\partial \varrho}{\partial t} &\approx \{H_c(\Phi), \varrho\} - i[H_m + +H_I, \varrho] + \int d^3x \left[D_0(\Phi; x) e^{D_0^{-1}(\Phi; x) \frac{\delta}{\delta \pi_{\Phi}(x)}} m \psi(x) \varrho \psi^{\dagger}(x) \right. \\ &\left. - \frac{1}{2} \{m(x), \varrho\}_+ \right]. \end{split}$$

Equation (9.59) needs regularizing, which we assume is performed by computing smeared observables. I.e., instead of calculating quantities such as the variation in force $\partial_t \langle \nabla \Phi_{\Phi}(x) \nabla \Phi_{\Phi}(x) \rangle$ directly, we assume that one calculates smeared quantities, such the variations of the time and spatial averaged forces $\bar{F} = -\int d^3x dt L(x,t) \nabla \Phi(x,t)$ where L(t,x) is a regulator; we do this explicitly in Chapter 10.

Expanding Equation (9.59) to second order, we see that

$$\frac{\partial \varrho}{\partial t} \approx \{H_c(\Phi), \varrho\} - i[H_m + H_I, \varrho]
+ \int d^3x D_0(\Phi; x)[m\psi(x)\varrho\psi(x)^{\dagger} - \frac{1}{2}\{m(x), \varrho\}]
+ \int d^3x m\psi(x) \frac{\delta \varrho}{\delta\pi_{\Phi}(x)}\psi^{\dagger}(x) + \frac{1}{2}\int d^3x D_0(\Phi; x)m\psi(x) \frac{\delta^2 \varrho}{\delta\pi_{\Phi}(x)\delta\pi_{\Phi}(x)}\psi^{\dagger}(x).$$
(9.59)

The back-reaction is Hamiltonian, since we obtain Equation (9.25) under trace. The Lindblad operators are ψ , and in this basis we see that the back-reaction matrix $D_1^{\alpha\mu}$ has components

 $D_1^{\psi\psi} = 1$, whilst the diffusion coupling $D_2^{\mu\nu}$ is found to be $D_2^{\psi\psi} = \frac{1}{2}D_0$, which saturates the trade-off in Equation (9.53). The decoherence rate for a mass in superposition, as in Section 9.6, is found from the no-event term $-\frac{1}{2}\{m(x), \varrho\}$ and is given by

$$\lambda = D_0 M, \tag{9.60}$$

while for a system of N composite particles, the decoherence rate is additive and found to be $\lambda = D_0 N M.$

9.8 Discussion

In this chapter, we have considered, on general grounds, the weak field limit of classical-quantum theories of gravity, which give rise to linear, completely positive dynamics. This provides a template from which classical-quantum theories of gravity can be experimentally tested and theoretically developed.

The Newtonian limit we find generalizes previous approaches to coupling classical and quantum systems in the Newtonian limit. In [49, 52, 55], the Newtonian potential is sourced by the outcome of certain measurements of the mass operator. The behavior in these models is qualitatively the same as those presented here: the Newtonian potential diffuses by an amount that depends on the inverse of the strength of the measurement. At the same time, the quantum system decoheres into its mass eigenbasis because it is being continuously measured. In this chapter, we have arrived at this behavior in complete generality for CQ theories which agree with the Newtonian limit on expectation, with the diffusion of the Newtonian potential and decoherence on the quantum system described by the parameters $D_2(\Phi; x, y), D_0(\Phi; x, y)$ satisfying the decoherence-diffusion trade-off. Previously discussed models of Newtonian classical-quantum gravity [49, 52, 55] then arise as limits of the more general theory considered in this chapter.

The weak field CQ theories we studied gave a generic prediction: the Newtonian potential diffuses away from its average solution, and for the dynamics to be completely positive, the amount of the diffusion is lower bounded by the coherence time for masses in superposition. This feature is perhaps most elegantly described via Equation (9.31). This behavior has also been derived from the path integral formulation of CQ, ensuring the positivity of the state is preserved.

Several proposals to test the quantum nature of gravity via gravitationally induced entanglement are expected to be realizable within the next few decades with technological advancements [64, 53, 1, 2, 65, 66, 67, 68]. If the underlying theory is local, then witnessing entanglement would imply that gravity is not a classical field. Within the framework of consistent classical-quantum coupling, we can inquire from the other direction, asking about the general experimental signatures of treating the gravitational field as being classical.

The combination of imposing the Newtonian limit of GR together with the decoherencediffusion trade-off, which follows from complete positivity, means that CQ theories allow for very concrete predictions which can be experimentally tested with current experimental technologies, which we explore in detail in Chapter 10.

As an effective theory of Newtonian gravity, we still expect the path integrals and the unravellings derived in this paper to be valid to all dynamics with a time-local description [28]. However, in general, they can be non-Markovian, which means that the couplings D_0, D_2 need not be positive semi-definite for all times, [92, 93], nor satisfy the decoherence-diffusion trade-off for all times since this is a consequence of the Markovian assumption.

An open theoretical problem is to construct a complete diffeomorphism invariant theory of classical-quantum gravity and to study the low energy effects of such a theory in a top-down approach. We have made some progress towards this goal by considering the Newtonian limit of the theory introduced in Chapter 8. We have seen that the Newtonian limit of the theory gives rise to qualitatively and quantitatively similar behavior to the general approach considered in this chapter. A more detailed calculation would involve checking whether the constraints involved give rise to consistent dynamics away from the Newtonian limit and whether higher order corrections necessitate changing any of the assumptions outlined in this chapter; for example, if the Newtonian description should be altered even at low energies.

Chapter 10

The trade-off between decoherence and diffusion

This chapter discusses the experimental signatures that follow from consistent CQ dynamics. We prove a generic trade-off between the rate of decoherence and the amount of diffusion in the classical phase space, extending that found in the CQ Pawula theorem in Chapter 4. The stronger the interaction between the quantum and classical systems, the greater the trade-off. One cannot have quantum systems with long-coherence times without inducing much diffusion in the classical system. One can also generalize this result to a trade-off between the rate of diffusion and the strength of more general couplings to Lindblad operators, with decoherence being a special case. This is expressed as Equations (10.14) and (10.12), which bounds the product of diffusion coefficients and Lindblad coupling constants in terms of the strength of the CQ-interaction. It is precisely this trade-off which allows the theories considered here to evade the no-go arguments of Feynmann [17, 35], Aharonov [40], Eppley and Hannah [18] and others [17, 39, 221, 59, 48, 222, 37, 38, 41, 42, 43, 44, 46, 47]. The essence of arguments against quantum-classical interactions is that they would prohibit superpositions of quantum systems that source a classical field. Since different classical fields are perfectly distinguishable in principle, if the classical field is in a distinct state for each quantum state in the superposition, the classical field could always be used to determine the quantum system's state, causing it to decohere instantly. By satisfying the trade-off, the quantum system preserves coherence because diffusion of the classical degrees of freedom means that the state of the classical field does not determine the state of the quantum system [61, 28]. Equation (10.14) and other variants we derive quantify the amount of diffusion required to preserve any coherence. If space-time curvature is treated classically, then complete positivity of the dynamics means its interaction with quantum fields necessarily results in unpredictability and gravitationally induced decoherence.

This trade-off between the decoherence rate and diffusion provides an experimental signature, not only of models of hybrid Newtonian dynamics such as [60] or post-quantum theories of General Relativity such as [28] but of any theory which treats gravity as being fundamentally classical. The metric and their conjugate momenta necessarily diffuse away from what Einstein's General Relativity predicts, and this experimental signature squeezes classical-quantum theories of gravity from both sides. If one has shorter decoherence times for superpositions of different mass distributions, one necessarily has more diffusion of the metric and conjugate momenta. The latter effect causes imprecision in measurements of mass such as those undertaken in the Cavendish experiment [223, 224, 225] or in measurements of Newton's constant "Big G" [226, 227, 228]. The precision at which a mass can be measured in a short time thus provides an upper bound on the amount of gravitational diffusion, as quantified by Equation (10.30). At the same time, decoherence experiments place a lower bound on the diffusion. Our estimates suggest that experimental lower bounds on the coherence time of large molecules [229, 230, 231, 232, 233, 170], combined with gravitational experiments measuring the acceleration of small masses [234, 235, 236], already place substantial restrictions on theories where space-time is not quantized. In Section 10.3, we show that several realizations of CQ gravity are already ruled out. In contrast, other realizations produce enough diffusion away from General Relativity to be detectable by future table-top experiments. Although the absence of such deviations from General Relativity would not be a direct confirmation of the quantum nature of gravity, such as experiments proposed in [64, 53, 1, 2, 65, 66, 67, 68] to exhibit entanglement generated by gravitons, it would effectively rule out any sensible theory which treats space-time classically. In comparison, confirmation of gravitational diffusion would suggest that space-time is fundamentally classical.

This chapter is based on the paper [4], which is work done in collaboration with Carlo

Sparaciari, Barbara Šoda, and Jonathan Oppenheim.

10.1 A general Trade off between decoherence and diffusion

In this section, we use positivity conditions to prove that the trade-off between decoherence and diffusion seen in models such as those of [57, 28, 10] are, in fact, a general feature of *all* classicalquantum interactions. We shall also generalize this and derive a trade-off between diffusion and arbitrary Lindbladian coupling strengths. The trade-off is in relation to the strength of the dynamics and is captured by Equations (10.9), (10.12) and (10.14). In Section 10.2, we extend the trade-off to the case where the classical and quantum degrees of freedom can be fields and use this to show that treating the metric as being classical necessarily results in diffusion of the gravitational field. We consider the general form of autonomous CQ master equation. The master equation is given by Equation (2.91), and we reproduce it here for convenience

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{00}_{n,i_1\dots i_n}(z,\delta t) \varrho(z,t) \right)
- i[H(z), \varrho(z)] + D^{\alpha\beta}_0(z) L_{\alpha} \varrho(z) L^{\dagger}_{\beta} - \frac{1}{2} D^{\alpha\beta}_0 \{ L^{\dagger}_{\beta} L_{\alpha}, \varrho(z) \}_+
+ \sum_{\mu\nu \neq 00} \sum_{n=1}^{\infty} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{\mu\nu}_{n,i_1\dots i_n}(z) L_{\mu} \varrho(z,t) L^{\dagger}_{\nu} \right).$$
(10.1)

Where the moments D_n are related to the short-time expansion of the transition amplitude

$$\Lambda^{\mu\nu}(z,t+\delta t|z',t) = \delta^{\mu}_{0}\delta^{\nu}_{0} + \delta t W^{\mu\nu}(z|z',t), \qquad (10.2)$$

through the equation

$$D_{n,i_1\dots i_n}^{\mu\nu}(z',t) := \frac{1}{n!} \int dz W^{\mu\nu}(z|z',t), t(z-z')_{i_1}\dots(z-z')_{i_n}.$$
 (10.3)

There are two separate possible sources for the force (or drift) of the back-reaction of the quantum system on phase space – it can be sourced by either the $D_{1,i}^{0\alpha}$ components or the Lindbladian components $D_{1,i}^{\alpha\beta}$. We shall deal with both sources simultaneously by considering a CQ Cauchy-Schwartz inequality which arises from the positivity of

$$\operatorname{Tr}\left[\int dz dz' \Lambda^{\mu\nu}(z,t+\delta t|z',t) O_{\mu}(z,z')\rho(z')O_{\nu}^{\dagger}(z,z')\right] \ge 0,$$
(10.4)

which must be positive for any vector of CQ operators O_{μ} . One can verify that this must be positive directly from the positivity conditions on $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ and we go through the details in Appendix F. A common choice for O_{μ} would be the set of operators $L_{\mu} = \{\mathbb{I}, L_{\alpha}\}$ appearing in the master equation.

The inequality in Equation (10.4) turns out to be especially useful since it can be used to define a (pseudo) inner product on a vector of operators with components O_{μ} via

$$\langle \bar{O}_1, \bar{O}_2 \rangle = \int dz dz' \text{Tr} \left[\Lambda^{\mu\nu}(z, t + \delta t | z', t)(z | z') O_{1\mu} \varrho(z') O_{2\nu}^{\dagger} \right]$$
(10.5)

where $||\bar{O}|| = \sqrt{\langle \bar{O}, \bar{O} \rangle} \ge 0$ due to (10.4). Technically this is not positive definite, but this shall not be important for our purpose. Taking the combination $O_{\mu} = ||\bar{O}_2||^2 O_{1\mu} - \langle \bar{O}_1, \bar{O}_2 \rangle O_{2\mu}$ for vectors $O_{1\mu}, O_{2\mu}$, positivity of the norm gives

$$||\bar{O}||^{2} = ||\bar{O}_{2}||^{2}\bar{O}_{1} - \langle\bar{O}_{1},\bar{O}_{2}\rangle\bar{O}_{2}||^{2} = ||\bar{O}_{2}||^{2} \left(||\bar{O}_{1}||^{2}||\bar{O}_{2}||^{2} - |\langle\bar{O}_{1},\bar{O}_{2}\rangle|^{2}\right) \ge 0, \quad (10.6)$$

and as long as $||\bar{O}_2|| \neq 0$ we have a Cauchy- Schwartz inequality

$$||\bar{O}_1||^2 ||\bar{O}_2||^2 - |\langle \bar{O}_1, \bar{O}_2 \rangle|^2 \ge 0.$$
(10.7)

We can use (10.7) to get a trade-off between the observed diffusion and decoherence by picking $O_{2\mu} = \delta^{\alpha}_{\mu}L_{\alpha}$ and $O_{1\mu} = b^{i}(z - z')_{i}L_{\mu}$, where $L_{\mu} = \{\mathbb{I}, L_{\alpha}\}$ are the Lindblad operators appearing in the master equation. In this case, $||\bar{O}_{2}|| = \int dz \operatorname{Tr} \left[D_{0}^{\alpha\beta}L_{\alpha}\varrho L_{\beta}^{\dagger}\right]$ and one can verify using CQ Pawula theorem [5]¹ that in order to have non-trivial back-reaction on the quantum system, complete positivity demands that $||\bar{O}_{2}|| > 0$, meaning the Cauchy-Schwartz inequality in Equation (10.7) must hold. By using the short time moment expansion of $\Lambda^{\mu\nu}(z, t + \delta t | z', t)$ defined in Equation (10.2) and using integration by parts, we then arrive at the observational trade-off between decoherence and diffusion

$$\int dz \operatorname{Tr} \left[2b^{i*} D_{2,ij}^{\mu\nu} b^j L_{\mu} \varrho(z) L_{\nu}^{\dagger} \right] \int dz \operatorname{Tr} \left[D_0^{\alpha\beta} L_{\alpha} \varrho(z) L_{\beta}^{\dagger} \right] \ge \left| \int dz \operatorname{Tr} \left[b^i D_{1,i}^{\mu\alpha} L_{\mu} \varrho(z) L_{\alpha}^{\dagger} \right] \right|^2,$$
(10.8)

which must hold for any positive CQ state $\rho(z)$. Stripping out the b^i vectors, (10.8) is equivalent to the matrix positivity condition

$$0 \leq 2\langle D_2 \rangle \langle D_0 \rangle - \langle D_1^{br} \rangle \langle D_1^{br} \rangle^{\dagger}, \quad \forall \varrho(z),$$
(10.9)

¹In particular, to reach this conclusion one can insert the CQ state into the CQ Cauchy-Schwartz inequality and repeat the proof of the Pawula theorem [5], which must now hold once averaged over the state.

where we define

$$\langle D_0 \rangle = \int dz \operatorname{Tr} \left[D_0^{\alpha\beta} L_\alpha \varrho(z) L_\beta^\dagger \right], \ \langle D_1^{br} \rangle_i = \int dz \operatorname{Tr} \left[D_{1,i}^{\mu\alpha} L_\mu \varrho(z) L_\alpha^\dagger \right]$$

$$\langle D_2 \rangle_{ij} = \int dz \operatorname{Tr} \left[D_{2,ij}^{\mu\nu} L_\mu \varrho(z) L_\nu^\dagger \right]$$

$$(10.10)$$

and $D_1^{br} = D_{1,i}^{\mu\alpha}$ defines the *back-reaction matrix*, describing the back-reaction of the quantum system on the classical one. Since (10.9) holds for all states, the tightest bound is provided by the infimum over all states

$$0 \leq \inf_{\varrho(z)} \{ 2 \langle D_2 \rangle \langle D_0 \rangle - \langle D_1^{br} \rangle \langle D_1^{br} \rangle^{\dagger} \}.$$
(10.11)

The quantities $\langle D_2 \rangle$ and $\langle D_0 \rangle$ appearing in Equation (10.9) are related to observational quantities. In particular, $\langle D_2 \rangle$ is the expectation value of the classical diffusion observed, and $\langle D_0 \rangle$ is related to the amount of decoherence on the quantum system. The expectation value of the back-reaction matrix $\langle D_1^{br} \rangle$ quantifies the back-reaction on the classical system. In the trivial case $D_1^{br} = 0$, Equation (10.9) places little restriction on the diffusion and Lindbladian rates appearing on the left-hand side. We already knew from [62, 63] that the $D_0^{\alpha\beta}$ must be a positive semi-definite matrix, and we also know that diffusion coefficients must be positive semi-definite. However, in the non-trivial case, the larger the back-reaction exerted by the quantum system, the stronger the trade-off between the diffusion coefficients and Lindbladian coupling. Equation (10.9) gives a general trade-off between observed diffusion and Lindbladian rates. We call the trade-offs involving expectation values over the state observational trade-offs.

We can also find a trade-off regarding a theory's coupling coefficients alone. We show in Appendix F that the general matrix trade-off

$$D_1^{br} D_0^{-1} D_1^{br\dagger} \preceq 2D_2 \tag{10.12}$$

holds for the matrix whose elements are the couplings $D_{2,ij}^{\mu\nu}$, $D_{1,i}^{\alpha\mu}$, $D_0^{\alpha\beta}$ for any CQ dynamics. Moreover, $(\mathbb{I} - D_0 D_0^{-1}) D_1^{br} = 0$, which tells us that D_0 cannot vanish if there is non-zero back-reaction. Equation (10.12) quantifies the required amount of decoherence and diffusion for the dynamics to be completely positive. In Equation (10.12), and throughout, D_0^{-1} is the generalized inverse of $D_0^{\alpha\beta}$, since $D_0^{\alpha\beta}$ is only required to be positive semi-definite. In the special case of a single Lindblad operator $\alpha = 1$ and classical degree of freedom, and when the only non-zero couplings are $D_0^{11} := D_0$, $D_{2,pp}^{00} := 2D_2$ and $D_{1,q}^0 = 1$ this trade-off reduces to the condition $D_2D_0 \ge 1$ used in [57].

It is also useful to try to obtain an observational trade-off in terms of the total drift due to back-reaction as calculated in Equation (2.102)

$$\langle D_1^T \rangle_i = \sum_{\mu\nu \neq 00} \int dz \operatorname{Tr} \left[D_{1,i}^{\mu\nu} L_\mu \varrho(z) L_\nu^\dagger \right].$$
(10.13)

It follows directly from Equation (10.9) that when the back-reaction is sourced by either $D_{1,i}^{0\mu}$ or $D_{1,i}^{\alpha\beta}$ we can arrive at the observational trade-off in terms of the total drift²

$$0 \leq 8\langle D_2 \rangle \langle D_0 \rangle - \langle D_1^T \rangle \langle D_1^T \rangle^{\dagger}, \quad \forall \varrho(z),$$
(10.14)

where the quantities appearing in Equation (10.14) are now all observational quantities related to drift, decoherence, and diffusion.

In the case where the back-reaction is Hamiltonian at first order in the sense of Equation (2.103), then (10.14) can be written as

$$\langle \omega \cdot \frac{\partial H_I}{\partial \vec{z}} \rangle \langle \omega \cdot \frac{\partial H_I}{\partial \vec{z}} \rangle^{\dagger} \preceq 8 \langle D_2 \rangle \langle D_0 \rangle, \quad \forall \varrho(z).$$
 (10.15)

As a result, we can derive a trade-off between diffusion and decoherence for any theory that reproduces this classical limit and treats one of the systems classically.

To summarize, whenever the back-reaction of the quantum system on the classical system induces a force on the phase space, then we have a trade-off between the amount of diffusion on the classical system and the strength of decoherence on the quantum system (or, more precisely, the strength of the Lindbladian couplings $D_0^{\alpha\beta}$). This can be expressed both as a condition on the matrix of coupling coefficients in the master equation, via Equation (10.12) or in terms of observable quantities using Equation's (10.9) and (10.14). When the back-reaction is Hamiltonian, we further have Equation (10.15). We want to apply this trade-off to the case of gravity in the non-relativistic, Newtonian limit, which we considered in Chapter 9. In order to do so, we will need to generalize the trade-off to the case of quantum fields interacting with classical ones, which we do in Section 10.2. The goal will be to understand the implications of treating the metric (or Newtonian potential) as being classical by using the trade-off when the

 $^{^{2}}$ We believe that (10.14) should hold more generally, though we do not have a general proof.

quantum back-reaction induces a force on the gravitational field, which, on expectation, is the same as the weak field limit of general relativity.

10.2 Trade off in the presence of fields

We want to explore the trade-off in the gravitational setting and the consequences of treating the gravitational field as classical and matter quantum. Since gravity is a field theory, we must first discuss classical-quantum master equations in the presence of fields. In the field-theoretic case, both the Lindblad operators and the phase space degrees of freedom can have spatial dependence, $z(x), L_{\mu}(x)$ and a general bounded CP map which preserves the classicality of the two systems can be written [28]

$$\rho(z,t) = \int dz' dx dy \Lambda^{\mu\nu}(z,t+\delta t|z',t;x,y) L_{\mu}(x,z,z') \varrho(z',0) L_{\nu}^{\dagger}(y,z,z'), \qquad (10.16)$$

where, as is usually the case with fields, in Equation (10.16), it should be implicitly understood that a smearing procedure has been implemented. We elaborate on some details when fields are introduced in Appendix G. The condition for (10.16) to be completely positive on *all* CQ states is

$$\int dz dx dy A^*_{\mu}(x, z, z') \Lambda^{\mu\nu}(z, t + \delta t | z', t; x, y) A_{\nu}(y, z, z') \ge 0$$
(10.17)

meaning that $\Lambda^{\mu\nu}(z, t + \delta t | z', t; x, y)$ can be viewed as a positive matrix in $\mu\nu$ and a positive kernel in x, y. In the field-theoretic case, one can still perform a Kramers-Moyal expansion and find a trade-off between the coefficients $D_0(x, y), D_1(x, y), D_2(x, y)$ appearing in the master equation. Due to the Lindblad operators' spatial dependence, the coefficients now have an x, ydependence. The coefficients $D_1(x, y), D_2(x, y)$ still have a natural interpretation as measuring the amount of force (drift) and diffusion, while $D_0(x, y)$ describes the purely quantum evolution on the system and can be related to decoherence.

Using the positivity condition in Equation (10.17) we find the same trade of between coupling constants in Equation (10.12) but where now $D_2(x, y)$ is the $(p+1)n \times (p+1)n$ matrixkernel with elements $D_{2,ij}^{\mu\nu}(x, y)$, $D_1^{br}(x, y)$ is the $(p+1)n \times p$ matrix-kernel with rows labeled by μi , columns labelled by β , and elements $D_{1,i}^{\mu\beta}(x, y)$, and $D_0(x, y)$ is the $p \times p$ decoherence matrix-kernel with elements $D_0^{\alpha\beta}(x, y)$.³ In the field theoretic trade off we are treating

³Here $i \in \{1, ..., n\} \ \alpha \in \{1, ..., p\}$ and $\mu \in \{1, ..., p+1\}$,

the objects in Equation (10.12) as matrix-kernels, so that for any position dependent vector $b^i_{\mu}(x)$, $(D_2b)^{\mu}_i(x) = \int dy D^{\mu\nu}_{2,ij}(x,y) b^j_{\nu}(y)$, whilst for any position dependent vector $a_{\alpha}(x)$, $(D_0\alpha)^{\alpha}(x) = \int dy D^{\alpha\beta}_0(x,y) \alpha_{\beta}(y)$. Explicitly, we find that the complete positivity of the dynamics is equivalent to the matrix condition

$$\int dx dy [b^*(x), \alpha^*(x)] \begin{bmatrix} 2D_2(x, y) & D_1^{br}(x, y) \\ D_1^{br}(x, y) & D_0(x, y) \end{bmatrix} \begin{bmatrix} b(y) \\ \alpha(y) \end{bmatrix} \ge 0$$
(10.18)

which should be positive for any position dependent vectors $b^i_{\mu}(x)$ and $a_{\alpha}(x)$. This is equivalent to a trade-off between coupling constants in Equation (10.12) if we view (10.12) as a matrixkernel equation.

Though we make no assumption on the locality of the Lindbladian and diffusion couplings, we hereby assume that the drift back-reaction is local, so that $D_1^{br}(x,y) = \delta(x,y)D_1^{br}(x)$. As we shall see in the next section, this is a natural assumption if we want back-reaction given by a local Hamiltonian. However, one might not want to assume that the form of the Hamiltonian remains unchanged to arbitrarily small distances. With this locality assumption, Equation (10.18) gives rise to the same trade-off of Equation (10.12), where the trade-off is to be interpreted as a matrix kernel inequality. Writing this out explicitly, we have

$$\int dx dy \alpha_{\nu}^{i*}(x) D_{1,i}^{\mu\alpha}(x) (D_0^{-1})_{\alpha\beta}(x,y) D_{1,j}^{\beta\nu}(x') \alpha_{\nu}^i(x') \leq \int dx dy 2\alpha_{\mu}^{i*}(x) D_{2,ij}^{\mu\nu}(x,y) \alpha_{\nu}^j(y), \quad (10.19)$$

where asking that this inequality holds for all vectors $\alpha^i_{\mu}(x)$ is equivalent to the matrix-kernel trade-off condition of Equation (10.12). We saw examples of gravitational master Equations satisfying the coupling constant trade-off in Chapter 9. The decoherence-diffusion trade-off tells us how much diffusion and stochasticity is required to maintain coherence when the quantum system back-reacts on the classical one.

In the field-theoretic case, we can similarly find an observational trade-off, relating the expected value of the diffusion matrix $\langle D_2(x,y) \rangle$ to the expected value of the drift in a physical state ρ as we did in Section 10.1. This is done explicitly in Appendix G, using a field-theoretic version of the Cauchy-Schwartz inequality given by Equation (G.16), we find

$$2\langle D_2(x,x)\rangle \int dx' dy' \langle D_0(x',y')\rangle \succeq \langle D_1^{br}(x)\rangle \langle D_1^{br}(x)\rangle^{\dagger}, \qquad (10.20)$$

where equation (10.22) is to be understood as a matrix inequality with entries

$$\langle D_0(x,y) \rangle = \int dz \operatorname{Tr} \left[D_0^{\alpha\beta} L_\alpha(x) \varrho L_\beta^{\dagger}(y) \right], \langle D_1^{br}(x,y) \rangle_i = \int dz \operatorname{Tr} \left[D_{1,i}^{\mu\alpha} L_\mu(x) \varrho L_\alpha^{\dagger}(x) \right],$$

$$\langle D_2(x,y) \rangle_{ij} = \int dz \operatorname{Tr} \left[D_{2,ij}^{\mu\nu} L_\alpha(x) \varrho L_\beta^{\dagger}(y) \right].$$

$$(10.21)$$

Similarly, when the back-reaction is sourced by either $D_{1,i}^{0\mu}$ or $D_{1,i}^{\alpha\beta}$ it follows from Equation (10.20) we can arrive at the observational trade-off in terms of the total drift due to back-reaction

$$8\langle D_2(x,x)\rangle \int dx' dy' \langle D_0(x',y')\rangle \succeq \langle D_1^T(x)\rangle \langle D_1^T(x)\rangle^{\dagger}, \qquad (10.22)$$

where

$$\langle D_1^T(x)\rangle_i = \int dz \operatorname{Tr} \left[D_{1,i}^{0\alpha}(x)\varrho L_{\alpha}^{\dagger}(x) + D_{1,i}^{\alpha0}(x)L_{\alpha}\varrho(x) + D_{1,i}^{\alpha\beta}(x)L_{\alpha}(x)\varrho L_{\beta}^{\dagger}(x) \right].$$
(10.23)

We shall now use the trade-off to study the consequences of treating the gravitational field classically. We will consider the back-reaction of the mass on the gravitational field to be governed by the Newtonian interaction. We shall then find that experimental bounds on coherence lifetimes for particles in superposition require significant diffusion in the gravitational field to be maintained. This can be upper bounded by gravitational experiments.

To summarise this section, we have derived the trade-off between decoherence and diffusion for classical-quantum field theories, both in terms of coupling constants of the theory and in terms of observational quantities. This trade-off puts tight observational constraints on classical theories of gravity, which we now discuss.

10.3 Physical constraints on the classicality of gravity

In this section, we apply the trade-off of Equation (10.18) to the case of gravity. Since the trade-offs derived in the previous section depend only on the back-reaction or drift term, they are insensitive to the particulars of the theory. We shall consider the Newtonian, non-relativistic limit of a classical gravitational field which we studied in Chapter 9, and for completeness we outline all of our assumptions.

Assumption 1. We assume that the evolution of the combined classical-quantum system undergoes autonomous CQ dynamics. In particular, the theory is a completely positive norm-preserving autonomous map, and we can perform a short-time Kramers-Moyal expansion of the dynamics.

Assumption 2. We apply the theory to the weak field limit of General Relativity, where, as recalled in Chapter 9, the Newtonian potential interacts with matter through its mass density m(x). We assume that the purely classical part of the evolution is generated by the reduced Hamiltonian of Equation (9.24), that the interaction between classical and quantum degrees of freedom is Hamiltonian, and that it is governed by the reduced interaction Hamiltonian

$$H_I(\Phi) = \int d^3x \Phi(x) m(x), \qquad (10.24)$$

where the constraints $\pi_{\Phi}(t)$ should also be imposed.

Assumption 3. In this work, we will take the coefficients D_n entering the master equation to be minimally coupled, by which we mean they depend only on the Newtonian potential Φ , $D_n(\Phi)$ and not their conjugate momenta π_{Φ} .

Assumption 4. In relating D_0 to the decoherence rate of a particle in superposition, we shall assume that the state of interest is well approximated by a state living in a Hilbert space of fixed particle number. We believe this is a mild assumption: ordinary non-relativistic quantum mechanics is described via a single particle Hilbert space. We frequently place massive composite particles in superposition, and they do not typically decay into multiple particles.

These assumptions can be understood as describing the non-relativistic Newtonian limit of classical-quantum theories of gravity. In particular, we call such theories *non-relativistic* to emphasize that they assume the Newtonian approximation holds to short distances, i.e., sub millimeter scales where coherence experiments are performed, but general relativity has yet to be tested at distances shorter than the millimeter scale. It may be that higher order, or relativistic corrections, affect the low-energy behavior of the theory but we do not discuss such *relativistic* theories here.

With these assumptions and treating the matter density as a quantum operator, this tells

us that in order for the back-reaction term to reproduce the Newtonian interaction on average

$$\operatorname{Tr}\left[\{H_{I},\varrho\}\right] = \operatorname{Tr}\left[\int d^{3}x \ \hat{m}(x)\frac{\delta\varrho}{\delta\pi_{\Phi}}\right] = -\sum_{\mu\nu\neq00} \operatorname{Tr}\left[\int d^{3}x D_{1,\pi_{\Phi}}^{\mu\nu}(\Phi,\pi_{\Phi},x)L_{\mu}(x)\frac{\delta\varrho}{\delta\pi_{\Phi}}L_{\nu}^{\dagger}(x)\right],$$
(10.25)

then we must pick

$$\langle D_{1,\pi_{\phi}}^{T}(\Phi,\pi_{\Phi},x)\rangle = -\langle \hat{m}(x)\rangle, \qquad (10.26)$$

meaning that the back-reaction matrix $D_{1,\pi\Phi}^{\mu\alpha}$ is non vanishing. In the previous chapter (see also the appendices of [4]), we saw examples of master equations for which Equation (10.26) is satisfied. However, their details are irrelevant since we only require the expectation of the backreaction force to be the mass – a necessary condition for the theory to reproduce Newtonian gravity.

As a consequence of the coupling constant and observational trade-offs derived in Equations (10.19) and (10.20), a non-zero $D_{1,\pi_{\Phi}}$ implies that there must be diffusion in the momenta conjugate to π_{Φ} . For decohered mass, we saw in Chapter 9 that this diffusion is equivalent to adding a stochastic random process $\xi(x,t)$ to the equation of motion for $\dot{\pi}_{\Phi}$ to give

$$\dot{\pi}_{\Phi} = \frac{\nabla^2 \Phi}{4\pi G} - m(x) + u(\Phi, \hat{m})\xi(t, x), \qquad (10.27)$$

where we allow some *colouring* to the noise via a function $u(\Phi, \hat{m})$ which can depend on Φ , and the matter distribution \hat{m} . The noise process satisfies

$$\mathbb{E}_{m,\Phi}[u\xi(x,t)] = 0, \quad \mathbb{E}_{m,\Phi}[u\xi(x,t)u\xi(y,t')] = 2\langle D_2(x,y,\Phi)\rangle\delta(t,t'), \quad (10.28)$$

where we have defined $\langle D_2(x, y, \Phi) \rangle = \text{Tr} \left[D_2^{\mu\nu}(x, y, \Phi) L_{\mu}(x) \rho L_{\nu}^{\dagger}(y) \right]$, and ρ is the quantum state for the decohered mass density. Here the m, Φ subscripts of $\mathbb{E}_{m,\Phi}$ allow for the possibility that the statistics of the noise process can be dependent on the Newtonian potential and mass distribution of the particle. If $u\xi(x,t)$ is Gaussian, Equation (10.28) completely determines the noise process, but in general, higher-order correlations are possible, although they need not concern us here, since we are only interested in bounding the effects due to $D_2(x, y, \Phi)$. In this chapter, we often suppress the explicit dependence of the couplings D_n on the Newtonian potential for notational convenience.

In the non-relativistic limit, where $c \to \infty$, we impose the momentum constraint $\pi_{\Phi} \approx 0$ and recover Poisson's equation for gravity, but with a stochastic contribution to the mass. This is precisely as expected on purely physical grounds: in order to maintain the coherence of any mass in superposition, there must be noise in the Newtonian potential, and this must be such that we cannot tell which element of the superposition the particle will be in, meaning the Newtonian potential should look like it is being sourced in part by a random mass distribution. In other words, the trade-off requires that the stochastic component of the coupling obscures the amount of mass m at any point.

The solution to Equation (10.27) is given by

$$\Phi(t,x) \approx -G \int d^3x' \frac{[m(x',t) - u(\Phi,\hat{m})\xi(x',t)]}{|x - x'|},$$
(10.29)

and a formal treatment of solutions to non-linear stochastic integrals of the form of Equation (10.27) can be found in [237]. A higher precision calculation would involve a full simulation of CQ dynamics, for example, using unraveling methods [6, 28]. It may also be that relativistic corrections may constrain the degree of diffusion even at low energy. One should consider this when drawing conclusions from the models presented here.

We have seen that there are two classes of CQ dynamics, in the sense that there are those with continuous trajectories in phase space and those with discrete jumps. For the class of continuous CQ models, we know that $\xi(x,t)$ should be described by a white noise process in time, and its statistics should be independent of the mass density of the particle. For the discrete class, $\xi(x,t)$ can involve higher order moments and will generally be described by a jump process [5, 10]. Its statistics can also depend on the mass density since, generally, the diffusion matrix $D_{2,ij}^{\mu\nu}$ couples to Lindblad operators. It is worth noting that the discrete CQ theories considered in [28, 3, 10] generically suppress higher order moments, and often we expect that we can approximate the dynamics by a Gaussian process, but this need not be the case in general.

This variation in Newtonian potential leads to observational consequences, which can be used to experimentally test and constrain CQ theories of gravity for various choices of kernels appearing in the CQ master equation. One immediate consequence is that the variation in Newtonian potential leads to a variation of force experienced by a particle or composite mass via $\vec{F}_{tot} = -\int d^3x m(x) \nabla \Phi(x)$. We can also estimate the time-averaged force via $\frac{1}{\Delta T} \int_0^{\Delta T} \vec{F}_{tot}$ where ΔT is the time resolution over which the force is measured and is the useful quantity when comparing with experiments. Using Equation (10.29), in Appendix I.1, we find that the

Master Equa-	Diffusion Kernel	Experimental squeeze
tion		
Continuous (ultra-local)	$D_2(\Phi; x, y) = D_2(\Phi)\delta(x, y)$ $D_2(\Phi) = \sum_n c^n \Phi^n$	$10^{-41} \ge D_2 \ge 10^{-9} \ kg^2 sm^{-3}$ (Eqn (10.32))
Continuous (D.P)	$D_2(\Phi; x, y) = -l_p^2 D_2(\Phi) \nabla^2 \delta(x, y)$ $D_2(\Phi) = \sum_n c^n \Phi^n$	$10^{-9} \ge l_P^2 D_2 \ge 10^{-35} \ kg^2 sm^{-1}$ (Eqn (10.35))
Discrete (ultra-local)	$D_2(\Phi; x, y) = \frac{l_P^2}{m_P} D_2(\Phi) \delta(x, y)$ $D_2(\Phi) = \sum_n c^n \Phi^n$	$10^{-1} \ge \frac{l_P^2 D_2}{m_P} \ge 10^{-25} \ kgs$ (Eqn (10.34))

Table 10.1: Current experimental bounds on non-relativistic classical-quantum theories for different master equations and functional dependence on the diffusion coefficient. The diffusion coefficient is bounded from above by observed acceleration variations σ_a^2 seen in precision mass experiments via Equation (10.30). In all cases, the master equation is assumed to saturate the bound, which is used to find the lower bound the amount of diffusion on the quantum system by bounding D_0 from coherence rates via Equation (10.31). It is seen that minimally coupled continuous models, which are non-relativistic and do not create spatial correlations (we call these *ultra-local*) and have polynomial dependence on the Newtonian potential, are ruled out. In contrast, continuous models with non-local correlations, such as the Diosi-Penrose (D.P.) kernel, and ultra-local discrete models are less constrained. Here l_P, m_P denote the Planck length and Planck mass, respectively, which are required for the dimensions of $D_2(\Phi)$ to be the same in all cases.

variance of the magnitude of the time-averaged force experienced by a particle in a Newtonian potential is given by

$$\sigma_F^2 = \frac{2G^2}{\Delta T} \int d^3x d^3y d^3x' d^3y' m(x) m(y) \frac{(\vec{x} - \vec{x}') \cdot (\vec{y} - \vec{y}')}{|x - x'|^3 |y - y'|^3} \langle D_2(x', y', \Phi) \rangle, \tag{10.30}$$

where the variation is averaged by the time resolution ΔT . We will use this to estimate the variation in precision measurements of mass, such as modern versions of the Cavendish experiment for various choices of $\langle D_2(x', y', \Phi) \rangle$.

On the other hand, experimentally measured decoherence rates can be related to D_0 , which we calculated for both the continuous master equation and the jumping master equation in Chapter 9. In Appendix H, we show that for a mass whose quantum state is a superposition of two states $|L\rangle$ and $|R\rangle$ of approximately orthogonal mass densities $m_L(x), m_R(x)$, and whose separation we take to be larger than the correlation range of $D_0(x, y)$, the decoherence rate is given by

$$\lambda = \frac{1}{2} \int dx dy D_0^{\alpha\beta}(x, y) (\langle L | L_\beta^{\dagger}(y) L_\alpha(x) | L \rangle + \langle R | L_\beta^{\dagger}(y) L_\alpha(x) | R \rangle), \qquad (10.31)$$

which generalizes the calculations performed in Chapter 9. Via the coupling constant trade-off, Equations (10.30) and (10.31) give rise to a double-sided squeeze on the coupling D_2 . Equation (10.30) upper bounds D_2 in terms of the uncertainty of acceleration measurements seen in gravitational torsion experiments, while the coupling constant trade-off Equation (10.31) lower bounds D_2 in terms of experimentally measured decoherence rates arising from interferometry experiments.

We now show this for various choices of diffusion kernel for the models considered in Chapter 9. We have already calculated the decoherence rates, and we calculate the associated force variations in Appendix I.1. The bounds are summarized in Table 10.1. The diffusion coupling strength will be characterized by the coupling constant D_2 , which we consider a dimension-full quantity with units kg^2sm^{-3} , and is related to the rate of diffusion for the conjugate momenta of the Newtonian potential. We upper bound D_2 by considering the variation of the timeaveraged acceleration $\sigma_a = \frac{\sigma_F}{M}$ for a composite mass M which contains N atoms which we treat as spheres of constant density ρ with radius r_N and mass m_N . We lower bound D_2 via the coupling constant trade-off of Equation (10.18) and then by considering bounds on the coherence time for composite particles with total mass M_{λ} and which are made up of N_{λ} constituents, each with typical length scale when in superposition R_{λ} and volume V_{λ} .

For continuous dynamics $\langle D_2(x, y, \Phi) \rangle = D_2(x, y, \Phi)$ since the diffusion is not associated with any Lindblad operators. The full dynamics for these models were considered in Section 9.3 of Chapter 9. Let us now consider a very natural kernel, namely $D_2(x, y; \Phi) = D_2(\Phi)\delta(x, y)$, which is both translation invariant and does not create any correlations over space-like separated regions. We call dynamics which do not create correlations over space-like separated regions *ultra-local* since theories that are not of this form can still be non-signaling. The squeeze will generally depend on the functional choice of $D_2(\Phi)$ on the Newtonian potential. However, in the presence of a large background potential Φ_b , such as that of the Earth's, we will often be able to approximate $D_2(\Phi) = D_2(\Phi_b)$. This is true for kernels that depend on Φ and $\nabla \Phi$, though the approximation does not hold for all kernels, for example, $D_2 \sim -\nabla^2 \Phi$ which creates diffusion only where there is mass density. For diffusion kernels $D_2(\Phi_b)$

$$\frac{\sigma_a^2 N r_N^4 \Delta T}{V_b G^2} \ge D_2 \ge \frac{N_\lambda M_\lambda^2}{V_\lambda \lambda},\tag{10.32}$$

where V_b is the volume of space over which the background Newtonian potential is significant. V_b enters since the variation in acceleration is found to be

$$\sigma_a^2 \sim \frac{D_2 G^2}{r_N^4 N \Delta T} \int d^3 x' D_2(\Phi_b), \qquad (10.33)$$

where the d^3x' integral is over all space. This immediately rules out continuous theories with noise everywhere, i.e., with a diffusion coefficient independent of the Newtonian potential since the integral will diverge.

Standard Cavendish type classical torsion balance experiments [223] measure accelerations of the order $10^{-7}ms^{-2}$ over minutes $\Delta T \sim 10^2$, so a very conservative bound is $\sigma_a \sim 10^{-7}ms^{-2}$, whilst for a kg mass $N \sim 10^{26}$ and $r_N \sim 10^{-15}m$. Conservatively taking $V_b \sim r_E^2 h m^3$ where r_E is the radius of the Earth and h is the atmospheric height gives $D_2 \leq 10^{-41}kg^2sm^{-3}$. The decoherence rate λ is bounded by various experiments [238]. Typically, such experiments aim to witness interference patterns of molecules that are as massive as possible. Taking a conservative bound on λ , for example, that arising from the interferometry experiment of [233] which saw coherence in large organic fullerene molecules with total mass $M_{\lambda} = 10^{-24}kg$ over a timescale of 0.1s, gives an upper bound on the decoherence rate $\lambda < 10^1 s^{-1}$. Fullerene molecules comprise $N_{\lambda} \sim 10^3$ particles with typical atomic size $10^{-15}m$. After passing through the slits, the molecule becomes delocalized in the transverse direction on the order of $10^{-7}m$ before being detected. Since the interference effects are due to the superposition in the transverse x direction, which is the direction of alignment of the gratings, it seems like a reasonable assumption to take the size of the wave-packet in the remaining y, z direction to be the size of the fullerene, since we could imagine measuring the y, z directions without effecting the coherence. We, therefore, take the volume $V_{\lambda} \sim 10^{-15}10^{-15}10^{-7}m^3 = 10^{-37}m^3$, which gives $D_2 \geq 10^{-9}kg^2sm^{-3}$, and suggests that classical-quantum theories of gravity with ultra-local continuous noise are ruled out by experiment.

On the other hand, the discrete models appear less constrained due to the suppression of the noise away from the mass density. For example, consider the ultra-local discrete jumping models, such as the one given in Section 9.7 of Chapter 9, which have $\langle D_2(x, y, \Phi_b) \rangle = \frac{l_P^3 D_2(\Phi_b)}{m_P} m(x)$, where $m_P = \sqrt{\frac{\hbar c}{G}}$ is the Planck mass and $l_P = \sqrt{\frac{\hbar G}{c^3}}$ is the Planck length, required to ensure D_2 has the units of $kg^2 sm^{-3}$.

We find the squeeze on D_2

$$\frac{\sigma_a^2 N r_N^4 \Delta T}{m_N G^2} \ge \frac{l_P^3}{m_P} D_2 \ge \frac{M_\lambda}{\lambda},\tag{10.34}$$

and plugging in the numbers tells us that discrete theories of classical gravity are not ruled out by experiment, and we find $10^{-1}kgs \ge \frac{l_P^3}{m_P}D_2 \ge 10^{-25}kgs$.

We can also consider other noise kernels which are not ultra-local. A natural kernel is the Diosi-Penrose kernel for the class of continuous models studied in Section 9.3 of Chapter 9 with $D_2(x, y, \Phi_b) = -l_P^2 D(\Phi_b) \nabla^2 \delta(x, y)$. The inverse Lindbladian kernel satisfying the coupling constants trade-off is to zeroeth order in $\Phi(x)$, the Diosi-Penrose kernel $D_0(x, y, \Phi_b) = \frac{D_0(\Phi_b)}{|x-y|}$.

For this choice of dynamics, we find the squeeze for D_2 in terms of the variation in acceleration

$$\frac{\sigma_a^2 N r_N^3 \Delta T}{G^2} \ge l_P^2 D_2 \ge \frac{N_\lambda M_\lambda^2}{R_\lambda \lambda}.$$
(10.35)

Using the same numbers as for the ultra-local continuous model, with $R_{\lambda} \sim V_{\lambda}^{1/3} \sim 10^{-12} m$, we find that classical torsion experiments upper bound D_2 by $10^{-9} kg^2 sm^{-1} \geq l_P^2 D_2$, while interferometry experiments bound D_2 from below via $l_P^2 D_2 \geq 10^{-35} kg^2 sm^{-1}$.

Equations (10.32), (10.34), and (10.35) show that experiments squeeze classical theories

of gravity from both ways. We have been highly conservative here and anticipate that further analysis and near-term experiments can tighten the bounds by orders of magnitude. Several proposals for table-top experiments to measure gravity precisely have recently been performed and could give rise to tighter upper bounds on D_2 . Some of these experiments involve millimeter-sized masses whose gravitational coupling is measured via torsional pendula [234, 235], or rotating attractors [236]. With such devices, the gravitational coupling between small masses can be measured while limiting the amount of other noise sources. There are proposals for further mitigating the noise due to the environment, including the inertial noise, gas particles collisions, photon scattering on the masses, and curvature fluctuations due to other sources [239, 240, 241]. Other experiments are based on interference between masses; for example, atomic interferometers allow for measuring the curvature of space-time over a macroscopic superposition [242, 243, 244].

We can get stronger lower bounds via improved coherence experiments. Typically, such experiments aim to witness interference patterns of molecules that are as massive as possible. At the same time, we see that the experimental bound on CQ theories is generically obtained by maximizing the coherence time for massive particles with a small wave-packet size V_{λ} .

Thus far, we have considered local effects on particles due to diffusion. While this enables us to rule out some theories, the bounds are generally weak if one wants to rule out all of them. However, it may be possible to do so via cosmological considerations. In attempting to place experimental constraints on this diffusion, it is also worth considering other regimes, such as longer-range effects, which might be detected by gravitational wave detectors such as LIGO. We leave a detailed study of the effect of gravitational diffusion on cosmological scales and LIGO to future work. The effect will again depend on the form of the kernel $D_2(x, x')$. Our estimates [245] suggest that local effects from table-top experiments place a stronger bound on gravitational theories than LIGO. In particular, unlike gravitational wave measurements, which are reasonably high-frequency events requiring extraordinarily high precision in the relative displacement of the arm length from its average, it is preferential to have a lower precision measurement, which occurs over a longer period to allow for the diffusion in path length to build up, and with a smaller uncertainty in the average length of the arm itself. Furthermore, since the LIGO arm is kept in a vacuum, we do not expect strong bounds on discrete models where the diffusion is associated with an energy density.

10.4 Discussion

Several direct proposals to test the quantum nature of gravity are expected to come online in the next decade or two. These are based on detecting entanglement between mesoscopic masses inside matter-wave interferometers [64, 53, 1, 2, 65, 66, 67, 68]. For these experiments, some theoretical assumptions are needed: one requires that it is only gravitons that travel between the two masses and mediate the creation of entanglement. If this is the case, then the onset of entanglement implies that gravity is not a classical field. These can be thought of as experiments that, if successful, would confirm the quantum nature of gravity (although other alternatives to quantum theory are possible [51]).

Here, we come from the other direction by supposing that gravity is instead classical and then exploring the consequences. Theories in which gravity is fundamentally classical were thought to have been ruled out by various no-go theorems and conceptual difficulties. However, these no-go theorems are avoided if one allows for non-deterministic coupling in the dynamics.

We have here proven that this feature is necessary and made it quantitative by exploring the consequences of complete positivity on any dynamics that couples quantum and classical degrees of freedom. Complete positivity is required to ensure the probabilities of measurement outcomes remain positive throughout the dynamics. We have shown that any theory that preserves probabilities and treats one system classically must have fundamental decoherence of the quantum system and diffusion in phase space, which are signatures of information loss. Using a CQ version of the Kramers-Moyal expansion, we have derived a trade-off between decoherence on the quantum system and the system's diffusion in phase space. The trade-off is expressed in terms of the strength of the back-reaction of the quantum system on the classical one. We have derived the trade-off in terms of the theory's coupling constants and observational quantities that can be measured experimentally.

In the case of gravity, the trade-off places a lower bound on the rate of diffusion of the gravitational degrees of freedom in terms of the decoherence rate of particles in superposition. Current experiments do not rule out theories that treat gravity as fundamentally classical. However, we have been able to rule out a broad parameter space of such theories. This is done partly through table-top observations via Equations (10.32), (10.34), and (10.35). Given any diffusion kernel, we can compute the inaccuracy of mass measurements due to fluctuations in the gravitational field, and we can derive a bound on the associated decoherence rate using the trade-off. This allows us to rule out broad classes of theories in terms of their diffusion kernel. For example, we can rule out several ultra-local non-relativistic theories which backreact continuously in phase space.

Any theory which treats gravity classically has fairly limited freedom to evade the effects of the trade-off. There is the freedom to choose the diffusion or decoherence kernels $D_2(x, x')$ and $D_0(x, x')$, but the trade-off restricts one in terms of the other. Then, because of the results proven in [5], one can consider two classes of theory, those which are continuous realizations and whose diffusion can only depend on the gravitational degrees of freedom, and discrete theories whose diffusion can also depend directly on the matter fields. Chapter 9 gave examples of both theory classes.

Finally, one could consider theories that do not reproduce the weak field limit of General Relativity to all distances. We could imagine that the interaction Hamiltonian of Equation (10.24) does not hold to arbitrarily short distances or arbitrarily high mass densities. This would correspond to modifying $D_1(x, x')$ in some way, either by making it slightly non-local, by disallowing arbitrarily high mass densities, or by including an additional contribution such as a friction term. All of these modifications would seem to violate Lorentz invariance in some way. One caveat, however, to keep in mind is that relativistic corrections may affect the low-energy behavior of the theory, which we have not considered here.

We have only given an order of magnitude estimate of when gravitational diffusion will lead to appreciable deviations from Newtonian gravity or General Relativity. The most promising experiments bounding the diffusion appear to be table-top experiments that precisely measure the mass of an object. This area is important from the perspective of weight standards, for example, those undertaken by NIST on the 1kg mass standard K20 and K4 [246]. The increased precision and measuring time of Kibble Balances [247] and atomic interferometers [242, 243, 248, 249] would make such measurements an ideal testing ground, both to further constrain the diffusion kernel and to look for diffusion effects, whose dependence on the test mass is outlined in Appendix I. Here, we have found that the resolution time ΔT over which variations of acceleration are estimated affects the strength of the bound. It would be helpful if future experiments reported this value. Since we have found that CQ theories predict uncertainty in mass measurements, it is perhaps intriguing that different experiments to measure Newton's constant G yield results whose relative uncertainty differs by as much as $5 \cdot 10^{-4} \text{ m}^3 \text{kg}^{-1} \text{s}^{-2}$, which is more than an order of magnitude larger than the average reported uncertainty [226, 227, 228]. If one were to try and explain the discrepancy in G measurements via gravitational diffusion, then for all the kernels we studied in Section 10.3, we find that the variation in acceleration depends on $\frac{1}{\sqrt{N}}$ the number of nucleons in the test mass so that masses with smaller volume should yield more considerable uncertainty and this would be the effect to look for in measurement discrepancies. The relatively large uncertainty in such measurements makes it challenging for table-top experiments to place strong upper bounds on gravitational diffusion.

Now turning to the other side of the trade-off. Improved decoherence times would further squeeze theories in which gravity remains classical. While a current experimental challenge is to demonstrate interference patterns using larger and larger mass particles, we here find the bounds in Equation's (10.32) (10.34) depend on the expectation of the particle's mass density in ways which depend on the particular kernel. Thus interference experiments with particles of high mass density rather than mass can be preferable. There are also kernels, for which the relevant quantity is the expectation of the mass density, which will depend on the size of the wave-packet used in the interference experiment. This quantity is rarely obtainable from most papers that report on such experiments. While this dependence might initially appear counter-intuitive, it follows from the fact that in order to relate the trade-off in terms of coupling constants to observational quantities to get a trade-off in terms of only averages. And indeed, the decoherence rate, which is an expectation value, can easily depend on the wave-packet density, as we see from examples in Chapter 9

Since we here show that all theories which treat gravity classically necessarily decohere the quantum system, another constraint on theories that treat gravity classically is given by constraints from anomalous heating of the quantum system [206, 250, 251, 252, 253, 254, 255, 256, 257, 258, 259, 260, 261, 262]. In standard collapse models, these constraints are not in themselves very strong, since fundamental decoherence effects can be made arbitrarily weak by appropriate scaling of the decoherence parameter. Here, however, we see less freedom than one might imagine. If the Lindbladian coupling constants are made small to reduce heating, the gravitational diffusion must be large. Thus, heating constraints that place bounds on $D_0(x, x')$ place additional constraints on $D_2(x, x')$.

While the absence of diffusion could rule out theories where gravity is fundamentally classical, the presence of such deviations, at least on short time scales, might not confirm gravity's classical nature. Such effects could instead be caused by quantum theories of gravity, whose classical limit is effectively described by a CQ theory. In other words, one might expect some gravitational diffusion because, from an effective theory point of view, one is in a regime where space-time behaves classically. However, the trade-off we have derived directly results from treating the background space-time as fundamentally classical. In a fully quantum theory of gravity, the interaction of the gravitational field with particles in a superposition of two trajectories will cause decoherence. However, coherence can then be restored when the two trajectories converge. This happens when electrons interact with the electromagnetic field while passing through a diffraction grating yet still form an interference pattern at the screen. This is a non-Markovian effect, and the trade-off we derived is a direct consequence of the positivity condition, which is a direct consequence of the Markovian assumption. In the time-local non-Markovian theory where General Relativity is treated classically, one still expects the master equation to take the form found in [28], but without the matrix whose elements are $D_n^{\mu\nu}$ needing to be positive semi-definite [92, 93].

Appendix A

CQ states with continuous classical degrees of freedom

In this appendix, we formulate CQ dynamics in terms of continuous degrees of freedom more rigorously by defining a CQ state as an operator-valued measure, which formalizes the notion "to each z we associate a sub-normalized density matrix such that its trace defines a probability distribution over phase space". We then use this to argue why any completely positive CQ dynamics can be written in the form of Equation (2.80), even in the case where the classical degrees of freedom are continuous.

Let Ω be a set and \mathcal{A} be a σ algebra. A map $\varrho : \mathcal{A} \to S_{\leq \infty}(\mathcal{H})$, where $S_{\leq \infty}(\mathcal{H})$ denotes the space of un-normalized density matrices, is called a CQ state if

- 1. For each $A \in \mathcal{A}$, $\varrho(A)$ is an un-normalised, density operator on \mathcal{H} .
- 2. $\rho(\emptyset) = 0$ and $\rho(\Omega)$ is a normalised density matrix.
- 3. If E_i are disjoint then $\rho\left(\bigcup_{j=1}^{\infty} A_j\right) = \sum_{j=1}^{\infty} \rho\left(A_j\right)$
- 4. $\mu_{\varrho}(A) = \text{Tr} \left[\varrho(A) \right]$ defines a probability measure on \mathcal{A}

Now from any CQ state ϱ we can form a real-valued measure by setting $\varrho_v(A) = \langle v, \varrho(A)v \rangle$ for any $v \in \mathcal{H}$. Then, there exists a unique linear map, denoted $f \to \int_{\Omega} f \varrho(d\omega)$ with the property that

$$\left\langle v, \left(\int_{\Omega} f\varrho(d\omega)\right) v \right\rangle = \int_{\Omega} f\varrho_v(d\omega)$$
 (A.1)

for all bounded measurable complex functions $f : \Omega \to \mathbb{C}$ and all $v \in \mathcal{H}$, where the right-hand side of (A.1) is the ordinary Lebesgue integral; this follows in the same way as the proof of unique integration for projector-valued operators [263], after all a density matrix can be written as a sum of projectors. We can compute the operator-valued integral of an arbitrary bounded measurable function as follows. Take a sequence s_n of simple functions converging uniformly to f, then the integral of f is the limit, in the operator norm topology, of the integral of the s_n [263].

CQ evolution is a map taking CQ states to CQ states. In order to make sense of this, we need to define a notion of measurable super-operators. We first define the space of completely positive super-operators, which maps $S_{\leq\infty}(\mathcal{H})$ to itself as P. We want to write ultimately

$$\varrho'(A) = \int_{\Omega} \Lambda(\omega, A)(\varrho(d\omega)), \tag{A.2}$$

where $\Lambda : \Omega \times \mathcal{A} \to P$ is such that $\Lambda(\omega, A)$ is a completely positive super-operator for $\omega \in \Omega$ and $A \in \mathcal{A}$. In the rest of the section, and occasionally in the next section, we write the integral in Equation (A.2) as

$$\varrho(z) = \int dz' \Lambda(z|z') \varrho(z'). \tag{A.3}$$

This section aims to give a slightly more precise definition of the integral so that we can prove a CQ version of Kraus' theorem in the case where the classical degrees of freedom are continuous.

For CQ dynamics, we further ask that $\varrho'(A)$ define a CQ state. We give meaning to the integral $\varrho'(A) = \int_{\Omega} \Lambda(\omega, A)(\varrho(d\omega))$ by taking the inner product with a Hilbert space vector. In particular, if we let $\{|a\rangle\}$ denote an arbitrary basis of vectors in \mathcal{H} , and we write the super-operator as $\Lambda(\omega, A)(\varrho) = \sum_{abcd} \Lambda^{abcd}(\omega, A)|a\rangle\langle b|\varrho|c\rangle\langle d|$, then we can make sense of the integral as follows

$$\langle a|\varrho'(A)|d\rangle = \sum_{bc} \langle b| \left[\int_{\Omega} \Lambda^{abcd}(\omega, A)\varrho(d\omega) \right] |c\rangle, \tag{A.4}$$

and we loosely define measurable CQ dynamics as dynamics for which Equation (A.4) is well defined. For simplicity, we assume here that the Hilbert space dimension is finite. However, by analogy with Kraus' theorem for quantum operations, we expect that this can be extended to any bounded trace-class operation.

A.1 Proof of Kraus theorem for CQ dynamics

In this section we sketch a proof of a CQ Kraus theorem when the classical degrees of freedom are allowed to be continuous, and the Hilbert space is finite-dimensional; that is, we give an outline of a proof that every completely positive CQ map can be written in the form of Equation (2.80), with normalization conditions in Equation (2.82).

We assume we have a completely positive linear CQ map Λ . By linearity, the most general form of the dynamics can be written in the form $\varrho'(A) = \int_{\Omega} \Lambda(\omega|A)(\varrho(d\omega))$ – we take it given that $\Lambda(\omega|A)$ is measurable in the sense that (A.4) is well defined.

If it is completely positive, then it is certainly n positive. To that end, consider the Choi matrix

$$\varrho_{\bar{z}}'(A) = \int_{\Omega} \sum_{ab} (I \otimes \Lambda(\omega, A))(E_{ab} \otimes E_{ab} \delta_{\bar{z}}(d\omega)) = \sum_{ab} E_{ab} \otimes \Lambda(\bar{z}|A)(E_{ab}), \tag{A.5}$$

where $\delta_{\bar{z}}$ is the delta measure $(\delta_{\bar{z}}(A) = 1 \text{ if } \bar{z} \in A \text{ and } 0 \text{ otherwise})$, E_{ab} is the natural basis of operators on \mathcal{H} , $E_{ab} = |a\rangle\langle b|$ and $\bar{z} \in \Omega$. Since $\sum_{ab} E_{ab} \otimes E_{ab} \delta_{\bar{z}}$ is positive, and Λ is assumed to be a completely positive CQ evolution map, $\varrho'_{\bar{z}}(A)$ defines a CQ state on $\mathcal{H}_R \otimes \mathcal{H}$, where \mathcal{H}_R is a reference Hilbert space. Hence, for each $A \in \mathcal{A}$, $\varrho_{\bar{z}}(A)$ can be diagonalized

$$\varrho_{\bar{z}}'(A) = \sum_{\mu} \lambda^{\mu}(\bar{z}, A) |\phi_{\mu}(\bar{z}, A)\rangle \langle \phi_{\mu}(\bar{z}, A)|, \qquad (A.6)$$

where the eigenvalues $\lambda^{\mu}(\bar{z}, A)$ are positive for each A, \bar{z} . This is equivalent to the statement that $\int dz dz' \lambda(z, z') P_{\mu}(z, z') \geq 0$ for any positive $P_{\mu}(z, z')$. Here $|\phi_{\mu}(\bar{z}, A)\rangle$ is an element of the product Hilbert space $\mathcal{H}_R \otimes \mathcal{H}$. Now we can find the map $\Lambda(\bar{z}|A)(E_{ab})$ by projection of the Choi matrix on the reference system

$$\operatorname{Tr}_{R}\left[(E_{ba}\otimes I)\varrho_{\bar{z}}'(A)\right] = \langle a_{R}|\varrho_{\bar{z}}'(A)|b_{R}\rangle = \Lambda(\bar{z}|A)(E_{ab}).$$
(A.7)

If we define the operator $V_{\mu} : \mathcal{H} \to \mathcal{H}$ via its action on the basis of \mathcal{H} , $\{|a\rangle\}$, via $V_{\mu}(\bar{z}, A)|a\rangle = \langle a_R | \phi_{\mu}(\bar{z}, A) \rangle$ then

$$\Lambda(\bar{z}|A)(E_{ab}) = \sum_{\mu} \lambda^{\mu}(\bar{z}, A) V_{\mu}(\bar{z}, A) E_{ab} V_{\mu}^{\dagger}(\bar{z}, A)$$
(A.8)

Since \bar{z} is arbitrary and E_{ab} is a basis of operators on \mathcal{H} we conclude

$$\Lambda(\omega|A) = \sum_{\mu} \lambda^{\mu}(\omega, A) V_{\mu}(\omega, A) \odot V_{\mu}^{\dagger}(\omega, A)$$
(A.9)

for all $\omega \in \Omega$. Hence, we can write any complete positive, measurable CQ dynamics in the form

$$\varrho'(A) = \int_{\Omega} \Lambda(\omega|A)(\varrho(d\omega)) = \sum_{\mu} \int_{\Omega} \lambda^{\mu}(\omega, A) V_{\mu}(\omega, A) \varrho(d\omega) V_{\mu}^{\dagger}(\omega, A).$$
(A.10)

By changing basis to an arbitrary basis of operators $V_{\mu}(\omega, A) = U_{\mu\nu}(\omega, A)V_{\nu}^{\dagger}(\omega, A)$ the CQ map can be written in the form of (2.80).

We can recover the normalization conditions in (2.82) as follows. We first note, since the $|\phi_{\mu}(\bar{z}, A)\rangle$ are orthogonal, so are the matrices $V_{\mu}(\bar{z}, A)$

$$\sum_{a,b} \langle \phi_{\mu}(\bar{z},A) | a_R, b \rangle \langle a_R, b | \phi_{\nu}(\bar{z},A) \rangle = \operatorname{Tr}_{\mathcal{H}} \left[V_{\mu}^{\dagger}(\bar{z},A) V_{\nu}(\bar{z},A) \right] = \delta_{\mu\nu}$$
(A.11)

Finally, we note that

$$\sum_{\mu} \lambda^{\mu}(\bar{z}, A) V^{\dagger}_{\mu}(\bar{z}, A) V_{\mu}(\bar{z}, A) = \sum_{\mu ab} \lambda^{\mu}(\bar{z}, A) E_{ab} \langle \phi_{\mu}(\bar{z}, A) | (E_{ab} \otimes I) | \phi_{\mu}(\bar{z}, A) \rangle$$

$$= \sum_{ab} \operatorname{Tr}_{\mathcal{H}_{R} \otimes \mathcal{H}} \left[(E_{ab} \otimes I) \varrho_{\bar{z}}(A) \right] E_{ab}$$
(A.12)

which using equation (A.5) gives

$$\sum_{\mu} \lambda^{\mu}(\bar{z}, A) V^{\dagger}_{\mu}(\bar{z}, A) V_{\mu}(\bar{z}, A) = \sum_{\mu} \lambda^{\mu}(\bar{z}, A) I$$
(A.13)

Since, $\operatorname{Tr}_{\mathcal{H}_{\mathcal{R}}\otimes\mathcal{H}}[\varrho_{\bar{z}}(A)] = \sum_{\mu} \lambda^{\mu}(\bar{z}, A)$ defines a probability measure on \mathcal{A} we deduce that

$$\int_{\Omega} \sum_{\mu} \lambda^{\mu}(\bar{z}, d\omega) V^{\dagger}_{\mu}(\bar{z}, \omega) V_{\mu}(\bar{z}, \omega) = I$$
(A.14)

which is the normalization condition in (K.3).

The main lesson here is that once we treat the CQ state as an operator valued measure, then the intuition of $\Lambda(z|z')$ as describing a quantum operator for each z, z', holds, even when the degrees of freedom are continuous.

Appendix B

Continuous CP evolution with arbitrary Lindblad operators

We have shown that any continuous CP CQ map can be written as Equation (4.13) where the Lindblad operators are traceless. This appendix shows that one can pick arbitrary Lindblad operators, L_{α} , in (4.13), and the map will still be completely positive. In other words, we show that Equation (4.13) where the Lindblad operators are arbitrary is also completely positive, so long as the moments satisfy the positivity conditions.

We first write the arbitrary Lindblad operators, L_{α} , in terms of a set of traceless matrices $L_{\alpha} = \bar{L}_{\alpha} + b_{\alpha}\mathbb{I}$. The equation then takes the same form

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{n=2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(D^{00}_{n,i_1\dots i_n} \varrho(z,t) \right) + \frac{\partial}{\partial z_i} \left(D^{0\alpha}_{1,i} \varrho(z,t) \bar{L}^{\dagger}_{\alpha} \right) + \frac{\partial}{\partial z_i} \left(D^{\alpha 0}_{1,i} \bar{L}_{\alpha} \varrho(z,t) \right) \\ - i [H(z), \varrho(z,t)] + D^{\alpha \beta}_0(z) \bar{L}_{\alpha} \varrho(z) \bar{L}^{\dagger}_{\beta} - \frac{1}{2} D^{\alpha \beta}_0 \{ \bar{L}^{\dagger}_{\beta} \bar{L}_{\alpha}, \varrho(z) \}_+, \tag{B.1}$$

but with a re-scaled Hamiltonian

$$H(z) \to H(z) + \frac{1}{2i} (D_0^{\alpha\beta} b_\beta^* \bar{L}_\alpha - D_0^{\alpha\beta} b_\alpha \bar{L}_\beta^\dagger), \tag{B.2}$$

and a re-scaled *classical* drift coefficient

$$D_{1,i}^{00} \to D_{1,i}^{00} + D_{1,i}^{0\alpha} b_{\alpha}^* + D_{1,i}^{\alpha 0} b_{\alpha}.$$
(B.3)

We can then write the traceless Lindblad operators in terms of a basis of traceless Lindblad operators $\bar{L}_{\alpha} = V_{\alpha}^{\beta} \tilde{L}_{\beta}$ which span the same vector space. We can always choose V_{α}^{β} is invertible since \tilde{L}_{β} form a basis for the traceless operators. Defining $\tilde{D}_{n,i_1...i_n}^{00} = D_{n,i_1...i_n}^{\beta0}$, $\tilde{D}_{1,i}^{\beta0} = D_{1,i}^{\alpha0} V_{\alpha}^{\beta}$ and $\tilde{D}_{0}^{\gamma\sigma} = V_{\gamma}^{\alpha} D_{0}^{\gamma\sigma} (V^{\dagger})_{\sigma}^{\beta}$ we find the master equation takes the form

$$\frac{\partial \varrho(z,t)}{\partial t} = \sum_{n=1}^{n=2} (-1)^n \left(\frac{\partial^n}{\partial z_{i_1} \dots \partial z_{i_n}} \right) \left(\tilde{D}_{n,i_1\dots i_n}^{00} \varrho(z,t) \right) + \frac{\partial}{\partial z_i} \left(\tilde{D}_{1,i}^{0\alpha} \varrho(z,t) \tilde{L}_{\alpha}^{\dagger} \right) + \frac{\partial}{\partial z_i} \left(\tilde{D}_{1,i}^{\alpha0} \tilde{L}_{\alpha} \varrho(z,t) \right) + i [H(z), \varrho(z,t)] + \tilde{D}_0^{\alpha\beta} (z) \tilde{L}_{\alpha} \varrho(z) \tilde{L}_{\beta}^{\dagger} - \frac{1}{2} \tilde{D}_0^{\alpha\beta} \{ \tilde{L}_{\beta}^{\dagger} \tilde{L}_{\alpha}, \varrho(z) \}_+$$
(B.4)

which is now of the form in Equation (4.13). Furthermore,

$$2\tilde{D}_2 = 2D_2 \succeq D_1^{\dagger} \tilde{D}_0^{-1} \tilde{D}_1 = D_1^{\dagger} D_0^{-1} D_1, \tag{B.5}$$

where we have used the invertibility of V^{α}_{β} . Hence any equation of the form (4.13) with arbitrary Lindblad operators and coefficients satisfying the positivity conditions will be completely positive.

Appendix C

Unraveling of classical-quantum field theory

Since gravity is a field theory, we discuss unravelings in the context of fields in this appendix. The field theoretic version of the continuous master equation in (4.13) of Chapter 4 is [4]

$$\frac{\partial \varrho(z,t)}{\partial t} = -\int dx \frac{\delta}{\delta z_i(x)} \left(D_{1,i}^{00}(z;x)\varrho(z,t) \right) - \int dx dy \frac{\delta^2}{\delta z_i(x)\delta z_j(y)} \left(D_{2,ij}^{00}(z;x,y)\varrho(z,t) \right)
- i[H(z), \varrho(z,t)] + \int dx dy \left[D_0^{\alpha\beta}(z;x,y)L_\alpha(x)\varrho(z)L_\beta^{\dagger}(y) - \frac{1}{2}D_0^{\alpha\beta}(z;x,y)\{L_\beta^{\dagger}(y)L_\alpha(x),\varrho(z)\}_+ \right]
\int dx \frac{\delta}{\delta z_i(x)} \left(D_{1,i}^{0\alpha}(z;x)\varrho(z,t)L_\alpha^{\dagger}(x) \right) + \frac{\delta}{\delta z_i(x)} \left(D_{1,i}^{\alpha0}(z;x)L_\alpha(x)\varrho(z,t) \right).$$
(C.1)

The complete positivity of the dynamics is enforced by the positivity of the matrix

$$\int dx dy [b^*(x), \alpha^*(x)] \begin{bmatrix} 2D_2(x, y) & D_1(x, y) \\ D_1(x, y) & D_0(x, y) \end{bmatrix} \begin{bmatrix} b(y) \\ \alpha(y) \end{bmatrix} \ge 0,$$
(C.2)

which should be positive for any position dependent vectors $b^i_{\mu}(x)$ and $a_{\alpha}(x)$ [4]. Using the same methods as for the derivation of the unraveling in the case of continuous classical-quantum dynamics but replacing derivatives by functional derivatives, Equation (C.1) can be unraveled
by the coupled stochastic differential equations

$$d\rho_{t} = \mathcal{L}(Z_{t})(\rho_{t})dt + \int dxdy \left[D_{1,j}^{\alpha 0}(Z_{t};x)\sigma_{ij}^{-1}(Z_{t};x,y)(L_{\alpha}(x) - \langle L_{\alpha}(x) \rangle)\rho_{t}dW_{i}(y) + D_{1,j}^{0\alpha}(Z_{t};x)\sigma_{ij}^{-1}(Z_{t};x,y)\rho_{t}(L_{\alpha}^{\dagger}(x) - \langle L_{\alpha}^{\dagger}(x) \rangle)dW_{i}(y) \right] \\ dZ_{t,i}(x) = (D_{1,i}^{00}(Z_{t};x) + \langle D_{1,i}^{\alpha 0}(Z_{t};x)L_{\alpha}(x) + D_{1,i}^{0\alpha}(Z_{t};x)L_{\alpha}^{\dagger}(x) \rangle)dt + \int dy\sigma_{ij}(Z_{t};x,y)dW_{j}(y).$$
(C.3)

Where now $W_i(x)$ is a spatially dependent Wiener process satisfying

$$\mathbb{E}[W_i(x)] = 0, \quad \mathbb{E}[dW_i(x)dW_j(y)] = \delta_{ij}\delta(x,y)dt, \quad (C.4)$$

and we have used the notation $\mathcal{L}(Z)(\rho)$ as shorthand for the pure Lindbladian term appearing in Equation (C.1). In (C.3) $\sigma^{-1}(x,y)$ denotes the generalized kernel inverse, so that $\int dy dz \ \sigma(x,y)\sigma^{-1}(y,z)\sigma(z,w) = \sigma(z,w)$. The equations will be local if one picks $\sigma(x,y) \sim \delta(x,y)$, but we can also allow for the more general case. In Equation (C.4) $\frac{dW_i(x)}{dt}$ is a white noise process, and as a result, the solutions to the dynamics will, in general, require regularization. Studying this is beyond the scope of the current work. However, a promising approach would be to study the renormalization properties of classical-quantum path integrals in Chapters 6 and 8.

C.1 A gravitational CQ theory example

As an example, we can use the theory of [28] to formally write down dynamics that agree with the ADM equations of motion on expectation for minimally coupled matter (we consider the Newtonian limit of this theory elsewhere [9, 4]). The idea of [28] was to assume that the dynamics are approximately Einstein's gravity in the classical limit. Here we see that this fixes the drift terms so that the Hamiltonian form of the semi-classical Einstein's equations (1.1) are obeyed on average, and we arrive at a general form for the evolution of the 3-metric γ_{ij} and its conjugate momenta π^{ij}

$$\begin{split} d\gamma_{ij} &= \{\gamma_{ij}, H_{ADM}[N, N]\} dt, \\ d\pi^{ij} &= \{\pi^{ij}, H_{ADM}[N, \vec{N}]\} dt - \langle \frac{\delta \hat{H}^{(m)}[N, \vec{N}]}{\delta \gamma_{ij}} \rangle dt + \int dy \sigma_{kl}^{ij}[\gamma, \pi; x, y] dW_{kl}(x) \\ d\rho_t &= -i [\hat{H}^{(m)}[N, \vec{N}], \rho_t] - \frac{1}{2} \int dx dy D_0^{ij;kl}[\pi; x, y] [\frac{\partial \hat{H}^{(m)}[N, \vec{N}]}{\delta \gamma_{ij}(x)}, [\frac{\partial H_m}{\delta \gamma_{kl}(y)}, \rho_t]] dt, \qquad (C.5) \\ &+ \frac{1}{2} \int dy (\sigma^{-1})_{ij}^{kl}[\gamma, \pi; x, y] (\frac{\delta \hat{H}^{(m)}[N, \vec{N}]}{\delta \gamma_{ij}(y)} - \langle \frac{\delta \hat{H}^{(m)}[N, \vec{N}]}{\delta \gamma_{ij}(y)} \rangle) \rho_t dW_{kl}(y) \\ &+ \frac{1}{2} \int dy (\sigma^{-1})_{ij}^{kl}[\gamma, \pi; x, y] \rho_t (\frac{\delta \hat{H}^{(m)}[N, \vec{N}]}{\delta \gamma_{ij}(y)} - \langle \frac{\delta \hat{H}^{(m)}[N, \vec{N}]}{\delta \gamma_{ij}(y)} \rangle) dW_{kl}(y). \end{split}$$

We obtain the semi-classical equation (5.36) when the dynamics are taken to be ultra-local, $\sigma \sim \delta(x, y)$, where we use equations of motion to invert $\pi_{ij}[\dot{\gamma}]$ and obtain the expression for G_{ij} . Equation (5.36) is sourced by a white noise term, since $G_{ij} \sim \ddot{\gamma}_{ij} \sim \frac{dW_{kl}}{dt}$ which is a white noise process. In Equation (C.5), we can also consider the case where the lapse and shift N, N^i and their conjugate momenta are included as phase space degrees of freedom. While adding them does nothing in the purely classical case when the constraints are satisfied, it does have some advantages concerning the weak field limit [9] and the constraint algebra [3] of the CQ theory. In this case, one has additional diffusion and Lindbladian terms. We could also add a term that describes any information loss, classical or quantum, not due to the decoherence diffusion trade-off, but we have omitted such terms.

The dynamics will generally depend on the lapse and shift functions N, N^i as in the Hamiltonian formulation of general relativity. On each realization of the noise process, we now have entire trajectories for each of the variables $(\gamma_{ij}, \pi^{ij}, N, N^i)$ each associated with a quantum state, $\rho(t|\gamma_{ij}, \pi^{ij}, N, N^i)$. This allows us to define a tuple $(g_{\mu\nu}, \rho_{\Sigma_t}(t))$ via the ADM embedding

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -N^{2}(t,x)dt^{2} + \gamma_{ij}(t,x)(N^{i}(t,x)dt + dx^{i})(N^{j}(t,x)dt + dx^{j}).$$
(C.6)

This associates to each trajectory a 4-metric and quantum state on a 1-parameter family of hypersurfaces Σ . The unraveling provides a method for studying the dynamics of classical gravity interacting with quantum matter.

In the gravitational context, one also expects that one should consider theories that retain diffeomorphism symmetry. Treated as an effective theory, we expect that the coefficients D_0, D_1, D_2 entering into the dynamics result from integrating out high energy modes of dynamical fields. As such, we should demand that an effective theory be diffeomorphism *covariant*, meaning that a solution to the dynamics in one frame should be a solution to the dynamics in any other frame, where we should transform the parameters entering into the master equation by hand since they arise from hidden dynamical degrees of freedom.

On the other hand, if there are no degrees of freedom that have been integrated out, as would be the case in a fundamental theory of classical-quantum gravity, it is natural to impose diffeomorphism invariance on the dynamics and the coefficients entering the master equation should be constructed out of the gravitational degrees of freedom alone. It remains to be seen if such dynamics can be made diffeomorphism invariant and give rise to full general relativity with preserved constraints. However, we progressed towards this in Chapter 8. Such dynamics could, in principle, be taken as fundamental, and we leave it as an open question of whether or not such dynamics can be made diffeomorphism invariant with a full classical-quantum constraint surface.

Appendix D

Perturbative methods for CQ path integrals

In this appendix, we study a simple model of CQ interaction to illustrate how one can use standard perturbative methods to calculate classical-quantum correlation functions via CQ Feynman diagrams.

In Chapters 6 and 8, we considered the path integral, which constructs a CQ state at a time t_f from a CQ state at time t_i . In computing correlation functions of classical-quantum observables, the final state is not important, so we can perform a t_f integral over the classical-quantum fields to arrive at the partition function

$$Z = \int dq_f \operatorname{Tr} \left[\varrho(q_f, t_f) \right]$$
(D.1)

which for the configuration space path integral takes the form

$$Z_0 = \mathcal{N} \int \mathcal{D}\phi^- \mathcal{D}\phi^+ \mathcal{D}q \ e^{\mathcal{I}(\phi^-, \phi^+, q, t_i, t_f)} \varrho(q_i, \phi_i^+, \phi_i^-, t_i),$$
(D.2)

where now there are no final boundary conditions imposed on the path integral.

Formally, we can calculate correlation functions by inserting sources J^+ , J^- , J_q into the path integral and taking functional derivatives with respect to the sources. The partition function of interest is, therefore

$$Z[J^+, J^-, J_q] = \mathcal{N} \int \mathcal{D}\phi^- \mathcal{D}\phi^+ \mathcal{D}q \ e^{\mathcal{I}(\phi^-, \phi^+, q, t_i, t_f) - iJ_+\phi^+ + iJ^-\phi^- - J_q q} \varrho(q_i, \phi_i^+, \phi_i^-, t_i).$$
(D.3)

In general, the form of the path integral depends on the initial CQ state $\rho(q_i, \phi_i^+, \phi_i^-, t_i)$ and any calculation of correlation function must be performed on a case-by-case basis depending on the initial state.

However, often we are interested in stationary states, and we would like to obtain information on correlation functions over arbitrary long times by taking the limit $t_i \to -\infty, t_f \to \infty$. In open systems, as well as when calculating scattering amplitudes, it is often assumed that the initial state in the infinite past does not affect the stationary state of the system so that there is a complete loss of memory of the initial state [97]. Under this assumption, it is possible to ignore the boundary term containing the initial CQ state $\rho(q_i, \phi_i^+, \phi_i^-, t_i)$ and we arrive at the partition function

$$Z[J^+, J^-, J_q] = \mathcal{N} \int \mathcal{D}\phi^- \mathcal{D}\phi^+ \mathcal{D}q \ e^{\mathcal{I}(\phi^-, \phi^+, q, -\infty, \infty) - iJ_+\phi^+ + iJ^-\phi^- - J_q q}.$$
 (D.4)

Using equation (D.4), we can then use standard perturbation methods for computing correlation functions in CQ theories.

As a simple example, consider the zero-dimensional CQ theory with CQ proto-action

$$W_{CQ} = S_Q - \frac{m_q^2 q^2}{2} - \frac{\lambda q^2 \phi^2}{2},$$
 (D.5)

and a pure quantum action given by $S_Q = -\frac{m_{\phi}^2 \phi^2}{2}$. Assuming the decoherence diffusion trade-off is saturated, we arrive at the total action

$$\mathcal{I}[\phi^{\pm},q] = -\frac{i}{\hbar} \frac{m_{\phi}^2(\phi^+)^2}{2} + \frac{i}{\hbar} \frac{m_{\phi}^2(\phi^-)^2}{2} - \frac{1}{4D_2} \left(q^2 m_q^4 + \frac{1}{2} \lambda^2 q^2 ((\phi^+)^4 + (\phi^-)^4) + \frac{1}{2} \lambda q m_q^2 ((\phi^+)^2 + (\phi^-)^2) \right)$$
(D.6)

We see from Equation (D.6) that D_2 in an interacting CQ theory plays the same role as \hbar in an interacting quantum theory. To compute correlation functions, we can therefore work perturbatively in D_2 . Note, the double limit $D_2 \rightarrow 0, D_2^{-1}\lambda \rightarrow 0$ defines a deterministic quantum theory with no classical back-reaction.

We define the free theory as the action independent of any CQ back-reaction

$$I_{free} = iS^{+} - iS^{-} - I_{C} = -i\frac{m_{\phi}^{2}(\phi^{+})^{2}}{2\hbar} + i\frac{m_{\phi}^{2}(\phi^{-})^{2}}{2\hbar} - \frac{1}{4D_{2}}q^{2}m_{q}^{4}.$$
 (D.7)

Inserting sources, we find the partition function

$$Z_{free}[J_+, J_-, J_q] = \int d\phi^{\pm} dq e^{I_{free} - iJ_+ \phi^+ iJ_- \phi^- - J_q q},$$
 (D.8)

which can be performed exactly by performing each Gaussian integral individually

$$Z_{free}[J_+, J_-, J_q] = \left(\int d\phi^+ e^{-i\frac{m_{\phi}^2(\phi^+)^2}{2\hbar} - iJ_+\phi^+}\right)\left(\int d\phi^- e^{+i\frac{m_{\phi}^2(\phi^-)^2}{2\hbar} + iJ_-\phi^-}\right)\left(\int dq e^{-\frac{1}{4D_2}q^2m_q^4 - J_qq}\right).$$
(D.9)

Equation (D.9) is evaluated as

$$Z_{free}[J_+, J_-, J_q] = \left(\frac{-2\pi i\hbar}{m_{\phi}^2}\right) e^{\frac{iJ_+^2}{2\hbar m_{\phi}^2}} \left(\frac{2\pi i\hbar}{m_{\phi}^2}\right) e^{\frac{-iJ_-^2}{2\hbar m_{\phi}^2}} \left(\frac{\pi 2D_2}{m_q^4}\right) e^{\frac{J_q^2}{4D_2m_q^4}} = Z_0 e^{\frac{iJ_+^2}{2\hbar m_{\phi}^2}} e^{\frac{-iJ_-^2}{2\hbar m_{\phi}^2}} e^{\frac{J_q^2}{4D_2m_q^4}}.$$
 (D.10)

From Equation (D.10), we can then define the propagators for the free theory

$$\langle \phi^+ \phi^+ \rangle = -\frac{i\hbar}{m_{\phi}^2}, \ \langle \phi^- \phi^- \rangle = \frac{i\hbar}{m_{\phi}^2}, \ \langle qq \rangle = \frac{2D_2}{m_q^4}, \tag{D.11}$$

and we can represent each of the propagators by the following Feynman diagrams

$$\phi^{+} - \frac{i\hbar}{m_{\phi}^{2}} \phi^{+} \phi^{-} \frac{i\hbar}{m_{\phi}^{2}} \phi^{-} q \bullet \frac{2D_{2}}{m_{q}^{4}} q \qquad (D.12)$$

The full partition function with the CQ interaction turned on then takes the form

$$Z[J_+, J_-, J_q] = \langle e^{\mathcal{I}_{CQ}} \rangle = \langle e^{-\frac{1}{4D_2} \left(\frac{1}{2}\lambda^2 q^2 ((\phi^+)^4 + (\phi^-)^4) + \frac{1}{2}\lambda q m_q^2 ((\phi^+)^2 + (\phi^-)^2)\right)} \rangle$$
(D.13)

and we can perform an asymptotic expansion of the CQ interaction in terms of D_2 to arrive at the usual Feynman rules for computing correlation functions. Specifically, for terms in the action like $\lambda_{nml}\phi^n_+\phi^m_-q^l$, we assign the vertex with value $\lambda_{nml}n!m!l!$ to each topologically distinct diagram.

As an example, the CQ interaction term $q(\phi^{\pm})^2$ in Equation (D.6) has two tri-vertices with strength $-\frac{2!}{8D_2}\lambda m_q^2$ and can be represented by the diagrams



We also have the sextic $q^2(\phi^{\pm})^4$ interaction with vertex value $-\frac{\lambda^2 4! 2!}{8D_2}$ which is assigned to each of the following diagrams



Appendix E

Complete postivity of classical-quantum path integrals

In this appendix, we prove the statement made in Chapter 8 that Equation (8.4) defines completely positive CQ dynamics and that the dynamics defined by Equation (8.20) takes the form of Equation (8.4) and is hence completely positive. We also show that normalization of the path integral occurs via the inclusion of appropriate classical and quantum kinetic terms in Section E.3.

E.1 Proof of complete positivty

The proof of complete positivity of Equation (8.9) is almost immediate from an expansion of the path integral. To see this in detail, we can perform a short-time expansion of the full path integral, which we can always do since we assume the dynamics are time-local.

Let us first consider the case where the quantum state remains pure, so that $c^{\gamma} = 0$ in Equation (8.9). Defining $t_f = t_0 + K\delta t$, $t_i = t_0 + i\delta t$, and discretizing the path integral into steps of size δt we have that

$$\varrho_{i+1} = \int d\phi_i^+ d\phi_i^- dz_i (e^{\mathcal{I}_{i+1,i}^+}) (e^{\mathcal{I}_{i+1,i}^-})^* e^{-\mathcal{I}_{C,i+1,i}} \varrho_i,$$
(E.1)

where we use the shorthand $\varrho_i = \varrho(\phi_i^+, \phi_i^-, z_i, t_i), \ \mathcal{I}_{i+1,i} = \mathcal{I}[\phi_{i+1}, \phi_i, z_{i+1}, z_i] \text{ and } \mathcal{I}_{C,i+1,i} = \mathcal{I}_C[z_{i+1}, z_i].$

More generally, we can allow for the case where the action contains higher time derivatives, in which case we have $\mathcal{I}[\phi_{i+k_q}, \ldots, \phi_i, z_{i+k_c}, \ldots, z_i]$ and $\mathcal{I}_C[z_{i+k_c}, z_i]$ with $k_c, k_q \geq 2$. In order to retain the usual composition law for the path integral, we must also let increasingly higher derivative terms describe the state $\varrho(\phi_i^{\pm}, \ldots, \frac{d^{k_q-1}\phi_i^{\pm}}{dt^{k_q-1}}, z_i, \frac{d^{k_c-1}z_i}{dt^{k_c-1}})$. [200]. The final state then imposes boundary conditions on the components of the action, which contain higher derivative terms so that Equation (E.1) is still well defined.

With this in mind, we can take the trace with respect to an arbitrary vector $|v(q)\rangle$, and for complete positivity, we need to show

$$\int d\phi_{i+1}^+ d\phi_{i+1}^- v_{i+1}^{+*} \varrho_{i+1} v_{i+1}^- \ge 0.$$
(E.2)

Denoting $\tilde{v}_{i+1,i}^+ = e^{\mathcal{I}_{i+1,i}^+} v_{i+1}^{+*}$, then inserting Equation (E.1) into Equation (E.2) we have

$$\int d\phi_{i+1}^+ d\phi_{i+1}^- d\phi_i^+ d\phi_i^- dz_i \tilde{v}_{i+1,i}^+ \tilde{v}_{i+1,i}^{-*} e^{-\mathcal{I}_{C,i+1,i}} \varrho_i.$$
(E.3)

Because the integral factorizes into \pm conjugates, Equation (E.3) will always be positive. To see this explicitly, we first perform the ϕ_i^{\pm} integrals to obtain

$$c_{i+1} = \int d\phi_i^+ d\phi_i^- \tilde{v}_{i+1,i}^+ \tilde{v}_{i+1,i}^{-*} \varrho_{\cdot} \ge 0,$$
(E.4)

where we have used the positivity of the state CQ $\rho(\phi_i^+, \phi_i^-, q_i, t_i)$. What remains is the integral

$$\int d\phi_{i+1}^+ d\phi_{i+1}^- dz_i e^{-\mathcal{I}_{C,i+1,i}} c_{i+1} \ge 0,$$
(E.5)

which is positive since both c_{i+1} and the exponential are both positive. In Equation (E.5), there is still a free z_{i+1} variable, which corresponds to the fact that positivity of the CQ state demands that the CQ dynamics keep quantum states positive conditioned on the classical degrees of freedom. We thus see that the state after applying the time-evolved state will also be positive. Hence the dynamics are positive. When we consider the dynamics as part of a larger system, we apply the identity map to the larger system. The dynamics still factorize this way - we perform a delta function path integral on the auxiliary system. Hence, Equation (6.67) defines completely positive dynamics.

In the more general case, we can have non-zero $c^{\gamma}[z, x]$, and there is information loss since the dynamics can send pure states to mixed states. In this case, the only thing which changes is the definition of $\tilde{v}_{i+1,i}^+$ in Equation (E.2). In particular, in the general case, we must also expand out the terms involving c_n^{γ} in the action of Equation (8.4)

$$e^{\delta t \sum_{\gamma} c_{i+1,i}^{\gamma} (L_{\gamma}^{+})_{i+1,i} (L_{\gamma}^{-})_{i+1,i}^{*}} = 1 + \delta t \sum_{\gamma} c_{i+1,i}^{\gamma} (L_{\gamma}^{+})_{i+1,i} (L_{\gamma}^{-})_{i+1,i}^{*} + \dots,$$
(E.6)

where $c_{i+1,i}^{\gamma} = c^{\gamma}[q_{i+1}, q_i], L_{i+1,i}^+ = L^+[\phi_{i+1}^+, \phi_i^+]$ and similarly for the ⁻ branch.

With this in mind, the integrand of the path integral in Equation (6.67) factorizes according to Equation (E.3), and the steps to prove complete positivity are the same but with

$$\tilde{v}_{i+1,i}^{\gamma+} = (\sqrt{\delta t c_{i+1,i}^{\gamma}} (L_{\gamma}^{+})_{i+1,i}) e^{\mathcal{I}_{i+1,i}^{+}} v_{i+1}^{+*}, \tag{E.7}$$

from which the complete positivity of the dynamics follows from the same arguments outlined in Equation's (E.3) and (E.4), where we now also sum over γ . Note, though we need only work to first order in δt , had we included them, the higher order δt terms also factorize similarly.

In the field-theoretic case, the total CQ action is

$$\mathcal{I}(\phi^{-},\phi^{+},q,t_{i},t_{f}) = \mathcal{I}_{CQ}(q,\phi^{+},t_{i},t_{f}) + \mathcal{I}_{CQ}^{*}(q,\phi^{-},t_{i},t_{f}) - \mathcal{I}_{C}(q,t_{i},t_{f}) + \int_{t_{i}}^{t_{f}} dx \sum_{\gamma} c^{\gamma}(q,t,x) (L_{\gamma}[\phi^{+}](x)L_{\gamma}^{*}[\phi^{-}](x))$$
(E.8)

and we can repeat the argument for complete positivity, which again follows from the factorization of the path integral integrand. In this case, complete positivity follows from the fact that

$$\int \mathcal{D}\phi_{i+1}^{+} \mathcal{D}\phi_{i+1}^{-} \mathcal{D}\phi_{i}^{+} \mathcal{D}\phi_{i}^{-} \mathcal{D}z_{i} \times \sum_{\gamma} \int d\vec{x} c_{i+1,i}^{\gamma}(x) \left(v_{i+1,i}^{+}(L_{\gamma}^{+})_{i+1,i}(x)e^{\mathcal{I}_{i+1,i}^{+}}\right) \left(L_{\gamma}^{-}\right)_{i+1,i}(x)v_{i+1,i}^{-}e^{\mathcal{I}_{i+1,i}^{-}}\right)^{*} e^{-\mathcal{I}_{C,i+1,i}} \varrho_{i},$$
(E.9)

is positive when $c^{\gamma} \geq 0$ and ρ_i is a positive density matrix.

E.2 Showing the natural class of CQ dynamics is CP

In this section, we show that the dynamics defined by Equation (8.20) takes the form of Equation (8.4) and is hence completely positive. Since the purely quantum Lagrangian terms appearing in Equation (8.4) are manifestly CP, we shall focus on the CQ interaction term

$$-\frac{1}{2}\int dx\Delta X^{i}(q,x)D_{0,ij}(q,x)\Delta X^{j}(x) - \frac{1}{4}\int dx\bar{X}^{i}(q,x)D_{2,ij}^{-1}(q,x)\bar{X}^{j}(q,x),$$
(E.10)

where we use the shorthand notation $X^i(q, x) = \frac{\delta W_{CQ}}{\delta q_i(x)}$, which we assume is Hermitian since it is generated by a real proto-action W_{CQ} . For ease of presentation, we will here suppress any potential q, x dependence from X^i, D_0, D_2 , but these can be added back in.

Expanding Equation (E.10), we can group terms according to D_0, D_2^{-1} as

$$-\frac{1}{16}\int dx d\vec{y} (8D_{0,ij} + D_{2,ij}^{-1})((X^+)^i (X^+)^j + (X^-)^i (X^-)^j) +\frac{1}{16}\int dx (8D_{0,ij} - D_{2,ij}^{-1})((X^+)^i (X^-)^j + (X^-)^i (X^+)^j).$$
(E.11)

We see that the first line in Equation (E.11) is of the form $I_{CQ}^+ + (I_{CQ}^-)^*$ and so adheres to the form in Equation (8.4). If the trade-off is saturated, this completes the proof that (8.20) takes the form of Equation (8.4). When not saturated, we can write $8D_0 - D_2 = c \succeq 0$, where $c_{ij}(q, x)$ is a real, symmetric positive semi-definite matrix. We can then expand the second line of Equation (E.11) as

$$\frac{1}{16} \int dx \ c_{ij}(q,x)((X^+)^i (X^-)^j + (X^-)^i (X^+)^j), \tag{E.12}$$

which, after diagonalizing c_{ij} , takes the form of Equation (E.8), and hence defines CP dynamics whenever the condition $8D_0 - D_2 = c \succeq 0$ is satisfied.

When the trade-off $8D_0 - D_2 = c \succeq 0$ is saturated, i.e., when c = 0, we can reproduce the full action for the gravity theory of Equation (8.33)

$$\mathcal{I}[\phi^{-},\phi^{+},g_{\mu\nu}] = \int dx \bigg[i\mathcal{L}_{KG}^{+} - i\mathcal{L}_{KG}^{-} - \frac{\det(-g)}{8} (T^{\mu\nu+} - T^{\mu\nu-}) D_{0,\mu\nu\rho\sigma} (T^{\rho\sigma+} - T^{\rho\sigma-}) \\ - \frac{\det(-g)}{128\pi^{2}} (G^{\mu\nu} - \frac{1}{2} (8\pi (T^{\mu\nu})^{+} + 8\pi (T^{\mu\nu})^{-}) D_{0,\mu\nu\rho\sigma} [g] (G^{\rho\sigma} - \frac{1}{2} (8\pi (T^{\rho\sigma})^{+} + 8\pi (T^{\rho\sigma})^{-}) \bigg],$$
(E.13)

which we have now verified takes the form of Equation (8.4).

E.3 Ensuring the CQ path integral is normalized

In this section, we show that the CQ action defined by Equation's (8.22) and (8.21) are normalized.

To see the problem of normalization of higher derivative path integrals in more detail, we will review how the normalization of quantum states occurs in Lindbladian path integrals with a Feynman-Vernon action [79], and how probabilities are conserved in higher-order classical path integrals. Let us first consider higher-order classical path integrals. We refer the reader to Chapter 6 for a complete derivation of normalized CQ path integrals from master equations.

E.3.1 Normalization of higher derivative classical path integrals

When considering a classical path integral that contains higher derivatives, we should treat q, \dot{q} as independent variables. This is outlined in detail in [200]. To that end, we will show how the normalization of the path integral

$$p(q_f, \dot{q}_f, t_f) = \int^B \mathcal{D}q e^{-\int_{t_i}^{t_f} dt [\ddot{q} - f(\dot{q}, q)]^2} p(q_i, t_i)$$
(E.14)

occurs. In Equation (E.14), note that the boundary conditions are given by $B = \{q(t_f) = q_f, \dot{q}(t_f) = \dot{q}_f\}$, which involve both q and \dot{q} .

To check normalization, we consider Equation (E.14) for small δt , with $t_n = \delta t + t_{n-1}$

$$p(q_{n+1}, q_{n+2}, t_{n+1}) = \int dq_n e^{-\delta t \left[\frac{q_{n+2}-2q_n+q_{n+1}}{\delta t} - f(q_{n+1}, q_n)\right]^2} p(q_n, q_{n+1}, t_n).$$
(E.15)

The norm of the probability distribution is found by performing the integral over the final variables q_{n+1}, q_{n+2}

$$\int dq_n dq_{n+1} dq_{n+1} e^{-\delta t \left[\frac{q_{n+2}-2q_n+q_{n+1}}{\delta t} - f(q_{n+1},q_n)\right]^2} \times p(q_n, q_{n+1}, t_n).$$
(E.16)

Equation (E.16) defines a standard Gaussian integral over the q_{n+2} coordinate. Hence, the q_{n+2} integral eats the action up to a Gaussian normalization factor that we can calculate exactly, and we are left with

$$1 = \int dq_n dq_{n+1} N p(q_n, q_{n+1}, t_n),$$
(E.17)

so we can simply absorb N into the measure, and the path integral will be normalized. If we were to include a q dependent diffusion coefficient $D_2(q, \dot{q})$ in Equation (E.14), then the Gaussian integral will be q dependent, and this will need to be included in the measure for $\mathcal{D}q$. One can also re-exponentiate this term via a Faddeev-Popov action, as in Equation (8.23). The message is that the higher derivative terms in the classical path integral are standard Gaussian integrals if we consider q and \dot{q} as independent variables. Hence, the classical contribution to the path integral defined by Equation (8.20), which also involves the \pm bra-ket branch average, will be normalized in a similar way. We show this explicitly in Section E.3.3

E.3.2 Normalization of higher derivative Feynman-Vernon path integrals

Let us now consider a Feynman-Vernon quantum path integral with a decoherence term. Consider first the path integral for a quantum state σ

$$\sigma(\phi_f^+, \phi_f^-, t_f) = \int^B \mathcal{D}\phi^+ \mathcal{D}\phi^- e^{\int_{t_i}^{t_f} dt i [\dot{\phi_+}^2 + V(\phi_+)] - i [\dot{\phi_-}^2 + V(\phi_-)] - \frac{D_0}{2} (L(\phi_+) - L(\phi_-))^2} \sigma(\phi_i^+, \phi_i^-, t_i),$$
(E.18)

where B imposes the final state boundary conditions on the bra and ket fields, and $L(\phi)$ is an arbitrary operator of ϕ but not of its derivatives.

For Equation (E.18), it will prove insightful to show how the kinetic term enforces the normalization of the quantum state. To that end, consider the short time version of Equation (E.19)

$$\sigma(\phi_{n+1}^{+}, \phi_{n+1}^{-}, t_{f}) = \int d\phi_{n}^{+} d\phi_{n}^{-} e^{\delta t [i(\frac{\phi_{n+1}^{+} - \phi_{n}^{+}}{\delta t})^{2} + iV(\phi_{n}^{+}) - i(\frac{\phi_{n+1}^{-} - \phi_{n}^{-}}{\delta t})^{2} - iV(\phi_{n}^{-})]} \times e^{-\delta t \frac{D_{0}}{2} (L(\phi_{n}^{+}) - L(\phi_{n}^{-}))^{2}} \sigma(\phi_{n}^{+}, \phi_{n}^{-}, t_{n}).$$
(E.19)

The trace of the quantum state is found by matching the $\phi_{n+1}^+ = \phi_{n+1}^- = \phi$ fields and integrating over ϕ

$$\int d\phi_n^+ d\phi_n^- d\phi e^{\frac{i}{\delta t}\phi(\phi_n^+ - \phi_n^-)} e^{i\delta t [V(\phi_n^+) - iV(\phi_n^-)]} e^{-\delta t \frac{D_0}{2} (L(\phi_n^+) - L(\phi_n^-))^2} \sigma(\phi_n^+, \phi_n^-, t_n).$$
(E.20)

Performing the integration over ϕ gives rise to a delta function $\delta(\phi_n^+ - \phi_n^-)$. Hence, the quantum state is normalized to constant factors that can be absorbed.

However, had we included higher-order kinetic terms in the decoherence sector, we would not have found this normalization. In particular, if the decoherence term was instead

$$\int_{t_i}^{t_f} dt \frac{1}{2} D_0 (\dot{\phi}_+^2 - \dot{\phi}_-^2)^2, \qquad (E.21)$$

then the delta function integral is not imposed, and the state is no longer normalized to constant factors.

As such, it seems inevitable that for the path integral to be normalized with higher derivative decoherence terms, one needs to also add higher derivative kinetic terms in the action. In this case, the action

$$S = \int_{t_i}^{t_f} dt i [\dot{\phi_+}^2 + \ddot{\phi_+}^2 - V(\phi_-)] - i [\dot{\phi_-}^2 + \ddot{\phi_-}^2 - V(\phi_-)] - \frac{1}{2} D_0 (\dot{\phi_+}^2 - \dot{\phi_-}^2)^2$$
(E.22)

is normalized up to constant factors by the same argument, so long as we treat ϕ and $\dot{\phi}$ as independent variables to be specified in the quantum state; this is also argued for independent reasons in [200]. To see this, one does the short time expansion, treating ϕ and $\dot{\phi}$ as independent variables as in the higher derivative classical path integral. Computing the trace then sets ϕ_{n+2}^{\pm} equal to each other, as well as the setting the ϕ_{n+1}^{\pm} fields equal to be the same. The ϕ_{n+2} integral then enforces a delta function over $\delta(\phi_n^+ - \phi_n^-)$, which kills the decoherence term and means that the path integral is normalized up to constant factors.

E.3.3 Normalization of CQ path integrals

In this section, we show that Equation's (8.21) give rise to normalized CQ dynamics (8.22). The proofs will follow in the same way as the discussion of normalization of classical and quantum path integrals.

Let us start with the higher derivative case. We show that any CQ path integral with action

$$\begin{split} I[q,\phi^{+},\phi^{-}] &= \int dt i \ddot{\phi}_{+}^{2} + i V(\phi^{+},\dot{\phi}^{+}) - i \ddot{\phi}_{-}^{2} - i V(\phi^{-},\dot{\phi}^{-}) \\ &- \frac{D_{0}(q,\dot{q},\phi^{+},\dot{\phi}^{+})}{2} (\ddot{q} + f(q,\dot{q},\phi^{+},\dot{\phi}^{+}))^{2} - \frac{D_{0}(q,\dot{q},\phi^{-},\dot{\phi}^{-})}{2} (\ddot{q} + f(q,\dot{q},\phi^{-},\dot{\phi}^{-}))^{2} \end{split}$$
(E.23)

is normalized up-to constant factors when $D_0 > 0$. In the case where D_0 has a functional dependence on the fields one must make sure to also include a factor of $\sqrt{\det(D_0(q, \dot{q}, \phi))}$ in the path integral measure. Equation (E.23) is a generic type of action one gets from varying Equation (8.20) with a CQ proto action that has second order equations of motion for the classical degree of freedom.

The steps in showing Equation (E.23) follow in the same way as the discussions of classical and quantum path integrals. Firstly, because the action is higher derivative, the CQ state is specified through $\rho(q, \dot{q}, \phi^{\pm}, \dot{\phi}^{\pm})$.

Taking the trace at the $t_{n+1} = t_n + \delta t$ time-step therefore involves identifying $\phi_{n+2}^+ = \phi_{n+2}^- = \phi_{n+2}$ and $\phi_{n+1}^+ = \phi_{n+1}^- = \phi_{n+1}$. We then integrate over the ϕ_{n+2} and ϕ_{n+1} variables, as well as over the q_{n+2}, q_{n+1} classical degrees of freedom.

Let us first look at the higher derivative quantum kinetic term. This can be expanded as

$$\ddot{\phi}_{+}^{2} - \ddot{\phi}_{-}^{2} \sim (\phi_{n+2} - 2\phi_{n}^{+} + \phi_{n+1})^{2} - (\phi_{n+2} - 2\phi_{n}^{-} + \phi_{n+1})^{2}$$

$$= 4 \left[(\phi_{n}^{+})^{2} - (\phi_{n}^{-})^{2} + \phi_{n+2}(\phi_{n}^{-} - \phi_{n}^{+}) + \phi_{n+1}(\phi_{n}^{-} - \phi_{n}^{+}) \right].$$
(E.24)

Hence, integrating over ϕ_{n+2} gives a delta function in $\delta(\phi_n^- - \phi_n^+)$. As a consequence of this, all the bra and ket fields in the path integral are identified. We are therefore left with the action

$$I'[q,\phi] = -\int dt D_0(q,\dot{q},\phi,\dot{\phi})(\ddot{q} + f(q,\dot{q},\phi,\dot{\phi}))^2.$$
 (E.25)

Since all the bra and ket quantum fields are identified, normalization of Equation (E.23) is equivalent to ensuring that Equation (E.25) is normalized.

As we saw for the classical path integrals, integrating Equation (E.25) over the \ddot{q} at second time step implements a standard Gaussian integral. If D_0 is dependent on the fields, we therefore pick up a term $(\sqrt{\det(D_0(q,\dot{q},\phi))})^{-1/2}$, which we must cancel in the measure by including a $\sqrt{\det(D_0(q,\dot{q},\phi))}$ term. It can also be exponentiated into the action by introducing Bosonic and Fermionic Faddeev-Poppov fields [201]. This term commonly arises in the study of Fokker-Plank type equations when the noise is multiplicative [153, 161, 201]. With this in mind, once we have integrated over \ddot{q} , the action vanishes and we are left with the normalization of the initial CQ state. Hence the path integral preserves the normalization of CQ states.

In a similar manner, we can also show that the path integral of Equation (8.21) is also normalized

$$I[q,\phi^{+},\phi^{-}] = \int dt i \dot{\phi}_{+}^{2} + iV(\phi^{+}) - i \dot{\phi}_{-}^{2} - iV(\phi^{-}) - \frac{D_{0}(q,\dot{q},\phi^{+})}{2} (\ddot{q} + f(q,\dot{q},\phi^{+}))^{2} - \frac{D_{0}(q,\dot{q},\phi^{-})}{2} (\ddot{q} + f(q,\dot{q},\phi^{-}))^{2},$$
(E.26)

where $D_0 > 0$. To see this, we first take the trace of the system, setting $\phi_{n+1}^+ = \phi_{n+1}^- = \phi$. Integrating over ϕ then enforces a delta function $\delta(\phi^+ - \phi^-)$. We are then left with the action

$$I'[q,\phi] = -\int dt D_0(q,\dot{q},\phi)(\ddot{q}+f(q,\dot{q},\phi))^2,$$
(E.27)

and we can again perform the Gaussian integral over \ddot{q} to arrive at a normalized path integral if $\sqrt{\det(D_0(q, \dot{q}, \phi))}$ is included in the measure.

Appendix F

The trade-off between decoherence and diffusion coupling constants

This appendix will introduce the positivity conditions used to prove the decoherence diffusion trade-off of chapter 10.

Recall from the definition of positivity of the transition amplitude that $\Lambda^{\mu\nu}(z + \delta t | z't)$ must satisfy

$$\int dz A^*_{\mu}(z, z') \Lambda^{\mu\nu}(z + \delta t | z', t) A_{\nu}(z, z') \ge 0,$$
(F.1)

for any $A_{\mu}(z, z')$ for which (F.1) is well defined: i.e. so that the distributional derivatives in (F.1) are well defined.

As a consequence of Equation (F.1) being positive, we also know that

$$\operatorname{Tr}\left[\int dz \Lambda^{\mu\nu}(z+\delta t|z',t)O_{\mu}(z,z')\rho(z')O_{\nu}^{\dagger}(z,z')\right] \ge 0$$
(F.2)

will be positive for any vector of operators (potentially phase space dependent) $O_{\mu}(z, z')$. This follows from the cyclicity of the trace and the fact that $\Lambda^{\mu\nu}(z + \delta t | z', t) O_{\nu}^{\dagger}(z, z') O_{\mu}(z, z')$ will be a positive operator so long as (F.1) holds. A common choice of O_{μ} would be the Lindblad operators L_{μ} appearing in the master equation.

The inequality in Equation (F.1) proves useful to derive positivity conditions on the coupling constants appearing in the master equation. In contrast, Equation (F.2) is useful in deriving the observational trade-off for the continuous master equations.

We can get a general trade-off between the decoherence and diffusion coefficients which appear in the master equation, arriving at a trade-off between the decoherence and diffusion coefficients in terms of the back-reaction drift coefficient $D_{1,i}^{\mu\alpha}$.

Consider Equation (F.1), with $A_{\mu} = \delta^{\alpha}_{\mu} a_{\alpha} + b^{i}_{\mu} (z - z')_{i}$. By integrating by parts over the phase space degrees of freedom, we find

$$2b_{\mu}^{i*}D_{2,ij}^{\mu\nu}b_{\nu}^{j} + b_{\mu}^{i*}D_{1,i}^{\mu\beta}a_{\beta} + a_{\alpha}^{*}D_{1,i}^{\alpha\mu}b_{\mu}^{i} + a_{\alpha}^{*}D_{0}^{\alpha\beta}a_{\beta} \ge 0$$
(F.3)

Taking $i \in \{1, ..., n\}$ $\alpha \in \{1, ..., p\}$ and $\mu \in \{1, ..., p+1\}$, we can write this as a matrix positivity condition

$$\begin{bmatrix} b^*, \alpha^* \end{bmatrix} \begin{bmatrix} 2D_2 & D_1^{br} \\ D_1^{br} & D_0 \end{bmatrix} \begin{bmatrix} b \\ \alpha \end{bmatrix} \ge 0$$
 (F.4)

where D_2 is the $(p+1)n \times (p+1)n$ matrix with elements $D_{2,ij}^{\mu\nu}$, D_1^{br} is the $(p+1)n \times p$ matrix with rows labeled by μi and columns labelled by β with elements $D_{1,i}^{\mu\beta}$ and D_0 is the $p \times p$ decoherence matrix with elements $D_0^{\alpha\beta}$. $D_{1,i}^{br}$ describes the quantum back-reacting components of the drift. Equation (F.4) is equivalent to the condition that the $((p+1)n+p) \times ((p+1)n+p)$ matrix

$$\begin{bmatrix} 2D_2 & D_1^{br} \\ D_1^{br} & D_0 \end{bmatrix} \succeq 0 \tag{F.5}$$

Since we know D_2 and D_0 must be positive semi-definite, we know from Schur decomposition that

$$2D_2 \succeq D_1^{br} D_0^{-1} D_1^{br\dagger}$$
(F.6)

where D_0 is the generalized inverse of D_0 . Furthermore, if D_0 vanishes, then clearly D_1^{br} must vanish for (F.5) to be positive semi-definite.

Appendix G

Classical-quantum dynamics with fields

In this appendix, we describe CQ dynamics in the case where the Lindblad operators and the phase-space degrees of freedom can have spatial dependence $z(x), L_{\mu}(x)$.

For the case of fields, operators O(x) constructed out of local fields $\phi(x)$ will generally be unbounded; hence, the Stinespring dilation theorem does not hold. This problem is common in the study of algebraic quantum field theory. We can get around it by considering the case in which operators of interest are obtained by smearing the local fields over bounded functionals F. For example, we can first smear the local fields over a smearing function f, $\phi_f = \int dx \phi(x) f(x)$ and then consider bounded functions of ϕ_f such as $F(\phi_f) = e^{i\phi_f}$. In doing this, we can write a CQ version of the Stinespring dilation theorem exactly and proceed along the lines of [28] to show that any completely positive CQ map can be written in the form

$$\rho'(z) = \int dz dx dy \Lambda^{\mu\nu}(z|z';x,y) L_{\mu}(x,z,z') \varrho(z') L_{\nu}^{\dagger}(y,z,z')$$
(G.1)

where the positivity condition states

$$\int dz dx dy A^*_{\mu}(x, z, z') \Lambda^{\mu\nu}(z|z'; x, y) A_{\nu}(y, z, z') \ge 0.$$
(G.2)

We shall assume that we deal with dynamics which can be written in Lindblad form, as is usually assumed in the unbounded case [264].

G.1 CQ Kramers-Moyal expansion for fields

Just as in Chapter 2, we can formally introduce the moments of the transition amplitude

$$M_{n,i_1\dots i_n}^{\mu\nu}(w_1,\dots,w_n;x,y,\delta t) = \int Dz \Lambda^{\mu\nu}(z|z';x,y,\delta t)(z-z')_{i_1}(w_1)\dots(z-z')_{i_n}(w_n) \quad (G.3)$$

which we assume to exist, which might involve the smearing of the operators z(x). Defining $L_0(x) = \delta(x)\mathbb{I}$, we can define the coefficients $D_{n,i_1...i_n}^{\mu\nu}$ implicitly via

$$M_{n,i_1...i_n}^{\mu\nu}(z',w_1,...w_n;x,y,\delta t) = \delta_0^{\mu}\delta_0^{\nu} + \delta t n! D_{n,i_1...i_n}^{\mu\nu}(w_1,...w_n;x,y,\delta t).$$
(G.4)

The characteristic function then takes the form

$$C^{\mu\nu}(u,z';x,y) = \int Dz e^{i \int dw u(w) \cdot (z(w) - z'(w))} \Lambda^{\mu\nu}(z|z';x,y)$$
(G.5)

and expanding out the exponential, this takes the form

$$C^{\mu\nu}(u,z';x,y) = \sum_{n=0}^{\infty} \int dw_1 \dots dw_n \frac{u_i(w_1) \dots u_{i_n}(w_n)}{n!} M^{\mu\nu}_{n,i_1\dots i_n}(z',w_1,\dots,w_n;x,y,\delta t) \quad (G.6)$$

performing the inverse Fourier transform allows us to write the transition amplitude in terms of functional derivatives of the delta function

$$\Lambda^{\mu\nu}(z|z';x,y,\delta t) = \sum_{n=0}^{\infty} \int dw_1 \dots dw_n \frac{M_{n,i_1\dots i_n}^{\mu\nu}(z',w_1,\dots,w_n;x,y,\delta t)}{n!} \frac{\delta^n}{\delta z'_{i_1}(w_1)\dots z'_{i_n}(w_n)} \delta(z,z')$$
(G.7)

and we can use this to write a CQ master equation in the form

$$\frac{\partial \varrho(z,\delta t)}{\partial t} = \sum_{n=1}^{\infty} \int dw_1 \dots dw_n (-1)^n \frac{\delta^n}{\delta z_{i_1}(w_1) \dots z_{i_n}(w_n)} \left(D^{00}_{n,i_1\dots i_n}(z,w_1,\dots w_n)\varrho(z) \right)
- i[H,\varrho(z)] + \int dx dy D^{\alpha\beta}_0(z;x,y) L_\alpha(x)\varrho(z) L_\beta(y) - \frac{1}{2} D^{\alpha\beta}_0(z;x,y) \{L^{\dagger}_{\beta}(y) L_\alpha(x),\varrho\}
+ \sum_{n=0}^{\infty} \sum_{\mu\nu\neq00} \int dx dy dw_1 \dots dw_n (-1)^n \frac{\delta^n}{\delta z_{i_1}(w_1) \dots z_{i_n}(w_n)} \left(D^{\mu\nu}_{n,i_1\dots i_n}(z,w_1,\dots w_n;x,y) L_\mu(x)\varrho(z) L^{\dagger}_{\nu}(y) \right)$$
(G.8)

Since we are interested in studying dynamics with local back-reaction, we shall take $D_1^{\mu\nu}(z, w; x, y) = D_1^{\mu\nu}(x)\delta(x, y)\delta(x, w)$. By the decoherence diffusion trade-off, which we derive in the next subsection¹, this also means that the diffusion matrix $D_{2,ij}^{\mu\nu}(z, w_1, w_2, x, y)$ is lower bounded by the

¹More precisely, take Equation (G.2) with $A_{\mu}(x) = \delta^{\alpha}_{\mu} \alpha_{\alpha}(x) + \int dw b^{i}_{\mu}(x, w)(z - z')(x, w)$ and apply the same methods as in subsection G.2.

matrix $D_1^{\mu\alpha}(x)(D_0^{-1})_{\alpha\beta}(x,y)D_1^{\beta\nu*}(y)\delta(w_1,x)\delta(w_2,y)$. Without loss of generality we thus take $D_2(z,w_1,w_2,x,y) = D_2(z,x,y)\delta(x,w_1)\delta(y,w_2)$

G.2 Trade-off between diffusion and decoherence couplings for fields

In the field-theoretic case, the positivity condition is given by Equation (G.2), and we can find a trade-off between decoherence and diffusion by considering $A_{\mu}(x) = \delta^{\alpha}_{\mu}\alpha_{\alpha}(x) + \int dx b^{i}_{\mu}(x)(z - z')(x)$. So that

$$\int dx dy \left[2b_{\mu}^{i*}(x) D_{2,ij}^{\mu\nu}(x,y) b_{\nu}^{j}(y) + b_{\mu}^{i*}(x) D_{1,i}^{\mu\beta}(x,y) a_{\beta}(y) + a_{\alpha}^{*}(x) D_{1,i}^{\alpha\mu}(x,y) b_{\mu}^{i}(y) + a_{\alpha}^{*}(x) D_{0}^{\alpha\beta}(x,y) a_{\beta}(y) \right] \ge 0$$
(G.9)

where we use the shorthand notation $D_{2,ij}^{\mu\nu}(z,x,y) := D_{2,ij}^{\mu\nu}(x,y)$ and similarly $D_{1,i}^{\alpha\mu}(z;x,y) := D_{1,i}^{\alpha\mu}(x,y)$.

Taking $i \in \{1, ..., n\}$ $\alpha \in \{1, ..., p\}$ and $\mu \in \{1, ..., p+1\}$, we can write this as a matrix positivity condition

$$\int dx dy [b^*(x), \alpha^*(x)] \begin{bmatrix} 2D_2(x, y) & D_1^{br}(x, y) \\ D_1^{br}(x, y) & D_0(x, y) \end{bmatrix} \begin{bmatrix} b(y) \\ \alpha(y) \end{bmatrix} \ge 0$$
(G.10)

where $D_2(x, y)$ is the $(p+1)n \times (p+1)n$ matrix-kernel with elements $D_{2,ij}^{\mu\nu}(x, y)$, $D_1^{br}(x, y)$ is the $(p+1)n \times p$ matrix-kernel with rows labeled by μi and columns labelled by β with elements $D_{1,i}^{\mu\beta}(x, y)$ and $D_0(x, y)$ is the $p \times p$ decoherence matrix-kernel with elements $D_0^{\alpha\beta}(x, y)$. $D_{1,i}^{br}$ describes the quantum back-reacting components of the drift.

Equation (G.10) is equivalent to the condition that the $((p+1)n + p) \times ((p+1)n + p)$ matrix of operators

$$\begin{bmatrix} 2D_2 & D_1^{br} \\ D_1^{br} & D_0 \end{bmatrix} \succeq 0 \tag{G.11}$$

be positive semi-definite. Here we are viewing the objects of (G.11) as matrix-kernels, so that for any position dependent vector $b^i_{\mu}(x)$, $(D_2 b)^{\mu}_i(x) = \int dy D^{\mu\nu}_{2,ij}(x,y) b^j_{\nu}(y)$. Since we know D_2 and D_0 must be positive semi-definite, we know from Schur decomposition that

$$2D_2 \succeq D_1^{br} D_0^{-1} D_1^{br\dagger} \tag{G.12}$$

and

$$(\mathbb{I} - D_0 D_0^{-1}) D_1^{br} = 0, (G.13)$$

where D_0^{-1} is the generalized inverse of D_0 . Furthermore, from Equation (G.13), we see if D_0 vanishes, then clearly D_1^{br} must also vanish in order for (G.11) to be positive semi-definite.

G.3 Observational trade-off for fields

We can use the same methods to arrive at an observational trade-off using the field-theoretic version of the Cauchy-Schwartz inequality in (10.7). This arises from the positivity of

$$\operatorname{Tr}\left[\int dz dz' dx dy \Lambda^{\mu\nu}(z|z',x,y) O_{\mu}(z,z',x) \rho(z') O_{\nu}^{\dagger}(z,z',y)\right] \ge 0$$
(G.14)

for any local vector of CQ operators $O_{\mu}(z, z', x)$. We have to be careful since (G.14) is not in general well defined since O_{μ} may not be trace-class. We hence assume that we consider states $\rho(z)$ and operators $O_{\mu}(z, z', x)$ for which (G.14) is well defined. Since we are interested in getting an observational trade-off, we expect this always to be the case for physical classical-quantum states $\rho(z)$.

We shall use Equation (G.14) to arrive at a (pseudo) inner product on a vector of operators O_{μ} via

$$\langle \bar{O}_1, \bar{O}_2 \rangle = \int dz dz' dx dy \operatorname{Tr} \left[\Lambda^{\mu\nu}(z|z'x, y) O_{1\mu}(x) \varrho(z') O_{2\nu}^{\dagger}(y) \right]$$
(G.15)

where $||\bar{O}|| = \sqrt{\langle \bar{O}, \bar{O} \rangle} \ge 0$ due to (G.14). Technically this is not positive definite, but again, this will not worry us. Hence, so long as $||\bar{O}_2|| \ne 0$ we again have a Cauchy-Schwartz inequality

$$||\bar{O}_1||^2 ||\bar{O}_2||^2 - |\langle \bar{O}_1, \bar{O}_2 \rangle|^2 \ge 0.$$
(G.16)

Choosing $O_{1,\mu}(x) = \delta^{\alpha}_{\mu}L_{\alpha}(x)$ and $O_{2,\mu}(x) = \int dx' b^i(x)(z-z')_i(x)L_{\mu}(x)$, one finds

$$\begin{split} ||\bar{O}_{1}||^{2} &= \int dz dx dy \operatorname{Tr} \left[D_{0}^{\alpha\beta}(z;x,y) L_{\alpha} \varrho(z) L_{\beta}^{\dagger} \right] := \langle D_{0} \rangle \\ ||\bar{O}_{2}||^{2} &= 2 \int dz dx dy \operatorname{Tr} \left[b^{j*}(x) D_{2,ij}^{\mu\nu}(z;x,y) L_{\mu}(x) \varrho(z) L_{\nu}^{\dagger}(y) b^{i}(y) \right] \\ |\langle \bar{O}_{1}, \bar{O}_{2} \rangle|^{2} &= |\int dz dx \operatorname{Tr} \left[b^{i*}(x,x) D_{1,i}^{\alpha\nu}(z;x) L_{\alpha}(x) \varrho(z) L_{\nu}^{\dagger}(x) \right] |^{2} := |\langle \int dx b^{i*}(x) D_{1,i}^{br}(x) \rangle|^{2} \\ (G.17) \end{split}$$

Taking the limit $b^i(x) \to \delta(x, \bar{x})b^i(\bar{x})$, we arrive at a local trade-off between diffusion, drift, and the total decoherence. In particular, using G.17 and the fact that for back-reaction, the expectation value of D_0 cannot vanish, we arrive at the observational trade-off of Equation (10.22)

$$b^{i}(\bar{x})\left[2\langle D_{2,ij}(\bar{x},\bar{x})\rangle\langle D_{0}\rangle - |\langle D_{1,i}^{br}(\bar{x})\rangle|^{2}\right]b^{j}(\bar{x}) \ge 0$$
(G.18)

which we write in matrix form as

$$2\langle D_2(\bar{x},\bar{x})\rangle\langle D_0\rangle \succeq \langle D_1^{br}(\bar{x})\rangle\langle D_1^{br}(\bar{x})\rangle^{\dagger}$$
(G.19)

We obtain the trade-off between decoherence, diffusion, and total drift

$$8\langle D_2(\bar{x},\bar{x})\rangle\langle D_0\rangle \succeq \langle D_1^T(\bar{x})\rangle\langle D_1^T(\bar{x})\rangle^{\dagger}$$
(G.20)

using the same methods as in Chapter 10 for the continuous case, which we do not reproduce here.

Appendix H

Relating decoherence rates to the observational trade-off

This appendix studies the decoherence rates for general CQ master equations. We consider the case of a quantum mass initially in a partially decohered superposition of state $|L\rangle$ and $|R\rangle$. We describe the quantum state using creation and annihilation operators $\psi(x), \psi^{\dagger}(x)$ on a Fock space, related to the usual momentum-based Fock operators as $\psi(x) = \int dp e^{i\vec{p}\cdot\vec{x}}a_{\vec{p}}$. We assume that the state remains well approximated by a state with a fixed particle number. The superposition can be taken to be distributions centered around $x = x_L$ and $x = x_R$ with total mass m, i.e., for a one-particle state we could take $|L/R\rangle = \int d^3x f_{L/R}(x)\psi^{\dagger}(x)|0\rangle$. We will take them to be well separated so that $f_L(x)f_R(x) \approx 0$, and we take the separation distance to be larger than the scale of the non-locality in $D_0(x, y)$. Mathematically this means that $\langle L|D_0^{\alpha\beta}(z;x,y)L_{\beta}^{\dagger}(y)L_{\alpha}(x)|R\rangle \approx 0$ for any local operators $L_{\alpha}(x)$ and $L_{\beta}(y)$.

With this orthogonality condition, we can then (at least initially) consider the joint quantum classical state restricted to the 2 dimensional Hilbert space of these two states so that the total quantum-classical system can be written as

$$\varrho(\Phi, \pi_{\Phi}, t) = \begin{pmatrix} u_L(\Phi, \pi_{\Phi}, t) & \alpha(\Phi, \pi_{\Phi}, t) \\ \alpha^{\star}(\Phi, \pi_{\Phi}, t) & u_R(\Phi, \pi_{\Phi}, t) \end{pmatrix},$$
(H.1)

where $u_L(\Phi, \pi_{\Phi}, t)$ and $u_R(\Phi, \pi_{\Phi}, t)$ corresponds to some sub-normalized probability distribution over the classical states of the gravitational field. We define the total quantum state ρ_Q by integrating over the classical degrees of freedom

$$\rho_Q = \int D\Phi D\pi_\Phi \varrho(\Phi, \pi_\Phi, t) \tag{H.2}$$

and we shall relate $\langle D_0 \rangle$ appearing in the trade-off to the decoherence rate of the off diagonals of ρ_Q . Integrating over the classical phase space in Equation (10.1), one finds the follows expression for the evolution of ρ_Q

$$\frac{\partial \rho_Q}{\partial t} = \int D\phi D\pi_{\Phi} - i[H(\Phi, \pi_{\Phi}), \varrho(\Phi, \pi_{\Phi})]
+ \int D\phi D\pi_{\Phi} \int dx dy \left[D_0^{\alpha\beta}(\Phi, \pi_{\Phi}; x, y) L_{\alpha}(x) \varrho(\Phi, \pi_{\Phi}, t) L_{\beta}^{\dagger}(y) \right]
- \frac{1}{2} D_0^{\alpha\beta}(\Phi, \pi_{\Phi}; x, y) \{ L_{\beta}^{\dagger}(y) L_{\alpha}(x), \varrho(\Phi, \pi_{\Phi}, t) \} .$$
(H.3)

In particular, one finds that the off-diagonals $\langle L|\frac{\partial \rho_Q}{\partial t}|R\rangle$ evolve in part according to standard unitary evolution, and in part due to the Lindbladian term

$$\int D\phi D\pi_{\Phi} \int dx dy \bigg[\langle L|D_{0}^{\alpha\beta}(\Phi, \pi_{\Phi}; x, y) L_{\alpha}(x) \varrho(\Phi, \pi_{\Phi}, t) L_{\beta}^{\dagger}(y)|R \rangle \\ - \frac{1}{2} D_{0}^{\alpha\beta}(\Phi, \pi_{\Phi}; x, y) \langle L| \{L_{\beta}^{\dagger}(y) L_{\alpha}(x), \varrho(\Phi, \pi_{\Phi}, t)\}|R \rangle \bigg].$$
(H.4)

We shall now study the two terms appearing in Equation (H.4) separately, starting with the first term. Since we assume that the state is well approximated by a state with a fixed particle number, then the contributions to the first term in Equation (H.4) only come from terms where $L_{\alpha}(x)$ and $L_{\beta}(y)$ have the same number of creation and annihilation operators. To compute the expression, one commutes through the creation operators to act on the $\langle L|$ bra and picks up a term $f_L(x)$. Similarly, one commutes the annihilation operators to the act on the $|R\rangle$ ket and picks up a term $f_R(y)$. As a consequence

$$\langle L|D_0^{\alpha\beta}(\Phi,\pi_{\Phi};x,y)L_{\alpha}(x)\varrho(\Phi,\pi_{\Phi},t)L_{\beta}^{\dagger}(y)|R\rangle \sim D_0^{\alpha\beta}(\Phi,\pi_{\Phi};x,y)f_L(x)f_R(y)\approx 0$$
(H.5)

where the last equality follows from the fact that we are taking the masses to be well separated, and the range of D_0 is assumed to be much less than the separation between the masses.

Hence, the evolution of the off-diagonals comes from the unitary evolution and the second term in Equation (H.4), the so-called *no-event* term

$$-\frac{1}{2}\int D\phi D\pi_{\Phi}\int dxdy D_{0}^{\alpha\beta}(\Phi,\pi_{\Phi};x,y)\langle L|\{L_{\beta}^{\dagger}(y)L_{\alpha}(x),\varrho(\Phi,\pi_{\Phi},t)\}|R\rangle,$$
(H.6)

-

which is negative definite and acts to suppress the off-diagonals exponentially. To see this, note that expanding out $\rho(\Phi, \pi_{\Phi}, t)$ in terms of the approximate 2 dimensional Hilbert space

$$\varrho(\Phi, \pi_{\Phi}, t) = u_L(\Phi, \pi_{\Phi}, t) |L\rangle \langle L| + u_R(\Phi, \pi_{\Phi}, t) |R\rangle \langle R| + \alpha(\Phi, \pi_{\Phi}, t) |L\rangle \langle R| + \alpha^*(\Phi, \pi_{\Phi}, t) |R\rangle \langle L|,$$
(H.7)

and using the fact that the range of D_0 is much less than the separation between the left and right masses, we can write the no-event term as

$$-\frac{1}{2}\int D\Phi D\pi D_0^{\alpha\beta}(\Phi,\pi_{\Phi};x,y)\left(\langle L|L_{\beta}^{\dagger}(y)L_{\alpha}(x)|L\rangle+\langle R|L_{\beta}^{\dagger}(y)L_{\alpha}(x)|R\rangle\right)\langle L|\varrho(\Phi,\pi_{\Phi})|R\rangle.$$
(H.8)

Equation (H.8) already expresses that the off-diagonal terms will decay, and the particle will decohere at a rate determined by the integrand of Equation (H.8). We can go slightly further in the presence of a background Newtonian potential such as the Earth's Φ_b . The Earth's background potential dominates over small fluctuations in Φ due to the particles, and we can approximate Equation (H.8) by

$$-\frac{1}{2}D_0^{\alpha\beta}(\Phi_b,\pi_{\Phi_b};x,y)(\langle L|L_{\beta}^{\dagger}(y)L_{\beta}^{\dagger}(y)L_{\alpha}(x)|L\rangle + \langle R|L_{\beta}^{\dagger}(y)L_{\beta}^{\dagger}(y)L_{\alpha}(x)|R\rangle)\langle L|\rho_Q|R\rangle.$$
(H.9)

The result is to exponentially decrease the coherence $\langle L|\rho_Q|R\rangle$ with a rate λ_{LR} determined by

$$\lambda_{LR} = \frac{1}{2} \int dx dy D_0^{\alpha\beta}(\Phi_b, \pi_{\Phi_b}; x, y) (\langle L | L_\beta^{\dagger}(y) L_\alpha(x) | L \rangle + \langle R | L_\beta^{\dagger}(y) L_\alpha(x) | R \rangle).$$
(H.10)

Note Equation (H.10) is the same decoherence rate as if we choose not to integrate out the Newtonian potential, which gives rise to the same bound for the decoherence rate and is what we use for the experimental bounds considered in Chapter 10.

We can also show that the $\langle D_0 \rangle$ term appearing in the observational trade-off of Equation (10.22) is always less than (twice) this decoherence rate, though we do not use it in Chapter 10.

Specifically, we show that

$$\int D\Phi D\pi_{\Phi} \int dx dy \operatorname{Tr} \left[D_0^{\alpha\beta}(\Phi, \pi_{\Phi}; x, y) L_{\beta}^{\dagger}(y) L_{\alpha}(x) \varrho(\Phi, \pi_{\Phi}) \right] \le 2\lambda_{LR}$$
(H.11)

To see this, we first expand out the CQ state in terms of Equation (H.7) and use the fact that D_0 has a range less than the separation of the masses. We then arrive at the following expression for the left-hand side of Equation (H.11)

$$\int D\Phi D\pi_{\Phi} \int dx dy D_0^{\alpha\beta}(\Phi, \pi_{\Phi}; x, y) (\langle L|L_{\beta}^{\dagger}(y)L_{\alpha}(x)|L\rangle u_L(\Phi, \pi_{\Phi}, t) + \langle R|L_{\beta}^{\dagger}(y)L_{\alpha}(x)|R\rangle u_R(\Phi, \pi_{\Phi}, t))$$
(H.12)

Due to the positivity of the CQ density matrix, u_L and u_R must both be positive. Furthermore, $u_L + u_R \leq 1$. Hence this must be less than Equation (H.8), from which (H.10) directly follows.

It is also important to note that though λ_{LR} is the decoherence rate of a particle in a superposition of L/R states, the bound of Equation (H.11) holds even for fully decohered masses in any mixture of $|L\rangle\langle L|, |R\rangle\langle R|$ states. This can be seen directly from Equation (H.12), which depends only on u_L , u_R .

Appendix I

Detecting gravitational diffusion

This appendix shows how the diffusion induced on the Newtonian potential can be measured experimentally, giving rise to the bounds in Chapter 10.

As shown in Chapter 9, in the non-relativistic limit, $c \to \infty$, the CQ dynamics can be approximated by sourcing the Newtonian potential by a random mass term, and in order to maintain the coherence of any mass in superposition, there must be noise in the Newtonian potential such that we cannot tell which element of the superposition the particle will be in

$$\nabla^2 \Phi = 4\pi G[m(x,t) + u(\Phi,\hat{m})\xi(x,t)], \qquad (I.1)$$

with

$$\mathbb{E}_m[\xi(x,t)] = 0, \ \mathbb{E}_m[u\xi(x,t)u\xi(y,t')] = 2\langle D_2(x,y,\Phi)\rangle\delta(t,t'),$$
(I.2)

where $\langle D_2(x, y, \Phi) \rangle := \text{Tr} \left[D_2^{\mu\nu}(x, y, \Phi_b) L_{\mu}(x) \varrho L_{\nu}^{\dagger}(y) \right]$ and ϱ is the quantum state for the decohered mass density.

The solution to Equation (I.1) is given by

$$\Phi(t,x) = -G \int d^3x' \frac{[m(x',t) - u(\Phi,\hat{m})\xi(x',t)]}{|x - x'|},$$
(I.3)

where the statistics of ξ are described by Equation (I.2). A formal treatment of solutions to non-linear stochastic integrals of the form Equation (I.1) can be found in [237].

I.1 Table-top experiments

In this section, we estimate the variation in force that would be seen in table-top experiments. This bounds the diffusion of classical theories of gravity from above, giving a squeezed bound on D_2 due to lower bounds on diffusion arising from coherence experiments. We do this for dynamics in Equation (I.1), but the methodology is general and could also be used in a full simulation of CQ dynamics.

The variation in force induced on a composite mass is found via

$$\vec{F}_{tot} = -\int d^3 x m(x) \nabla \Phi. \tag{I.4}$$

Using the solution in Equation (I.3), the total force can be written

$$\vec{F}_{tot} = -G \int d^3x d^3x' m(x) \frac{(\vec{x} - \vec{x}')}{|x - x'|^3} [m(x', t) - J(x', t)].$$
(I.5)

In reality, we measure time-averaged force by measuring time averaged accelerations over the time resolution of the experiment $\Delta T \frac{1}{\Delta T} \int_0^{\Delta T} dt F_{tot}$. The total variation in the forces time averaged magnitude¹ $\sigma_F^2 := \vec{F}_{tot} \cdot \vec{F}_{tot}$ can be written as

$$\sigma_F^2 = \frac{1}{\Delta T} 2G^2 \int d^3x d^3y d^3x' d^3y' m(x) m(y) \frac{(\vec{x} - \vec{x}') \cdot (\vec{y} - \vec{y}')}{|x - x'|^3 |y - y'|^3} \langle D_2(x', y', \Phi) \rangle.$$
(I.6)

We shall use Equation (I.6) to provide an upper bound on coupling constants of CQ theories for different choices of kernels $D_2(x', y', \Phi)$. Given a choice of functional form of the kernel, all that remains is the strength of the diffusion coupling, which for the translation invariant kernels we consider here takes the form of a single coupling constant D_2 . We take D_2 as a dimension-full quantity with units kg^2sm^{-3} , which characterizes the diffusion rate for the conjugate momenta of the Newtonian potential.

For a composite mass, we can approximate the mass density by summing over N individual atoms of mass density $m_i(x)$, $m(x) = \sum_i m_i(x)$. The total force is the given by $\vec{F}_{tot} = \sum_i \vec{F}_i$, where \vec{F}_i is the force on each individual atom $\vec{F}_i = -\int_V dx m_i(x) \nabla \Phi(x)$, and the total variation of force is then $\sigma_F^2 = \mathbb{E}[\sum_{ij} F_i F_j] - \mathbb{E}[\sum_i F_i]^2$.

The squeeze will generally depend on the functional choice of $D_2(x, y, \Phi)$ on the Newtonian potential. As mentioned in the main body, in the presence of a large background potential Φ_b ,

¹The full covariance matrix for various kernels is given in [265]

such as that of the Earth's, we will often be able to approximate $D_2(x, y, \Phi) = D_2(x, y, \Phi_b)$. This is true for the kernels with functional dependence of the form $D_2 \sim \Phi^n, D_2 \sim \nabla \Phi$, though the approximation does not hold for all kernels, for example, $D_2 \sim \nabla^2 \Phi$ which creates diffusion only where there is mass density. We shall only consider diffusion kernels $D_2(x, y, \Phi_b)$ where the background potential is dominant, leaving more general considerations for future work.

I.1.1 Ultra-local continuous models

For local translation invariant dynamics for which the background Newtonian potential is dominant, for example, $D_2 \sim \Phi^n$, we have $\langle D_2(x, y, \Phi_b) \rangle = \langle D_2(\Phi_b) \rangle \delta(x, y)$, and we arrive at the expression for the total variation in time-averaged force

$$\sigma_F^2 = \frac{2G^2}{\Delta T} \sum_{ij} \int d^3x d^3y d^3x' m_i(x) m_j(y) \frac{(\vec{x} - \vec{x}') \cdot (\vec{y} - \vec{x}')}{|x - x'|^3 |y - x'|^3} \langle D_2(x', \Phi_b) \rangle. \tag{I.7}$$

To leading order, the integral in Equation (I.7) is dominated by the self variation term where i = j, since nuclear scales $10^{-15}m$ dominate over inter-atomic scales $10^{-9}m$, so that $\mathbb{E}[\sum_{ij} F_i F_j] \sim \sum_i \mathbb{E}[F_i^2]$. Approximating the mass density of the atoms as coming from their nucleus and taking them to be spheres of constant density ρ with radius r_N and mass m_N , we find that the integral in Equation (I.7) is approximately

$$\sigma_F^2 \sim \frac{NG^2 \rho^2 r_N^2}{\Delta T} \int d^3 x' \langle D_2(\Phi_b) \rangle. \tag{I.8}$$

For the class of ultra-local continuous dynamics studied in Section 9.3 of Chapter 9, we have $\langle D_2(\Phi_b) \rangle = D_2(\Phi_b)$ since the diffusion is not associated to any Lindblad operators. If noise is everywhere throughout space, then the integral in Equation (I.8) diverges and gives evidence that continuous CQ theories with noise everywhere should be ruled out.

As such, we expect that continuous CQ theory must contain non-linear terms proportional to the Newtonian potential appearing in Equation (I.1), in which case we can approximate $\int dx' D_2$ by $V_b D_2$ where V_b is the volume of the region over which the background Newtonian potential is significant. In total, we find for continuous local CQ dynamics

$$\sigma_F^2 \sim \frac{D_2 N G^2 \rho^2 r_N^2 V_b}{\Delta T}.$$
(I.9)

From this, we can calculate D_2 in terms of the total variance of the acceleration $\sigma_a^2 = \frac{\sigma_F^2}{m_{tot}^2}$ to get a lower bound

$$D_2 \le \frac{\sigma_a^2 N r_N^4 \Delta T}{V_b G^2}.\tag{I.10}$$

Standard Cavendish-type classical torsion experiments measure accelerations of the order $10^{-7}ms^{-2}$, and we can take the time over which the acceleration is averaged to be that of minutes $\Delta T \sim 10^2 s$, so a very conservative bound is $\sigma_a \sim 10^{-7}ms^{-2}$, while N will be $N \sim 10^{26}$ and $r_N \sim 10^{-15}m$. We take the background Newtonian potential to be that of the Earth's, and we (conservatively) take V_b to be $V_b \sim r_E^2 h \sim 10^{15} m^3$ where r_E is the Earths radius, and h is the atmospheric height. We see that this bounds D_2 from above by $D_2 \leq 10^{-41} kg^2 sm^{-3}$.

On the other hand, D_2 is bounded from below from interferometry experiments which bound the decoherence rate. We calculated the decoherence rate in Section 9.6 of Chapter 9. Using the coupling constant trade-off, for the kernel $D_2(x, y) = D_2\delta(x, y)$, we see (ignoring constant factors) that the decoherence rate is found to be

$$\lambda \sim \frac{N_{\lambda} M_{\lambda}^2}{V_{\lambda} D_2},\tag{I.11}$$

where M_{λ} is the mass of a composite particle in the interferometry experiment, which is made up of N_{λ} particles, each with volume V_{λ} . This gives rise to the squeeze

$$\frac{\sigma_a^2 N r_N^4 \Delta T}{V_b G^2} \ge D_2 \ge \frac{N_\lambda M_\lambda^2}{V_\lambda \lambda}.$$
(I.12)

Using the numbers from [233], with $M_{\lambda} \sim 10^{-24} kg$, $N_{\lambda} \sim 10^{3}$, and $V_{\lambda} \sim 10^{-15} 10^{-15} 10^{-7} m^{3} = 10^{-37} m^{3}$, $\lambda \sim 10^{1} s^{-1}$ we find that $D_{2} \geq 10^{-9} kg^{2} sm^{-3}$. This suggests that the $D_{2}(x, y) = D_{2}\delta(x, y)$ kernel for classical gravity is already ruled out by experiment.

I.1.2 Ultra-local jumping models

For the local discrete models, such as that of Equation (9.59) studied in Section 9.7 of Chapter 9, the theory is less constrained due to the dependence of the diffusion on the mass density. In this case, $\langle D_2(\Phi_b) \rangle = \frac{l_P^3}{m_P} D_2(\Phi_b) m(x)$, where the factors of Planck length and Planck mass are to ensure that $D_2(\Phi_b)$ has the required units. We arrive at the upper bound for D_2

$$\frac{\sigma_a^2 N r_N^4 \Delta T m_P}{m_N G^2 l_P^3} \ge D_2. \tag{I.13}$$

Meanwhile, we calculated the decoherence rate in Section 9.7 of Chapter 9, which, using the coupling constant trade-off (C.2) goes as $\lambda \sim \frac{M_{\lambda}m_P}{l_P^3 D_2}$, which lower bounds D_2 . From this, we arrive at the squeeze

$$\frac{\sigma_a^2 N r_N^4 \Delta T}{m_N G^2} \ge \frac{l_P^3 D_2}{m_P} \ge \frac{M_\lambda}{\lambda},\tag{I.14}$$

and plugging in the numbers, we find the bound given by Equation (10.34), which gives rise to the squeeze for local discrete models $10^{-1}kgs \ge \frac{l_P^3}{m_P}D_2 \ge 10^{-25}kgs$.

I.1.3 Continous Diosi-Penrose model

We can also consider other diffusion kernels; for example, we can consider the continuous dynamics of Section 9.3 of Chapter 9 with D_0 the Diosi-Penrose kernel. In this case we have that $D_2(x,y) = -l_P^2 D_2(\Phi_b) \nabla^2 \delta(x,y)$. The Lindbladian kernel saturating the coupling constants trade-off at zeroeth order in $\Phi(x)$ is the Diosi-Penrose kernel $D_0(x, y, \Phi_b) = \frac{D_0(\Phi_b)}{|x-y|}$, as we saw in Table 9.3. Approximating the masses as spheres of constant density, we find from a substitution of the kernel into Equation (I.6) that the variation in time-averaged force is given by

$$\sigma_F^2 \sim \frac{l_P^2 G^2 m_N^2 N D_2}{\Delta T r_N^3}.$$
 (I.15)

We, therefore, find a lower bound for D_2 in terms of the variation in acceleration

$$D_2 \le \frac{\Delta T l_P^2 \sigma_a^2 N r_N^3}{G^2},\tag{I.16}$$

which for classical torsion experiments $\sigma_a \sim 10^{-7} m s^{-2}$, $T \sim 10^2 s$, $N \sim 10^{26}$ and $r_N \sim 10^{-15} m$ gives $D_2 l_p^2 \leq 10^{-9} kg sm^{-1}$. On the other hand, for this kernel, the decoherence rate was calculated in Section 9.6 of Chapter 9, which via the trade-off reads

$$\lambda \sim \frac{NM_{\lambda}^2}{l_P^2 D_2 R_{\lambda}},\tag{I.17}$$

which gives the squeeze on D_2

$$\frac{\Delta T \sigma_a^2 N r_N^3}{G^2} \ge l_P^2 D_2 \ge \frac{N_\lambda M_\lambda^2}{R_\lambda \lambda}.$$
(I.18)

For the numbers used in the main body of the text [233] $M_{\lambda} \sim 10^{-24} kg$, $N_{\lambda} \sim 10^3$, $R_{\lambda} \sim V^{1/3} = 10^{-12} m$, $\lambda \sim 10^1 s$, this yields $D_2 l_P^2 \geq 10^{-35} kg sm^{-1}$ and so this model is not ruled out by experiment.

In general, we can squeeze D_2 from above and below by simulating full CQ dynamics satisfying the decoherence-diffusion trade-off. We bound D_2 from above by studying the effects of diffusion on gravitational experiments. We bound D_2 from below using the coupling constant trade-off and coherence experiments lower bounding the decoherence rate. As we have seen in this section, it appears that classes of continuous CQ hybrid theories of gravity obeying the assumptions outlined in Chapter 10, including models without spatial correlations, are already experimentally ruled out, while others, such as the ultra-local jumping models, require stronger bounds from both gravitational and coherence experiments. We have been very conservative in our estimates, so we expect a more thorough analysis to tighten the bounds by orders of magnitude.

Appendix J

Positivity constraints in open quantum field theory

In the path integral approach, the space-time Lorentz invariance of the open quantum field theory manifests. However, in the open quantum theory literature, path integrals that are not completely positive, but are thought to be, are often considered. This chapter shows that the path integral introduced in [165] is not completely positive.

The path integral

$$S_{\phi} = -\int d^{d}x \left[\frac{1}{2} z \left(\partial \phi_{\rm R} \right)^{2} + \frac{1}{2} m^{2} \phi_{\rm R}^{2} + \frac{\lambda_{3}}{3!} \phi_{\rm R}^{3} + \frac{\lambda_{4}}{4!} \phi_{\rm R}^{4} + \frac{\sigma_{3}}{2!} \phi_{\rm R}^{2} \phi_{\rm L} + \frac{\sigma_{4}}{3!} \phi_{\rm R}^{3} \phi_{\rm L} \right] + \int d^{d}x \left[\frac{1}{2} z^{*} \left(\partial \phi_{\rm L} \right)^{2} + \frac{1}{2} m^{2*} \phi_{\rm L}^{2} + \frac{\lambda_{3}^{*}}{3!} \phi_{\rm L}^{3} + \frac{\lambda_{4}^{*}}{4!} \phi_{\rm L}^{4} + \frac{\sigma_{3}^{*}}{2!} \phi_{\rm L}^{2} \phi_{\rm R} + \frac{\sigma_{4}^{*}}{3!} \phi_{\rm L}^{3} \phi_{\rm R} \right]$$
$$+ i \int d^{d}x \left[z_{\Delta} \left(\partial \phi_{\rm R} \right) \cdot \left(\partial \phi_{\rm L} \right) + m_{\Delta}^{2} \phi_{\rm R} \phi_{\rm L} + \frac{\lambda_{\Delta}}{2!2!} \phi_{\rm R}^{2} \phi_{\rm L}^{2} \right],$$
(J.1)

is Lorentz invariant and was shown to be renormalizable [165]. However, we now show that the $\partial_{\mu}\phi_{L}\partial^{\mu}\phi_{R}$ terms give rise to non-positive dynamics, severely restricting the allowed dissipation to be of the form $\phi_{L}^{n}\phi_{R}^{m}$.

As an example, we can consider the Lindblad equation

$$\frac{\partial\rho}{\partial t} = -i[H,\rho] - \frac{1}{2}\pi\rho\pi + \frac{1}{2}\partial_i\phi\rho\partial_i\rho - \frac{1}{2}\{-\frac{1}{2}\pi^2 + \frac{1}{2}\partial_i\phi\partial_i\phi,\rho\} \equiv \mathcal{L}(\rho), \qquad (J.2)$$

which corresponds to the Lorentz invariant path integral

$$iS = \int d^4x \left[\frac{i}{2}(\partial_\mu\phi^L\partial^\mu\phi^L - \partial_\mu\phi^R\partial^\mu\phi^R) + \frac{1}{2}(\partial_\mu\phi^R\partial^\mu\phi^L) - \frac{1}{4}(\partial_\mu\phi^R\partial^\mu\phi^R + \partial_\mu\phi^L\partial^\mu\phi^L)\right].$$
(J.3)

However, Equation (J.2) is not positive. To be precise, the density operator ρ needs to be positive for all vectors $|v\rangle$ in the Hilbert space $\langle v|\rho|v\rangle \geq 0$. For the dynamics to be positive means that positive operators are mapped to positive operators. To show that the dynamics in Equation (J.2) is not positive, we need to find a positive operator mapped to a non-positive operator under the dynamics.

The dynamics of equation (J.2) can be written in terms of creation and annihilation operators

$$\frac{\partial\rho}{\partial t} = -i[H,\rho] + \int \frac{1}{2} \frac{d^3p}{(2\pi)^3} \omega_{\vec{p}} \left[a_{\vec{p}}\rho a_{-\vec{p}} + a_{\vec{p}}^{\dagger}\rho a_{-\vec{p}}^{\dagger} - \frac{1}{2} \{ a_{-\vec{p}}a_{\vec{p}} + a_{-\vec{p}}^{\dagger}a_{\vec{p}}^{\dagger}, \rho \} \right] \equiv \mathcal{L}(\rho) \tag{J.4}$$

Under short time evolution, we have $\rho(t + \delta t) = \rho(t) + \delta t \mathcal{L}(\rho)$. To show non-positivity, we shall construct explicitly a positive operator σ for which $\sigma + \delta t \mathcal{L}(\sigma)$ is not positive. Note, for the proof of non-positivity, we do not have to normalize, and for notational convenience, we deal with un-normalized states and vectors.

Now, consider the positive operator $a_{\vec{q}}^{\dagger}|0\rangle\langle 0|a_{\vec{q}} \equiv |\vec{q}\rangle\langle \vec{q}|$. We can compute $\mathcal{L}(|\vec{q}\rangle\langle \vec{q}|)$, which gives

$$\mathcal{L}(|\vec{q}\rangle\langle\vec{q}|) = \frac{1}{2} \frac{1}{(2\pi)^3} \omega_{\vec{q}} \left[|0\rangle\langle 0|a_{\vec{q}}a_{-\vec{q}} + a_{\vec{q}}^{\dagger}a_{-\vec{q}}^{\dagger}|0\rangle\langle 0| \right] - \frac{1}{2} \int \frac{1}{2} \frac{d^3p}{(2\pi)^3} \omega_{\vec{p}} \left[a_{\vec{q}}^{\dagger}|0\rangle\langle 0|a_{\vec{q}}a_{\vec{p}}a_{\vec{p}} + a_{\vec{p}}^{\dagger}a_{\vec{p}}^{\dagger}a_{\vec{q}}^{\dagger}|0\rangle\langle 0|a_{\vec{q}} \right].$$
(J.5)

We note that this takes a diagonal operator to a non-diagonal operator, which can now be used to show the non-positivity of the dynamics. To that end consider $\langle v | \mathcal{L}(|\vec{q}\rangle\langle\vec{q}|) | v \rangle$ where the (un-normalised) vector $|v\rangle$ is defined via

$$|v\rangle = |0\rangle - \int d^3k d^3l \ c_{\vec{k}} c_{\vec{l}} a^{\dagger}_{\vec{k}} a^{\dagger}_{\vec{l}} |0\rangle, \qquad (J.6)$$

with $c_{\vec{k}}, c_{\vec{l}} > 0$. We then compute $\langle v | (|\vec{q}\rangle \langle \vec{q}| + \delta t \mathcal{L}(|\vec{q}\rangle \langle \vec{q}|) | v \rangle$ which gives

$$\langle v|(|\vec{q}\rangle\langle\vec{q}| + \delta t \mathcal{L}(|\vec{q}\rangle\langle\vec{q}|)|v\rangle = -\delta t \frac{\omega_{\vec{q}} c_{\vec{q}} c_{\vec{-q}}}{(2\pi)^3} |\langle 0|0\rangle|^2, \tag{J.7}$$

which is negative. To summarize, the operator defined via the dynamics has taken a positive operator to a non-positive operator. So the dynamics defined via Equation (J.2) is not positive.

In general, if one includes a $-\pi\rho\pi$ term in the Lindblad equation as a jump operator, which corresponds to a $\partial_{\mu}\phi^{L}\partial_{\mu}\phi^{R}$ terms in the path integral, while the dynamics are Lorentz invariant then it will not be completely positive.

Appendix K

Symmetry generators with information loss

In this appendix, we study a simple model of a Lorentz invariant theory with information loss, originally studied in [211]. Our goal will be to understand the role of the Lorentz symmetry generators in theories with information loss, which are currently not well understood [211]. It would be interesting to further explore the consequences for the generators of symmetries in the covariant path integrals found in Chapter 8, and this appendix helps us gain intuition for some of the subtleties that arise in transforming states when information is lost.

We start by recalling the definition of Lorentz invariance in unitary quantum systems. For unitary quantum evolution, one can consider the equations of motion for the fields Φ : $M \to \mathbb{R}$ in the Heisenberg representation, in which case the definition of Lorentz invariance and covariance is unchanged relative to the classical case: Φ solves the equations of motion if and only if $\Phi' = \Phi(\Lambda^{-1}x)$ solves the same equations of motion, where Λ is a Lorentz boost. Consequently, Hermitian operators Q_a represent the Lorentz algebra (and, more generally, the Poincare algebra). The Q^a are such that $\Phi(t, x)$ solves the equations of motion if and only if $\Phi - i\epsilon_a[Q^a, \Phi]$ also solve the equation of motion.

When there is information loss in the quantum system, there are two cases we can consider. When there is an underlying classical noise, we can easily extend the notion of space-time symmetry of the dynamical fields. We denote a *realization* of a noise process as $\xi(t, x)$. We denote a particular trajectory of the dynamical fields obeying the stochastic equations of motion as Φ_{ξ} . We say that a stochastic theory is Lorentz invariant if the trajectory Φ_{ξ} occurs with probability $P[\xi(t,x)]$ if and only if the trajectory $\Phi'_{\xi'} = \Phi_{\xi}(\lambda^{-1}x)$ is a solution to the same equations of motion, with a realization of the noise ξ' which occurs with the same probability as ξ . With this definition, two Lorentz-related observers use the same equations of motion. Although they both see different realizations of the noise process, the statistics of the dynamics remain the same, so they cannot decipher which frame they are in without reference to some external system.

On the other hand, we could consider the case where the noise is intrinsically quantum. The dynamics are then generated by Lindbladian \mathcal{L} on the quantum system, which does not have a unique unraveling in terms of a classical noise process. We shall be interested in the case where the dynamics are considered fundamental. Our goal will be to understand the role of the generators of Lorentz symmetry in this case.

We first study a concrete example of a Lorentz invariant theory with a classical noise process and use this to gain intuition into the case of fundamental Lindbladian evolution.

K.1 Transformations with a classical noise process

As an example, we consider a theory whose dynamics are generated by the random Hamiltonian [206, 211]

$$H_{\xi} = H + \int d^3x \xi(t, x)\phi(x), \qquad (K.1)$$

with $H = \int d^3x \mathcal{H}(x) = \frac{1}{2}\pi^2 + \frac{1}{2}\partial_i\phi\partial_i\phi$. We take $\xi(t,x)$ to be a white noise process

$$\mathbb{E}[\xi(t,x)] = 0, \ \mathbb{E}[\xi(t,x)\xi(t',x')] = \delta(t-t')\delta^3(x-x').$$
(K.2)

The model is not physical [206, 266, 211] since it gives rise to infinite particle production, however, it provides a simple model where we are able to explore the consequences of Lorentz invariance in detail.

For white noise, it is useful to note that the probability of a particular realization of ξ is given by

$$\operatorname{Prob}[\xi(t,x)] = \mathcal{N} \exp\left(-\int d^4x \frac{\xi(t,x) \cdot \xi(t,x)}{2}\right),\tag{K.3}$$

where \mathcal{N} is to ensure that probabilities are normalised $\int D\xi(t,x) \operatorname{Prob}[\xi(t,x)] = 1$. It is clear from Equation (K.3) that the realization $\xi(t,x)$ occurs with the same probability as $\xi(\Lambda^{-1}(t,x))^1$. In the Heisenberg representation, the dynamics generated by Equation (K.1) give rise to a Langevin type of equation for ϕ, π

$$\dot{\phi} = \pi, \ \dot{\pi} = \partial_i \partial_i \phi + \phi \xi(t, x),$$
 (K.4)

which can be written in a manifestly covariant manner as

$$-\partial^{\mu}\partial^{\mu}\phi = \phi\xi(t,x). \tag{K.5}$$

We denote solutions to the equations of motion in the Heisenberg representation as $\phi_{\xi}(t, x)$; we shall not construct them explicitly, and we refer the reader to [237] for an in-depth discussion of solutions to non-linear Equations of the form in Equation (K.5).

By averaging over the noise ξ , Equation (K.4) corresponds to the Lindblad equation

$$\frac{\partial \rho}{\partial t} = -i[H,\rho] + \frac{1}{2} \int d^3 x \phi(x) \rho \phi(x) - \frac{1}{2} \{\phi^2(x),\rho\}_+.$$
 (K.6)

Let us now discuss the case of symmetry generators for the dynamics generated by Equation (K.5). On a solution to the dynamics, the active space-time Lorentz transformation relating two observers is given by

$$\phi'_{\xi}(x) = \phi_{\xi}(\Lambda^{-1}x), \tag{K.7}$$

which depends on the realization of the noise process $\xi(t, x)$ since it involves time derivatives of the fields: to transform between frames, we perform a space-time transformation conditioned on the realization of the classical variable ξ . In particular, for the case of a classical noise process, we find the algebra of Lorentz boosts are the same as for the deterministic quantum case, but with $H \to H_{\xi}$ and $Q^i \to Q^i_{\xi}$. In particular, the Lorentz boosts Q^i_{ξ} now contain an explicit dependence on the noise process

$$Q_{\xi}^{ti} = \int d^3x \ tP^{0i} - x^i (\frac{1}{2}\pi^2 + \frac{1}{2}\nabla\phi\nabla\phi + \xi(t,x)\phi(x))$$
(K.8)

where $P^{0i} = \int d^3x \pi \partial^i \phi$ is standard the momentum generator.

¹We have to be careful here since the measure is not well defined on its own, but $D\xi(t, x) \operatorname{Prob}[\xi(t, x)]$ is well defined.
An important point is that these Lorentz transformations act on the solution to the dynamics, not on a time slice. They do not project to the phase space since they involve the time coordinate; this can be seen explicitly from the time dependence in Equation (K.8). As a consequence, an observer (call them Alice) who has initial data on a space-time slice Σ , $(\phi(0, x), \pi(0, x))$, at t = 0, cannot transform their state to a Lorentz boosted observer (Bob) in their own t = 0 frame without knowing the realization of the noise process ξ , since Bob has initial data at space-time points which lie in the future and past of Alice.

K.1.1 Best guess for transformed states: prediction vs. retrodiction

Since the Lorentz transformation depends on the realization which has occurred, one can ask what we should do if we do not know the realization of the noise process ξ everywhere: what is Alice's best guess for transforming to Bob's state? Since averaging over the noise process for the theory gives rise to a Lindblad equation, this "best guess" Lorentz transformation gives intuition for the generator of the Lorentz boost when fundamental Lindbladian evolution is considered – we now study it in more detail.

To be concrete, let us consider the case where Alice has a known state at some time t, Alice cannot see into the future, and she has forgotten the past, but she is persistent and so tries to come up with an estimate for the state Bob sees at his own time. The best thing Alice can do is run the dynamics forward and backward in time to find $\phi_{\xi}(t, x)$ everywhere. She simulates the state $\phi_{\xi}(t, x)$ to the future of \bar{t} using her state as an initial condition, and for $t < \bar{t}$, she uses her known state as a final state. Bob is related to Alice by a Lorentz transformation. So for each realization of the noise process, she performs a Lorentz boost, obtaining the field $\phi_{\xi}(\Lambda^{-1}x)$, which represents the state Bob would see if the realization ξ occurs. Alice then averages over all possible realizations, arriving at a best estimate for the boosted state. As we shall see, we can explicitly construct the generator taking Alice's state to Bob's, represented by a CP map in Equation (K.9). To understand this, it will first prove helpful to comment on some points pertaining to prediction vs. retrodiction in stochastic theories.

Given initial data at some time t, we use the Lindbladian dynamics \mathcal{L} to predict the system's state in the future. Conversely, given initial data at t, we can use adjoint \mathcal{L}^{\dagger} to retrodict what the state was in the past. To understand this, let us first consider the simple case of time-

translation symmetry. Alice has an initial state $\rho(t)$ and wishes to guess what Bob, who is to Alice's future or past, sees. First, we consider the case where Bob lives in the future of Alice. In this case, Alice simulates over all classical trajectories ξ for time ϵ and averages over the result; this is found via the Lindbladian evolution in Equation (K.6). However, if Bob lives to Alice's past, she must retrodict her state. Alice uses her data as a final condition and evolves backward for all possible realizations of the noise ξ in the past to time $t - \epsilon$. Averaging over the noise process, she evolves her state with the backward equation to find the best guess for Bob's state. Both maps define CP evolution on initial quantum states.

Note, this is *not* what one would do if the Lindblad equation were treated as fundamental; this is to do with the omelet of inference and causation arising from information loss. If the Lindblad equation is taken to be fundamental, then to predict the past state $\rho(t - \epsilon)$, one uses the inverse generator $-\mathcal{L}$: it is known that $\rho(t) = e^{\mathcal{L}t}\rho(0)$ is a solution to the Lindblad equation with some earlier initial state $\rho(\bar{t})$, $\bar{t} < t$. In this case, the back-wards generators $-\mathcal{L}$ is not a CP map, but it is CP for all t > 0 on states which are solutions to the dynamics $\rho(t) = e^{\mathcal{L}t}\rho(0)$. We shall consider this point in detail in section K.2.

Moving back to the best guess, Lorentz boost generator. We consider an infinitesimal Lorentz boost $(t, x) \rightarrow (t + \epsilon x, x + \epsilon t)$ and a scalar field undergoing the dynamics in Equation (K.4). We take the space-time representation of the Lorentz group on the trajectories $\phi'_{\xi}(t, x) =$ $\phi_{\xi}(t - \epsilon x, x - \epsilon t)$, which implicitly defines $\pi'_{\xi}(t, x)$ as its time derivative from Equation (K.4). Alice has initial data at some time \bar{t} , we take this to be given by $(\phi(\bar{t}, x), \pi(\bar{t}, x))$. We evolve forwards and backward in time to get $\phi_{\xi}(t, x)$ everywhere.

The Lorentz transformation sends $\phi_{\xi}(t,x) \to \phi'_{\xi}(t,x) = \phi_{\xi}(t-\epsilon x, x-\epsilon t)$. We want to use this to get the best estimate for Bob's state at his time \bar{t} . Importantly, for x > 0, $\phi'_{\xi}(\bar{t},x) = \phi_{\xi}(\bar{t}-\epsilon x, x-\epsilon \bar{t})$ is to the past of \bar{t} , and so is obtained by retrodiction on Alices's initial state, while for x < 0 $\phi'_{\xi}(\bar{t},x) = \phi_{\xi}(\bar{t}-\epsilon x, x-\epsilon \bar{t})$ is obtained from the state which Alice has predicted from his initial state. Taking this into account, the resulting best guess boosts at time t is found to be

$$\frac{\partial\rho(t)}{\partial\epsilon} = -i[Q^{ti},\rho(t)] + \frac{1}{2}\int d^3x |x^i| \left[\phi(x)\rho\phi(x) - 1/2\{\phi^2(x),\rho\}_+\right] \equiv \mathcal{Q}(\rho(t)) \tag{K.9}$$

Where $Q^{ti} = \int d^3x \left[t\phi \partial^i \pi - x^i \left(\pi^2 + \partial_i \phi \partial_i \phi \right) \right]$ is the free Lorentz charge. We should note that after boosting $\rho'(t) = \rho(t) + \epsilon \mathcal{Q}(\rho(t))$ does not satisfy the same equation of motion as $\rho(t)$, but this is to be expected since for x > 0 we are retrodicting on the state.

Equation (K.9) takes the same form as one would expect if including in the boost the component due to its Lindbladian evolution

$$\frac{\partial\rho(t)}{\partial\epsilon} = -i[Q^{ti},\rho(t)] - \frac{1}{2}\int d^3x x^i \left[\phi(x)\rho\phi(x) - 1/2\{\phi^2(x),\rho\}_+\right] \equiv \mathcal{Q}(\rho(t)). \tag{K.10}$$

The difference is that for x > 0, the best guess state is obtained by retrodiction, so the operator \mathcal{L}^{\dagger} should be used, which leads to the modulus $|x^i|$ appearing in Equation (K.9) as opposed to x^i alone, which does not define a completely positive operator on all states $\rho(t)$.

However, the best guess generator for the state is not what one would do if the Lindblad equation

$$\frac{\partial \rho}{\partial t} = -i[H,\rho] + \frac{1}{2} \int d^3 x \phi(x) \rho \phi(x) - \frac{1}{2} \{\phi^2(x),\rho\}_+$$
(K.11)

was treated to be fundamental. As discussed, we can consider the simple example of timetranslation invariance: if the Lindblad equation is taken to be fundamental, then to predict the past state $\rho(t - \epsilon)$ one uses the inverse generator $-\mathcal{L}$. In this case, the backward-in-time generators $-\mathcal{L}$ is not a CP map. However, it is clearly CP for all t > 0 on the subset of states which are solutions to the dynamics $\rho(t) = e^{\mathcal{L}t}\rho(0)$. We might then expect that Equation (K.10) can define Lorentz boosts for a theory with fundamental information loss. We now show this is the case.

K.2 Lorentz invariance for Lindblad equations

To discuss the Lorentz invariance of Lindblad equations, we note that Equation (K.11) has an equivalent path integral description. The path integral tells us how to relate states at two different times via

$$\langle \tilde{\phi}_{+} | \rho(t_f) | \tilde{\phi}_{-} \rangle = \int_{\phi^{\pm}(t_i) = \bar{\phi}^{\pm}}^{\phi^{\pm}(t_f) = \tilde{\phi}^{\pm}} D\phi_{+} D\phi_{-} e^{iS_{t_i}^{t_f}[\phi_{+}\phi_{-}]} \langle \bar{\phi}_{+} | \rho(t_i) | \bar{\phi}_{-} \rangle, \tag{K.12}$$

where ϕ_L is the ket field, and ϕ_R is the bra field arising due to the density matrix's two-sided evolution. The action is given by

$$iS_{t_i}^{t_f}[\phi_L, \phi_R] = \int_{t_i}^{t_f} dt \int d^3x \; [i\mathcal{L}_{KG}^+ - i\mathcal{L}_{KG}^- - \frac{1}{2}(\phi^+ - \phi^-)^2)]. \tag{K.13}$$

and can be found using the methods introduced in Chapter 2. We can find the generator of the Lorentz boost from the charge associated with the Lorentz invariance of the action in the path integral. We find

$$Q^{ti} = \int d^3x t T^{0i} - x^i T^{00}, \qquad (K.14)$$

with

$$-iT^{\mu\nu}[\phi^+,\phi^-] = \partial^{\mu}\phi^-\partial^{\nu}\phi^- - \partial^{\mu}\phi^+\partial^{\nu}\phi^+ - \frac{1}{2}n^{\mu\nu}(\partial_{\sigma}\phi^-\partial^{\sigma}\phi^- - \frac{i}{2}\phi^-\phi^- - \partial_{\sigma}\phi^+\partial^{\sigma}\phi^+ - \frac{i}{2}\phi^+\phi^+ + i\phi^+\phi^-).$$
(K.15)

We can write the Hilbert space operator corresponding to the Lorentz boost as

$$Q^{ti}(\rho) = -i[Q_H^{ti}, \rho] - \frac{1}{2} \int d^3x x^i \left[\phi(x)\rho\phi(x) - \frac{1}{2} \{\phi^2(x), \rho\}_+ \right]$$
(K.16)

Where Q_H^{tx} is the unitary Lorentz charge, $Q^{tx} = \int d^3x \left[t\phi \partial_i \pi - x \left(\pi^2 + \partial_i \phi \partial_i \phi \right) \right]$, which takes exactly the form of Equation (K.10).

One can easily verify that it is a symmetry of the equations of motion. That is, $\rho(t)$ solves $\frac{\partial \rho}{\partial t} = \mathcal{L}(\rho)$ if and only if $\frac{\partial Q^{tx}(\rho)}{\partial t} = \mathcal{L}(Q^{tx}(\rho)).$

More generally, in combination with the time evolution operator \mathcal{L} , and the standard momentum P_i and rotation generators R_i , we see that the Lorentz boosts defined via Equation (K.14), (K.15) satisfy the same algebra as the Poincare algebra

$$[Q_i, P_k] = i\eta_{ik}\mathcal{L}, \ [Q_i, \mathcal{L}] = -iP_i,$$

$$[R_m, Q_n] = i\epsilon_{mnk}Q_k, \ [Q_m, Q_n] = -i\epsilon_{mnk}R_k, \ [\mathcal{L}, P_i] = 0, [\mathcal{L}, R_i] = 0,$$
(K.17)

with all of the other commutation relations unchanged.

Consequently, we see that the dynamics with these generators are Lorentz invariant - as is to be expected via the manifestly invariant path integral. In particular, ρ is a solution to the Lindblad equation of motion given by Equation (K.11) if and only if $\rho(t) + \epsilon_a Q^a(\rho)$ is a solution to the equation of motion, where Q^a is a generator of the Lorentz boosts (or more generally Poincare boosts) satisfying Equation (K.17). The fact that the boosts are Lindbladian represents the facts that the dynamics are not unitary, and since the boosts involve mixing space and time, they are not unitary either. The generators defined in Equation (K.17) define a generalization of the Poincare algebra to the case of information loss. Notably, the generator defined by Equation (K.16) is *not* CP, which initially looks worrying. However, if the Lindblad equation is fundamental, then states are found by solving $\rho(t) = e^{\mathcal{L}(t-t_i)}\rho(t_i)$ and the generator will be CP on states which are solutions to the dynamics. In particular, consider applying an infinitesimal tx Lorentz boost, parameterized by ϵ . In the classical case, we have $\phi'(t, x) = \phi(t - \epsilon x, x - \epsilon t)$. For x > 0, the state comes in from the past, and to describe the boosted state on a sub-region $-X \leq x \leq X$ we will therefore have to know the original state at time $t - \epsilon X$. For example, suppose we have a quantum state which only has support on a subregion X, which we denote ρ_X . We then expect that (K.17) describes CP evolution on $\rho_X(t) = \rho_X e^{(t-t_i)\mathcal{L}}\rho_X(t_i)$, so long as $(t-t_i) \geq \epsilon X$. Taking the $t_i \to -\infty$ limit, we see that the evolution is CP on all time evolved states. If we want to further want to demand consistency with initial states in the infinite past $\rho(-\infty)$, we should consider the subset of initial states for which the action of Q^{tx} is completely positive which is not completely. For the dynamics of Equation (K.11), this is satisfied by the states which are initially decohered in the ϕ basis.

To summarize, we have seen that for simply the theory defined by Equation (K.6), we find a Lorentz invariant path integral, which has led us to Equation (K.17), which defines a generalization of the Poincare algebra to the case where there is fundamental information loss. Notably, we see that because the Lindbladian dynamics are not unitary, the generators of the symmetry group are not unitary. Exploring the consequence of this in more detail, as well as studying the analogous algebra for the covariant path integrals studied in Chapter 8, would be interesting, which we leave for future work.

Commonly used notation

This section summarizes notation commonly used in the thesis and where the notation is introduced.

Classical stochastic mechanics

- $D_{1,i}$ Drift coefficient, page 32
- $D_{2,ij}$ Diffusion coefficient, page 32
- z Classical degree of freedom, page 29
- z_i d dimensional classical degree of freedom, page 29
- \approx Weakly vanishing, page 156
- $\delta(z,z'), \delta(z-z')\,$ Dirac delta functional, page 32
- δ_{ij} Kronecker delta, page 35
- \mathbb{E} Expectation value, page 29
- ω Phase space symplectic form, page 60
- $\xi(t, x)$ White noise process, page 202
- $\{,\}$ Poisson bracket, page 60
- $C(u, z, \delta t)$ Characteristic function, page 32
- *d* Dimension of the classical system, page 29

 $D(z,t)_{n,i_1\ldots i_n}\,$ Moments of the Kramers-Moyal expansion, page 31

 d_i Phase space exterior derivative, page 60

 $M_{n,i_1...i_n}(z,t,\delta t)$ Moments of the probability transition amplitude, page 31

 $P(z,t+\delta t|z',t)$ Probability transition amplitude, page 31

- p(z,t) Probability density, page 29
- S_C Classical path integral action, page 36
- u Response variables, page 36
- W(z|z',t) Short time expansion of the probability transition amplitude, page 31
- $W_i(t)$ Wiener process, page 35
- $Z(t), Z_t$ Classical trajectory, page 29

Fokker-Plank equation Equation (2.19), page 33

Quantum theory

- $\lambda^{\alpha\beta}$ Lindblad coupling, page 44
- $\Lambda^{\mu\nu}$ Kraus matrix, page 43
- $\langle \psi | \phi \rangle$ Inner product, page 37
- $\langle O \rangle$ Expectation value of a quantum observable, page 37
- I Identity operator, page 43
- \mathcal{H} Hilbert space, page 37
- \mathcal{L}_t Lindblad generator, page 44
- \otimes Tensor product, page 38
- Φ_t Dynamical quantum map, page 43

- ρ Density matrix, page 39
- $\operatorname{Tr}_{\mathcal{H}}[]$ Trace over over Hilbert space \mathcal{H} , page 39
- $\operatorname{Tr}_B[]$ Partial trace over system B, page 39
- $|\psi\rangle$ Quantum state, page 37
- ξ_{α} Complex Wiener process, page 47
- $\{,\}_+$ Anti-commutator, page 46
- E_i Projection operator onto the eigenvector $|i\rangle$, page 37
- *H* Quantum Hamiltonian, page 46
- K_{μ} Kraus operators, page 43
- L_{α} Lindblad operator, page 44
- Q_a Generator of the Lorentz boosts, page 287
- S Path integral action, page 50
- S_{FV} Feynman-Vernon action, page 50
- U Unitary operator, page 37
- x^+, x^- Ket and bra variables, page 49
- GKSL equation Equation (2.58), page 46

Classical-quantum theory

 $\frac{1}{2}(\{H(z),\varrho\}-\{\varrho,H(z)\})$ Alexandrov-Gerasimenko bracket, page 65

- $\Lambda^{\mu\nu}(z,t+\delta t|z',t)$ Classical-quantum transition amplitude, page 53
- \mathcal{I} Full classical-quantum path integral action, page 122

 $D^{\mu\nu}_{n,i_1\ldots i_n}(z,t)\,$ Moments of the classical-quantum Kramers-Moyal expansion, page 54

 $\bar{O} = \frac{1}{2}(O^- + O^+)$ The bra-ket average of a classical-quantum function, page 135

 ϱ Classical-quantum state, page 50

 $\Delta O = (O^- - O^+)$ The bra-ket difference of a classical-quantum function, page 135

- λ Decoherence rate, page 218
- Λ^{vec} Vectorization of a superoperator, page 59
- $\langle A, \rho \rangle$ Smeared classical-quantum state, page 158

 $\langle O(z) \rangle := \int dz \operatorname{Tr} [O(z)\varrho]$ Expectation value of a classical-quantum operator O(z), page 59

 $\mathcal{L}(x)$ Local generator of CQ dynamics $\partial_t \varrho = \int d^3 x \mathcal{L}(x)(\varrho)$, page 155

det λ^+ Determinant of the positive semi-definite block of $D_{2,ij}^{00}$, page 126

 $\sigma, D^{00}_{2,ij}=\frac{1}{2}(\sigma\sigma^T)_{ij}\,$ Decomposition of the classical diffusion coefficient, page 97

 $C^{\mu\nu}(u,z,t,\delta t)$ Classical-quantum characteristic function, page 53

 $D_0^{\alpha\beta}(z)$ Lindbladian coefficient, page 55

 $D_{1.ii}^{\mu\nu}(z)\,$ Diffusion coefficient, page 55

 $D_{1,i}^{diff}$ Difference vector between the classical path $\frac{dz_i}{dt}$ and its expected drift, page 126

- $D_{1,i}^{\mu\nu}(z)$ Drift coefficient, page 55
- D_1^{br} Back-reaction matrix, page 228
- f(z; x, y) Classical-quantum matrix kernel, page 208
- H(z) Classical-quantum Hamiltonian, page 54
- L_{μ} Lindblad operators, page 53

 $M^{\mu\nu}_{n,i_1\ldots i_n}(z,t,\delta t)\,$ Moments of the classical-quantum transition amplitude, page 53

 S_{CQ} Classical-quantum path integral coupling, page 122

 $W^{\mu\nu}(z|z',t)$ Short time expansion of the classical-quantum transition amplitude, page 53

 W_{CQ} Classical-quantum proto-action, page 135

 $X \succeq 0$ Positive semi-definite matrix, page 82

 $X^{-1}, X \succeq 0$ Generalized inverse of a positive semi-definite matrix, page 82

General form of classical-quantum master equation Equation (2.91), page 54

General form of continuous classical-quantum master equation Equation (2.93), page 56

Gravity

- Γ^a_{bc} Christoffel symbol of the covariant derivative D, page 69
- (M,g) Lorentzian manifold, page 66
- γ_{ab} Intrinsic metric on Σ_t , page 69
- $\mathcal{C}[\underline{M}]$ Smeared classical-quantum momentum constraint, page 169
- C^a Classical-quantum momentum constraint, page 167
- \mathcal{H} Hamiltonian constraint, page 71
- \mathcal{H}_a Momentum constraint, page 71

 $\mathcal{L}_{constraint}$ Classical-quantum Hamiltonian constraint, page 172

- ∇_{μ} Levi-Civita covariant derivative of g, page 67
- \mathcal{R}_{AB} Gravitational classical-quantum transformation couplings, page 171
- Φ Newtonian potential, page 78
- π^{ab} Momentum conjugate to γ_{ab} , page 71
- π_{Φ} Momentum conjugate to Φ , page 200
- Σ_t Hypersurface of a foliation, page 68

- C_J, C_N, C_H, C_{JN} Jump, no-event, Hamiltonian, and jump-no-event components of the CQ momentum constraint, page 167
- D Covariant derivative of γ , page 69
- $D_{0,\mu\nu\rho\sigma}$ Generalized de-Witt metric, page 192
- $G_{\mu\nu}$ Einstein tensor, page 18
- $g_{\mu\nu}$ Metric tensor, page 18
- G_{abcd} de-Witt metric, page 71

 $H[\vec{N}] = \int d^3x N^a(x) \mathcal{H}_a(x)$ Smeared momentum constraint, page 71

 $H[N] = \int d^3x N(x) \mathcal{H}(x)$ Smeared Hamiltonian constraint, page 71

- $h^{\alpha\beta}$ Decomposition of a matter super-Hamiltonian into Lindblad operators, page 151
- $H_m(x)$ Matter super-Hamiltonian, page 151
- $H_m[N,\vec{N}]\,$ Matter Hamiltonian, page 151
- $h_{\mu\nu}$ Perturbation of the metric tensor, page 199
- $H_{ADM}[N,\vec{N}]$ ADM Hamiltonian, page 71
- $H_{m,a}(x)$ Matter super-momentum, page 151
- K_{ab} Extrinsic curvature of the foliation Σ_t , page 70
- m(x) Mass density, page 78
- N Lapse function, page 68
- N^i Shift vector, page 68
- $p_a^{\alpha\beta}$ Decomposition of a matter super-momenta into Lindblad operators, page 151
- P_N, P_{N^a} Momentum conjugate to the lapse and shift vectors, page 71
- R Ricci scalar, page 66

- $R_{\mu\nu}$ Ricci tensor, page 66
- S_{EH} Einstein-Hilbert action, page 67
- $T_{\mu\nu}$ Stress-energy tensor, page 18
- ${}^{(3)}R^a_{bcd}\,$ Riemann tensor of $\gamma,$ page 69

Constants

$$\begin{split} &\hbar = 6.626 \times 10^{-34} \ m^2 kg/s \ \text{Planck's constant, page 239} \\ &c = 2.997 \times 10^8 \ m/s \ \text{Speed of light, page 18} \\ &G = 6.674 \times 10^{-11} \ m^3 kg^{-1}s^{-2} \ \text{Newtons constant, page 18} \\ &l_P = 1.616 \times 10^{-35} \ m \ \text{Planck's length, page 239} \\ &m_P = 2.176 \times 10^{-8} \ kg \ \text{Planck's mass, page 239} \end{split}$$

Experiment

- M_{λ} Mass of a composite particle, page 238
- V_{λ} Volume of a composite particle, page 238
- ΔT Time resolution of an experiment, page 237
- λ Decoherence rate, page 237
- ρ Density of a single particle, page 238
- σ_a Variation of the time-averaged acceleration, page 238
- σ_F Variation of the time-averaged force, page 237
- \vec{F}_{tot} Time-averaged force, page 235

 $D_0(x,y) = \frac{D_0}{|x-y|}$ Diosi-Penrose (D.P) kernel, page 219

 $D_0(x,y) = D_0\delta(x,y)$ Ultra-local kernel, page 219

- m_N Mass of a single particle, page 238
- N Number of particles in a composite particle, page 238
- r_N Radius of a single particle, page 238
- R_{λ} Radius of a composite particle, page 238
- V_b Volume of space with significant background Newtonian potential, page 238

Abbreviations

- *CP* Completely positive, page 43
- $CPTP\$ Completely positive and trace preserving, page 42
- CQ Classical-quantum, page 17
- D.P Diosi-Penrose, page 219
- GIE Gravity induced entanglement, page 217
- GR General relativity, page 66
- $POVM\,$ Positive operator valued measure, page 104
- QFT Quantum field theory, page 152

Bibliography

- Sougato Bose, Anupam Mazumdar, Gavin W Morley, Hendrik Ulbricht, Marko Toroš, Mauro Paternostro, Andrew A Geraci, Peter F Barker, MS Kim, and Gerard Milburn. Spin entanglement witness for quantum gravity. *Physical review letters*, 119(24):240401, 2017.
- [2] Chiara Marletto and Vlatko Vedral. Gravitationally induced entanglement between two massive particles is sufficient evidence of quantum effects in gravity. *Physical review letters*, 119(24):240402, 2017.
- Jonathan Oppenheim and Zachary Weller-Davies. The constraints of post-quantum classical gravity. JHEP, 02:080, 2022. doi:10.1007/JHEP02(2022)080.
- [4] Jonathan Oppenheim, Carlo Sparaciari, Barbara Soda, and Zachary Weller-Davies. Gravitationally induced decoherence vs space-time diffusion: testing the quantum nature of gravity. *Nature Commun.*, 14(1):7910, 2023. doi:10.1038/s41467-023-43348-2.
- [5] Jonathan Oppenheim, Carlo Sparaciari, Barbara Šoda, and Zachary Weller-Davies. The two classes of hybrid classical-quantum dynamics, 2022. URL https://arxiv.org/abs/ 2203.01332.
- [6] Isaac Layton, Jonathan Oppenheim, and Zachary Weller-Davies. A healthier semiclassical dynamics, 2022. URL https://arxiv.org/abs/2208.11722.
- [7] Jonathan Oppenheim and Zachary Weller-Davies. Path integrals for classical-quantum dynamics, 2023. URL https://arxiv.org/abs/2301.04677.

- [8] Jonathan Oppenheim and Zachary Weller-Davies. Covariant path integrals for quantum fields back-reacting on classical space-time, 2023. URL https://arxiv.org/abs/2302. 07283.
- [9] Isaac Layton, Jonathan Oppenheim, Andrea Russo, and Zachary Weller-Davies. The weak field limit of quantum matter back-reacting on classical spacetime. *JHEP*, 08:163, 2023. doi:10.1007/JHEP08(2023)163.
- [10] Jonathan Oppenheim, Carlo Sparaciari, Barbara Šoda, and Zachary Weller-Davies. Objective trajectories in hybrid classical-quantum dynamics. *Quantum*, 7:891, 2023. doi:10.22331/q-2023-01-03-891.
- [11] Juan F. Pedraza, Andrea Russo, Andrew Svesko, and Zachary Weller-Davies. Lorentzian Threads as Gatelines and Holographic Complexity. *Phys. Rev. Lett.*, 127(27):271602, 2021. doi:10.1103/PhysRevLett.127.271602.
- [12] Juan F. Pedraza, Andrea Russo, Andrew Svesko, and Zachary Weller-Davies. Sewing spacetime with Lorentzian threads: complexity and the emergence of time in quantum gravity. JHEP, 02:093, 2022. doi:10.1007/JHEP02(2022)093.
- [13] Juan F. Pedraza, Andrea Russo, Andrew Svesko, and Zachary Weller-Davies. Computing spacetime. Int. J. Mod. Phys. D, 31(14):2242010, 2022. doi:10.1142/S021827182242010X.
- [14] Rafael Carrasco, Juan F. Pedraza, Andrew Svesko, and Zachary Weller-Davies. Gravitation from optimized computation: Einstein and beyond. JHEP, 09:167, 2023. doi:10.1007/JHEP09(2023)167.
- [15] H Georgi. Effective field theory. Annual Review of Nuclear and Particle Science, 43(1):
 209-252, 1993. doi:10.1146/annurev.ns.43.120193.001233. URL https://doi.org/10.
 1146/annurev.ns.43.120193.001233.
- [16] Kenneth G. Wilson. The renormalization group: Critical phenomena and the kondo problem. Rev. Mod. Phys., 47:773-840, Oct 1975. doi:10.1103/RevModPhys.47.773. URL https://link.aps.org/doi/10.1103/RevModPhys.47.773.

- [17] Cécile M. DeWitt and Dean Rickles. The role of gravitation in physics: Report from the 1957 Chapel Hill Conference, volume 5. epubli, 2011.
- [18] Kenneth Eppley and Eric Hannah. The necessity of quantizing the gravitational field. Foundations of Physics, 7(1-2):51–68, 1977.
- [19] Christian Møller et al. Les théories relativistes de la gravitation. Colloques Internationaux CNRS, 91(1), 1962.
- [20] Leon Rosenfeld. On quantization of fields. Nuclear Physics, 40:353–356, 1963.
- [21] B. L. Hu and E. Verdaguer. Stochastic Gravity: Theory and Applications. Living Rev. Rel., 11:3, 2008. doi:10.12942/lrr-2008-3.
- [22] Gary T. Horowitz. Semiclassical relativity: The weak-field limit. Phys. Rev. D, 21:1445– 1461, Mar 1980. doi:10.1103/PhysRevD.21.1445. URL https://link.aps.org/doi/10. 1103/PhysRevD.21.1445.
- [23] R. D. Jordan. Stability of flat spacetime in quantum gravity. *Phys. Rev. D*, 36:3593-3603, Dec 1987. doi:10.1103/PhysRevD.36.3593. URL https://link.aps.org/doi/10.1103/ PhysRevD.36.3593.
- [24] L. H. Ford. GRAVITATIONAL RADIATION BY QUANTUM SYSTEMS. Annals Phys., 144:238, 1982. doi:10.1016/0003-4916(82)90115-4.
- [25] Chung-I Kuo and L. H. Ford. Semiclassical gravity theory and quantum fluctuations. *Phys. Rev. D*, 47:4510–4519, 1993. doi:10.1103/PhysRevD.47.4510.
- [26] Robert M Wald. The back reaction effect in particle creation in curved spacetime. Communications in Mathematical Physics, 54(1):1–19, 1977.
- [27] James B. Hartle and Gary T. Horowitz. Ground-state expectation value of the metric in the ¹/_N or semiclassical approximation to quantum gravity. *Phys. Rev. D*, 24:257-274, Jul 1981. doi:10.1103/PhysRevD.24.257. URL https://link.aps.org/doi/10.1103/ PhysRevD.24.257.

- [28] Jonathan Oppenheim. A post-quantum theory of classical gravity? arXiv:1811.03116 [hep-th, physics:quant-ph], 2018.
- [29] Don N Page and CD Geilker. Indirect evidence for quantum gravity. Physical Review Letters, 47(14):979, 1981.
- [30] Yaakov Fein, Philipp Geyer, Patrick Zwick, Filip Kialka, Sebastian Pedalino, Marcel Mayor, Stefan Gerlich, and Markus Arndt. Quantum superposition of molecules beyond 25 kda. *Nature Physics*, 15:1–4, 12 2019. doi:10.1038/s41567-019-0663-9.
- [31] B. L. Hu and Nicholas G. Phillips. Fluctuations of energy density and validity of semiclassical gravity. Int. J. Theor. Phys., 39:1817–1830, 2000. doi:10.1023/A:1003689630751.
- [32] Alejandro Perez, Hanno Sahlmann, and Daniel Sudarsky. On the quantum origin of the seeds of cosmic structure. *Class. Quant. Grav.*, 23:2317–2354, 2006. doi:10.1088/0264-9381/23/7/008.
- [33] Claus Kiefer and David Polarski. Why do cosmological perturbations look classical to us? Adv. Sci. Lett., 2:164–173, 2009. doi:10.1166/asl.2009.1023.
- [34] Claus Kiefer, Ingo Lohmar, David Polarski, and Alexei A. Starobinsky. Pointer states for primordial fluctuations in inflationary cosmology. *Class. Quant. Grav.*, 24:1699–1718, 2007. doi:10.1088/0264-9381/24/7/002.
- [35] Richard P Feynman, Fernando B Morinigo, and William G Wagner. Feynman lectures on gravitation. European Journal of Physics, 24(3):330, may 2003. doi:10.1088/0143-0807/24/3/702. URL https://dx.doi.org/10.1088/0143-0807/24/3/702.
- [36] Niels Bohr and Léon Rosenfeld. On the question of the measurability of electromagnetic field quantities. *Quantum theory and measurement*, pages 478–522, 1933.
- [37] Gordon Baym and Tomoki Ozawa. Two-slit diffraction with highly charged particles: Niels bohr's consistency argument that the electromagnetic field must be quantized. Proceedings of the National Academy of Sciences, 106(9):3035-3040, 2009. ISSN 0027-8424. doi:10.1073/pnas.0813239106. URL http://www.pnas.org/content/106/9/3035.

- [38] Alessio Belenchia, Robert Wald, Flaminia Giacomini, Esteban Castro-Ruiz, Caslav Brukner, and Markus Aspelmeyer. Quantum superposition of massive objects and the quantization of gravity. arXiv preprint arXiv:1807.07015, 2018.
- [39] Bryce S DeWitt. Definition of commutators via the uncertainty principle. Journal of Mathematical Physics, 3(4):619–624, 1962.
- [40] Yakir Aharonov and Daniel Rohrlich. Quantum Paradoxes: Quantum Theory for the Perplexed. Wiley-VCH, 09 2003. doi:10.1002/9783527619115. pp212-213.
- [41] J Caro and LL Salcedo. Impediments to mixing classical and quantum dynamics. Physical Review A, 60(2):842, 1999.
- [42] LL Salcedo. Absence of classical and quantum mixing. Physical Review A, 54(4):3657, 1996.
- [43] Debendranath Sahoo. Mixing quantum and classical mechanics and uniqueness of planck's constant. Journal of Physics A: Mathematical and General, 37(3):997, 2004.
- [44] Daniel R Terno. Inconsistency of quantum—classical dynamics, and what it implies. Foundations of Physics, 36(1):102–111, 2006.
- [45] LL Salcedo. Statistical consistency of quantum-classical hybrids. *Physical Review A*, 85 (2):022127, 2012.
- [46] Carlos Barceló, Raúl Carballo-Rubio, Luis J Garay, and Ricardo Gómez-Escalante. Hybrid classical-quantum formulations ask for hybrid notions. *Physical Review A*, 86(4): 042120, 2012.
- [47] Chiara Marletto and Vlatko Vedral. Why we need to quantise everything, including gravity. npj Quantum Information, 3(1):29, 2017.
- [48] Nicolas Gisin. Stochastic quantum dynamics and relativity. Helv. Phys. Acta, 62(4): 363–371, 1989.
- [49] Antoine Tilloy and Lajos Diósi. Sourcing semiclassical gravity from spontaneously localized quantum matter. *Physical Review D*, 93(2):024026, 2016.

- [50] Adrian Kent. Nonlinearity without superluminality. *Physical Review A*, 72(1), Jul 2005.
 ISSN 1094-1622. doi:10.1103/physreva.72.012108. URL http://dx.doi.org/10.1103/ PhysRevA.72.012108.
- [51] Thomas D. Galley, Flaminia Giacomini, and John H. Selby. A no-go theorem on the nature of the gravitational field beyond quantum theory, 2021.
- [52] D Kafri, JM Taylor, and GJ Milburn. A classical channel model for gravitational decoherence. New Journal of Physics, 16(6):065020, 2014.
- [53] D Kafri, G J Milburn, and J M Taylor. Bounds on quantum communication via newtonian gravity. New Journal of Physics, 17(1):015006, jan 2015. doi:10.1088/1367-2630/17/1/015006. URL https://doi.org/10.1088/1367-2630/17/1/015006.
- [54] Antoine Tilloy and Lajos Diósi. On gkls dynamics for local operations and classical communication. Open Systems & Information Dynamics, 24:1740020, 2017. doi:10.1142/S1230161217400200.
- [55] Daniel Carney and Jacob M. Taylor. Strongly incoherent gravity, 2023. URL https: //arxiv.org/abs/2301.08378.
- [56] Philippe Blanchard and Arkadiusz Jadczyk. On the interaction between classical and quantum systems. *Physics Letters A*, 175(3-4):157–164, 1993.
- [57] Lajos Diosi. Quantum dynamics with two planck constants and the semiclassical limit. arXiv preprint quant-ph/9503023, 1995.
- [58] Robert Alicki and Stanisław Kryszewski. Completely positive bloch-boltzmann equations. Physical Review A, 68(1):013809, 2003.
- [59] Lajos Diósi, Nicolas Gisin, and Walter T Strunz. Quantum approach to coupling classical and quantum dynamics. *Physical Review A*, 61(2):022108, 2000.
- [60] Lajos Diósi. The gravity-related decoherence master equation from hybrid dynamics. In Journal of Physics-Conference Series, volume 306, page 012006, 2011.

- [61] David Poulin and John Preskill. Information loss in quantum field theories. Frontiers of Quantum Information Physics, KITP, 2017. URL http://online.kitp.ucsb.edu/ online/qinfo-c17/poulin/.
- [62] V. Gorini, A. Kossakowski, and E. C. G. Sudarshan. Completely positive dynamical semigroups of n-level systems. *Journal of Mathematical Physics*, 17:821–825, may 1976. doi:10.1063/1.522979.
- [63] G. Lindblad. On the generators of quantum dynamical semigroups. Commun. Math. Phys., 48:119–130, 1976.
- [64] Dvir Kafri and J. M. Taylor. A noise inequality for classical forces, 2013. URL https: //arxiv.org/abs/1311.4558.
- [65] Ryan J. Marshman, Anupam Mazumdar, and Sougato Bose. Locality and entanglement in table-top testing of the quantum nature of linearized gravity. *Physical Review A*, 101 (5), May 2020. doi:10.1103/physreva.101.052110.
- [66] Julen S. Pedernales, Kirill Streltsov, and Martin B. Plenio. Enhancing Gravitational Interaction between Quantum Systems by a Massive Mediator. *Phys. Rev. Lett.*, 128(11): 110401, 2022. doi:10.1103/PhysRevLett.128.110401.
- [67] Daniel Carney, Holger Müller, and Jacob M. Taylor. Using an Atom Interferometer to Infer Gravitational Entanglement Generation. *PRX Quantum*, 2(3):030330, 2021. doi:10.1103/PRXQuantum.2.030330.
- [68] Marios Christodoulou, Andrea Di Biagio, Markus Aspelmeyer, Caslav Brukner, Carlo Rovelli, and Richard Howl. Locally mediated entanglement through gravity from first principles, 2022. URL https://arxiv.org/abs/2202.03368.
- [69] Ludovico Lami, Julen S. Pedernales, and Martin B. Plenio. Testing the quantumness of gravity without entanglement, 2023. URL https://arxiv.org/abs/2302.03075.
- [70] Richard Howl, Vlatko Vedral, Devang Naik, Marios Christodoulou, Carlo Rovelli, and Aditya Iyer. Non-Gaussianity as a signature of a quantum theory of gravity. *PRX Quantum*, 2:010325, 2021. doi:10.1103/PRXQuantum.2.010325.

- [71] Thomas Thiemann. Loop Quantum Gravity: An Inside View. Lect. Notes Phys., 721: 185–263, 2007. doi:10.1007/978-3-540-71117-9_10.
- [72] Hermann Nicolai, Kasper Peeters, and Marija Zamaklar. Loop quantum gravity: an outside view. Classical and Quantum Gravity, 22(19):R193-R247, sep 2005. doi:10.1088/0264-9381/22/19/r01. URL https://doi.org/10.1088%2F0264-9381% 2F22%2F19%2Fr01.
- [73] H. Risken and H. Haken. The Fokker-Planck Equation: Methods of Solution and Applications Second Edition. Springer, 1989.
- [74] Lajos Diósi. Hybrid quantum-classical master equations. *Physica Scripta*, 2014(T163): 014004, 2014.
- [75] Lajos Diósi and Jonathan J. Halliwell. Coupling classical and quantum variables using continuous quantum measurement theory. *Physical Review Letters*, 81(14):2846-2849, Oct 1998. ISSN 1079-7114. doi:10.1103/physrevlett.81.2846. URL http://dx.doi.org/10.1103/PhysRevLett.81.2846.
- [76] RF Pawula. Rf pawula, phys. rev. 162, 186 (1967). Phys. Rev., 162:186, 1967.
- [77] Richard Arnowitt, Stanley Deser, and Charles W Misner. Republication of: The dynamics of general relativity. *General Relativity and Gravitation*, 40(9):1997–2027, 2008.
- [78] Bryce S DeWitt. Quantum theory of gravity. i. the canonical theory. *Physical Review*, 160(5):1113, 1967.
- [79] R.P. Feynman and F.L. Vernon. The theory of a general quantum system interacting with a linear dissipative system. Annals of Physics, 281(1):547-607, 2000. ISSN 0003-4916. doi:https://doi.org/10.1006/aphy.2000.6017. URL https://www.sciencedirect.com/science/article/pii/S0003491600960172.
- [80] Markus F Weber and Erwin Frey. Master equations and the theory of stochastic path integrals. Reports on Progress in Physics, 80(4):046601, Mar 2017. ISSN 1361-6633. doi:10.1088/1361-6633/aa5ae2. URL http://dx.doi.org/10.1088/1361-6633/aa5ae2.

- [81] SAMUEL KARLIN. Preface to first edition. In SAMUEL KARLIN and HOWARD M. TAYLOR, editors, A First Course in Stochastic Processes (Second Edition), pages xv-xvi. Academic Press, Boston, second edition edition, 1975. ISBN 978-0-08-057041-9. doi:https://doi.org/10.1016/B978-0-08-057041-9.50004-0. URL https://www. sciencedirect.com/science/article/pii/B9780080570419500040.
- [82] Heinz-Peter Breuer and Francesco Petruccione. The Theory of Open Quantum Systems. Oxford University Press, 01 2007. ISBN 9780199213900.
 doi:10.1093/acprof:oso/9780199213900.001.0001. URL https://doi.org/10.1093/acprof:oso/9780199213900.001.0001.
- [83] Hendrik Anthony Kramers. Brownian motion in a field of force and the diffusion model of chemical reactions. *Physica*, 7(4):284–304, 1940.
- [84] JE Moyal. Stochastic processes and statistical physics. Journal of the Royal Statistical Society. Series B (Methodological), 11(2):150–210, 1949.
- [85] M. M. Wolf, J. Eisert, T. S. Cubitt, and J. I. Cirac. Assessing non-markovian quantum dynamics. *Phys. Rev. Lett.*, 101:150402, Oct 2008. doi:10.1103/PhysRevLett.101.150402. URL https://link.aps.org/doi/10.1103/PhysRevLett.101.150402.
- [86] Kiyosi Itô. Stochastic integral. Proceedings of the Imperial Academy, 20(8):519 524, 1944. doi:10.3792/pia/1195572786. URL https://doi.org/10.3792/pia/1195572786.
- [87] Hagen Kleinert. Path Integrals in Quantum Mechanics, Statistics, Polymer Physics, and Financial Markets. WORLD SCIENTIFIC, 5th edition, 2009. doi:10.1142/7305. URL https://www.worldscientific.com/doi/abs/10.1142/7305.
- [88] John Preskill. Lecture notes for physics 219: Quantum computation. California Institute of Technology, 1999.
- [89] Lane P. Hughston, Richard Jozsa, and William K. Wootters. A complete classification of quantum ensembles having a given density matrix. *Physics Letters A*, 183(1):14–18, 1993. ISSN 0375-9601. doi:https://doi.org/10.1016/0375-9601(93)90880-9. URL https://www.sciencedirect.com/science/article/pii/0375960193908809.

- [90] J. S. Bell. On the Einstein-Podolsky-Rosen paradox. *Physics Physique Fizika*, 1:195–200, 1964. doi:10.1103/PhysicsPhysiqueFizika.1.195.
- [91] Karl Kraus, A. Böhm, J. D. Dollard, and W. H. Wootters. States, Effects, and Operations Fundamental Notions of Quantum Theory, volume 190. Springer Berlin, Heidelberg, 1983. doi:10.1007/3-540-12732-1.
- [92] Michael J. W. Hall, James D. Cresser, Li Li, and Erika Andersson. Canonical form of master equations and characterization of non-markovianity. *Physical Review A*, 89(4), apr 2014. doi:10.1103/physreva.89.042120. URL https://doi.org/10.1103%2Fphysreva. 89.042120.
- [93] Heinz-Peter Breuer, Elsi-Mari Laine, Jyrki Piilo, and Bassano Vacchini. Colloquium: Non-markovian dynamics in open quantum systems. *Rev. Mod. Phys.*, 88:021002, Apr 2016. doi:10.1103/RevModPhys.88.021002. URL https://link.aps.org/doi/10.1103/ RevModPhys.88.021002.
- [94] D. Tamascelli, A. Smirne, S. F. Huelga, and M. B. Plenio. Nonperturbative treatment of non-markovian dynamics of open quantum systems. *Physical Review Letters*, 120 (3), jan 2018. doi:10.1103/physrevlett.120.030402. URL https://doi.org/10.1103% 2Fphysrevlett.120.030402.
- [95] Philip Pearle. Simple derivation of the lindblad equation. European Journal of Physics, 33(4):805-822, apr 2012. doi:10.1088/0143-0807/33/4/805. URL https://doi.org/10. 1088%2F0143-0807%2F33%2F4%2F805.
- [96] H.M. Wiseman and L. Diósi. Complete parameterization, and invariance, of diffusive quantum trajectories for markovian open systems. *Chemical Physics*, 268(1-3):91-104, Jun 2001. ISSN 0301-0104. doi:10.1016/s0301-0104(01)00296-8. URL http://dx.doi.org/10.1016/S0301-0104(01)00296-8.
- [97] L M Sieberer, M Buchhold, and S Diehl. Keldysh field theory for driven open quantum systems. *Reports on Progress in Physics*, 79(9):096001, Aug 2016. ISSN 1361-6633. doi:10.1088/0034-4885/79/9/096001. URL http://dx.doi.org/10.1088/0034-4885/79/9/096001.

- [98] David Poulin, 2017. private communication (result announced in [61]).
- [99] V. P. Belavkin. A stochastic posterior Schrödinger equation for counting nondemolition measurement. Letters in Mathematical Physics, 20(2):85–89, August 1990. doi:10.1007/BF00398273.
- [100] C. W. Gardiner, A. S. Parkins, and P. Zoller. Wave-function quantum stochastic differential equations and quantum-jump simulation methods. *Phys. Rev. A*, 46:4363-4381, Oct 1992. doi:10.1103/PhysRevA.46.4363. URL https://link.aps.org/doi/10.1103/ PhysRevA.46.4363.
- [101] Jean Dalibard, Yvan Castin, and Klaus Mølmer. Wave-function approach to dissipative processes in quantum optics. *Phys. Rev. Lett.*, 68:580-583, Feb 1992. doi:10.1103/PhysRevLett.68.580. URL https://link.aps.org/doi/10.1103/ PhysRevLett.68.580.
- [102] Michael A. Nielsen and Isaac L. Chuang. Quantum Computation and Quantum Information. Cambridge University Press, 2000.
- [103] IV Aleksandrov and Z Naturf. 36a, 902 (1981); a. anderson. Phys. Rev. Lett, 74:621, 1995.
- [104] V. I. Gerasimenko. Dynamical equations of quantum-classical systems. Theoretical and Mathematical Physics, 50(1):49-55, 1982. doi:10.1007/BF01027604. URL https://doi. org/10.1007/BF01027604.
- [105] Eric Gourgoulhon. 3+1 formalism and bases of numerical relativity, 2007. URL https: //arxiv.org/abs/gr-qc/0703035.
- [106] P.A.M. Dirac. Lectures on Quantum Mechanics. Belfer Graduate School of Science, monograph series. Dover Publications, 2001. ISBN 9780486417134. URL https://books. google.co.uk/books?id=GVwzb1rZW9kC.
- [107] C.J. Isham. Canonical quantum gravity and the problem of time. NATO Sci. Ser. C, 409:157–287, 1993.

- [108] S. A. Hojman, K. Kuchar, and C. Teitelboim. Geometrodynamics Regained. Annals Phys., 96:88–135, 1976. doi:10.1016/0003-4916(76)90112-3.
- [109] J M Pons, D C Salisbury, and K A Sundermeyer. Observables in classical canonical gravity: Folklore demystified. Journal of Physics: Conference Series, 222(1):012018, apr 2010. doi:10.1088/1742-6596/222/1/012018. URL https://dx.doi.org/10.1088/ 1742-6596/222/1/012018.
- [110] J. M. Pons, D. C. Salisbury, and L. C. Shepley. Gauge transformations in the Lagrangian and Hamiltonian formalisms of generally covariant theories. *Phys. Rev. D*, 55:658–668, 1997. doi:10.1103/PhysRevD.55.658.
- Bryce S. DeWitt. Quantum theory of gravity. ii. the manifestly covariant theory. *Phys. Rev.*, 162:1195–1239, Oct 1967. doi:10.1103/PhysRev.162.1195. URL https://link.aps.org/doi/10.1103/PhysRev.162.1195.
- [112] Claus Kiefer. Quantum gravity, volume 124. Clarendon, Oxford, 2004. ISBN 978-0-19-958520-5.
- [113] M. J. G. Veltman. Quantum Theory of Gravitation. Conf. Proc. C, 7507281:265–327, 1975.
- [114] David Wallace. Quantum gravity at low energies, 2021. URL https://arxiv.org/abs/ 2112.12235.
- [115] REMO RUFFINI and SILVANO BONAZZOLA. Systems of self-gravitating particles in general relativity and the concept of an equation of state. *Phys. Rev.*, 187:1767-1783, Nov 1969. doi:10.1103/PhysRev.187.1767. URL https://link.aps.org/doi/10.1103/ PhysRev.187.1767.
- [116] Lajos Diósi. Models for universal reduction of macroscopic quantum fluctuations. *Physical Review A*, 40(3):1165, 1989.
- [117] Roger Penrose. On gravity's role in quantum state reduction. General relativity and gravitation, 28(5):581–600, 1996.

- [118] André Großardt. Three little paradoxes: Making sense of semiclassical gravity. AVS Quantum Science, 4(1):010502, mar 2022. doi:10.1116/5.0073509. URL https://doi. org/10.1116%2F5.0073509.
- [119] Daniela Kohen, Frank H Stillinger, and John C Tully. Model studies of nonadiabatic dynamics. The Journal of chemical physics, 109(12):4713–4725, 1998.
- [120] Oleg V. Prezhdo. Mean field approximation for the stochastic schrödinger equation. The Journal of Chemical Physics, 111(18):8366-8377, 1999. doi:10.1063/1.480178. URL https://doi.org/10.1063/1.480178.
- [121] Kurt Jacobs and Daniel A. Steck. A straightforward introduction to continuous quantum measurement. Contemporary Physics, 47(5):279-303, Sep 2006. ISSN 1366-5812. doi:10.1080/00107510601101934. URL http://dx.doi.org/10.1080/00107510601101934.
- [122] Howard M. Wiseman and Gerard J. Milburn. Quantum Measurement and Control. Cambridge University Press, 2009. doi:10.1017/CBO9780511813948.
- [123] N. Gisin. Quantum measurements and stochastic processes. *Phys. Rev. Lett.*, 52:1657–1660, May 1984. doi:10.1103/PhysRevLett.52.1657. URL https://link.aps.org/doi/10.1103/PhysRevLett.52.1657.
- [124] Antoine Tilloy and Lajos Diósi. Principle of least decoherence for newtonian semiclassical gravity. *Physical Review D*, 96(10):104045, 2017.
- T. N. Sherry and E. C. G. Sudarshan. Interaction between classical and quantum systems: A new approach to quantum measurement.i. *Phys. Rev. D*, 18:4580-4589, Dec 1978. doi:10.1103/PhysRevD.18.4580. URL https://link.aps.org/doi/10.1103/PhysRevD. 18.4580.
- T. N. Sherry and E. C. G. Sudarshan. Interaction between classical and quantum systems: A new approach to quantum measurement. ii. theoretical considerations. *Phys. Rev. D*, 20:857-868, Aug 1979. doi:10.1103/PhysRevD.20.857. URL https://link.aps.org/ doi/10.1103/PhysRevD.20.857.

- [127] E. C. G. Sudarshan. Interaction between classical and quantum systems and the measurement of quantum observables. *Pramana*, 6(3):117–126, 1976. doi:10.1007/BF02847120.
 URL https://doi.org/10.1007/BF02847120.
- [128] Philippe Blanchard and Arkadiusz Jadczyk. Event-enhanced quantum theory and piecewise deterministic dynamics. Annalen der Physik, 507(6):583–599, 1995. https://arxiv.org/abs/hep-th/9409189.
- [129] Rafael D. Sorkin. Impossible measurements on quantum fields. In Directions in General Relativity: An International Symposium in Honor of the 60th Birthdays of Dieter Brill and Charles Misner, 2 1993.
- [130] Henning Bostelmann, Christopher J. Fewster, and Maximilian H. Ruep. Impossible measurements require impossible apparatus. *Phys. Rev. D*, 103(2):025017, 2021. doi:10.1103/PhysRevD.103.025017.
- [131] T. O. Philips. Lorentz invariant localized states. *Phys. Rev.*, 136:B893-B896, Nov 1964. doi:10.1103/PhysRev.136.B893. URL https://link.aps.org/doi/10.1103/PhysRev. 136.B893.
- [132] Donald Marolf and Carlo Rovelli. Relativistic quantum measurement. *Physical Review* D, 66(2), Jul 2002. ISSN 1089-4918. doi:10.1103/physrevd.66.023510. URL http://dx. doi.org/10.1103/PhysRevD.66.023510.
- [133] Lluis Masanes, Thomas Galley, and Markus Müller. The measurement postulates of quantum mechanics are operationally redundant. *Nature Communications*, 10:1361, 03 2019. doi:10.1038/s41467-019-09348-x.
- [134] Esteban Calzetta and B. L. Hu. Noise and fluctuations in semiclassical gravity. *Phys. Rev. D*, 49:6636-6655, Jun 1994. doi:10.1103/PhysRevD.49.6636. URL https://link.aps.org/doi/10.1103/PhysRevD.49.6636.
- [135] B. L. Hu and A. Matacz. Back reaction in semiclassical gravity: The einstein-langevin equation. Phys. Rev. D, 51:1577-1586, Feb 1995. doi:10.1103/PhysRevD.51.1577. URL https://link.aps.org/doi/10.1103/PhysRevD.51.1577.

- [136] Rosario Martin and Enric Verdaguer. Metric fluctuations in semiclassical gravity, 1997.
 URL https://arxiv.org/abs/gr-qc/9710137.
- [137] Rosario Martin and Enric Verdaguer. On the semiclassical Einstein-Langevin equation.
 Phys. Lett. B, 465:113–118, 1999. doi:10.1016/S0370-2693(99)01068-0.
- [138] Claus Kiefer, David Polarski, and Alexei A. Starobinsky. Quantum to classical transition for fluctuations in the early universe. Int. J. Mod. Phys. D, 7:455–462, 1998. doi:10.1142/S0218271898000292.
- [139] Jonathan J. Halliwell. Decoherence in Quantum Cosmology. *Phys. Rev. D*, 39:2912, 1989. doi:10.1103/PhysRevD.39.2912.
- [140] A. A. Starobinsky. Stochastic de sitter (inflationary) stage in the early universe. In H. J. de Vega and N. Sánchez, editors, *Field Theory, Quantum Gravity and Strings*, pages 107–126, Berlin, Heidelberg, 1986. Springer Berlin Heidelberg. ISBN 978-3-540-39789-2.
- [141] Alexei A Starobinsky and Jun'ichi Yokoyama. Equilibrium state of a self-interacting scalar field in the de sitter background. *Physical Review D*, 50(10):6357, 1994.
- [142] David Polarski and Alexei A. Starobinsky. Semiclassicality and decoherence of cosmological perturbations. *Class. Quant. Grav.*, 13:377–392, 1996. doi:10.1088/0264-9381/13/3/006.
- Misao Sasaki. Large Scale Quantum Fluctuations in the Inflationary Universe.
 Progress of Theoretical Physics, 76(5):1036-1046, 11 1986. ISSN 0033-068X.
 doi:10.1143/PTP.76.1036. URL https://doi.org/10.1143/PTP.76.1036.
- [144] Alan H. Guth and So-Young Pi. Fluctuations in the new inflationary universe. Phys. Rev. Lett., 49:1110–1113, Oct 1982. doi:10.1103/PhysRevLett.49.1110. URL https: //link.aps.org/doi/10.1103/PhysRevLett.49.1110.
- [145] Alan H. Guth and So-Young Pi. Quantum mechanics of the scalar field in the new inflationary universe. *Phys. Rev. D*, 32:1899–1920, Oct 1985. doi:10.1103/PhysRevD.32.1899.
 URL https://link.aps.org/doi/10.1103/PhysRevD.32.1899.

- B J Berne and D Thirumalai. On the simulation of quantum systems: Path integral methods. Annual Review of Physical Chemistry, 37(1):401-424, 1986. doi:10.1146/annurev.pc.37.100186.002153. URL https://doi.org/10.1146/annurev. pc.37.100186.002153.
- [147] M. F. Herman, E. J. Bruskin, and B. J. Berne. On path integral monte carlo simulations. *The Journal of Chemical Physics*, 76(10):5150-5155, 1982. doi:10.1063/1.442815. URL https://doi.org/10.1063/1.442815.
- [148] A. Chantasri, J. Dressel, and A. N. Jordan. Action principle for continuous quantum measurement. *Phys. Rev. A*, 88:042110, Oct 2013. doi:10.1103/PhysRevA.88.042110.
 URL https://link.aps.org/doi/10.1103/PhysRevA.88.042110.
- [149] Areeya Chantasri and Andrew N. Jordan. Stochastic path-integral formalism for continuous quantum measurement. *Physical Review A*, 92(3), Sep 2015. ISSN 1094-1622. doi:10.1103/physreva.92.032125. URL http://dx.doi.org/10.1103/PhysRevA. 92.032125.
- [150] Carlton M. Caves. Quantum mechanics of measurements distributed in time. a path-integral formulation. *Phys. Rev. D*, 33:1643-1665, Mar 1986. doi:10.1103/PhysRevD.33.1643. URL https://link.aps.org/doi/10.1103/PhysRevD. 33.1643.
- [151] Hongduo Wei and Yuli V. Nazarov. Statistics of measurement of noncommuting quantum variables: Monitoring and purification of a qubit. *Phys. Rev. B*, 78:045308, Jul 2008. doi:10.1103/PhysRevB.78.045308. URL https://link.aps.org/doi/10.1103/ PhysRevB.78.045308.
- [152] Heinz-Peter Breuer, Bernd Kappler, and Francesco Petruccione. Stochastic wave-function approach to the calculation of multitime correlation functions of open quantum systems. *Phys. Rev. A*, 56:2334–2351, Sep 1997. doi:10.1103/PhysRevA.56.2334. URL https: //link.aps.org/doi/10.1103/PhysRevA.56.2334.
- [153] L. Onsager and S. Machlup. Fluctuations and irreversible processes. Phys. Rev., 91:

1505-1512, Sep 1953. doi:10.1103/PhysRev.91.1505. URL https://link.aps.org/doi/ 10.1103/PhysRev.91.1505.

- [154] Mark Iosifovich Freidlin and Alexander D Wentzell. Random perturbations. In Random perturbations of dynamical systems, pages 15–43. Springer, 1998.
- [155] Nancy Makri. Quantum-classical path integral: A rigorous approach to condensed phase dynamics. International Journal of Quantum Chemistry, 115(18):1209-1214, 2015. doi:https://doi.org/10.1002/qua.24975. URL https://onlinelibrary.wiley.com/doi/ abs/10.1002/qua.24975.
- [156] Roberto Lambert and Nancy Makri. Quantum-classical path integral. i. classical memory and weak quantum nonlocality. *The Journal of Chemical Physics*, 137(22):22A552, 2012. doi:10.1063/1.4767931. URL https://doi.org/10.1063/1.4767931.
- [157] Roberto Lambert and N. Makri. Quantum-classical path integral. ii. numerical methodology. The Journal of chemical physics, 137 22:22A553, 2012.
- [158] Fei Wang and Nancy Makri. Quantum-classical path integral with a harmonic treatment of the back-reaction. The Journal of Chemical Physics, 150(18):184102, 2019. doi:10.1063/1.5091725. URL https://doi.org/10.1063/1.5091725.
- [159] Gerard McCaul and Denys I. Bondar. How to win friends and influence functionals: Deducing stochasticity from deterministic dynamics, 2020.
- [160] H. F. Trotter. On the product of semi-groups of operators. Proceedings of the American Mathematical Society, 10(4):545-551, 1959. ISSN 00029939, 10886826. URL http:// www.jstor.org/stable/2033649.
- [161] H. Dekker. Functional integration and the onsager-machlup lagrangian for continuous markov processes in riemannian geometries. *Phys. Rev. A*, 19:2102–2111, May 1979. doi:10.1103/PhysRevA.19.2102. URL https://link.aps.org/doi/10.1103/PhysRevA. 19.2102.
- [162] R. Graham. Path integral formulation of general diffusion processes. Zeitschrift für Physik B Condensed Matter, 26:281–290, 1977.

- [163] I. V. Aleksandrov. The statistical dynamics of a system consisting of a classical and a quantum subsystem. Zeitschrift für Naturforschung A, 36(8):902-908, 1981. doi:doi:10.1515/zna-1981-0819. URL https://doi.org/10.1515/zna-1981-0819.
- [164] H. M. Wiseman and G. J. Milburn. Quantum theory of optical feedback via homodyne detection. *Phys. Rev. Lett.*, 70:548-551, Feb 1993. doi:10.1103/PhysRevLett.70.548. URL https://link.aps.org/doi/10.1103/PhysRevLett.70.548.
- [165] Avinash Baidya, Chandan Jana, R Loganayagam, and Arnab Rudra. Renormalization in open quantum field theory. part i. scalar field theory. *Journal of High Energy Physics*, 2017(11):204, 2017. Also, initial work by Poulin and Preskill (unpublished note).
- [166] G. J. Milburn. Decoherence and the conditions for the classical control of quantum systems. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 370(1975):4469–4486, Sep 2012. ISSN 1471-2962. doi:10.1098/rsta.2011.0487. URL http://dx.doi.org/10.1098/rsta.2011.0487.
- [167] D. Dong and I.R. Petersen. Quantum control theory and applications: a survey. IET Control Theory and Applications, 4(12):2651-2671, Dec 2010. ISSN 1751-8652. doi:10.1049/iet-cta.2009.0508. URL http://dx.doi.org/10.1049/iet-cta.2009.0508.
- [168] Philip M. Pearle. Combining Stochastic Dynamical State Vector Reduction With Spontaneous Localization. Phys. Rev. A, 39:2277–2289, 1989. doi:10.1103/PhysRevA.39.2277.
- [169] Gian Carlo Ghirardi, Philip Pearle, and Alberto Rimini. Markov processes in hilbert space and continuous spontaneous localization of systems of identical particles. *Phys. Rev. A*, 42:78–89, Jul 1990. doi:10.1103/PhysRevA.42.78. URL https://link.aps.org/ doi/10.1103/PhysRevA.42.78.
- [170] Angelo Bassi and GianCarlo Ghirardi. Dynamical reduction models. *Physics Reports*, 379(5-6):257-426, Jun 2003. ISSN 0370-1573. doi:10.1016/s0370-1573(03)00103-0. URL http://dx.doi.org/10.1016/S0370-1573(03)00103-0.
- [171] G. C. Ghirardi, A. Rimini, and T. Weber. Unified dynamics for microscopic and macro-

scopic systems. *Phys. Rev. D*, 34:470-491, Jul 1986. doi:10.1103/PhysRevD.34.470. URL https://link.aps.org/doi/10.1103/PhysRevD.34.470.

- [172] Philip Pearle. Relativistic dynamical collapse model. Physical Review D, 91(10), May 2015. ISSN 1550-2368. doi:10.1103/physrevd.91.105012. URL http://dx.doi.org/10.1103/PhysRevD.91.105012.
- [173] Roderich Tumulka. A relativistic version of the ghirardi-rimini-weber model. Journal of Statistical Physics, 125(4):821-840, Dec 2006. ISSN 1572-9613. doi:10.1007/s10955-006-9227-3. URL http://dx.doi.org/10.1007/s10955-006-9227-3.
- [174] Avinash, Chandan Jana, and Arnab Rudra. Renormalisation in open quantum field theory ii: Yukawa theory and pv reduction, 2019. URL https://arxiv.org/abs/1906.10180.
- [175] R. Haag. Local quantum physics: Fields, particles, algebras. Springer Berlin, Heidelberg, 1992.
- [176] Stefan Hollands and Robert M. Wald. Quantum fields in curved spacetime. Phys. Rept., 574:1–35, 2015. doi:10.1016/j.physrep.2015.02.001.
- [177] I. Gelfand and M. Neumark. On the imbedding of normed rings into the ring of operators in Hilbert space. *Rec. Math. [Mat. Sbornik] N.S.*, 12(54):197–217, 1943.
- [178] I. E. Segal. Irreducible representations of operator algebras. Bulletin of the American Mathematical Society, 53(2):73 - 88, 1947. doi:bams/1183510397. URL https://doi.org/.
- [179] Petr Horava. Quantum Gravity at a Lifshitz Point. Phys. Rev., D79:084008, 2009. doi:10.1103/PhysRevD.79.084008.
- [180] William Donnelly and Ted Jacobson. Hamiltonian structure of hořava gravity. Phys. Rev. D, 84, 11 2011. doi:10.1103/PhysRevD.84.104019.
- [181] Miao Li and Yi Pang. A trouble with hořava-lifshitz gravity. Journal of High Energy Physics, 2009(08):015-015, aug 2009. doi:10.1088/1126-6708/2009/08/015. URL https: //doi.org/10.1088%2F1126-6708%2F2009%2F08%2F015.

- [182] Flavio Mercati. A shape dynamics tutorial, 2014. URL https://arxiv.org/abs/1409. 0105.
- [183] E Anderson, J Barbour, B Z Foster, B Kelleher, and N O Murchadha. The physical gravitational degrees of freedom. *Classical and Quantum Gravity*, 22(9):1795–1802, Apr 2005. ISSN 1361-6382. doi:10.1088/0264-9381/22/9/020. URL http://dx.doi.org/10.1088/0264-9381/22/9/020.
- [184] Henrique Gomes and Tim Koslowski. The link between general relativity and shape dynamics. *Classical and Quantum Gravity*, 29(7):075009, Mar 2012. ISSN 1361-6382. doi:10.1088/0264-9381/29/7/075009. URL http://dx.doi.org/10.1088/0264-9381/29/7/075009.
- [185] Karel Kuchar and Joseph Romano. Gravitational constraints which generate a lie algebra. *Physical review D: Particles and fields*, 51, 05 1996. doi:10.1103/PhysRevD.51.5579.
- [186] L. Smolin. Quantum gravity with a positive cosmological constant. arXiv: High Energy Physics - Theory, 2002.
- [187] N. Bretón, Jorge Cervantes-Cota, and Marcelo Salgad. The early universe and observational cosmology. *Lecture Notes in Physics*, 01 2004. doi:10.1007/b97189.
- [188] T. Thiemann. Quantum spin dynamics (QSD). Class. Quant. Grav., 15:839–873, 1998. doi:10.1088/0264-9381/15/4/011.
- [189] Marcus Gaul and Carlo Rovelli. A Generalized Hamiltonian constraint operator in loop quantum gravity and its simplest Euclidean matrix elements. *Class. Quant. Grav.*, 18: 1593–1624, 2001. doi:10.1088/0264-9381/18/9/301.
- [190] Thomas Thiemann. Modern Canonical Quantum General Relativity. Cambridge Monographs on Mathematical Physics. Cambridge University Press, 2007. doi:10.1017/CBO9780511755682.
- [191] Máximo Bañados and Ignacio Reyes. A short review on noether's theorems, gauge symmetries and boundary terms. International Journal of Modern Physics D, 01 2016. doi:10.1142/S0218271816300214.

- [192] J M Pons, D C Salisbury, and K A Sundermeyer. Observables in classical canonical gravity: Folklore demystified. Journal of Physics: Conference Series, 222:012018, Apr 2010. ISSN 1742-6596. doi:10.1088/1742-6596/222/1/012018. URL http://dx.doi.org/ 10.1088/1742-6596/222/1/012018.
- [193] Albert Einstein. Do gravitational fields play an essential part in the structure of the elementary particles of matter? *prel*, pages 189–198, 1952.
- [194] Jochum J Van der Bij, H Van Dam, and Yee Jack Ng. The exchange of massless spin-two particles. *Physica A: Statistical Mechanics and its Applications*, 116(1-2):307–320, 1982.
- [195] Steven Weinberg. The cosmological constant problem. Reviews of modern physics, 61(1): 1, 1989.
- [196] William G Unruh. Unimodular theory of canonical quantum gravity. Physical Review D, 40(4):1048, 1989.
- [197] Enrique Alvarez. Can one tell einstein's unimodular theory from einstein's general relativity? Journal of High Energy Physics, 2005(03):002, 2005.
- [198] Lee Smolin. Quantization of unimodular gravity and the cosmological constant problems. *Physical Review D*, 80(8):084003, 2009.
- [199] Mikhail Shaposhnikov and Daniel Zenhäusern. Scale invariance, unimodular gravity and dark energy. *Physics Letters B*, 671(1):187–192, 2009.
- [200] S. W. Hawking and Thomas Hertog. Living with ghosts. Phys. Rev. D, 65:103515, 2002. doi:10.1103/PhysRevD.65.103515.
- [201] Fiorenzo Bastianelli, Olindo Corradini, and Edoardo Vassura. Quantum mechanical path integrals in curved spaces and the type-a trace anomaly. Journal of High Energy Physics, 2017(4), April 2017. ISSN 1029-8479. doi:10.1007/jhep04(2017)050. URL http://dx.doi.org/10.1007/JHEP04(2017)050.
- [202] R Alicki, M Fannes, and A Verbeure. Unstable particles and the poincare semigroup in quantum field theory. Journal of Physics A: Mathematical and General, 19(6):919-927, 1986. URL http://stacks.iop.org/0305-4470/19/919.

- [203] David Poulin and John Preskill. Information loss in quantum field theories. Frontiers of Quantum Information Physics, KITP, 2017. URL http://online.kitp.ucsb.edu/ online/qinfo-c17/poulin/.
- [204] Domenico Giulini. What is the geometry of superspace? Physical Review D, 51(10): 5630-5635, may 1995. doi:10.1103/physrevd.51.5630. URL https://doi.org/10.1103/ physrevd.51.5630.
- [205] Domenico Giulini and Claus Kiefer. Wheeler-dewitt metric and the attractivity of gravity. *Physics Letters A*, 193(1):21-24, 1994. ISSN 0375-9601. doi:https://doi.org/10.1016/0375-9601(94)00651-2. URL https://www.sciencedirect. com/science/article/pii/0375960194006512.
- [206] T. Banks, M. E. Peskin, and L. Susskind. Difficulties for the evolution of pure states into mixed states. *Nuclear Physics B*, 244:125–134, September 1984. doi:10.1016/0550-3213(84)90184-6.
- [207] Wojciech H Zurek. Environment-induced superselection rules. *Physical review D*, 26(8): 1862, 1982.
- [208] Juan Pablo Paz and Wojciech Hubert Zurek. Environment-induced decoherence, classicality, and consistency of quantum histories. *Physical Review D*, 48(6):2728, 1993.
- [209] Wojciech Hubert Zurek. Decoherence and the transition from quantum to classical—revisited. In *Quantum Decoherence*, pages 1–31. Springer, 2006.
- [210] Todd A. Brun. A simple model of quantum trajectories. American Journal of Physics, 70(7):719-737, jul 2002. doi:10.1119/1.1475328. URL https://doi.org/10.1119%2F1. 1475328.
- [211] Mark Srednicki. Is purity eternal? Nucl. Phys., B410:143–154, 1993. doi:10.1016/0550-3213(93)90576-B.
- [212] J. Fuchs and C. Schweigert. Symmetries, Lie algebras and representations: A graduate course for physicists. Cambridge University Press, 10 2003. ISBN 978-0-521-54119-0.

- [213] Antoine Tilloy. Does gravity have to be quantized? Lessons from non-relativistic toy models. J. Phys. Conf. Ser., 1275(1):012006, 2019. doi:10.1088/1742-6596/1275/1/012006.
- [214] Sean M. Carroll. Lecture notes on general relativity, 1997. URL https://arxiv.org/ abs/gr-qc/9712019.
- [215] M. Fierz and W. Pauli. On relativistic wave equations for particles of arbitrary spin in an electromagnetic field. Proc. Roy. Soc. Lond., A173:211–232, 1939. doi:10.1098/rspa.1939.0140.
- [216] K. R. Green, N. Kiriushcheva, and S. V. Kuzmin. Analysis of Hamiltonian formulations of linearized General Relativity. *Eur. Phys. J. C*, 71:1678, 2011. doi:10.1140/epjc/s10052-011-1678-2.
- [217] Waldemar Ulmer. Deconvolution of a linear combination of gaussian kernels by an inhomogeneous fredholm integral equation of second kind and applications to image processing, 2011.
- [218] Jonathan Oppenheim and Benni Reznik. Fundamental destruction of information and conservation laws, 2009. URL https://arxiv.org/abs/0902.2361.
- [219] G.C. Ghirardi, A. Rimini, and T Weber. A model for a unified quantum description of macroscopic and microscopic systems. In *Quantum Probability and Applications, L. Accardi et al. (eds).* Springer, Berlin., 1985.
- [220] Michael JW Hall and Marcel Reginatto. Interacting classical and quantum ensembles. Physical Review A, 72(6):062109, 2005.
- [221] Wayne Boucher and Jennie Traschen. Semiclassical physics and quantum fluctuations. *Physical Review D*, 37(12):3522, 1988.
- [222] Andrea Mari, Giacomo De Palma, and Vittorio Giovannetti. Experiments testing macroscopic quantum superpositions must be slow. Sci. Rep., 6:22777, 2016. doi:10.1038/srep22777.
- [223] Henry Cavendish. Xxi. experiments to determine the density of the earth. Philosophical Transactions of the Royal Society of London, 88:469–526, 1798.
- [224] Gabriel G Luther and William R Towler. Redetermination of the newtonian gravitational constant g. *Physical Review Letters*, 48(3):121, 1982.
- [225] Jens H. Gundlach and Stephen M. Merkowitz. Measurement of newton's constant using a torsion balance with angular acceleration feedback. *Phys. Rev. Lett.*, 85:2869–2872, Oct 2000. doi:10.1103/PhysRevLett.85.2869. URL https://link.aps.org/doi/10.1103/PhysRevLett.85.2869.
- [226] Terry Quinn. Measuring big g. Nature, 408(6815):919–920, 2000.
- [227] GT Gillies and CS Unnikrishnan. The attracting masses in measurements of g: an overview of physical characteristics and performance. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 372(2026):20140022, 2014.
- [228] Christian Rothleitner and Stephan Schlamminger. Invited review article: Measurements of the newtonian constant of gravitation, g. *Review of Scientific Instruments*, 88(11): 111101, 2017.
- [229] Markus Arndt, Olaf Nairz, Julian Vos-Andreae, Claudia Keller, Gerbrand Van der Zouw, and Anton Zeilinger. Wave-particle duality of c60 molecules. *nature*, 401(6754):680–682, 1999.
- [230] Stefan Nimmrichter, Klaus Hornberger, Philipp Haslinger, and Markus Arndt. Testing spontaneous localization theories with matter-wave interferometry. *Phys. Rev. A*, 83: 043621, Apr 2011. doi:10.1103/PhysRevA.83.043621. URL https://link.aps.org/doi/ 10.1103/PhysRevA.83.043621.
- [231] Thomas Juffmann, Adriana Milic, Michael Müllneritsch, Peter Asenbaum, Alexander Tsukernik, Jens Tüxen, Marcel Mayor, Ori Cheshnovsky, and Markus Arndt. Real-time single-molecule imaging of quantum interference. *Nature Nanotechnology*, 7(5):297–300, 2012. doi:10.1038/nnano.2012.34. URL https://doi.org/10.1038/nnano.2012.34.
- [232] Thomas Juffmann, Stefan Nimmrichter, Markus Arndt, Herbert Gleiter, and Klaus Hornberger. New prospects for de broglie interferometry. *Foundations of Physics*, 42

(1):98-110, 2012. doi:10.1007/s10701-010-9520-5. URL https://doi.org/10.1007/ s10701-010-9520-5.

- [233] Stefan Gerlich, Sandra Eibenberger, Mathias Tomandl, Stefan Nimmrichter, Klaus Hornberger, Paul Fagan, Jens Tüxen, Marcel Mayor, and Markus Arndt. Quantum interference of large organic molecules. *Nature communications*, 2:263, 04 2011. doi:10.1038/ncomms1263.
- [234] Tobias Westphal, Hans Hepach, Jeremias Pfaff, and Markus Aspelmeyer. Measurement of gravitational coupling between millimetre-sized masses. *Nature*, 591(7849):225–228, 2021. doi:10.1038/s41586-021-03250-7.
- [235] Jonas Schmöle, Mathias Dragosits, Hans Hepach, and Markus Aspelmeyer. A micromechanical proof-of-principle experiment for measuring the gravitational force of milligram masses. *Classical and Quantum Gravity*, 33(12):125031, May 2016. doi:10.1088/0264-9381/33/12/125031.
- [236] J. G. Lee, E. G. Adelberger, T. S. Cook, S. M. Fleischer, and B. R. Heckel. New test of the gravitational 1/r² law at separations down to 52 μm. *Phys. Rev. Lett.*, 124:101101, Mar 2020. doi:10.1103/PhysRevLett.124.101101. URL https://link.aps.org/doi/10.1103/PhysRevLett.124.101101.
- [237] Daniel Conus and Robert Dalang. The Non-Linear Stochastic Wave Equation in High Dimensions. *Electronic Journal of Probability*, 13(none):629 - 670, 2008. doi:10.1214/EJP.v13-500. URL https://doi.org/10.1214/EJP.v13-500.
- [238] Angelo Bassi, Kinjalk Lochan, Seema Satin, Tejinder P. Singh, and Hendrik Ulbricht. Models of wave-function collapse, underlying theories, and experimental tests. *Rev. Mod. Phys.*, 85:471–527, Apr 2013. doi:10.1103/RevModPhys.85.471. URL https://link.aps.org/doi/10.1103/RevModPhys.85.471.
- [239] Hadrien Chevalier, A. J. Paige, and M. S. Kim. Witnessing the nonclassical nature of gravity in the presence of unknown interactions. *Physical Review A*, 102(2), Aug 2020. ISSN 2469-9934. doi:10.1103/physreva.102.022428. URL http://dx.doi.org/10.1103/ PhysRevA.102.022428.

- [240] Thomas W. van de Kamp, Ryan J. Marshman, Sougato Bose, and Anupam Mazumdar. Quantum Gravity Witness via Entanglement of Masses: Casimir Screening. *Phys. Rev.* A, 102(6):062807, 2020. doi:10.1103/PhysRevA.102.062807.
- [241] Marko Toroš, Thomas W. Van De Kamp, Ryan J. Marshman, M. S. Kim, Anupam Mazumdar, and Sougato Bose. Relative acceleration noise mitigation for nanocrystal matter-wave interferometry: Applications to entangling masses via quantum gravity. *Phys. Rev. Res.*, 3(2):023178, 2021. doi:10.1103/PhysRevResearch.3.023178.
- [242] C. W. Chou, D. B. Hume, T. Rosenband, and D. J. Wineland. Optical clocks and relativity. *Science*, 329(5999):1630–1633, 2010. doi:10.1126/science.1192720.
- [243] Peter Asenbaum, Chris Overstreet, Tim Kovachy, Daniel D. Brown, Jason M. Hogan, and Mark A. Kasevich. Phase shift in an atom interferometer due to spacetime curvature across its wave function. *Phys. Rev. Lett.*, 118:183602, May 2017. doi:10.1103/PhysRevLett.118.183602. URL https://link.aps.org/doi/10.1103/PhysRevLett.118.183602.
- [244] Chris Overstreet, Peter Asenbaum, Joseph Curti, Minjeong Kim, and Mark A. Kasevich. Observation of a gravitational aharonov-bohm effect. *Science*, 375(6577):226-229, 2022. doi:10.1126/science.abl7152. URL https://www.science.org/doi/abs/10.1126/ science.abl7152.
- [245] J. Oppenheim, A.Russo, and Z. Weller-Davies. Estimating space-time diffusion in ligo, 2022. Unpublished note.
- [246] Patrick J Abbott and Zeina Kubarych. Mass calibration at nist in the revised si. Metrolologist, 21:1, 2019.
- [247] Leon Chao, Frank Seifert, Darine Haddad, Julian Stirling, David Newell, and Stephan Schlamminger. The design and development of a tabletop kibble balance at nist. *IEEE Transactions on Instrumentation and Measurement*, 68(6):2176–2182, 2019. doi:10.1109/TIM.2019.2901550.

- [248] Achim Peters, Keng Yeow Chung, and Steven Chu. High-precision gravity measurements using atom interferometry. *Metrologia*, 38(1):25, 2001.
- [249] Vincent Ménoret, Pierre Vermeulen, Nicolas Le Moigne, Sylvain Bonvalot, Philippe Bouyer, Arnaud Landragin, and Bruno Desruelle. Gravity measurements below 10- 9 g with a transportable absolute quantum gravimeter. Scientific reports, 8(1):1–11, 2018.
- [250] Gian Carlo Ghirardi, Alberto Rimini, and Tullio Weber. Unified dynamics for microscopic and macroscopic systems. *Physical review D*, 34(2):470, 1986.
- [251] LE Ballentine. Failure of some theories of state reduction. *Physical Review A*, 43(1):9, 1991.
- [252] Philip Pearle, James Ring, Juan I Collar, and Frank T Avignone. The csl collapse model and spontaneous radiation: an update. *Foundations of physics*, 29(3):465–480, 1999.
- [253] Angelo Bassi, Emiliano Ippoliti, and Bassano Vacchini. On the energy increase in spacecollapse models. Journal of Physics A: Mathematical and General, 38(37):8017, 2005.
- [254] Stephen L Adler. Lower and upper bounds on csl parameters from latent image formation and igm heating. Journal of Physics A: Mathematical and Theoretical, 40(12):2935, 2007.
- [255] Kinjalk Lochan, Suratna Das, and Angelo Bassi. Constraining continuous spontaneous localization strength parameter λ from standard cosmology and spectral distortions of cosmic microwave background radiation. *Physical Review D*, 86(6):065016, 2012.
- [256] Stefan Nimmrichter, Klaus Hornberger, and Klemens Hammerer. Optomechanical sensing of spontaneous wave-function collapse. *Physical review letters*, 113(2):020405, 2014.
- [257] Mohammad Bahrami, Angelo Bassi, and Hendrik Ulbricht. Testing the quantum superposition principle in the frequency domain. *Physical Review A*, 89(3):032127, 2014.
- [258] Franck Laloë, William J Mullin, and Philip Pearle. Heating of trapped ultracold atoms by collapse dynamics. *Physical Review A*, 90(5):052119, 2014.

- [259] Mohammad Bahrami, M Paternostro, Angelo Bassi, and H Ulbricht. Proposal for a noninterferometric test of collapse models in optomechanical systems. *Physical Review Letters*, 112(21):210404, 2014.
- [260] Daniel Goldwater, Mauro Paternostro, and PF Barker. Testing wave-function-collapse models using parametric heating of a trapped nanosphere. *Physical Review A*, 94(1): 010104, 2016.
- [261] Antoine Tilloy and Thomas M Stace. Neutron star heating constraints on wave-function collapse models. *Physical Review Letters*, 123(8):080402, 2019.
- [262] Sandro Donadi, Kristian Piscicchia, Catalina Curceanu, Lajos Diósi, Matthias Laubenstein, and Angelo Bassi. Underground test of gravity-related wave function collapse. *Nature Physics*, pages 1–5, 2020.
- [263] Brian C. Hall. Quantum Theory for Mathematicians. Springer New York, NY, 2013. doi:10.1007/978-1-4614-7116-5.
- I. Siemon, A. S. Holevo, and R. F. Werner. Unbounded generators of dynamical semigroups. Open Systems and Information Dynamics, 24(04):1740015, Dec 2017. ISSN 1793-7191. doi:10.1142/s1230161217400157. URL http://dx.doi.org/10.1142/S1230161217400157.
- [265] Jonathan Oppenheim and Andrea Russo. Gravity in the diffusion regime, 2023. manuscript in preparation.
- [266] Lajos Diósi. Is there a relativistic Gorini-Kossakowski-Lindblad-Sudarshan master equation? Phys. Rev. D, 106(5):L051901, 2022. doi:10.1103/PhysRevD.106.L051901.